

## Boundary controllability, stabilization and synchronization for 1-D hyperbolic balance laws Long Hu

## ▶ To cite this version:

Long Hu. Boundary controllability, stabilization and synchronization for 1-D hyperbolic balance laws. General Mathematics [math.GM]. Université Pierre et Marie Curie - Paris VI; Université de Fudan (Shanghai, Chine), 2015. English. NNT: 2015PA066401. tel-01275752

## HAL Id: tel-01275752 https://theses.hal.science/tel-01275752

Submitted on 18 Feb 2016

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Université Pierre et Marie Curie Paris 6



### Thèse de Doctorat de l'Université Pierre et Marie Curie

Présentée et soutenue publiquement le 16 Août 2015

pour l'obtention du grade de

Docteur de l'Université Pierre et Marie Curie

Spécialité : Mathématiques Appliquées

 $\operatorname{par}$ 

Long HU

sous la direction de Jean-Michel CORON et Tatsien LI

# Contrôle frontière, stabilisation et synchronisation pour des systèmes de lois de bilan en dimension un d'espace

après avis des rapporteurs Mme. Fatiha Alabau-Boussouira & M. Günther Leugering

devant le jury composé de

Mme.	Fatiha Alabau-Boussouira	Rapporteur
М.	Georges BASTIN	Examinateur
М.	Jean-Michel Coron	Directeur de Thèse
М.	Tatsien LI	Directeur de Thèse
М.	Shige Peng	Examinateur
М.	Bopeng RAO	Examinateur
М.	Emmanuel Trélat	Examinateur
М.	Yi Zhou	Examinateur

École Doctorale de Sciences Mathématiques de Paris Centre ED 386





Long HU :

Sorbonne Universités, UPMC Univ Paris 06, UMR 7598, Laboratoire Jacques-Louis Lions, F-75005, Paris, France.

School of Mathematical Sciences, Fudan University, Shanghai 200433, China.

Adresse électronique: hu@ann.jussieu.fr, hul10@fudan.edu.cn

# Acknowledgements

I would like to take this opportunity to express my deepest and foremost gratitude to my two supervisors, Jean-Michel Coron and Tatsien Li, who have introduced me into the field of control theory. I extremely appreciate many fruitful discussions with them, from which I gradually come to understand how excellent mathematicians perceive and deal with mathematical problems. Without their patient supervision and constant encouragement, this dissertation would not have been possible. Besides this, their extraordinary personality also greatly impresses me.

I am immensely indebted to Fatiha Alabau-Boussouira and Günther Leugering for their very positive reports and useful comments on my thesis.

I also enthusiastically thank Georges Bastin, Shige Peng, Bopeng Rao, Emmanuel Trélat and Yi Zhou for their coming to my thesis defense and becoming as the jury members.

My special thank are owe to Florent Di Meglio, Miroslav Krstic, Peng Qu, Rafael Vazquez and Zhiqiang Wang for their friendly cooperation on the related topics of this thesis and to Bingyu Zhang and Xu Zhang for the meaningful discussions during their visit in Paris. I benefit a lot by communicating with them.

I also thank my fellow classmates in Fudan and talented colleagues in LJLL as well as some friends, Abdellah Chkifa, Yan Cui, Ludovick Gagnon, Xumin Gu, Lianhua He, Cunming Liu, Xing Lu, Qi Lü, Frédéric Marbach, Carlo Marcati, Thomas Michel, Ivan Moyano, Guillaume Olive, Chaoyu Quan, Peipei Shang, Pierre-Henri Tournier, Philippe Ung, Fan Wang, Ke Wang, Shuyang Xiang, Wei Yang, Dongbin Zha, Can Zhang, Hai Zhang, Zhiyuan Zhang, Qi Zhou and Jiamin Zhu, who are always giving me their generous help whenever I have a trouble.

Last but never at least, my thanks would go to my beloved family and my GF Yuming Sun for their encouragement and confidence in me all through these years. 4\_\_\_\_\_

# Contents

Introd	luction	9
1	Classical Topics in Control Theory for Dynamical Systems	9
2	Hyperbolic Systems of Balance Laws and Their Related Control Prob-	11
3	Motivation and Outline of the Thesis	11

## I Exact Boundary Controllability for Hyperbolic Balance Laws 23

Sharp Boundary Control Time Estimates for 1-D Homogeneous Qua		
line	ar Hyperbolic Systems	25
1.1	Introduction and Main Results	25
1.2	Some Estimates on Linear Hyperbolic Systems with Nonlinear Bound-	
	ary Conditions	33
1.3	Proof of Main results	35
1.4	The Existence of Operator $\mathfrak{F}$	39
1.5	Appendix: Proof of Lemma 1.2.1	46
0		
Un	boundary control of a hyperbolic system with a vanishing char-	
acte	eristic speed	57
2.1	Introduction and main results	57
	<ul> <li>Sha</li> <li>line</li> <li>1.1</li> <li>1.2</li> <li>1.3</li> <li>1.4</li> <li>1.5</li> <li>On</li> <li>acted</li> <li>2.1</li> </ul>	<ul> <li>Sharp Boundary Control Time Estimates for 1-D Homogeneous Quas linear Hyperbolic Systems</li> <li>1.1 Introduction and Main Results</li></ul>

2.2	Well-posedness of the system $(2.1.5)$ and the system $(2.1.9)$	61
2.3	Proof of Theorem 2.1.1	62
2.4	Proof of Theorem 2.1.2	65
2.5	Proof of Theorem 2.1.3	71

## II Boundary Stabilization for Inhomogeneous Hyperbolic Systems 73

3	3 Feedback Control Design of Homodirectional and General Hetero		
	rect	ional Linear Coupled Hyperbolic PDEs	75
	3.1	Introduction	76
	3.2	System description	78
	3.3	Finite-time stabilization of heterodirectional systems	79
	3.4	Uncollocated observer design and output feedback controller $\ . \ . \ .$	85
	3.5	Motion planning for homodirectional systems	89
	3.6	Proof of Theorem 3.3.1: well-posedness of the kernel equations	95
	3.7	Concluding remarks	105
	3.8	Appendix: Explicit definition of the characteristic lines	106
4	Boi	undary Exponential Stabilization of 1-D Inhomogeneous Quasi-	_
-	line	ar Hyperbolic Systems	109
	4.1	Introduction and Main Result	110
	4.2	Preliminaries–Linear Case	113
	4.3	Backstepping boundary control design for nonlinear system	121
	4.4	Proof of Theorem 4.1.1	125
	4.5	Appendix A	134
	4.6	Appendix B	138

III	Exact Boundary Synchronization for a Coupled Sys-	,
$\mathbf{tem}$	of Wave Equations 1	<b>43</b>

5	Exa Syst	ct Boundary Controllability and Observability for a Coupled tem of Quasilinear Wave Equations	$\frac{1}{145}$
	5.1	Introduction	145
	5.2	Existence and Uniqueness of Semi-global $C^2$ Solution	148
	5.3	Local Exact Boundary Controllability	153
	5.4	Local Exact Boundary Observability	155
6	Exa Way	ct Boundary Synchronization for a Coupled System of Linea ve Equations	r 159
	6.1	Introduction	159
	6.2	Exact boundary synchronization	163
	6.3	Exact boundary synchronization by groups	168
	6.4	Exact boundary null controllability and synchronization by groups	173
	6.5	Generalized exact boundary synchronization	178
	6.6	Remarks	187
7	Exa Wav Typ	act Boundary Synchronization for a Coupled System of Linea we Equations with Coupled Boundary Conditions of Dissipative be	r e 189
	7.1	Introduction	189
	7.2	Exact boundary synchronization	192
	7.3	Exact boundary synchronization by groups	197
	7.4	Exact boundary null controllability and synchronization by groups	203
	7.5	Generalized exact boundary synchronization	208

8 Exact Boundary Synchronization for a Coupled System of Quasilin-

ear	Wave Equations	215
8.1	Introduction and Main Result	215
8.2	Preliminaries	220
8.3	Proof of Theorem 8.1.1	227
8.4	Remarks	234
8.5	Appendix: Proof of Lemma 8.3.2	236
Bibliography		252

## Introduction

## 1 Classical Topics in Control Theory for Dynamical Systems

What is the controllability and stabilization for a dynamical system? Let us start with the case of Ordinary Differential Equations (ODEs). Consider the following finite dimensional dynamical system

$$\frac{dx}{dt} = f(x, u), \tag{1.1}$$

where  $t \in \mathbb{R}$  is the independent variable (time),  $x \in \mathbb{R}^n$  is the state variable and  $u \in \mathbb{R}^m$  is the control (note that in many cases m < n), f is a function of class  $C^{\infty}$  with respect to x and u. For convenience, we assume that

$$f(0,0) = 0. (1.2)$$

Let us recall some basic definitions of local exact controllability and stabilization for the system (1.1). One can refer to [19] for an introduction and fruitful results on the controllability and stabilization of linear and nonlinear dynamical systems.

**Definition 1.1** The control system (1.1) is locally controllable at the equilibrium (0,0) on the interval  $[T_0, T_1]$   $(T_0 < T_1)$  if, for every real number  $\varepsilon > 0$ , there exists a real number  $\eta > 0$  such that, for every  $x^0 \in B_\eta := \{x \in \mathbb{R}^n; |x| < \eta\}$  and for every  $x^1 \in B_\eta$ , there exists a measurable bounded function  $u : [T_0, T_1] \to \mathbb{R}^m$  such that the solution x = x(t) to the Cauchy problem

$$\begin{cases} \frac{dx}{dt} = f(x, u(t)), \\ x(T_0) = x^0 \end{cases}$$
(1.3)

verifies the final condition

$$x(T_1) = x^1. (1.4)$$

**Remark 1.1** The dynamical system  $\dot{x} = f(x, u(t))$  is usually called an open-loop system and u is called the corresponding open-loop control.

**Definition 1.2** The control system (1.1) is locally asymptotically (resp. exponentially) stabilizable at the equilibrium (0,0) if, there exists a feedback law, i.e. a continuous map,  $u : \mathbb{R}^n \to \mathbb{R}^m$ , satisfying

$$u(0) = 0, (1.5)$$

such that  $0 \in \mathbb{R}^n$  is a locally asymptotically (resp. exponentially) stable point for the system

$$\frac{dx}{dt} = f(x, u(x)). \tag{1.6}$$

**Remark 1.2** The dynamical system  $\dot{x} = f(x, u(x))$  is usually called a closed-loop system and  $u : \mathbb{R}^n \to \mathbb{R}^m$  is called the corresponding closed-loop control.

It is well-known that if we focus on the linear case, *i.e.* 

$$f(x,u) := Ax + Bu, \tag{1.7}$$

where  $A \in \mathcal{M}_{n,n}(\mathbb{R})$  and  $B \in \mathcal{M}_{n,m}(\mathbb{R})$ , one can find explicit necessary and sufficient conditions for the aforementioned controllability and stabilization properties (see, for example, [19, 34, 62] *et. al.*):

**Theorem 1.1** System (1.1) with (1.7) is controllable if and only if

$$rank(\lambda I - A, B) = n, \quad \forall \lambda \in \mathbb{C}.$$
 (1.8)

Here and in what follows,  $rank(\cdot)$  denotes the rank of the corresponding matrix.

**Theorem 1.2** System (1.1) with (1.7) is asymptotically (resp. exponentially) stabilizable if and only if

$$rank(\lambda I - A, B) = n$$
, for each  $\lambda \in \mathbb{C}$  with  $\Re \lambda \ge 0$ . (1.9)

**Remark 1.3** Condition (1.8) is known as the Hautus criterion, which is equivalent to the famous Kalman rank condition:

$$rank(B, AB, \cdots, A^{n-1}B) = n.$$
(1.10)

In the finite dimensional case, from Theorem 1.1 and Theorem 1.2, one can easily see that :

- (1) For linear control system (1.1) with (1.7) which satisfies (1.8) or equivalently (1.10), the open-loop controllability can be realized almost immediately. In other words, a linear control system which is controllable in large time (i.e.  $T_1 T_0$  is large) is also controllable in small time (i.e.  $T_1 T_0$  is small);
- (2) Every linear control system which is controllable can be asymptotically stabilized by means of linear feedback laws.

However, (1)-(2) mentioned above usually do not hold for general nonlinear system and for infinite dimensional dynamical systems, which leaves us much research space. In fact, in what follows, we will study three related control problems, i.e. controllability, stabilization and synchronization (a weak kind of controllability) for the linear and nonlinear hyperbolic systems of balance laws, and we will show some new mechanisms for such systems.

## 2 Hyperbolic Systems of Balance Laws and Their Related Control Problems

The concepts in control theory for ODEs in Introduction 1 can be automatically adapted to Partial Differential Equations (PDEs). Before this, let us briefly review our hyperbolic systems of balance laws.

### 2.1 System Description

Consider the following first-order 1-D hyperbolic systems of balance laws

$$\frac{\partial u}{\partial t} + A(u)\frac{\partial u}{\partial x} = F(u), \quad (t,x) \in [0,T] \times [0,L], \tag{2.1}$$

where,  $u = (u_1, \ldots, u_n)^T$  is a vector function of (t, x), A(u) is a given  $n \times n$  matrix with suitably smooth entries  $a_{ij}(u)$   $(i, j = 1, \cdots, n)$ , and  $F(u) = (f_1(u), \cdots, f_n(u))^T$ is a given vector function of u with suitably smooth elements  $f_i(u)(i = 1, \cdots, n)$ .

**Remark 2.1** In general, we call the following systems

$$\frac{\partial u}{\partial t} + \frac{\partial g(u)}{\partial x} = F(u) \tag{2.2}$$

to be hyperbolic balance laws, where the flux  $g := (g_1, \dots, g_n)$  is a vector function of u with suitably smooth elements. Obviously, system (2.2) can be written in the quasilinear form as (2.1) with the Jacobian matrix

$$A(u) := D(g(u)). \tag{2.3}$$

Therefore, throughout this thesis, unless otherwise indicated, we consider the inhomogeneous quasilinear hyperbolic system (2.1) as the hyperbolic balance laws.

**Remark 2.2** In the special case where there are no source terms (i.e.  $F(u) \equiv 0$ ), the homogeneous quasilinear system (2.1) reduces to

$$\frac{\partial u}{\partial t} + A(u)\frac{\partial u}{\partial x} = 0, \qquad (2.4)$$

which constitutes a so-called hyperbolic system of conservation laws.

**Definition 2.1** (2.1) is a hyperbolic system, if, for any given u on the domain under consideration, A(u) has n real eigenvalues  $\lambda_i(u)$   $(i = 1, \dots, n)$  and a complete set of left (resp. right) eigenvectors  $l_i(u) = (l_{i1}(u), \dots, l_{in}(u))$  (resp.  $r_i(u) = (r_{1i}(u), \dots, l_{ni}(u))^T$ ,  $(i = 1, \dots, n)$ ):

$$l_i(u)A(u) = \lambda_i(u)l_i(u).$$
(2.5)

and

$$A(u)r_i(u) = \lambda_i(u)r_i(u).$$
(2.6)

Here, by a complete set we mean a basis for  $\mathbb{R}^n$ .

In what follows, we assume that, maybe after a suitable change of variables,

$$\lambda_r(u) < \lambda_p(u) \equiv 0 < \lambda_s(u)$$
  

$$r = 1, \cdots, l, \ p = l+1, \cdots, m, \ s = m+1, \cdots, n,$$
(2.7)

and

$$l_{ij}(0) = \delta_{ij} \quad (i, j = 1, \cdots, n),$$
 (2.8)

where  $\delta_{ij}$  stands for the Kronecker symbol. In particular,

$$A(0) = \begin{pmatrix} \lambda_1(0) & 0 \\ & \ddots & \\ 0 & & \lambda_n(0) \end{pmatrix}.$$
 (2.9)

**Definition 2.2** If, for any given u on the domain under consideration, A(u) has n distinct real eigenvalues

$$\lambda_1(u) < \lambda_2(u) < \dots < \lambda_n(u), \tag{2.10}$$

then, the set of left (resp. right) eigenvectors forms a complete set. The corresponding hyperbolic systems (2.1) satisfying (2.10) are called strictly hyperbolic systems.

**Remark 2.3** Here and in what follows, we always suppose that all the real eigenvalues  $\lambda_i(u)(i = 1, \dots, n)$  and all left and right eigenvectors  $l_i(u)$  and  $r_i(u)(i = 1, \dots, n)$  have the same regularity as A(u). However, one can easily check that this is not an issue for strictly hyperbolic systems (see, for example, [72, Appendix A, Pages 195-210]).

To ensure the well-posedness of system (2.1), we consider the following general kind of boundary conditions:

$$x = 0: u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \ (s = m + 1, \cdots, n), \tag{2.11}$$

$$x = L : u_r = G_r(t, u_{l+1}, \cdots, u_n) + H_r(t) \ (r = 1, \cdots, l),$$
(2.12)

where  $G_j$  and  $H_j$   $(j = 1, \dots, l, m+1, \dots, n)$  are functions of t with suitably smooth components. Here and hereafter, we suppose that

$$G_j(t, 0, \cdots, 0) \equiv 0 \quad (j = 1, \cdots, l, m + 1, \cdots, n).$$
 (2.13)

**Remark 2.4** Boundary conditions (2.11)-(2.12) are prescribed under the hypothesis (2.8). We could also replace  $u_i$  by  $v_i = l_i(u)u$  in (2.11)-(2.12) in the general situation (see also [69, Remark 1.3 on page 171], [72], [111] and [52]).

Hyperbolic balance laws (2.1), even systems modeled by PDEs, are typically infinite dimensional dynamical system, which not only contains all the cases of conservation laws such as the Saint–Venant equation (including subcritical, supercritical, and critical cases), the isentropic and full Euler equation for one-dimensional gas dynamics (including subsonic, supersonic, and sonic cases), the nonlinear vibrating string equations *et. al.* (see, for example, [18, 23, 40, 41, 44, 48, 69, 102] and the references therein), but also the fundamental dynamics of open conservative systems such as the telegrapher equations for electrical lines, the Saint–Venant–Exner equations *et.al.* (see [12, 32, 101]). It is well-known that the controllability theory of hyperbolic systems (2.1) depends strongly on the matrix A(u) and how the controls act to the systems (in the equations or to the boundary conditions).

In this thesis, we study the following three control problems, i.e. Problem of Exact Boundary Controllability (EBC), Problem of Boundary Stabilization (BS) and Problem of Weak Boundary Controllability (WBC), for the system (2.1) with boundary conditions (2.11)-(2.12):

• Problem (EBC). Let  $\mathfrak{C} \subseteq \{1, \dots, n\}$ . For any given initial data  $\varphi := (\varphi_1, \dots, \varphi_n)^T \in (C^1[0, L])^n$  and final date  $\psi := (\psi_1, \dots, \psi_n)^T \in (C^1[0, L])^n$  with small  $C^1$  norm, can we find a time T > 0 and boundary controls  $H_i \in C^1[0, T] (i \in \mathfrak{C})$  (for  $i \notin \mathfrak{C}$ ,  $H_i(t)$  is given and is assumed to be small) with small  $C^1$  norm, satisfying the  $C^1$  compatibility conditions at the points (t, x) = (0, 0)

and (0, L) (see Remark 2.5), such that the mixed initial-boundary value problem (2.1), (2.11)-(2.12) with the initial condition

$$t = 0: u = \varphi(x), 0 \le x \le L \tag{2.14}$$

admits a  $C^1$  solution u(t, x) on the domain

$$R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\},$$
(2.15)

which satisfies the final condition

$$t = T : u = \psi(x), 0 \le x \le L?$$
 (2.16)

• Problem (BS). Suppose that  $H_i(t) \equiv 0$   $(i = 1, \dots, n)$  and that  $\lambda > 0$  is an arbitrary given constant. Suppose furthermore that  $C^1$  compatibility conditions are satisfied at the point (t, x) = (0, 0) (see Remark 2.5). Does there exist a linear feedback control  $\mathcal{B} : (H^2(0, L))^n \to \mathbb{R}^l$ , without loss of generality verifying the  $C^1$  compatibility conditions at the point (t, x) = (0, 1), such that for some  $\varepsilon > 0$ , every solution u of the mixed initial-boundary value problem (2.1), (2.11)-(2.12) and (2.14) with  $H(t) = \mathcal{B}(u(t, \cdot))$  satisfies

$$\|u(t,\cdot)\|_{H^2(0,L)} \le C e^{-\lambda t} \|\varphi(\cdot)\|_{H^2(0,L)},\tag{2.17}$$

for some C > 0, provided that  $\|\varphi(\cdot)\|_{H^2(0,L)} \leq \varepsilon$ ?

Moreover, generally speaking, if there are essentially not enough boundary controls, then it is not possible to realize the exact boundary (null) controllability (see [80]). Then, it is nature to ask what happens in the case of lack of boundary controls to the hyperbolic systems (2.1)? Therefore, we are also concerned with

• Problem (WBC). Which kind of controllability in a weaker sense can be realized by means of less boundary controls for the general hyperbolic balance laws (2.1) or its special form?

**Remark 2.5** The  $C^1$  compatibility conditions at the point (t, x) = (0, 0) are given by

$$\begin{aligned} \varphi_s(0) &= G_s(0,\varphi_1(0),\cdots,\varphi_m(0)) + H_s(0) \quad (s = m + 1,\cdots,n), \\ f_s(\varphi(0)) &- \sum_{j=1}^n a_{sj}(\varphi(0))\varphi_{jx}(0) = \frac{\partial G_s}{\partial t}(0,\varphi_1(0),\cdots,\varphi_m(0)) + H_{st}(0) \\ &+ \sum_{r=1}^m \sum_{j=1}^n \frac{\partial G_s}{\partial u_r}(0,\varphi_1(0),\cdots,\varphi_m(0)) \Big( f_r(\varphi(0)) - a_{rj}(\varphi(0))\varphi_{jx}(0) \Big) \ (s = m + 1,\cdots,n). \end{aligned}$$

The  $C^1$  compatibility conditions at the point (t, x) = (0, L) are similar.

### 2.2 Concise Review of the Literature and Open Problems

### Exact Boundary Controllability

There are many publications concerning the exact controllability for linear hyperbolic systems (see [88], [100] and the references therein). As a special form of second order hyperbolic equations, the exact boundary controllability for linear wave equations was obtained in a complete manner by using the so-called Hilbert Uniqueness Method (HUM) (see [88]-[89]). In fact, the essence of J.-L. Lions's HUM is to use the duality to get the controllability by a corresponding observability inequality. Using HUM as well as other technical developments, Zuazua [122]-[123], Emanuilov [37] and Lasiecka & Triggiani [66] subsequently established the exact boundary controllability for some semilinear wave equations. As for the general first order hyperbolic systems, one can refer to the prior works [15, 86] and the recent monograph of Li [72] for almost complete results on several kinds of exact boundary controllability for the 1-D quasilinear inhomogeneous hyperbolic systems (2.1) in the context of classical solution (see also Part I). Let us mention [121] for the local boundary controllability of quasilinear wave equations in the multidimensional space. In the framework of BV solutions, the corresponding controllability for hyperbolic conservation laws has been studied in [8, 9, 13, 40, 41, 51, 96].

Since the hyperbolic wave has a finite speed of propagation, the exact boundary control time T > 0 should be suitably large. However, from the point of view of applications, T should be chosen as small as possible. Thus, one may ask

(i). Is it possible to get an optimal control time estimate for the hyperbolic balance laws (2.1), homogeneous case or inhomogeneous case?

On the other hand, it is well known that if all the characteristic speeds of the system do not vanish, i.e., all the eigenvalues of A(u) are nonzero in the domain under consideration, then (2.1) is exactly controllable (at least locally in  $C^1$  class) by boundary controls provided that the control time is sufficiently large. (see Li and Rao [75, 74]; see also Wang [111] for the nonautonomous case). If A(u) has zero eigenvalue, one can add some internal controls acting on the components corresponding to the zero eigenvalues and boundary controls acting on the other components [82, 111, 76]. It is worth mentioning that Glass recently obtained the controllability (not exact boundary controllability) of 1-D non-isentropic Euler equation, a hyperbolic system of conservation laws with a vanishing characteristic speed, in both Eulerian and Lagrangian coordinates in the context of entropy solutions [41]. Another open question naturally arises:

(ii). Is it possible to establish the exact controllability for a hyperbolic system of balance laws with vanishing characteristic speeds by only boundary controls?

This situation seems to be complicated. Notice that the solution u(t, x) of the degenerate case of the equation (essentially an ordinary differential equation)

$$\begin{cases} u_t = 0, \quad (t, x) \in (0, T) \times (0, 2\pi), \\ u(0, x) = u_0(x), \quad x \in (0, 2\pi), \end{cases}$$
(2.18)

obeys

$$u(t, \cdot) \equiv u_0, \quad \forall t \in [0, T].$$

$$(2.19)$$

Obviously, it is impossible to change the value u by using the boundary controls acting on the end x = 0 and/or  $x = 2\pi$ . In this direction, Gugat and Leugering [48] proved a global boundary controllability result which covers the critical case in which one of the characteristic speed is zero. In spite of the degenerate model (2.18), it is still possible to prove the controllability for some nonlinear systems. The return method of Coron has been applied to many situations, where the linearized system is not controllable while the nonlinearity enables the corresponding nonlinear systems to be controllable (see, in particular, [19] and the references therein). One can apply the return method to realize exact boundary controllability for quasilinear hyperbolic systems with a zero characteristic speed [23], when the possible vanishing characteristic speed can be driven to nonzero after sufficiently long time. However, this approach seems to be no longer valid for the system with identically zero characteristic speeds (see [23, Remark 1.3]).

**Remark 2.6** Let us also mention [36, 38] for the distributed (internal) controllability of linear/semilinear hyperbolic systems in any space dimension and [90] for the corresponding stochastic problem. The optimal control problems for wave equation are considered with controls in both Dirichlet and Neumann boundary conditions and with pointwise constraints on the control and the state (see [92, 93]).

#### **Boundary Stabilization**

In the last three decades or so, there has been a resurgence in the analysis of the asymptotical behavior for quasilinear hyperbolic systems (2.1). During this time, three parallel mathematical approaches have emerged. The first one is the so-called "Characteristic method", i.e. computing corresponding bounds by using explicit evolution of the solution along the characteristic curves. With this method, Problem (BS) has been previously investigated by Greenberg and Li (see [42]) for  $2 \times 2$  (i.e.  $A \in \mathcal{M}_{2,2}(\mathbb{R})$ ) systems and Li and Qin (see [69, 98]) for a generalization to  $n \times n$ homogeneous systems in the framework of  $C^1$  norm. Also, this method was developed by Li and Rao [75] to study the exact boundary controllability for general inhomogeneous quasilinear hyperbolic systems.

The second method is the "Control Lyapunov Functions method", which is a useful tool to analyze the asymptotic behavior of dynamical systems. This method was first used by Coron et.al. to enable the design of dissipative boundary conditions for nonlinear homogeneous hyperbolic systems in the context of both  $C^1$  norm and  $H^2$  norm [11, 20, 21, 22, 25]. Although both Characteristic method and Control Lyapunov Functions method have been driven to study exponential stability for hyperbolic conservation laws or even some hyperbolic balance laws, however, these two methods fail when dealing with Problem (BS) for the general inhomogeneous hyperbolic systems (see [10], [11, Section 5.6], [19, Pages 314 and 361–371], [32], [69], [21] and [22]).

The third one is the "Backstepping method", which has been proven to be effective to stabilize inhomogeneous quasilinear hyperbolic systems (2.1) (see [26] and [31]) and other equations (see [14, 24, 65, 103]). In [26], a full-state feedback control law, with actuation on only one end of the domain, which achieves  $H^2$  exponential stability of the closed-loop 2 × 2 linear and quasilinear hyperbolic system is derived using a backstepping method. Unfortunately, the method presented in [26] can not be directly extended to  $n \times n$  systems, even in linear cases, especially when several states convecting in the same direction are controlled (see also [31]). Therefore, the following question is of great interest and remained open:

(iii). Can we stabilize the  $3 \times 3$  linear or quasilinear inhomogeneous hyperbolic systems (2.1) by multi-boundary feedback controls? and what about  $n \times n$  cases?

**Remark 2.7** One can also refer [28], [46], [68] and [97] to the results on stabilization for hyperbolic systems on networks.

#### Weak Boundary Controllability-Exact Boundary Synchronization

As being said before, it is a significant issue from both theoretical and practical points of view that whether we can shorten the control time or number of controls for hyperbolic systems (2.1). For instance, consider the following coupled system of wave equations

$$U_{tt} - U_{xx} + AU = 0, (2.20)$$

where  $U \in \mathbb{R}^N$  and  $A \in \mathcal{M}_{N,N}(\mathbb{R})$ . Based on the results in [55], we can see that one needs N boundary controls to achieve controllability for N coupled wave systems (2.20). In view of less controls, although one can not realize the exact controllability in general (see [80] and Remark 2.8), however, the indirect controllability and asymptotical controllability (another kind of controllability which is not as mentioned in Definition 1.2), both the boundary controllability in a weaker sense, can be realized by means of M(< N) boundary controls for some special coupled systems of N wave equations (2.20) (cf. [2], [3], [6], [78]). An interesting open question rises:

(iv). Can we define another kind of weak boundary controllability without enough controls for hyperbolic systems (2.20), such as the exact boundary synchronization and its related topics? and how to construct boundary controls achieving these synchronizations?

Synchronization is actually a widespread natural phenomenon. Thousands of fireflies may twinkle at the same time; audiences in the theater can applaud with a rhythmic beat; pacemaker cells of the heart function simultaneously; and field crickets give out a unanimous cry—all these are phenomena of synchronization [104], [115].

In principle, synchronization happens when different individuals possess likeness in nature, that is, they conform essentially to the same governing equation, and meanwhile, the individuals should bear a certain coupled relation.

The phenomenon of synchronization was first observed by Huygens in 1665 [60]. The theoretical research on synchronization phenomena dates back to H. Fujisaka and T. Yamada's study of synchronization for coupled equations in 1983 [39]. The previous studies focused on systems described by ODEs, such as

$$\frac{\mathrm{d}X_i}{\mathrm{d}t} = f(X_i, t) + \sum_{j=1}^N A_{ij} X_j \qquad (i = 1, \cdots, N),$$
(2.21)

and considered the asymptotical synchronization behavior, i.e.,

$$X_i(t) - X_j(t) \to 0$$
, as  $t \to +\infty \quad (\forall i \neq j)$ , (2.22)

where  $X_i (i = 1, \dots, N)$  are *n*-dimensional vectors,  $A_{ij} (i, j = 1, \dots, N)$  are  $n \times n$  matrices, and f(X, t) is an n-dimensional vector function independent of *i*.

Our goal is to synchronize a hyperbolic system (2.20) or its quasilinear form through boundary controls. Different from the ODE situation, the coupling of systems can then be fulfilled by coupling of the equations or (and) the boundary conditions, which has richer research implications, moreover, boundary controls can be used to realize our goal in a finite time.

Roughly speaking, we want to find a finite time T > 0 and few boundary controls on [0, T], such that from the time t = T on, the system states tend to a state having components which are equal. That is to say, we hope to achieve the synchronization of the system states not only at the moment t = T under the action of few boundary controls on [0, T], but also when  $t \ge T$  and withdrawing all the controls. This is forever, instead of short-lived, synchronization, as is desired in many actual applications. All the topics in synchronization are almost blank and uncultivated in the field of PDEs, especially for hyperbolic systems of balance laws (2.1) or its special form (2.20).

**Remark 2.8** One can also find some results on the exact controllability of cascade coupled multi-dimensional hyperbolic systems by a reduced number of controls in [4, 5].

## **3** Motivation and Outline of the Thesis

This thesis is motivated by the above four questions in Introduction 2.2 and brings precise answers to all of them. It is composed of eight chapters and comprises three main parts numbered I,II and III, where Part I deals with the Questions (i) and (ii), while Parts II and III deal with the Questions (iii) and (iv), respectively. Most of the results presented in the thesis are published or submitted for publication in our papers [56, 52, 59, 58, 53, 54, 55, 57, 81].

### 3.1 Main Work in Part I

In Part I (Chapters 1–2), we are concerned with the exact boundary controllability problem, i.e. Problem (EBC), for hyperbolic systems of balance laws (2.1).

For this, in Chapter 1, we present our results in [52], which is devoted to give sharp time estimates for local exact boundary controllability of 1-D homogeneous quasilinear hyperbolic systems without zero characteristic speeds. We improve the time for the controllability found by Li [72, Theorem 3.2 and Theorem 3.3 on pages 39-41] and Zhang [120] both for the case of one-sided control and for the case of both-sided controls when the dimension of the controls is the same as the case of one-sided control. We prove that this control time is sharp. Our approach relies on the utilization of constructive methods, the characteristic theory together with a suitable iteration scheme on a suitable time interval.

As being said, despite of the fact that distributed (internal) controls are usually used to obtain controllability for a hyperbolic system (2.1) with vanishing characteristic speeds, Chapter 2 is, however, devoted to study the case where only boundary controls are considered. Following our approach developed in [59], we first prove that a weak coupled hyperbolic system of balance laws (2.1) is not (null) controllable in finite time. Meanwhile, we give a sufficient and necessary condition for the asymptotic stabilization of the system under a natural feedback. This result somehow gives a negative answer to Question (ii), that is even for the coupled hyperbolic system (2.1) with zero characteristic speed, it is also impossible to achieve the corresponding exact boundary controllability. However, one can stabilize these systems in infinite time by means of boundary feedback. This extends the previous works [82, 111, 76] and [23].

### 3.2 Main Work in Part II

Part II (Chapters 3–4) is devoted to study boundary stabilization for inhomogeneous hyperbolic systems. Two chapters are established in order to illustrate this topic.

In Chapter 3, we first introduce the definition of heterodirectional and homodirectional hyperbolic systems, and present our results from [54] that deal with backstepping control design for linear first-order heterodirectional and homodirectional hyperbolic systems featuring arbitrary large (local) coupling coefficients. A particular choice of the target system, featuring a cascade structure similar to [26, Section 3.5], enables the use of a classical Volterra integral transformation. Well-posedness of the system of kernel equations, which is the main technical challenge of this chapter, is proved by a method of successive approximations using a novel recursive bound. Moreover, we design a stabilizing full-state feedback law for heterodirectional systems. Along with a dual observer design, this yields an output feedback controller amenable to implementation. Besides, we solve the problem of boundary motion planning for homodirectional systems. Our results in Chapter 3 not only give a positive answer to Question (iii) and the open question proposed in [26] and [31], but also open the door for a large number of related problems to be solved, e.g. collocated observer design, disturbance rejection, similarly to [1], parameter identification as in [30], stabilization of quasilinear systems as in [26], which directly results in the research activities in Chapter 4.

As an apparent generalization of Chapter 3, in Chapter 4, we present the results of our paper [53], which deal with the problem of boundary stabilization of first-order  $n \times n$  inhomogeneous quasilinear hyperbolic systems (2.1). Based on the results of the linear case [54] and Chapter 3, we use the linearized feedback control to stabilize the nonlinear system as it is mentioned in [26]. Although the target system is a little different from the one in [26] with a linear term involved in the equations, thanks to its special structure, we still find that the natural control Lyapunov functions do lead to arbitrarily large exponential decay rate to the target system (2.1) with more technical developments than that was used in [26]. The main result of this chapter supplements the previous works on how to design multi-boundary feedback controllers to realize (rapid) exponential stability of the original nonlinear system in the spatial  $H^2$  sense. Therefore, Part II completely answers the Question (iii).

### 3.3 Main Work in Part III

In Part III (Chapters 5-6-7-8), we focus on the study of a kind of weak boundary controllability, i.e. exact boundary synchronization, for a coupled system of wave equations (2.20) and its quasilinear form, which, to our best knowledge, is first introduced to the field of PDEs.

In Chapter 5, based on the theory of semi-global classical solutions to quasilinear hyperbolic systems, we first recall our results in [55] that apply a unified constructive method to establish the local exact boundary (null) controllability and the local boundary (weak) observability for the following coupled system of 1-D quasilinear wave equations

$$\frac{\partial^2 w_i}{\partial t^2} - a_i^2(w) \frac{\partial^2 w_i}{\partial x^2} + \sum_{j=1}^n a_{ij}(w) w_j = 0, \qquad (3.1)$$

where  $w = (w_1, \ldots, w_n)^T$  is the unknown vector function of (t, x),  $a_i(w)$  and  $a_{ij}(w)$  $(i, j = 1, \cdots, n)$  are all  $C^1$  functions of w on the domain under consideration, satisfying

$$a_i(0) > 0, \quad i = 1, \cdots, n.$$
 (3.2)

Four kinds of boundary conditions, i.e. Dirichlet type, Neumann type, Coupled third type and Coupled dissipative type are considered. The conclusions we obtain in Chapter 5 provide significant basis for studying the exact boundary synchronization for a coupled system of wave equations.

In Chapter 6 and Chapter 7, inspired by the natural phenomena of synchronization and combining the null controllability results in Chapter 5, several kinds of exact synchronizations and the generalized exact synchronization are introduced for a coupled system of 1-D wave equations (2.1) with boundary conditions of Dirichlet type, Neumann type, Coupled third type and Coupled dissipative type, respectively. Our results shown in Chapter 6 (see also [81]) indicate that all these synchronizations can be realized by means of less boundary controls with the boundary conditions of Dirichlet type, Neumann type and Coupled third type, while similar results are proven in Chapter 7 (see also [57]) for the hyperbolic system (2.1) with boundary conditions of coupled dissipative type. It should be noted that sufficient and necessary conditions are obtained in Chapter 6 for the cases of exact boundary synchronization and exact boundary null controllability and synchronization by groups for coupled wave systems (2.1) with the first three boundary conditions.

In Chapter 8, we consider the following general kind of coupled system of wave equations

$$\frac{\partial^2 U}{\partial t^2} - a^2 (U, U_t, U_x) \frac{\partial^2 U}{\partial x^2} + A(U, U_t, U_x) U = 0, \qquad (3.3)$$

where,  $U = (u_1, \ldots, u_N)^T$  is an unknown vector function of (t, x),  $A(U, U_t, U_x) = (a_{ij}(U, U_t, U_x))$  is an  $N \times N$  coupling matrix, whose elements are  $C^1$  functions with respect to their arguments.  $a(U, U_t, U_x)$ , the common velocity for the coupled system, is also a  $C^1$  function with respect to its arguments, satisfying

$$a(0,0,0) > 0. (3.4)$$

As being said before, it is a significant issue from both theoretical and practical points of view to see if we can shorten the number of controls for the coupled systems of wave equations (3.3). Based on the theory of semi-global classical  $C^2$  solution for coupled wave equations, using a unified constructive method for one-sided exact boundary controllability (see [55], [74], [75], [72]), by a suitable iteration procedure as well as certain estimates for coupled systems of wave equations with afore mentioned four kinds of boundary conditions, we establish its local exact boundary synchronization around U = 0. The main work in this chapter is also presented in [56].

# Part I

# Exact Boundary Controllability for Hyperbolic Balance Laws

# Chapter 1

# Sharp Boundary Control Time Estimates for 1-D Homogeneous Quasilinear Hyperbolic Systems

### Contents

1.1 Introduction and Main Results	<b>25</b>
1.2 Some Estimates on Linear Hyperbolic Systems with Non- linear Boundary Conditions	33
1.3 Proof of Main results	35
1.3.1 Proof of Theorem 1.1.1 $\ldots$	35
1.3.2 Proof of Theorem 1.1.2 $\ldots$	37
1.4 The Existence of Operator $\mathfrak{F}$	39
<b>1.5</b> Appendix: Proof of Lemma 1.2.1	46

## **1.1** Introduction and Main Results

Consider the following 1-D quasilinear system

$$\frac{\partial u}{\partial t} + A(u)\frac{\partial u}{\partial x} = 0 \quad (t, x) \in [0, T] \times [0, L], \tag{1.1.1}$$

where,  $u = (u_1, \ldots, u_n)^T$  is a vector function of (t, x), A(u) is an  $n \times n$  matrix with  $C^1$  elements  $a_{ij}(u)$   $(i, j = 1, \cdots, n)$ . By the definition of hyperbolicity, A(u) has n real eigenvalues  $\lambda_i(u)$   $(i = 1, \cdots, n)$  and a complete set of left eigenvectors  $l_i(u) =$ 

 $(l_{i1}(u), \cdots, l_{in}(u)) \ (i = 1, \cdots, n):$ 

$$l_i(u)A(u) = \lambda_i(u)l_i(u), \qquad (1.1.2)$$

and, consequently, a complete set of right eigenvectors  $r_i(u) = (r_{1i}(u), \cdots, r_{ni}(u))^T$   $(i = 1, \cdots, n)$ :

$$A(u)r_i(u) = \lambda_i(u)r_i(u). \tag{1.1.3}$$

Without loss of generality, here we assume that

$$l_{ij}(0) = \delta_{ij} \quad (i, j = 1, \cdots, n), \tag{1.1.4}$$

where  $\delta_{ij}$  stands for the Kronecker symbol. Moreover, for simplicity of statement, we suppose that the set of eigenvalues  $\Lambda(u) = (\lambda_1(u), \dots, \lambda_n(u))$  satisfies the following condition:

$$\lambda_1(0) < \lambda_2(0) < \dots < \lambda_m(0) < 0 < \lambda_{m+1}(0) < \dots < \lambda_n(0),$$
 (1.1.5)

and the number of positive eigenvalues is not bigger than the number of negative eigenvalues, i.e.

$$\overline{m} \stackrel{\text{def.}}{=} n - m \le m. \tag{1.1.6}$$

The boundary conditions are given as follows:

$$x = 0: u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \ (s = m + 1, \cdots, n), \tag{1.1.7}$$

$$x = L : u_r = G_r(t, u_{m+1}, \cdots, u_n) + H_r(t) \ (r = 1, \cdots, m), \tag{1.1.8}$$

where  $G_i$  and  $H_i$   $(i = 1, \dots, n)$  are  $C^1$  functions, without loss of generality, such that

$$G_i(t, 0, \cdots, 0) \equiv 0 \quad (i = 1, \cdots, n).$$
 (1.1.9)

Let us first recall the following well-posedness result due to Li (see [72, Chapter 2 on pages 19-36]) for the system (1.1.1) with (1.1.7)-(1.1.8):

**Lemma 1.1.1** Suppose that  $\lambda_i$ ,  $l_i$ ,  $r_i$ ,  $G_i$  and  $H_i$   $(i = 1, \dots, n)$  are all  $C^1$  functions with respect to their arguments and (1.1.4) holds. Suppose furthermore that the conditions of  $C^1$  compatibility are satisfied at the points (t, x) = (0, 0) and (0, L), respectively. For any given  $T_0 > 0$ , the mixed initial-boundary value problem (1.1.1), (1.1.7)-(1.1.8) with

$$t = 0: u = \varphi(x), 0 \le x \le L \tag{1.1.10}$$

admits a unique  $C^1$  solution u(t, x) with sufficiently small  $C^1$  norm on the domain

$$R(T_0) = \{(t, x) | 0 \le t \le T_0, 0 \le x \le L\},$$
(1.1.11)

provided that  $\|\varphi\|_1$  and  $\|H\|_1$  are small enough (depending on  $T_0$ ).

**Remark 1.1.1** Here and throughout this chapter, for simplicity, we denote  $\|\cdot\|_0$ and  $\|\cdot\|_1$  as the corresponding  $C^0$  and  $C^1$  norms, respectively.

**Remark 1.1.2** Boundary conditions (1.1.7)-(1.1.8) are prescribed under the hypothesis (1.1.4). We could also replace  $u_i$  by the  $v_i = l_i(u)u$  in (1.1.7)-(1.1.8) in the general situation (see also [69, Remark 1.3 on page 171]).

**Remark 1.1.3** The  $C^1$  compatibility conditions at the point (t, x) = (0, 0) are given by

$$\varphi_{s}(0) = G_{s}(0,\varphi_{1}(0),\cdots,\varphi_{m}(0)) + H_{s}(0) \quad (s = m + 1,\cdots,n),$$

$$\sum_{j=1}^{n} a_{sj}(\varphi(0))\varphi_{jx}(0) = \sum_{r=1}^{m} \sum_{j=1}^{n} \frac{\partial G_{s}}{\partial u_{r}}(0,\varphi_{1}(0),\cdots,\varphi_{m}(0))a_{rj}(\varphi(0))\varphi_{jx}(0)$$

$$- \frac{\partial G_{s}}{\partial t}(0,\varphi_{1}(0),\cdots,\varphi_{m}(0)) - H_{st}(0) \quad (s = m + 1,\cdots,n).$$
(1.1.13)

The  $C^1$  compatibility conditions at the point (t, x) = (0, L) are similar.

Based on Lemma 1.1.1, we are concerned with the corresponding exact controllability problem for the system (1.1.1) and (1.1.7)-(1.1.8):

**Problem (EBC).** Let  $\mathfrak{C} \subseteq \{1, \dots, n\}$ . For any given initial data  $\varphi \in (C^1[0, L])^n$ and final date  $\psi \in (C^1[0, L])^n$  with small  $C^1$  norm, can we find a time T > 0 and boundary controls  $H_i \in C^1[0, T] (i \in \mathfrak{C})$  (for  $i \notin \mathfrak{C}$ ,  $H_i(t)$  is given) with small  $C^1$ norm, such that the mixed initial-boundary value problem (1.1.1), (1.1.7)-(1.1.8) and (1.1.10) admits a  $C^1$  solution u(t, x) on the domain

$$R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\},$$
(1.1.14)

which satisfies exactly the final condition

$$t = T : u = \psi(x), 0 \le x \le L? \tag{1.1.15}$$

The exact boundary controllability problems for both linear and quasilinear 1-D hyperbolic systems have attracted considerable attention over the past several decades (see, for example, [19, 72, 100] and the numerous references therein). Eventually,

three cases have been distinguished in the literatures, depending on the number of boundary controls, the boundary conditions on non-control side as well as the control time, to study **Problem (EBC)**(see, for instance, [72, Chapter 3 on pages 37-62] and [75, 120]):

(1) Both-sided control ( $\mathfrak{C} = \{1, \dots, n\}$ ). One can choose

$$T > L \max\left(\frac{1}{|\lambda_m(0)|}, \frac{1}{\lambda_{m+1}(0)}\right) \text{ and } H_i(i=1,\cdots,n)$$
 (1.1.16)

to realize the (local) exact boundary controllability for the general first order 1-D quasilinear hyperbolic system (1.1.1), (1.1.7)-(1.1.8). Since enough boundary controls are acting on both ends, we can drive any initial data  $\varphi$  to any desired data  $\psi$  around equilibrium without any additional assumptions to the boundary conditions (1.1.7)-(1.1.8).

(2) **One-sided control (** $\mathfrak{C} = \{1, \dots, m\}$ **).** In this situation, in order to handle the **Problem (EBC)**, one should utilize all the boundary functions  $H_r(t)(r = 1, \dots, m)$  as boundary controls on x = L, meanwhile, suppose that the boundary condition (1.1.7) on the non-control side x = 0 in a neighborhood of u = 0 satisfies

$$u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \quad (s = m + 1, \cdots, n)$$
  
$$\Leftrightarrow u_{\overline{r}} = \overline{G_{\overline{r}}}(t, u_{\overline{m}+1}, \cdots, u_m, u_{m+1}, \cdots, u_n) + \overline{H_{\overline{r}}}(t) \quad (\overline{r} = 1, \cdots, \overline{m}),$$
(1.1.17)

where  $\overline{G}_{\overline{r}}$  and  $\overline{H}_{\overline{r}}$  are also  $C^1$  functions with respect to their arguments, and

$$\overline{G}_{\overline{r}}(t,0,\cdots,0) \equiv 0 \quad (\overline{r}=1,\cdots,\overline{m}), \tag{1.1.18}$$

provided that the control time satisfies

$$T > L \left( \frac{1}{|\lambda_m(0)|} + \frac{1}{\lambda_{m+1}(0)} \right).$$
(1.1.19)

(3) Both-sided with less controls ( $\mathfrak{C} = \{\overline{m} + 1, \dots, n\}$ ). In this case, under the same assumption (1.1.19) as in the case of "One-sided control", one needs all (i.e.  $\overline{m}$ ) boundary controls  $H_s(s = m + 1, \dots, n)$  acting on x = 0 together with  $m - \overline{m}$  boundary controls  $H_q(=\overline{m} + 1, \dots, m)$  acting on x = L, provided that in a neighborhood of u = 0 the first  $\overline{m}$  boundary conditions (1.1.8), the non-control components, on x = L satisfies

$$u_{\overline{r}} = G_{\overline{r}}(t, u_{m+1}, \cdots, u_n) + H_{\overline{r}}(t) \quad (\overline{r} = 1, \cdots, \overline{m})$$
  
$$\Leftrightarrow u_s = \overline{G}_s(t, u_1, \cdots, u_{\overline{m}}) + \overline{H}_s(t) \quad (s = m+1, \cdots, n),$$
  
(1.1.20)

where  $\overline{G}_s$  and  $\overline{H}_s$  are also  $C^1$  functions with respect to their arguments, and

$$\overline{G}_s(t, 0, \cdots, 0) \equiv 0 \quad (s = m + 1, \cdots, n).$$
 (1.1.21)

Compared with the case "Both-sided control", the latter two cases reduce the number of boundary controls to m, however, at the cost of the larger control time. It is easy to show that the control time given by (1.1.16) is sharp for the first case (see [72, Remark 3.11 on page 47]). However, in the cases of "One-sided control" and "Both-sided with less controls", one verifies that the control time (1.1.19) can be improved for linear hyperbolic systems (see Remark 1.1.4). Taking into account this phenomenon, one may ask the following natural question: is it still valid for the general nonlinear systems? Here, we give a positive answer.

In this chapter, based on the constructive methods, the characteristic theory together with a suitable iteration scheme on a suitable time interval, we shorten the control time for the quasilinear hyperbolic systems (1.1.1) with boundary controls acting on one side or both sides with less controls that was proposed by Li [72, Theorem 3.2 and Theorem 3.3 on pages 39-41] and Zhang [120]. Our main results, which can be applied to control Saint-Venant-Exner system (see [32]), the full Euler equation for 1-D gas dynamics ([23, section 4])etc., are given as follows:

**Theorem 1.1.1** (One-sided control) Let us assume that the basic hypotheses of Lemma 1.1.1 hold and (1.1.17)-(1.1.18) is valid in a neighborhood of u = 0. Let T > 0 satisfies

$$T > L \max\left(\frac{1}{|\lambda_{\overline{m}}(0)|} + \frac{1}{\lambda_{m+1}(0)}, \frac{1}{|\lambda_m(0)|}\right).$$
(1.1.22)

Then there exists  $\varepsilon > 0$  such that for any given initial data  $\varphi \in (C^1[0,L])^n$ , any boundary functions  $H_s \in C^1[0,T](s=m+1,\cdots,n)$ ,  $\overline{H}_{\overline{r}} \in C^1[0,T]$  ( $\overline{r} = 1,\cdots,\overline{m}$ ) and any  $\psi \in (C^1[0,L])^n$  satisfying the  $C^1$  compatibility conditions at the point (t,x) =(0,0) and (T,0) respectively, and

$$\|\varphi\|_{1} + \|\psi\|_{1} + \max_{\overline{r}=1,\cdots,\overline{m}} \|\overline{H}_{\overline{r}}\|_{1} + \max_{s=m+1,\cdots,n} \|H_{s}\|_{1} \le \varepsilon,$$
(1.1.23)

there exist boundary controls  $H_r \in C^1[0,T]$   $(r = 1, \dots, m)$  on x = L, such that the mixed initial-boundary value problem (1.1.1), (1.1.7)-(1.1.8) and (1.1.10) admits a unique  $C^1$  solution u = u(t,x) on the domain R(T), which satisfies the final condition (1.1.15).

**Theorem 1.1.2** (Both-sided with less controls) Let us assume that the basic hypotheses of Lemma 1.1.1 hold and (1.1.20)-(1.1.21) is valid in a neighborhood of u = 0. Furthermore, let T > 0 satisfies the sharp time estimate (1.1.22). Then there exists  $\varepsilon > 0$  such that for any given initial data  $\varphi \in (C^1[0, L])^n$ , any boundary functions  $H_{\overline{r}} \in C^1[0,T]$  ( $\overline{r} = 1, \dots, \overline{m}$ ),  $\overline{H}_s \in C^1[0,T](s = m+1,\dots,n)$  and any  $\psi \in (C^1[0,L])^n$  satisfying the  $C^1$  compatibility conditions at the point (t,x) = (0,L) and (T,L) respectively, and

$$\|\varphi\|_{1} + \|\psi\|_{1} + \max_{\overline{r}=1,\cdots,\overline{m}} \|H_{\overline{r}}\|_{1} + \max_{s=m+1,\cdots,n} \|\overline{H}_{s}\|_{1} \le \varepsilon,$$
(1.1.24)

there exist boundary controls  $H_p \in C^1[0,T](p = \overline{m} + 1, \dots, n)$ , such that the mixed initial-boundary value problem (1.1.1), (1.1.7)-(1.1.8) and (1.1.10) admits a unique  $C^1$  solution u = u(t,x) on the domain R(T), which satisfies the final condition (1.1.15).

In general, when considering our controllability problem for quasilinear hyperbolic systems, the difficulties usually come from two aspects. One is the nonlinearity from the boundary conditions, and the other is that from the governing matrix A(u). To overcome the first difficulty, we use the so-called constructive methods (see [72, 75, 111, 120]). It allows us to find a  $C^1$  solution u to the system (1.1.1) on the domain R(T), satisfying simultaneously any given initial condition (1.1.10) and any given final condition (1.1.15), as well as all the boundary conditions without control (i.e. (1.1.17) for the case "One-sided control" or (1.1.20) for the case "Both-sided with less controls"). On the other hand, one easily checks from (1.1.17) and (1.1.20) that the components  $u_q(q = \overline{m} + 1, \cdots, m)$  are in essence decoupled from  $u_p(p = 1, \cdots, \overline{m}, m + m)$  $1, \dots, n$  on these boundary conditions. Consequently the main difficulty of this chapter is from the second one, i.e. how to decouple these two parts of components from the governing matrix A(u) on the whole domain R(T). If so, we then respectively control the two different hyperbolic systems, the first are on  $u_q(q = \overline{m} + 1, \cdots, m)$ and the second are on  $u_p(p=1,\cdots,\overline{m},m+1,\cdots,n)$ . Therefore, the control time for the original system (1.1.1) is much smaller than the one in [72, Theorem 3.2 and Theorem 3.3 on pages 39-41] and [120]. To overcome the second difficulty, we will see below that the method used is strongly inspired from a suitable linearized iteration scheme on a suitable time interval. Then based on theory of the linearized hyperbolic systems with nonlinear boundary conditions, not only can we shorten the control time for the linearized system, but also prove the convergence by using constructive methods as well as some appropriate estimates.

**Remark 1.1.4** In this remark, we first show that the time estimates provided in Theorem 1.1.1 and Theorem 1.1.2 improves the one in [72, Theorem 3.2 and Theorem 3.3 on pages 39-41] and [120]. In fact, noting (1.1.5), one has

$$L\max\left(\frac{1}{|\lambda_{\overline{m}}(0)|} + \frac{1}{\lambda_{m+1}(0)}, \frac{1}{|\lambda_m(0)|}\right) \le L\left(\frac{1}{|\lambda_m(0)|} + \frac{1}{\lambda_{m+1}(0)}\right).$$
 (1.1.25)

Moreover, the two sides are equal if and only if  $\overline{m} = m$ . If  $\overline{m} < m$ , one can use less time given by (1.1.22) to control the nonlinear system (1.1.1). In this sense, our results indeed improve the one in [72, Theorem 3.2 and Theorem 3.3 on pages 39-41] and [120]. Next, we claim that the exact controllability time T > 0 given by (1.1.22) is sharp for the above two cases. Consider the following linear system

$$\frac{\partial u_i}{\partial t} + \lambda_i \frac{\partial u_i}{\partial x} = 0, \qquad (1.1.26)$$

with constant characteristic speeds  $\lambda_i$   $(i = 1, \dots, n)$  satisfying

$$\lambda_1 < \dots < \lambda_m < 0 < \lambda_{m+1} < \dots < \lambda_n. \tag{1.1.27}$$

(I) (One-sided Control) For the boundary conditions

$$x = 0: u_s = u_{n+1-s} \quad (s = m+1, \cdots, n), \tag{1.1.28}$$

$$x = L : u_r = H_r(t) \quad (r = 1, \cdots, m),$$
 (1.1.29)

when (1.1.6) holds, the whole system (1.1.26) can be divided into two systems, the one is

$$\begin{cases} \frac{\partial u_p}{\partial t} + \lambda_p \frac{\partial u_p}{\partial x} = 0 \quad (p = 1, \cdots, \overline{m}, m + 1, \cdots, n), \\ x = 0 : u_s = u_{n+1-s} \quad (s = m + 1, \cdots, n), \\ x = L : u_{\overline{r}} = H_{\overline{r}}(t) \quad (\overline{r} = 1, \cdots, \overline{m}), \end{cases}$$
(1.1.30)

and the other is

$$\begin{cases} \frac{\partial u_q}{\partial t} + \lambda_q \frac{\partial u_q}{\partial x} = 0 \quad (q = \overline{m} + 1, \cdots, m), \\ x = L : u_q = H_q(t) \quad (q = \overline{m} + 1, \cdots, m). \end{cases}$$
(1.1.31)

Noting (1.1.27), it is well known that the system (1.1.30) is controllable by boundary controls  $H_{\overline{r}}(\overline{r} = 1, \dots, \overline{m})$  if and only if

$$T > \frac{L}{|\lambda_{\overline{m}}|} + \frac{L}{\lambda_{m+1}}.$$
(1.1.32)

While the system (1.1.31) is controllable by boundary controls  $H_q(q = \overline{m} + 1, \dots, m)$ if and only if

$$T > \frac{L}{|\lambda_m|}.\tag{1.1.33}$$

Therefore, the system (1.1.26), (1.1.28)-(1.1.29) is controllable by the boundary con-

trols  $H_r(r = 1, \cdots, m)$  if and only if

$$T > L \max\left(\frac{1}{|\lambda_{\overline{m}}|} + \frac{1}{\lambda_{m+1}}, \frac{1}{|\lambda_m|}\right).$$
(1.1.34)

### (II) (Both-sided with less controls) For the boundary conditions

$$x = 0: u_s = H_s(t) \quad (s = m + 1, \cdots, n),$$
 (1.1.35)

$$x = L : \begin{cases} u_{\overline{r}} = u_{n+1-\overline{r}} & (\overline{r} = 1, \cdots, \overline{m}), \\ u_q = H_q(t) & (q = \overline{m} + 1, \cdots, m), \end{cases}$$
(1.1.36)

noting (1.1.6), the whole system (1.1.26) can be also divided into two systems, the one is

$$\begin{cases} \frac{\partial u_p}{\partial t} + \lambda_p \frac{\partial u_p}{\partial x} = 0 \quad (p = 1, \cdots, \overline{m}, m + 1, \cdots, n), \\ x = 0 : u_s = H_s(t) \quad (s = m + 1, \cdots, n), \\ x = L : u_{\overline{r}} = u_{n+1-\overline{r}} \quad (\overline{r} = 1, \cdots, \overline{m}), \end{cases}$$
(1.1.37)

and the other is (1.1.31). An argument similar to the one used in (I) shows that the system (1.1.26), (1.1.35)-(1.1.36) is controllable by the boundary controls  $H_p(p = \overline{m} + 1, \dots, n)$  if and only if (1.1.34) holds.

**Remark 1.1.5** We could also treat the case with source term (see, in particular, Remark 1.3.2 below):

$$\frac{\partial u}{\partial t} + A(u)\frac{\partial u}{\partial x} = F(u), \qquad (1.1.38)$$

in which  $F \in C^1$  and

$$\lim_{|u| \to 0} \frac{|F(u)|}{|u|} = 0.$$
(1.1.39)

However, if (1.1.39) is not satisfied, although one can refer [72, Chapter 3 on pages 37-62] and [120] to treat this case by larger control time (i.e. (1.1.19)), the approach of this chapter is not valid anymore. Can we still shorten control time as in (1.1.22) for such systems? To our best knowledge, this question remains open.

**Remark 1.1.6** The condition (1.1.5) implies that (1.1.1) is a strictly hyperbolic system around the equilibrium. For the more general case

$$\lambda_r(0) < 0 < \lambda_s(0) \quad (r = 1, \cdots, m, s = m + 1, \cdots, n),$$
 (1.1.40)

Theorem 1.1.1 and Theorem 1.1.2 still hold if we replace (1.1.22) by

$$T > L \max\left(\max_{\overline{r}=1,\cdots,\overline{m}} \frac{1}{|\lambda_{\overline{r}}(0)|} + \max_{s=m+1,\cdots,n} \frac{1}{\lambda_{s}(0)}, \max_{q=\overline{m}+1,\cdots,n} \frac{1}{|\lambda_{q}(0)|}\right).$$
(1.1.41)

The rest of this chapter is organized as follows: In §1.2, we show the  $C^0$ ,  $C^1$  and modulus of continuity estimates for general linear hyperbolic systems with nonlinear boundary conditions, which is proved in §1.5. In §1.3, we use a suitable iteration scheme together with fixed point theory to prove Theorem 1.1.1 and Theorem 1.1.2. In §1.4, we establish a technical result on controllability of the linearized system by using constructive methods.

## **1.2** Some Estimates on Linear Hyperbolic Systems with Nonlinear Boundary Conditions

In order to obtain the desired exact controllability for the nonlinear system (1.1.1), (1.1.7)-(1.1.8), we have to investigate  $C^0$ ,  $C^1$  norms and the modulus of continuity for the corresponding linear problem. Here, we introduce the modulus of continuity for f(t, x) to be the following non-negative function:

$$\omega(\eta|f) := \sup_{\substack{|t'-t''| \le \eta, \ |x'-x''| \le \eta \\ (t',x'), \ (t'',x'') \in R(T)}} |f(t',x') - f(t'',x'')|, \quad \eta \ge 0.$$
(1.2.1)

Similarly, we can define the modulus of continuity of a vector function  $f = (f_1, \dots, f_n)$  as

$$\omega(\eta|f) := \max_{i=1,\cdots,n} \omega(\eta|f_i), \qquad (1.2.2)$$

and the modulus of continuity of a matrix function  $A = (a_{ij})_{n \times n}$  can be

$$\omega(\eta|A) := \max_{i,j=1,\cdots,n} \omega(\eta|a_{ij}).$$
(1.2.3)

Consider the linear hyperbolic system as follows (see also [85, 111]):

$$\frac{\partial u_i}{\partial t} + \lambda_i(t,x)\frac{\partial u_i}{\partial x} = \sum_{k=1}^n b_{ik}(t,x) \Big(\frac{\partial c_k(t,x)}{\partial t} + \lambda_i(t,x)\frac{\partial c_k(t,x)}{\partial x}\Big),\tag{1.2.4}$$

where  $u = (u_1, \dots, u_n)^T$  is an unknown vector function of (t, x), we assume that  $\lambda_i, b_{ik}, c_k (i = 1, \dots, n; k = 1, \dots, \tilde{n})$  are all  $C^1$  functions with respect to their arguments. Moreover, without of generality, we assume also that, on the domain under

consideration, the set of eigenvalues  $\Lambda := \{\lambda_1(t, x), \dots, \lambda_n(t, x)\}$  satisfies the following condition:

$$\lambda_1(t,x) < \dots < \lambda_m(t,x) < 0 < \lambda_{m+1}(t,x) < \dots < \lambda_n(t,x), (t,x) \in R(T).$$
 (1.2.5)

Suppose furthermore that

$$\|\Lambda\|_1 + \|G_u\|_0 \le K < +\infty, \tag{1.2.6}$$

where K is a positive constant and

$$\|G_u\|_0 := \max_{\substack{r=1,\cdots,m\\s=m+1,\cdots,n}} \left\{ \left\| \frac{\partial G_r}{\partial u_s} \right\|_0, \left\| \frac{\partial G_s}{\partial u_r} \right\|_0 \right\}.$$
 (1.2.7)

For the mixed initial-boundary value problem (1.2.4) with the initial condition (1.1.10) and the nonlinear boundary conditions (1.1.7)-(1.1.8), where  $G_i (i = 1, \dots, n)$  and  $H_i$  are defined in §1.1, according to the characteristic theory for hyperbolic system (see [56, 84]), we have the following lemma (the details of the proof can be seen in §1.5):

**Lemma 1.2.1** Suppose  $G_i$  and  $H_i(i = 1, \dots, n)$  are  $C^1$  functions and (1.1.9) together with (1.2.5) hold. Suppose furthermore that the conditions of  $C^1$  compatibility are satisfied at the points (t, x) = (0, 0) and (0, L), respectively. For any given T > 0, the forward mixed initial-boundary problem (1.2.4), (1.1.7)-(1.1.8) and (1.1.10) admits a unique  $C^1$  solution u = u(t, x) on the domain  $R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\}$ , which depends continuously on  $(\varphi, \Lambda, B, c)$ . Moreover, we have

$$||u||_{0} \leq C_{1}(T, K)(||\varphi||_{0} + ||H||_{0} + ||B||_{0}||c||_{0}), \qquad (1.2.8)$$

$$|u||_{1} \leq C_{2}(T, K)(||\varphi||_{1} + ||H||_{1} + ||B||_{1}||c||_{1} + ||G_{t}||_{0}), \qquad (1.2.9)$$

where  $B := (b_{ik}), c := (c_1, \cdots, c_{\widetilde{n}}), H := (H_1, \cdots, H_n)$  and

$$\|G_t\|_0 := \max_{\substack{r=1,\cdots,m\\s=m+1,\cdots,n}} \left\{ \left\| \frac{\partial G_r}{\partial t} \right\|_0, \left\| \frac{\partial G_s}{\partial t} \right\|_0 \right\}.$$
 (1.2.10)

In addition, if

$$0 < \eta < \min_{i=1,\cdots,n} \inf_{(t,x)\in R(T)} \frac{L}{|\lambda_i(t,x)|},$$
(1.2.11)

one gets

$$\omega(\eta|u_t) + \omega(\eta|u_x) \le \Omega(\eta), \qquad (1.2.12)$$

where

$$\Omega(\eta) = C_3(T, K) \bigg( \omega(\eta | \varphi_x) + \|B\|_1 \big( \omega(\eta | c_t) + \omega(\eta | c_x) \big) + \|c\|_1 \big( \omega(\eta | B_t) + \omega(\eta | B_x) \big) \\
+ \big( 1 + \|\varphi\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \big) \big( \eta + \big( 1 + \|\varphi\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \big) \omega(\eta | \Gamma) \big) \\
+ \big( \|\varphi\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \big) \omega(\eta | \Lambda_x) + \omega(\eta | H_t) \bigg),$$
(1.2.13)

in which  $\Gamma$  denotes the following set of functions

$$\Gamma := \left\{ \frac{\partial G_r}{\partial u_s}, \frac{\partial G_r}{\partial t}, \frac{\partial G_s}{\partial u_r}, \frac{\partial G_s}{\partial t}, \begin{array}{l} r = 1, \cdots, m;\\ s = m+1, \cdots, n \end{array} \right\}.$$
(1.2.14)

**Remark 1.2.1**  $\omega(\eta|\Gamma)$  denotes the modulus of continuity for  $\Gamma$  on the domain  $\widetilde{R} = \{0 \le t \le T, 0 \le ||u||_0 \le C_1(T, K)(||\varphi||_0 + ||H||_0 + ||B||_0||c||_0)\}.$ 

**Remark 1.2.2** Different from the usual hyperbolic systems, the source term of (1.2.4) is essentially the directional derivative with respect to t along the i-th characteristic  $\frac{dx}{dt} = \lambda_i(t, x)$ . One can refer [85, 111] to the well-posedness of this kind of (nonlinear) hyperbolic systems. In this chapter, however, we still show the detail proof of Lemma 1.2.1 in order to (1) emphasize the explicit estimate on the matrix B, the smallness of which will be utilized when iteration in § 1.4; (2) obtain inequality (1.2.12) for the modulus of continuity under nonlinear boundary conditions (1.1.7)-(1.1.8) on the whole domain R(T) (not locally), which is not involved in previous references and will be used to prove the existence of fixed point below; (3) show the solution u depends continuously on ( $\varphi, \Lambda, B, c$ ), which guarantees the continuity of our iteration scheme in § 1.4.

### **1.3** Proof of Main results

### 1.3.1 Proof of Theorem 1.1.1

Since one can recover the boundary controls as the traces of u afterwards, in order to get Theorem 1.1.1, it suffices to establish the following

**Lemma 1.3.1** Under the assumptions of Theorem 1.1.1, let T > 0 be defined by (1.1.22). For any given initial data  $\varphi \in (C^1[0,L])^n$ , final data  $\psi \in (C^1[0,L])^n$ and the functions  $H_s \in C^1[0,T]$  ( $s = m + 1, \dots, n$ ) satisfying (1.1.23) and the  $C^1$ compatibility conditions at the points (t,x) = (0,0) and (T,0), respectively, system (1.1.1) together with the boundary conditions (1.1.7) on x = 0 admits a  $C^1$  solution u = u(t,x) on the domain R(T), which satisfies simultaneously the initial condition (1.1.10) and the final condition (1.1.15).
**Proof of Lemma 1.3.1** Based on the results in  $\S2$ , we next utilize a suitable iterative scheme, the constructive methods (see, in particular, [72, Proof of Theorem 3.2 on pages 47-52] and [111]) together with the fixed point theory (see, for instance, [19, Chapter 4 on pages 159-180]) to prove Lemma 1.3.1. Firstly, multiplying the *i*-th left eigenvector  $l_i(u)$   $(i = 1, \dots, n)$  on system (1.1.1) from the left and noting (1.1.2), we get the characteristic form of hyperbolic system (1.1.1):

$$l_i(u)\left(\frac{\partial u}{\partial t} + \lambda_i(u)\frac{\partial u}{\partial x}\right) = 0 \quad (i = 1, \cdots, n).$$
(1.3.1)

On the domain R(T), we introduce some sets of function

$$\Sigma_{1} := \left\{ y(t,x) \in (C^{1}[R(T)])^{n} \middle| \begin{array}{l} y(0,x) = \varphi(x), y(T,x) = \psi(x); \\ y_{t}(0,0) = -A(\varphi(0))\varphi_{x}(0); \\ y_{t}(0,L) = -A(\varphi(L))\varphi_{x}(L); \\ y_{t}(T,0) = -A(\psi(0))\psi_{x}(0); \\ y_{t}(T,L) = -A(\psi(L))\psi_{x}(L); \end{array} \right\}, \quad (1.3.2)$$

$$\Sigma_2 := \left\{ y(t,x) \in \Sigma_1 \big| \|y\|_0 \le C_4 \varepsilon, \|y\|_1 \le \kappa(\varepsilon) \right\},$$

$$(1.3.3)$$

$$\Sigma_3 := \left\{ y(t,x) \in \Sigma_2 \big| \omega(\eta|y_t) + \omega(\eta|y_x) \le \Omega_1(\eta) \right\},\tag{1.3.4}$$

Here and in what follows, for simplicity, when no confusion is possible, we put  $C_i(i =$  $(4, 5, \cdots)$  as positive constants which are independent of y, while  $\kappa(\varepsilon)$  and  $\Omega_1(\eta)(0 \le \varepsilon)$  $\eta \leq \min_{i=1,\cdots,n} \inf_{||u||_0 \leq C_4 \varepsilon} \frac{L}{|\lambda_i(u)|}$  denote two continuous nonnegative functions with respect

to their arguments, satisfying

$$\kappa(0) = 0, \tag{1.3.5}$$

and

$$\Omega_1(0) = 0. \tag{1.3.6}$$

For any given  $y \in \Sigma_2$ , we consider the following linear system

$$\begin{cases} \frac{\partial u_i}{\partial t} + \lambda_i(y) \frac{\partial u_i}{\partial x} = \sum_{k=1}^n \Theta_{ik}(y) \left(\frac{\partial y_k}{\partial t} + \lambda_i(y) \frac{\partial y_k}{\partial x}\right) & (i = 1, \cdots, n), \quad (1.3.7) \end{cases}$$

$$x = 0: u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \quad (s = m + 1, \cdots, n),$$
(1.3.8)

where

$$\Theta_{ik}(y) := \begin{cases} -\frac{l_{ik}(y)}{l_{ii}(y)}, & \text{if } k \neq i, \\ 0, & \text{if } k = i \end{cases}.$$
 (1.3.9)

It is easy to see that  $\Theta := (\Theta_{ik})_{n \times n}$  is a  $C^1$  matrix function of y, and

$$\Theta(0) = 0. \tag{1.3.10}$$

In addition, if  $y \in \Sigma_1$ , from(1.3.7), one can check that

$$u_t(0,0) = y_t(0,0) = -A(\varphi(0))\varphi_x(0),$$
  

$$u_t(0,L) = y_t(0,L) = -A(\varphi(L))\varphi_x(L),$$
  

$$u_t(T,0) = y_t(T,0) = -A(\psi(0))\psi_x(0),$$
  

$$u_t(T,L) = y_t(T,L) = -A(\psi(L))\psi_x(L),$$
  
(1.3.11)

provided  $u(0,x) = \varphi(x)$  and  $u(T,x) = \psi(x)$ . Therefore, the problem becomes whether there exist a positive constant  $C_4$  and two continuous nonnegative value functions  $\kappa(\varepsilon)$  and  $\Omega_1(\eta)$  satisfying (1.3.5)-(1.3.6), such that

(a)We can find a continuous iterative operator  $\mathfrak{F}: \Sigma_2 \to \Sigma_2$ :

$$u = \mathfrak{F}(y), \tag{1.3.12}$$

which satisfies

$$u \in \Sigma_3, \quad \text{if } y \in \Sigma_3;$$
 (1.3.13)

(b) (1.3.7)-(1.3.8) hold under the hypothesis (1.1.22).

In fact, by Ascoli's theorem,  $\Sigma_3$  is a compact subset of  $\Sigma_2$  (see [19, Page 170]). Clearly  $\Sigma_3$  is convex, then (1.3.13) imply that

$$\mathfrak{F}(\Sigma_3) \subset \Sigma_3. \tag{1.3.14}$$

By the Schauder fixed-point theorem, we know  $\mathfrak{F}$  has a fixed point. Thus, when T > 0 satisfies (1.1.22), there exists a  $C^1$  function u on the domain R(T) such that

$$\begin{cases} l_i(u)(u_t + \lambda_i(u)u_x) = 0, \ (t, x) \in R(T), \\ t = 0: \ u = \varphi(x), \\ t = T: \ u = \psi(x), \\ x = 0: u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \ (s = m + 1, \cdots, n). \end{cases}$$
(1.3.15)

This concludes the proof of Lemma 1.3.1, assuming the existence of operator  $\mathfrak{F}$ .

#### 1.3.2 Proof of Theorem 1.1.2

Assuming the existence of operator  $\mathfrak{F}$ , it is easy to see that we can find another continuous operator  $\mathfrak{B}: \Sigma_2 \to \Sigma_2$ , such that

(aa)

$$u = \mathfrak{B}(y), \tag{1.3.16}$$

which satisfy

$$u \in \Sigma_3, \quad \text{if } y \in \Sigma_3;$$
 (1.3.17)

(bb) under the hypothesis (1.1.22), we have

$$\left(\frac{\partial u_i}{\partial t} + \lambda_i(y)\frac{\partial u_i}{\partial x} = \sum_{k=1}^n \Theta_{ik}(y)\left(\frac{\partial y_k}{\partial t} + \lambda_i(y)\frac{\partial y_k}{\partial x}\right) \quad (i = 1, \cdots, n), \quad (1.3.18)$$

$$x = L : u_{\overline{r}} = G_{\overline{r}}(t, u_{m+1}, \cdots, u_n) + H_{\overline{r}}(t) \quad (\overline{r} = 1, \cdots, \overline{m}).$$
(1.3.19)

An argument similar to the one used in § 1.3.1 shows that  $\mathfrak{B}$  has a fixed point. When T > 0 satisfies (1.1.22), there exists a  $C^1$  function u on the domain R(T) such that

$$\begin{cases} l_i(u)(u_t + \lambda_i(u)u_x) = 0, \ (t, x) \in R(T), \\ t = 0: \ u = \varphi(x), \\ t = T: \ u = \psi(x), \\ x = L: u_{\overline{r}} = G_{\overline{r}}(t, u_{m+1}, \cdots, u_n) + H_{\overline{r}}(t) \ (\overline{r} = 1, \cdots, \overline{m}). \end{cases}$$
(1.3.20)

This concludes the proof of Theorem 1.1.2, assuming the existence of operator  $\mathfrak{F}$ .

**Remark 1.3.1** The purposes of our iteration (1.3.7)-(1.3.8) are (1) to guarantee the boundary conditions without control (1.3.8) are always satisfied. (2) to use the smallness of  $\Theta$  in order to make (a)-(b) work.

**Remark 1.3.2** For the system (1.1.38) with (1.1.39), the iteration scheme is

$$\frac{\partial u_i}{\partial t} + \lambda_i(y)\frac{\partial u_i}{\partial x} = \sum_{j=1}^n \Theta_{ij}(y) \left(\frac{\partial y_j}{\partial t} + \lambda_i(y)\frac{\partial y_j}{\partial x}\right) + \sum_{j=1}^n \frac{l_{ij}(y)}{l_{ii}(y)}F_j(y)$$
(1.3.21)

One can check, if (1.1.39) holds, by a similar manner, Theorem 1.1.1 and Theorem 1.1.2 are still valid. However, if (1.1.39) is not satisfied (for example, F(y) := Ay, where  $A \neq 0$  is a  $n \times n$  matrix with constant elements  $a_{ij}(i, j = 1, \dots, n)$  and  $||A|| := \max_{i,j=1,\dots,n} |a_{ij}|$  is large enough), through our iteration scheme (1.3.21), one can not expect that  $u = \mathfrak{F}(y) \in \Sigma_2$  and (1.3.13) hold.

# 1.4 The Existence of Operator $\mathfrak{F}$

In this section, we establish the existence of mapping  $\mathfrak{F}$  by a modified constructive method. Inspired from the iteration scheme (1.3.7)-(1.3.8), we can divide u into the following two systems:

$$\frac{\partial u_q}{\partial t} + \lambda_q(y)\frac{\partial u_q}{\partial x} = \sum_{k=1}^n \Theta_{qk}(y) \left(\frac{\partial y_k}{\partial t} + \lambda_q(y)\frac{\partial y_k}{\partial x}\right) \quad (q = \overline{m} + 1, \cdots, m), \tag{1.4.1}$$

and

$$\begin{cases} \frac{\partial u_p}{\partial t} + \lambda_p(y) \frac{\partial u_p}{\partial x} = \sum_{k=1}^n \Theta_{pk}(y) \left( \frac{\partial y_k}{\partial t} + \lambda_p(y) \frac{\partial y_k}{\partial x} \right) \ (p = 1, \cdots, \overline{m}, m+1, \cdots, n), \\ x = 0 : u_s = G_s(t, u_1, \cdots, u_m) + H_s(t) \ (s = m+1, \cdots, n). \end{cases}$$
(1.4.2)

Then the operator  $\mathfrak{F}$  can be established by the following three steps:

### Step 1: Control of the system (1.4.1)

In this step, we will construct a  $C^1$  solution  $u_q = u_q(t, x)(q = \overline{m} + 1, \dots, m)$  for the system (1.4.1), which satisfies

$$t = 0: \quad u_q = \varphi_q(x) \quad (q = \overline{m} + 1, \cdots, m), \tag{1.4.3}$$

$$t = T: \quad u_q = \psi_q(x) \quad (q = \overline{m} + 1, \cdots, m). \tag{1.4.4}$$

Before proceeding further, let us claim some uniform estimates. Noting (1.1.22), if  $\varepsilon$  is small enough, we have

$$\int_{0}^{T} \inf_{0 \le x \le L} |\lambda_m(y(t, x))| dt > L.$$
(1.4.5)

Then from the above estimate, if  $\varepsilon$  is small enough, there exist  $T_1^{(y)}$  and  $T_2^{(y)}$  satisfying

$$\int_{0}^{T_{1}^{(y)}} \inf_{0 \le x \le L} |\lambda_{m}(y(t,x))| dt = \int_{T_{2}^{(y)}}^{T} \inf_{0 \le x \le L} |\lambda_{m}(y(t,x))| dt = \frac{L}{2}.$$
 (1.4.6)

By (1.1.22), if  $\varepsilon$  is small enough, one can easily show that

$$0 < T_1^{(y)} < \frac{T-\delta}{2} < \frac{T}{2} < \frac{T+\delta}{2} < T_2^{(y)} < T,$$
(1.4.7)

in which we denote

$$\delta := \frac{1}{2} \left( T - L \max\left( \frac{1}{|\lambda_{\overline{m}}(0)|} + \frac{1}{\lambda_{m+1}(0)}, \frac{1}{|\lambda_m(0)|} \right) \right).$$
(1.4.8)

We now turn to the the construction of  $u_q(q = \overline{m} + 1, \dots, m)$ , which is based on the following three steps. The proof is similar to the one in [72], however, they are much more complicate in our situation.

(i) On the domain  $R_f = \{(t, x) | 0 \le t \le \frac{T+\delta}{2}, 0 \le x \le L\}$  (Not  $\hat{R}(y) = \{(t, x) | 0 \le t \le T_1^{(y)}, 0 \le x \le L\}$  as in [111]), we first consider a forward mixed initial-boundary problem (1.4.1) with the initial condition  $\varphi_q(q = \overline{m} + 1, \cdots, m)$  and artificial boundary ary conditions

$$x = L : u_q = f_q(t) \quad (q = \overline{m} + 1, \cdots, m),$$
 (1.4.9)

where  $f_q$  are any given  $C^1$  functions of t with

$$\max_{q=\overline{m}+1,\cdots,m} \|f_q\|_1 \le \|\widetilde{H}\|_1 \le \varepsilon, \tag{1.4.10}$$

$$\max_{q=\overline{m}+1,\cdots,m}\omega(\eta|\dot{f}_q) \le \omega(\eta|\widetilde{H}) \tag{1.4.11}$$

and satisfying the following conditions of  $C^1$  compatibility at the point (t, x) = (0, L):

$$\varphi_q(L) = f_q(0) \quad (q = \overline{m} + 1, \cdots, m), \tag{1.4.12}$$

$$\sum_{j=1}^{n} a_{qj}(\varphi(L))\varphi_{jx}(L) = -f_{qt}(0) \quad (q = \overline{m} + 1, \cdots, m).$$
(1.4.13)

Hereafter, we denote

$$\overline{H} := \{H_{m+1}, \cdots, H_n, \overline{H}_1, \cdots, \overline{H}_{\overline{r}}\}.$$
(1.4.14)

We point out here that, in our iteration,  $f_q(q = \overline{m} + 1, \dots, m)$  are considered as chosen: in particular, they do not depend on  $y \in \Sigma_2$ . This is crucial to guarantee the continuity of  $\mathfrak{F}$ . By Lemma 1.2.1, there exists a unique  $C^1$  solution  $u = u_f(t, x)$  on the domain  $R_f$  with

$$\begin{aligned} \|u_f\|_0 &\leq C_5(\|\Theta\|_0 \|y\|_0 + \max_q \|f_q\|_0 + \max_q \|\varphi_q\|_0) \\ &\leq C_5(\|\Theta\|_0 C_4 + 2)\varepsilon, \end{aligned}$$
(1.4.15)

$$\|u_{f}\|_{1} \leq C_{6}(\|\Theta\|_{1}\|y\|_{1} + \max_{q} \|f_{q}\|_{1} + \max_{q} \|\varphi_{q}\|_{1})$$
  
$$\leq C_{6}(\|\Theta\|_{1}\kappa(\varepsilon) + 2\varepsilon), \qquad (1.4.16)$$

and

$$\omega(\eta|u_{ft}) + \omega(\eta|u_{fx}) \le C_7 \Omega_2(\eta), \qquad (1.4.17)$$

where

$$\Omega_{2}(\eta) = \omega(\eta|\varphi_{x}) + \omega(\eta|\psi_{x}) + \omega(\eta|\tilde{H}_{t}) + \kappa^{2}(\varepsilon)(1 + \kappa(\varepsilon))\omega(\eta|\Theta_{y}) \\
+ \left(1 + \varepsilon + \tilde{G}(\varepsilon) + \|\Theta\|_{1}\kappa(\varepsilon)\right)\left(\eta + \left(1 + \varepsilon + \tilde{G}(\varepsilon) + \|\Theta\|_{1}\kappa(\varepsilon)\right)\omega(\eta|\Gamma^{*})\right) \\
+ \kappa(\varepsilon)(1 + \kappa(\varepsilon))\left(\varepsilon + \|\Theta\|_{1}\kappa(\varepsilon) + \tilde{G}(\varepsilon)\right)\omega(\eta|\Lambda_{u}) \\
+ \left(\varepsilon + \|\Theta\|_{1}\kappa(\varepsilon) + \tilde{G}(\varepsilon) + \|\Theta\|_{1} + \|\Theta_{y}\|_{0}\kappa(\varepsilon)\right)\Omega_{1}(\eta),$$
(1.4.18)

in which

$$\Lambda_{u} := \left\{ \frac{\partial \lambda_{i}}{\partial u_{j}} \right\}, \quad \Theta_{y} := \left\{ \frac{\partial \Theta_{ij}}{\partial y_{k}} \right\}, \|\Theta\|_{0} := \|\Theta(y(t,x))\|_{(C^{0}[R(T)])^{n}}, \|\Theta\|_{1} := \|\Theta(y(t,x))\|_{(C^{1}[R(T)])^{n}}, \|\Theta\|_{1} := \|\Theta(y(t,x))\|_{(C^{1}[R($$

$$\widetilde{G}(\varepsilon) := \max_{\substack{\overline{r}=1,\cdots,\overline{m}\\s=m+1,\dots,n}} \sup_{\substack{0 \le t \le T\\\|u\|_0 \le C_4 \varepsilon}} \left\{ \left| \frac{\partial G_s}{\partial t}(t, u_1, \cdots, u_m) \right|, \left| \frac{\partial \overline{G}_{\overline{r}}}{\partial t}(t, u_{\overline{m}+1}, \cdots, u_n) \right| \right\}, \quad (1.4.20)$$

$$\Gamma^* := \left\{ \frac{\partial G_s}{\partial u_r}, \frac{\partial \overline{G}_{\overline{r}}}{\partial u_s}, \frac{\partial G_s}{\partial t}, \frac{\partial \overline{G}_{\overline{r}}}{\partial t} \right\} \quad (i = 1, \cdots, n; \overline{r} = 1, \cdots, \overline{m}; s = m + 1, \cdots, n).$$

$$(1.4.21)$$

Here, we point out that  $\omega(\eta|\Theta_y)$ ,  $\omega(\eta|\Lambda_u)$  and  $\omega(\eta|\Gamma^*)$  denote the modulus of continuity for  $\Theta_y$ ,  $\Lambda_u$  and  $\Gamma^*$  on the domain  $R_1 = \{0 \leq ||y||_0 \leq C_4 \varepsilon\}$ ,  $R_2 = \{0 \leq ||u||_0 \leq C_4 \varepsilon\}$  and  $R_3 = \{0 \leq t \leq T, 0 \leq ||u||_0 \leq C_4 \varepsilon\}$ , respectively. One can easily check that  $\widetilde{G}(\varepsilon) \geq 0$  is a  $C^0$  function of  $\varepsilon$  and

$$\tilde{G}(0) = 0.$$
 (1.4.22)

Then we can dertermine the value of  $u_f$  at x = 0 as

$$x = 0: \quad u_f = a(t), \quad 0 \le t \le \frac{T+\delta}{2}.$$
 (1.4.23)

and

$$||a||_0 \le C_8(||\Theta||_0 C_4 + 2)\varepsilon, \tag{1.4.24}$$

$$\|a\|_1 \le C_9(\|\Theta\|_1 \kappa(\varepsilon) + 2\varepsilon), \tag{1.4.25}$$

$$\omega(\eta|a_t) \le C_{10}\Omega_2(\eta). \tag{1.4.26}$$

Here, we denote  $a(t) := (a_{\overline{m}+1}(t), \cdots, a_m(t))$ . Evidently, By Lemma 1.2.1, one obtains that  $u_f$  and a depends continuously on  $y \in \Sigma_2$ .

(ii) Similarly, on the domain  $R_b = \{(t, x) | \frac{T-\delta}{2} \le t \le T, 0 \le x \le L\}$ , we consider the backward mixed initial-boundary value problem (1.4.1) with the final condition  $\psi_q(q = \overline{m} + 1, \cdots, m)$  and artificial boundary conditions

$$x = 0: u_q = g_q(t), \tag{1.4.27}$$

where  $g_q(q = \overline{m} + 1, \cdots, m)$  are any given functions of t with

$$\max_{q=\overline{m}+1,\cdots,m} \|g_q\|_{C^1} \le \|\widetilde{H}\|_1 \le \varepsilon, \tag{1.4.28}$$

$$\max_{q=\overline{m}+1,\cdots,m} \omega(\eta|\dot{g}_q) \le \omega(\eta|\widetilde{H}), \tag{1.4.29}$$

such that the conditions of  $C^1$  compatibility are satisfied at the point (t, x) = (T, 0). Similarly, in order to have the continuity of  $\mathfrak{F}$ ,  $g_q(q = \overline{m} + 1, \cdots, m)$  are considered as chosen. By Lemma 1.2.1, there exists a unique  $C^1$  solution  $u = u_b(t, x)$  on the domain  $R_b$ , which allows us to determine the value of  $u_b$  at x = 0 as

$$x = 0: \quad u_b = b(t), \quad \frac{T - \delta}{2} \le t \le T,$$
 (1.4.30)

in which

$$\|b\|_{0} \le C_{11}(\|\Theta\|_{0}C_{4} + 2)\varepsilon, \qquad (1.4.31)$$

$$\|b\|_1 \le C_{12}(\|\Theta\|_1 \kappa(\varepsilon) + 2\varepsilon), \qquad (1.4.32)$$

$$\omega(\eta|b_t) \le C_{13}\Omega_2(\eta). \tag{1.4.33}$$

Here, we denote  $b(t) := (b_{\overline{m}+1}(t), \cdots, b_m(t))$ . An argument similar to the one used in (i) shows b depends continuously on  $y \in \Sigma_2$ .

(iii) Let  $(\alpha, \beta) \in (C^2[\frac{T-\delta}{2}, \frac{T+\delta}{2}])^2$  be such that

$$\alpha(\frac{T-\delta}{2}) = 1, \alpha(\frac{T+\delta}{2}) = 0, \alpha_t(\frac{T-\delta}{2}) = 0, \ \alpha_t(\frac{T+\delta}{2}) = 0;$$
(1.4.34)

$$\beta(\frac{T-\delta}{2}) = 0, \ \beta(\frac{T+\delta}{2}) = 1, \ \beta_t(\frac{T-\delta}{2}) = 0, \ \ \beta_t(\frac{T+\delta}{2}) = 0.$$
(1.4.35)

Again, the scalar functions  $\alpha(t)$  and  $\beta(t)$  as the functions  $f_q$  and  $g_q$   $(q = \overline{m} + 1, \dots, m)$  are considered as chosen. Next, Set

$$c(t) = \begin{cases} a(t), & 0 \le t \le \frac{T-\delta}{2}, \\ \alpha(t)a(t) + \beta(t)b(t), \frac{T-\delta}{2} \le t \le \frac{T+\delta}{2}, \\ b(t), & \frac{T+\delta}{2} \le t \le T. \end{cases}$$
(1.4.36)

Obviously,  $c \in (C^1[0,T])^{m-\overline{m}}$ . On the other hand, it can easily be verified from (1.4.24)-(1.4.26) and (1.4.31)-(1.4.33) that

$$||c||_0 \le C_{14} (||\Theta||_0 C_4 + 2)\varepsilon, \tag{1.4.37}$$

$$\|\varepsilon\|_{1} \le C_{15}(\|\Theta\|_{1}\kappa(\varepsilon) + \|\Theta\|_{0}C_{4}\varepsilon + 2\varepsilon), \qquad (1.4.38)$$

$$\omega(\eta|c_t) \le C_{16}\Omega_2(\eta), \tag{1.4.39}$$

and that c depends continuously on  $y \in \Sigma_2$ . Here, we denote  $c(t) := (c_{\overline{m}+1}(t), \cdots, c_m(t))$ .

Since there are no zero eigenvalues, changing the status of t and x, system (1.4.1) can be equivalently rewritten as

$$\frac{\partial u_q}{\partial x} + \frac{1}{\lambda_q(y)} \frac{\partial u_q}{\partial t} = \sum_{k=1}^n \Theta_{qk}(y) \Big( \frac{\partial y_k}{\partial x} + \frac{1}{\lambda_q(y)} \frac{\partial y_k}{\partial t} \Big).$$
(1.4.40)

Now we consider the rightward mixed initial-boundary value problem for the system (1.4.40) with the initial condition

$$x = 0: u_q = c(t), \quad 0 \le t \le T \ (q = \overline{m} + 1, \cdots, m),$$
 (1.4.41)

and the following boundary conditions reduced from the final data  $\psi_q$ :

$$t = T: \quad u_q = \psi_q(x) \quad (q = \overline{m} + 1, \cdots, m).$$
 (1.4.42)

We easily check that the mixed initial-boundary value problem (1.4.40)-(1.4.42) satisfies the condition of  $C^1$  compatibility at the point (t, x) = (T, 0). Therefore, by Lemma 1.2.1, there exists a unique  $C^1$  solution  $u_q = \tilde{u}_q(t, x)$   $(q = \overline{m} + 1, \dots, m)$  of (1.4.40)-(1.4.42) with

$$\max_{q=\overline{m}+1,\cdots,m} \|\tilde{u}_q\|_0 \le C_{17}(\|\Theta\|_0 C_4 + 2)\varepsilon, \tag{1.4.43}$$

$$\max_{q=\overline{m}+1,\cdots,m} \|\tilde{u}_q\|_1 \le C_{18}(\|\Theta\|_1\kappa(\varepsilon) + \|\Theta\|_0C_4\varepsilon + 2\varepsilon),$$
(1.4.44)

$$\max_{q=\overline{m}+1,\cdots,m} \Omega(\eta|\tilde{u}_{qt}) + \Omega(\eta|\tilde{u}_{qx}) \le C_{19}\Omega_2(\eta), \qquad (1.4.45)$$

and  $\tilde{u}_q$   $(q = \overline{m} + 1, \dots, m)$  depends continuously on y. Next, let  $u_q = \tilde{u}_q(t, x)(q = \overline{m} + 1, \dots, m)$ . By uniqueness (see also [72] and [111]),  $u_q(q = \overline{m} + 1, \dots, m)$  satisfies (1.4.3)-(1.4.4) and (1.4.43)-(1.4.45).

#### Step 2: Control of the system (1.4.2)

Once the system (1.4.1) is controlled, we can substitute  $u_q(t,0) = c_q(t)(q = \overline{m} + 1, \dots, m)$  to (1.3.8), which leads system (1.4.2) to a new linearized system as

follows:

$$\begin{cases} \frac{\partial u_p}{\partial t} + \lambda_p(y) \frac{\partial u_p}{\partial x} = \sum_{k=1}^n \Theta_{pk}(y) \left( \frac{\partial y_k}{\partial t} + \lambda_p(y) \frac{\partial y_k}{\partial x} \right), (p = 1, \cdots, \overline{m}, m + 1, \cdots, n), \\ x = 0 : u_s = \mathcal{G}_s(t, u_1, \cdots, u_{\overline{m}}) + H_s(t), s = m + 1, \cdots, n, \end{cases}$$
(1.4.46)

where

$$\mathcal{G}_s(t, u_1, \cdots, u_{\overline{m}}) := G_s(t, u_1, \cdots, u_{\overline{m}}, c_{\overline{m}+1}, \cdots, c_m), (s = m+1, \cdots, n).$$
(1.4.47)

Fortunately, noting (1.3.11), one can easily check that the above system (1.4.46) satisfies the same  $C^1$  compatibility conditions at the point (t, x) = (0, 0) and (T, 0) as the original one (1.1.1) (see Remark 1.1.3), which guarantees the well-posedness of the system (1.4.46).

Combining the one-side control approach in [72, Chapter 3 on pages 47-52] and a analogous heuristic argument used in **Step 1**, the details of which we omit, one can show that under the hypothesis (1.1.22), there exists a  $C^1$  solution  $u_p = \tilde{u}_p(t, x)(p = 1, \dots, \overline{m}, m+1, \dots, n)$  satisfying

$$t = 0: \quad \tilde{u}_p = \varphi_p(x) \quad (p = 1, \cdots, \overline{m}, m, \cdots, n), \tag{1.4.48}$$

$$t = T: \quad \tilde{u}_p = \psi_p(x) \quad (p = 1, \cdots, \overline{m}, m, \cdots, n).$$
 (1.4.49)

Moreover, we have

$$\max_{p} \|\tilde{u}_{p}\|_{0} \le C_{20} (\|\Theta\|_{0}C_{4} + 2)\varepsilon, \qquad (1.4.50)$$

$$\max_{n} \|\tilde{u}_{p}\|_{1} \leq C_{21} \big( \|\Theta\|_{1} \kappa(\varepsilon) + \|\Theta\|_{0} C_{4} \varepsilon + 2\varepsilon + \widetilde{G}(\varepsilon) \big), \tag{1.4.51}$$

$$\max_{\eta} \omega(\eta | \tilde{u}_{pt}) + \omega(\eta | \tilde{u}_{px}) \le C_{22} \Omega_2(\eta), \qquad (1.4.52)$$

and  $\tilde{u}_p(p=1,\cdots,\overline{m},m,\cdots,n)$  depends continuously on  $y \in \Sigma_2$ .

### Step 3: Proof of the existence and continuity of $\mathfrak{F}$

Let

$$C_{23} = \max\{C_{17}, C_{20}\}, \tag{1.4.53}$$

$$C_{24} = \max\{C_{18}, C_{21}\},\tag{1.4.54}$$

$$C_{25} = \max\{C_{19}, C_{22}\}.$$
 (1.4.55)

We then choose

$$C_4 = 3C_{23},\tag{1.4.56}$$

$$\kappa(\varepsilon) = 5C_{24}\varepsilon + 2C_{24}\widetilde{G}(\varepsilon), \qquad (1.4.57)$$

and

$$\Omega_{1}(\eta) = 2C_{25} \left( \omega(\eta | \dot{\varphi}) + \omega(\eta | \dot{\psi}) + \omega(\eta | \dot{\tilde{H}}) + 2\omega(\eta | \Theta_{y}) \right)$$

$$+ (2C_{25} + 1) \left( \eta + \left( 1 + \frac{1}{2C_{25}} \right) \omega(\eta | \Gamma^{*}) \right) + 2\omega(\eta | \Lambda_{u}),$$

$$(1.4.58)$$

by (1.1.4), if  $0 < \varepsilon \ll 1$  is small enough, one gets

$$\|\Theta\|_0 \le \frac{1}{2C_4}, \ \|\Theta\|_1 \le \frac{1}{2C_{24}},$$
 (1.4.59)

and

$$\kappa(\varepsilon) < 1, C_{25}\left(\varepsilon + \|\Theta\|_{1}\kappa(\varepsilon) + \widetilde{G}(\varepsilon) + \|\Theta\|_{1} + \|\Theta_{y}\|_{0}\kappa(\varepsilon)\right) \le \frac{1}{2}.$$
 (1.4.60)

Let

$$u(t,x) = (\tilde{u}_1, \cdots, \tilde{u}_n).$$
 (1.4.61)

Then noting (1.4.43)-(1.4.45) and (1.4.50)-(1.4.52), under the hypothesis (1.1.22), we have

$$\|u\|_0 \le C_4 \varepsilon, \|u\|_1 \le \kappa(\varepsilon), \tag{1.4.62}$$

and

$$\omega(\eta|u_t) + \omega(\eta|u_x) \le \Omega_1(\eta). \tag{1.4.63}$$

Therefore, combining (1.4.48)-(1.4.49) and (1.4.3)-(1.4.4), we show that

$$u \in \Sigma_3. \tag{1.4.64}$$

From the construction of the operator  $\mathfrak{F}$  in **Step 1-Step 2**, one can easily verify it is continuous on y. The construction of  $\mathfrak{F}$  is now completed.

## 1.5 Appendix: Proof of Lemma 1.2.1

In this section, we give a detail proof of Lemma 1.2.1. For the case of coupled system of wave equations, one can refer to [56].

Set  $w = (w_1, \dots, w_n), p = (p_1, \dots, p_n)$ , where

$$w_i = u_{ix} \ (i = 1, \cdots, n),$$
 (1.5.1)

$$p_i = u_{it} \ (i = 1, \cdots, n).$$
 (1.5.2)

Obviously, noting the equations (1.2.4), one can show that

$$p_i + \lambda_i(t, x)w_i = \sum_{k=1}^{\tilde{n}} b_{ik}(t, x) \Big(\frac{\partial c_k(t, x)}{\partial t} + \lambda_i(t, x)\frac{\partial c_k(t, x)}{\partial x}\Big).$$
(1.5.3)

Hence, by straightforward computations, we have

$$\|p\|_{0} \leq \widetilde{C}_{1}(K) \Big( \|w\|_{0} + \|B\|_{0} \|c\|_{1} \Big)$$
(1.5.4)

and

$$\omega(\eta|p) \le \widetilde{C}_2(K) \Big( \big( \|w\|_0 + \|B\|_1 \|c\|_1 \big) \eta + \omega(\eta|w) + \|B\|_0 \big( \omega(\eta|c_t) + \omega(\eta|c_x) \big) \Big).$$
(1.5.5)

Here and in what follows, we assume that  $\widetilde{C}_n(K)(n \in \mathbb{Z}^+)$  are positive real numbers depending on K.

Thus, if we want to prove that  $||u||_0$ ,  $||u||_1$  and  $\omega(\eta|u_t) + \omega(\eta|u_x)$  satisfy (1.2.8)-(1.2.9) and (1.2.12), respectively, it suffices to show (1.2.8),

$$\|w\|_{0} \leq \widetilde{C}_{3}(T, K) \left( \|\varphi\|_{1} + \|H\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} \right)$$
(1.5.6)

and

$$\omega(\eta|w) \leq \widetilde{C}_{4}(T,K) \Big( \omega(\eta|\varphi_{x}) + \|B\|_{1} \big( \omega(\eta|c_{t}) + \omega(\eta|c_{x}) \big) + \|c\|_{1} \big( \omega(\eta|B_{t}) + \omega(\eta|B_{x}) \big) \\
+ \big(1 + \|\varphi\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \big) \big(\eta + \big(1 + \|\varphi\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} \\
+ \|H\|_{1} \big) \omega(\eta|\Gamma) \big) + \big(\|\varphi\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \big) \omega(\eta|\Lambda_{x}) + \omega(\eta|H_{t}) \Big) \quad (1.5.7)$$

hold.

Utilizing the decomposition of waves (see [61] and [111]), one gets

$$\frac{du_i}{d_i t} = \beta_i(t, x, u), \tag{1.5.8}$$

$$\frac{dw_i}{d_i t} = \gamma_i(t, x, u) + \sum_{j=1}^n \gamma_{ij}(t, x) w_j, \qquad (1.5.9)$$

where

$$\frac{d}{d_i t} = \frac{\partial}{\partial t} + \lambda_i(t, x) \frac{\partial}{\partial x}$$
(1.5.10)

denotes the directional derivative along the *i*-th characteristic,

$$\beta_i(t, x, u) = \sum_{k=1}^{\tilde{n}} b_{ik}(t, x) \frac{dc_k(t, x)}{d_i t},$$
(1.5.11)

and

$$\gamma_{i}(t,x) = \sum_{k=1}^{\tilde{n}} \left[ \frac{d}{d_{i}t} \left( b_{ik}(t,x) \frac{\partial c_{k}(t,x)}{\partial x} \right) - \frac{\partial b_{ik}(t,x)}{\partial t} \frac{\partial c_{k}(t,x)}{\partial x} + \frac{\partial b_{ik}(t,x)}{\partial x} \frac{\partial c_{k}(t,x)}{\partial t} + b_{ik}(t,x) \frac{\partial \lambda_{i}(t,x)}{\partial x} \frac{\partial c_{k}(t,x)}{\partial x} \right],$$

$$\gamma_{ij}(t,x) = -\frac{\partial \lambda_{i}(t,x)}{\partial x} \delta_{ij}.$$
(1.5.12)
$$(1.5.13)$$

Let

$$T_1 = \min_{i=1,\cdots,n} \inf_{(t,x)\in R(T)} \frac{L}{|\lambda_i(t,x)|}.$$
 (1.5.14)

Next, we will prove (1.2.8), (1.5.6)-(1.5.7) on the domain

$$R(T_1) = \{(t, x) | 0 \le t \le T_1, 0 \le x \le L\}.$$
(1.5.15)

then claim they still hold on the whole domain R(T).

Step A (estimate of  $||u||_0$ ): For any given point  $(t, x) \in R(T_1)$ , we draw down the *r*-th characteristic  $\xi = f_r(\tau; t, x) (r = 1, \cdots, m)$ :

$$\begin{cases} \frac{df_r(\tau;t,x)}{d\tau} = \lambda_r(\tau, f_r(\tau;t,x)), \\ \tau = t : \xi = x. \end{cases}$$
(1.5.16)

passing through (t, x). Noting (1.2.5) and (1.5.14), there are only two possibilities:

(a) This r-th characteristic intersects the x-axis at a point  $(0, \alpha_r)$ . Integrating the r-th equation in (1.5.8) along this characteristic from  $(0, \alpha_r)$  to (t, x), noting (1.5.11),

we get

$$|u_r(t,x)| \le ||u(0,\cdot)||_0 + 2||B||_0 ||c||_0, \qquad (1.5.17)$$

(b) This *r*-characteristic intersects x = L at the point  $(t_r, L)$ , and all *s*-th characteristics passing through  $(t_r, L)$  intersect the *x*-axis at point  $(0, \beta_s)(s = m + 1, \dots, n)$  respectively. Then similarly to (1.5.17), we have

$$|u_r(t,x)| \le |u_r(t_r,L)| + 2||B||_0 ||c||_0.$$
(1.5.18)

Moreover, by means of the boundary conditions (1.1.8) and noting (1.1.9), it is easy to get

$$|u(t_r, L)| \le \widetilde{C}_5(K) \max_{m+1 \le s \le n} |u_s(t_r, L)| + ||H||_0,$$
(1.5.19)

without loss of generality, we may suppose  $\widetilde{C}_5(K) > 1$ .

Similar to (1.5.17), integrating the s-th equation in (1.5.8) along the s-th characteristic from  $(0, \beta_s)$  to  $(t_r, L)$  gives

$$|u_s(t_r, L)| \le ||u(0, \cdot)||_0 + 2||B||_0 ||c||_0.$$
(1.5.20)

The combination (1.5.18) to (1.5.20) leads to

$$|u_r(t,x)| \le \widetilde{C}_6(K) \Big( \|u(0,\cdot)\|_0 + \|H\|_0 + \|B\|_0 \|c\|_0 \Big).$$
(1.5.21)

Thus, it follows from (1.5.17) and (1.5.21) that

$$|u_r(t,x)| \le \widetilde{C}_7(K) \Big( ||u(0,\cdot)||_0 + ||H||_0 + ||B||_0 ||c||_0 \Big).$$
(1.5.22)

Similar estimates can be obtained for  $u_s(t, x)(s = m + 1, \dots, n)$ . Hence, we have

$$u(t) \le \widetilde{C}_8(K) \big( \|u(0, \cdot)\|_0 + \|H\|_0 + \|B\|_0 \|c\|_0 \big)$$
(1.5.23)

in which we may assume that  $\widetilde{C}_8(K) > 1$  and in what follows, we denote

$$u(\tau) := \sup_{0 \le t \le \tau} \|u(t, \cdot)\|_0, \tag{1.5.24}$$

$$w(\tau) := \sup_{0 \le t \le \tau} \|w(t, \cdot)\|_0.$$
(1.5.25)

Taking  $u(T_1, x)$  as the initial data on  $t = T_1$ , and repeating the previous procedure,

we obtain

$$u(t) \leq C_8(K) \left( \|u(T_1, \cdot)\|_0 + \|H\|_0 + \|B\|_0 \|c\|_0 \right)$$
  
$$\leq \widetilde{C}_8^2(K) \left( \|u(0, \cdot)\|_0 + \|H\|_0 + \|B\|_0 \|c\|_0 \right), \ T_1 \leq t \leq 2T_1$$

Repeating this procedure at most  $M = \begin{bmatrix} T \\ T_1 \end{bmatrix} + 1$  times, we get

$$\|u\|_{0} \leq \max_{t \in [0,T]} u(t) \leq \widetilde{C}_{8}^{M}(K) \big( \|u(0,\cdot)\|_{0} + \|H\|_{0} + \|B\|_{0} \|c\|_{0} \big), \quad \forall t \in [0,T].$$

This concludes the proof of (1.2.8).

Step B (estimate of  $||w||_0$ ): As before, for any given point  $(t, x) \in R(T_1)$ , there are still two possibilities for the r-th characteristic  $(r = 1, \dots, m)$ .

In case (a), integrating the r-th equation in (1.5.9) along this r-th characteristic from  $(0, \alpha_r)$  to (t, x) yields

$$w_r(t,x) - w_r(0,\alpha_r) = \int_0^t \gamma_r(\tau, f_r(\tau; t, x)) + \sum_{j=1}^n \gamma_{rj}(\tau, f_r(\tau; t, x)) w_j d\tau, \quad (1.5.26)$$

then by (1.5.12)-(1.5.13) and (1.5.25), one can show that

$$|w_r(t,x)| \le ||w(0,\cdot)||_0 + \widetilde{C}_9(K) \Big( ||B||_1 ||c||_1 + \int_0^t w(\tau) d\tau \Big).$$
(1.5.27)

In case (b), similar to (1.5.19), we have

$$|w_r(t,x)| \le |w_r(t_r,L)| + \widetilde{C}_{10}(K) \Big( ||B||_1 ||c||_1 + \int_{t_r}^t w(\tau) d\tau \Big).$$
(1.5.28)

In order to estimate  $|w_r(t_r, L)|$ , we seek the boundary conditions satisfied by  $w_r$ on x = L. Differentiating (1.1.8) with respect to t, we get

$$x = L: \quad \frac{\partial u_r}{\partial t} = \frac{\partial G_r}{\partial t} + \sum_{s=m+1}^n \frac{\partial G_r}{\partial u_s} \frac{\partial u_s}{\partial t} + H_{rt} \quad (r = 1, \cdots, m). \tag{1.5.29}$$

By (1.2.4), together with (1.5.29), we obtain

$$w_{r}(t,L) = \frac{1}{\lambda_{r}(t,L)} \left( \sum_{k=1}^{\tilde{n}} b_{rk} \left( \frac{\partial c_{k}}{\partial t} + \lambda_{r} \frac{\partial c_{k}}{\partial x} \right) - \frac{\partial G_{r}}{\partial t} - H_{rt} + \sum_{s=m+1}^{n} \frac{\partial G_{r}}{\partial u_{s}} \left( \lambda_{s} w_{s} - \sum_{k=1}^{\tilde{n}} b_{sk} \left( \frac{\partial c_{k}}{\partial t} + \lambda_{s} \frac{\partial c_{k}}{\partial x} \right) \right) \right) (t,L).$$

$$(1.5.30)$$

Then by noting (1.1.9), one can easily see that

$$|w_r(t_r, L)| \le \widetilde{C}_{11}(K) \Big( \max_{m+1 \le s \le n} |w_s(t_r, L)| + ||H_t||_0 + ||B||_0 ||c||_1 + ||G_t||_0 \Big), \quad (1.5.31)$$

in which we may assume that  $\widetilde{C}_{11}(K) \geq 1$ . Integrating the s-th equation in (1.5.9) along the corresponding s-th characteristic from  $(0, \beta_r)$  to  $(t_r, L)$  gives

$$|w_s(t_r, L)| \le ||w(0, \cdot)||_0 + \widetilde{C}_{12}(K) \Big( ||B||_1 ||c||_1 + \int_0^{t_r} w(\tau) d\tau \Big).$$
(1.5.32)

Combining (1.5.28), (1.5.31)-(1.5.32) yields

$$|w_r(t,x)| \le \widetilde{C}_{13}(K) \Big( \|\varphi\|_1 + \|B\|_1 \|c\|_1 + \|H\|_1 + \|G_t\|_0 + \int_0^t w(\tau) d\tau \Big), \quad (1.5.33)$$

Similar estimates can be obtained for  $w_s(t, x)(s = m + 1, \dots, n)$ . Hence we get

$$w(t) \le \widetilde{C}_{14}(K) \Big( \|\varphi\|_1 + \|B\|_1 \|c\|_1 + \|H\|_1 + \|G_t\|_0 + \int_0^t w(\tau) d\tau \Big).$$
(1.5.34)

Then, using Gronwall's inequality, for  $0 \le t \le T_1$ , one shows

$$w(t) \le \widetilde{C}_{15}(K)(\|\varphi\|_1 + \|B\|_1\|c\|_1 + \|H\|_1 + \|G_t\|_0), \qquad (1.5.35)$$

we assume here that  $\widetilde{C}_{15}(K) \ge 1$ .

Repeating above procedures, for  $0 \le t \le T$ , we have

$$\|w\|_{0} \leq \max_{t \in [0,T]} w(t) \leq \widetilde{C}_{16}(T,K) (\|\varphi\|_{1} + \|B\|_{1} \|c\|_{1} + \|H\|_{1} + \|G_{t}\|_{0}), \qquad (1.5.36)$$

which finishes the proof of (1.5.6).

Step C (estimate of  $\omega(\eta|w)$ ): By (1.2.5), when  $i = 1, \dots, m$ , let us consider the *i*-th characteristic from the point (0, L) which divides the domain  $R(T_1)$  into  $R_i^+$ and  $R_i^0$ ; Similarly, when  $i = m + 1, \dots, n$ , let us consider the *i*-th characteristic from (0,0) which divides  $R(T_1)$  into  $R_i^-$  and  $R_i^0$ . For any given (t,x) in  $R(T_1)$ , let us consider the *i*-th characteristic  $\xi_i = f_i(\tau; t, x)$  which intersects the boundary of  $R(T_1)$ (i.e. x = 0, x = L and t = 0) on  $(\tau_k(t, x), \xi_k(t, x))$ , then we have

$$\begin{cases} \tau_i(t,x) = 0, \xi_i(t,x) = f_i(0;t,x), \text{ if } (t,x) \in R_i^0, (i = 1, \cdots, n), \\ \xi_i(t,x) = L, f_i(\tau_i(t,x);t,x) = L, \text{ if } (t,x) \in R_i^+, (i = 1, \cdots, m), \\ \xi_i(t,x) = 0, f_i(\tau_i(t,x);t,x) = 0, \text{ if } (t,x) \in R_i^-, (i = m + 1, \cdots, n). \end{cases}$$
(1.5.37)

Before claiming (1.5.7), we need the following estimates, which can be found in many references (see, for instance, [84, Lemma 3.3 on pages 50-51]):

Chapter 1: Sharp Boundary Control Time Estimates for 1-D Homogeneous Quasilinear Hyperbolic Systems

**Proposition 1.5.1** On the domain  $R(T_1)$ , for any characteristic  $\xi = f_i(\tau; t, x)$ , we have

$$\left\|\frac{\partial f_i}{\partial x}\right\|_0, \left\|\frac{\partial f_i}{\partial t}\right\|_0, \|\tau_{it}\|_0, \|\tau_{ix}\|_0, \|\xi_{it}\|_0, \|\xi_{ix}\|_0 < \widetilde{C}_{17}(K) < +\infty \quad (i = 1, \cdots, n).$$
(1.5.38)

Integrating the r-th equation in (1.5.9) along the r-th characteristic  $\xi_r = f_r(\tau; t, x)$ , we obtain

$$w_r(t,x) = \overline{\psi}_r(\tau_r(t,x),\xi_r(t,x)) + \int_{\tau_r(t,x)}^t \left(\gamma_r + \sum_{j=1}^n \gamma_{rj} w_j\right)(\tau,f_r(\tau;t,x))d\tau, \quad (1.5.39)$$

where

$$\overline{\psi}_{r}(\tau_{r}(t,x),\xi_{r}(t,x)) = \begin{cases} w_{r}(\tau_{r}(t,x),L), \ \forall(t,x) \in R_{r}^{+}, \\ w_{r}(0,\xi_{r}(t,x)), \forall(t,x) \in R_{r}^{0}. \end{cases}$$
(1.5.40)

Obviously, by the basic proposition of the modulus of continuity (see, for example, [84, Pages 58-59 and 88-90]), one obtains

$$\omega(\eta | w_r(0, \xi_r(t, x))) \le \tilde{C}_{18}(K) \omega(\eta | w_r(0, \cdot)).$$
(1.5.41)

Let

$$\omega(\tau,\eta|w) = \sup_{\substack{(t_1,x_1), (t_2,x_2) \in R(\tau)\\ |t_1-t_2| \le \eta, |x_1-x_2| \le \eta}} |w(t_1,x_1) - w(t_2,x_2)|.$$
(1.5.42)

It is easy to see that

$$\omega(T_1, \eta | w) = \sup_{0 \le \tau \le T_1} \omega(\tau, \eta | w).$$
(1.5.43)

Utilizing Proposition 1.5.1, for any given  $0 \le \tau \le T_1$ , by (1.5.30), one gets

$$\begin{split} &\omega(\tau,\eta|w_{r}(\tau_{r}(\cdot,\cdot),L)) \leq \widetilde{C}_{19}(K)\omega(\tau,\eta|w_{r}(\cdot,L)) \\ \leq \widetilde{C}_{20}(K) \Big( \big( \|u\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \big)\eta + (1 + \|u\|_{1})(1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1})\omega(\eta|\Gamma) \\ &+ \|B\|_{0} \big(\omega(\eta|c_{t}) + \omega(\eta|c_{x})\big) + \omega(\eta|H_{t}) + \max_{s=m+1,\cdots,n} \omega(\tau,\eta|w_{s}(\cdot,L)) \Big) \\ \leq \widetilde{C}_{21}(K) \Big( \big( 1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \big) \big(\eta + (1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1})\omega(\eta|\Gamma) \big) \\ &+ \|B\|_{0} \big(\omega(\eta|c_{t}) + \omega(\eta|c_{x})\big) + \omega(\eta|H_{t}) + \max_{s=m+1,\cdots,n} \omega(\tau,\eta|w_{s}(\cdot,L)) \Big). \end{split}$$
(1.5.44)

For any given point (t, L) on x = L, integrating the s-th equation in (1.5.9) along the s-th characteristic  $\xi_s = f_s(\tau; t, x)$  form (t, L) to  $(0, \xi_s)$ , we obtain

$$w_s(t,L) = w_s(0,\xi_s(t,L)) + \int_0^t \gamma_s(\tau, f_s(\tau;t,x)) + \sum_{j=1}^n \gamma_{sj}(\tau, f_s(\tau;t,x)) w_j d\tau. \quad (1.5.45)$$

Next, we will use (1.5.45) in order to estimate  $\omega(\tau, \eta | w_s(\cdot, L))$ , where  $0 \leq \eta \leq T_1$ . Setting

$$\mathfrak{L}_{ijk}(t,x) := -\frac{\partial b_{ik}(t,x)}{\partial t} \frac{\partial c_k(t,x)}{\partial x} + \frac{\partial b_{ik}(t,x)}{\partial x} \frac{\partial c_k(t,x)}{\partial t} + b_{ik}(t,x) \frac{\partial \lambda_i(t,x)}{\partial x} \frac{\partial c_k(t,x)}{\partial x} - \frac{\partial \lambda_i(t,x)}{\partial x} \delta_{ij} w_j(t,x)$$
(1.5.46)

in which  $i, j = 1, \dots, n$  and  $k = 1, \dots, \tilde{n}$ . For any given  $0 \le \tau \le T_1$ , by Proposition 1.5.1, one gets

$$\begin{aligned} &\omega(\tau,\eta|\mathfrak{L}_{ijk}(\tau,f_{i}(\tau;t,L))) \\ \leq &\widetilde{C}_{22}(K)\omega(\tau,\eta|\mathfrak{L}_{ijk}(\tau,\xi)) \\ \leq &\widetilde{C}_{23}(K)\Big((\|u\|_{1}+\|B\|_{1}\|c\|_{1})\big(\eta+\omega(\eta|\Lambda_{x})\big)+\|B\|_{1}(\omega(\eta|c_{t})+\omega(\eta|c_{x})) \\ &+\|c\|_{1}(\omega(\eta|B_{t})+\omega(\eta|B_{x}))+\omega(\tau,\eta|w)\Big). \end{aligned}$$
(1.5.47)

Thus, noting (1.5.47), for any given  $0 \le \overline{t} \le T_1$ , we have

$$\begin{aligned} &\omega\Big(\bar{t},\eta\Big|\int_{0}^{t}\mathfrak{L}_{ijk}(\tau,f(\tau;t,L)d\tau)\Big) \\ \leq &\widetilde{C}_{24}(K)\Big(\Big(1+\|u\|_{1}+\|B\|_{1}\|c\|_{1}\Big)\eta+(\|u\|_{1}+\|B\|_{1}\|c\|_{1})\omega(\eta|\Lambda_{x}) \\ &+\|B\|_{1}(\omega(\eta|c_{t})+\omega(\eta|c_{x}))+\|c\|_{1}\Big(\omega(\eta|B_{t})+\omega(\eta|B_{x})\Big)+\int_{0}^{\bar{t}}\omega(\tau,\eta|w)d\tau\Big). \end{aligned}$$

$$(1.5.48)$$

By (1.5.45), it is easy to show that

$$\begin{aligned} &\omega(\bar{t},\eta|w_s(\cdot,L)) \\ \leq &\widetilde{C}_{25}(K) \Big( \omega(\eta|w(0,\cdot)) + \big(1 + \|u\|_1 + \|B\|_1 \|c\|_1 \big) \eta + (\|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Lambda_x) \\ &+ \|B\|_1 (\omega(\eta|c_t) + \omega(\eta|c_x)) + \|c\|_1 (\omega(\eta|B_t) + \omega(\eta|B_x)) + \int_0^{\bar{t}} \omega(\tau,\eta|w) d\tau \Big). \end{aligned}$$

$$(1.5.49)$$

From (1.5.44), we find that

$$\begin{aligned} &\omega(\bar{t},\eta|w_{r}(\tau_{r}(\cdot,\cdot),L)) \\ \leq &\widetilde{C}_{26}(K) \bigg( \omega(\eta|w(0,\cdot)) + \|B\|_{1} \big( \omega(\eta|c_{t}) + \omega(\eta|c_{x}) \big) + \|c\|_{1} \big( \omega(\eta|B_{t}) + \omega(\eta|B_{x}) \big) \\ &+ \big(1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \big) \big(\eta + (1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1}) \omega(\eta|\Gamma) \big) \\ &+ \big(\|u\|_{1} + \|B\|_{1} \|c\|_{1} \big) \omega(\eta|\Lambda_{x}) + \omega(\eta|H_{t}) + \int_{0}^{\bar{t}} \omega(\tau,\eta|w) d\tau \bigg). \end{aligned}$$

$$(1.5.50)$$

Combining (1.5.40)-(1.5.41) and (1.5.50), one has

$$\begin{aligned} &\omega(\bar{t},\eta|\overline{\psi}_{r}(\tau_{r}(\cdot,\cdot),\xi_{r}(\cdot,\cdot))) \\ \leq &\widetilde{C}_{27}(K) \left( \omega(\eta|w(0,\cdot)) + \|B\|_{1} \left( \omega(\eta|c_{t}) + \omega(\eta|c_{x}) \right) + \|c\|_{1} (\omega(\eta|B_{t}) + \omega(\eta|B_{x})) \\ &+ \left( 1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1} + \|G_{t}\|_{0} + \|H\|_{1} \right) (\eta + (1 + \|u\|_{1} + \|B\|_{1} \|c\|_{1}) \omega(\eta|\Gamma)) \\ &+ (\|u\|_{1} + \|B\|_{1} \|c\|_{1}) \omega(\eta|\Lambda_{x}) + \omega(\eta|H_{t}) + \int_{0}^{\bar{t}} \omega(\tau,\eta|w) d\tau \right). \end{aligned}$$

$$(1.5.51)$$

Next, noting (1.5.39) and (1.5.48) together with Proposition 1.5.1, one can show that

$$\begin{aligned} &\omega(\bar{t},\eta|w_r) \\ \leq &\widetilde{C}_{28}(K) \left( \omega(\eta|w(0,\cdot)) + \|B\|_1 \left( \omega(\eta|c_t) + \omega(\eta|c_x) \right) + \|c\|_1 (\omega(\eta|B_t) + \omega(\eta|B_x)) \\ &+ \left( 1 + \|u\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \right) (\eta + (1 + \|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Gamma) ) \\ &+ (\|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Lambda_x) + \omega(\eta|H_t) + \int_0^{\bar{t}} \omega(\tau,\eta|w) d\tau \right). \end{aligned}$$

$$(1.5.52)$$

Similar estimates can be obtained for  $w_s(t, x)(s = m + 1, \dots, n)$ . Thus,

$$\begin{aligned} &\omega(\bar{t},\eta|w) \\ \leq &\widetilde{C}_{29}(K) \bigg( \omega(\eta|w(0,\cdot)) + \|B\|_1 \big( \omega(\eta|c_t) + \omega(\eta|c_x) \big) + \|c\|_1 (\omega(\eta|B_t) + \omega(\eta|B_x)) \\ &+ \big(1 + \|u\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \big) (\eta + (1 + \|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Gamma) \big) \\ &+ (\|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Lambda_x) + \omega(\eta|H_t) + \int_0^{\bar{t}} \omega(\tau,\eta|w) d\tau \bigg), \end{aligned}$$

$$(1.5.53)$$

which, by Gronwall's inequality, leads to

$$\begin{aligned} &\omega(T_1,\eta|w) \\ \leq \widetilde{C}_{30}(K) \left( \omega(\eta|w(0,\cdot)) + \|B\|_1 \left( \omega(\eta|c_t) + \omega(\eta|c_x) \right) + \|c\|_1 (\omega(\eta|B_t) + \omega(\eta|B_x)) \\ &+ \left( 1 + \|u\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \right) (\eta + (1 + \|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Gamma)) \\ &+ (\|u\|_1 + \|B\|_1 \|c\|_1) \omega(\eta|\Lambda_x) + \omega(\eta|H_t) \right). \end{aligned}$$

$$(1.5.54)$$

Let

$$R_i := \{ (t, x) | t_{i-1} \le t \le t_i, 0 \le x \le L \}.$$
(1.5.55)

where,  $i = 1, \ldots, M$  and

$$t_i - t_{i-1} \le T_1, i = 1, \dots, M \ (t_0 = 0, t_M = T).$$
 (1.5.56)

Obviously, one has

$$R(T) \subseteq \bigcup_{i=1}^{M} R_i. \tag{1.5.57}$$

Before claiming (1.5.7), we denote that

$$\omega_i(\eta|w) := \sup_{\substack{(t_1, x_1), (t_2, x_2) \in R_i \\ |t_1 - t_2| \le \eta, |x_1 - x_2| \le \eta}} |w(t_1, x_1) - w(t_2, x_2)|.$$
(1.5.58)

Clearly,  $\omega_1(\eta|w) \leq \omega(T_1, \eta|w)$  and

$$\omega(\eta|w) = \omega(T, \eta|w) \le \sum_{i}^{M} \omega_i(\eta|w).$$
(1.5.59)

Then repeating the procedures in **Step C** at most M times, we can get similar estimates (like (1.5.54)) for  $\omega_i(\eta|w)(i = 2, \dots, M)$ . Therefore, noting (1.5.59), one obtains

$$\omega(\eta|w) \leq \widetilde{C}_{31}(T,K) \Big( \omega(\eta|w(0,\cdot)) + \|B\|_1 \big( \omega(\eta|c_t) + \omega(\eta|c_x) \big) + \|c\|_1 \big( \omega(\eta|B_t) + \omega(\eta|B_x) \big) \\
+ \big(1 + \|u\|_1 + \|B\|_1 \|c\|_1 + \|G_t\|_0 + \|H\|_1 \big) \big(\eta + \big(1 + \|u\|_1 + \|B\|_1 \|c\|_1 \big) \omega(\eta|\Gamma) \big) \\
+ \big( \|u\|_1 + \|B\|_1 \|c\|_1 \big) \omega(\eta|\Lambda_x) + \omega(\eta|H_t) \Big).$$
(1.5.60)

Applying (1.5.6) to (1.5.60) concludes the proof of (1.5.7).

Step D (Continuity on  $(\varphi, \Lambda, B, c)$ ): For any  $(\varphi^{(1)}, \Lambda^{(1)}, B^{(1)}, c^{(1)}), (\varphi^{(2)}, \Lambda^{(2)}, B^{(2)}, c^{(2)}) \in (C^1[R(T)])^n \times (C^1[R(T)])^n \times (C^1[R(T)])^n \times (C^1[R(T)])^n$ , we put  $u^{(1)}(t, x; \varphi^{(1)}, \Lambda^{(1)}, B^{(1)}, c^{(1)})$  and  $u^{(2)}(t, x; \varphi^{(2)}, \Lambda^{(2)}, B^{(2)}, c^{(2)})$  be the  $C^1$  solution for the corresponding mixed initial-boundary problems (1.2.4), (1.1.7)-(1.1.8) and (1.1.10). Taken

$$\begin{split} u^*(t,x) &:= u^{(1)}(t,x;\varphi^{(1)},\Lambda^{(1)},B^{(1)},c^{(1)}) - u^{(2)}(t,x;\varphi^{(2)},\Lambda^{(2)},B^{(2)},c^{(2)}),\\ \varphi^*(x) &:= \varphi^{(1)}(x) - \varphi^{(2)}(x),\\ \Lambda^*(t,x) &:= \Lambda^{(1)}(t,x) - \Lambda^{(2)}(t,x),\\ B^*(t,x) &:= B^{(1)}(t,x) - B^{(2)}(t,x),\\ c^*(t,x) &:= c^{(1)}(t,x) - c^{(2)}(t,x). \end{split}$$

Next, we will show that

$$\lim_{\|\varphi^*\|_1+\|\Lambda^*\|_1+\|B^*\|_1+\|c^*\|_1\to 0} \|u^*\|_1 = 0.$$
(1.5.61)

Since  $u^*$  satisfies the following mixed initial-boundary problems

$$\begin{cases} \frac{\partial u_i^*}{\partial t} + \lambda_i^{(1)}(t,x) \frac{\partial u_i^*}{\partial x} = \lambda_i^*(t,x) \frac{\partial u_i^{(2)}}{\partial x} + \sum_{k=1}^n b_{ik}^*(t,x) \left( \frac{\partial c_k^{(1)}}{\partial t} + \lambda_i^{(1)}(t,x) \frac{\partial c_k^{(1)}}{\partial x} \right) \\ + \sum_{k=1}^n b_{ik}^{(2)}(t,x) \left( \frac{\partial c_k^*}{\partial t} + \lambda_i^{(1)}(t,x) \frac{\partial c_k^*}{\partial x} + \lambda_i^*(t,x) \frac{\partial c_k^{(2)}}{\partial x} \right) \quad (i = 1, \cdots, n), \\ x = 0 : u_s^* = G_s(t, u_1^{(1)}, \cdots, u_m^{(1)}) - G_s(t, u_1^{(2)}, \cdots, u_m^{(2)}) \quad (s = m + 1 \cdots, n), \\ x = L : u_r^* = G_r(t, u_{m+1}^{(1)}, \cdots, u_n^{(1)}) - G_r(t, u_{m+1}^{(2)}, \cdots, u_n^{(2)}) \quad (r = 1, \cdots, m), \\ t = 0 : u_i^* = \varphi_i^*(x) \quad (i = 1, \cdots, n). \end{cases}$$

By the result in [111, inequality (2.23)] for the inhomogeneous system, one directly obtains

$$\lim_{\|\varphi^*\|_1 + \|\Lambda^*\|_1 + \|B^*\|_1 + \|c^*\|_1 \to 0} \|u^*\|_0 = 0.$$
(1.5.62)

On the other hand, on the domain  $R(T_1)$ , noting (1.5.39) and (1.5.3), a straightforward computation yields that

$$\lim_{\|\varphi^*\|_1+\|\Lambda^*\|_1+\|B^*\|_1+\|c^*\|_1\to 0} \|u_x^*\|_0 + \|u_t^*\|_0 = 0 \quad \text{(on the domain } R(T_1)\text{)}.$$
(1.5.63)

Then repeating the procedures in (1.5.63) at most M times, one can show that

$$\lim_{\|\varphi^*\|_1+\|\Lambda^*\|_1+\|B^*\|_1+\|c^*\|_1\to 0} \|u_x^*\|_0 + \|u_t^*\|_0 = 0 \quad (\text{on the domain } R(T)).$$
(1.5.64)

We have thus proved the Lemma 1.2.1.

# Chapter 2

# On boundary control of a hyperbolic system with a vanishing characteristic speed

Contents		
2.1	Introduction and main results	57
2.2	Well-posedness of the system $(2.1.5)$ and the system $(2.1.9)$	61
2.3	Proof of Theorem 2.1.1	<b>62</b>
2.4	Proof of Theorem 2.1.2	65
2.5	Proof of Theorem 2.1.3	71

## 2.1 Introduction and main results

There are many publications concerning the exact boundary controllability for linear and nonlinear hyperbolic systems, including wave equations, Saint-Venant equations, Euler equations, etc. (see [19, 72, 89, 100] and the references therein).

As for the general first order hyperbolic systems, one can refer to the recent monograph of Li [72] for almost complete results on the exact controllability of the following 1-D quasilinear hyperbolic system (including many conservation laws)

$$U_t + A(U)U_x = B(U) (2.1.1)$$

in the context of classical solution. It is well known that if all the characteristic speeds of the system does not vanish, i.e., all the eigenvalues of A(U) are nonzero in the domain under consideration, then (2.1.1) is exactly controllable (at least locally

in  $C^1$  class) by boundary controls provided that the control time is sufficiently large. (see Li and Rao [75, 74]; see also Wang [111] for the nonautonomous case).

On the other hand, the controllability problems have been studied also in the content of weak entropy solutions for hyperbolic conservation laws since the early result on attainable set by Ancona and Marson [8]. In the scalar case, one can refer to Horsin [51] for boundary controllability result of Burgers equation and to Perrollaz [96] when an additional distributed control appears in the right hand side. In the case of systems, the first result on controllability was due to Bressan and Coclite [13]. For general strictly hyperbolic systems of conservation laws with genuinely nonlinear or linearly degenerate and non-vanishing characteristic fields, it is shown in [13] that a small BV initial state can be driven to a constant state by an open-loop control asymptotically in time (see also 9 for one-sided control). Moreover, a negative result for a class of  $2 \times 2$  hyperbolic systems was also proved in [13] that an initial data with a dense distribution of shocks can not be driven to constant by using boundary control. Different from this negative result, Glass obtained the boundary controllability of 1-D isentropic Euler equation [40]. As a further study of [40], Glass recently obtained the boundary controllability for 1-D non-isentropic Euler equation in both Eulerian and Lagrangian coordinates. It is surprising because the equivalent controllability in  $C^1$ framework is false. Actually, the physical entropy can not be changed by boundary controls since the corresponding equation of the Lagrangian system has identically vanishing characteristic speed (See [41, Remark 1.3]).

A widely open question naturally arises that whether it is possible to establish exact controllability for a hyperbolic system with vanishing characteristic speeds. The situation seems different. Notice that the solution u(t, x) of the degenerate case of the equation (essentially an ordinary differential equation)

$$\begin{cases} u_t = 0, \quad (t, x) \in (0, T) \times (0, 2\pi), \\ u(0, x) = u_0(x), \quad x \in (0, 2\pi), \end{cases}$$
(2.1.2)

obeys

$$u(t, \cdot) \equiv u_0, \quad \forall t \in [0, T].$$

$$(2.1.3)$$

Obviously, it is impossible to change the value u by using the boundary controls acting on the end x = 0 and/or  $x = 2\pi$ .

There are several directions that have been attempted to answer the above open question.

One is to add some internal controls. For systems with identically zero characteristic speeds, a general result on exact controllability has been obtained by combining internal controls acted on the components corresponding to the zero eigenvalues and boundary controls acting on the other components [82, 111]. There the internal controls depend on both variables (t, x) and are globally distributed. Later on, an exact controllability result, with some constrains concerning the compatibility between the initial and final data, was established by using only the internal controls acted on the components corresponding to nonzero eigenvalues [76]. Again the internal controls are globally distributed. It is worthy of mentioning that exact controllability for a simplified model is also realized in [76] by switching controls where the internal control is locally distributed.

Another is to loose the requirements of exact controllability if one considers only the boundary controls because internal controls are usually not applicable for some physical reasons. It is possible to obtain *partial* exact controllability by boundary controls, if one aims to control only the values of the components corresponding to the nonzero eigenvalues. [113, 114].

While the most interesting problem is whether one could establish exact controllability for a hyperbolic system with zero characteristic speeds by only boundary controls. In this direction, Gugat and Leugering [48] proved a global boundary controllability result which covers the critical case in which one of the characteristic speed is zero. In spite of the degenerate model (2.1.2), it is still possible to prove the controllability for some nonlinear system. The return method of Coron has been applied to the many situations in [19], where the linearized system is not controllable while the nonlinearity enables the corresponding nonlinear system to be controllable. One can apply the return method to realize exact boundary controllability for quasilinear hyperbolic systems with a zero characteristic speed [23], when the possible vanishing characteristic speed can be driven nonzero after sufficiently long time. However, this approach seems to be no longer valid for the system with identically zero characteristic speeds (see [23, Remark 1.3]).

In this chapter, we consider the following linear hyperbolic system with a vanishing characteristic speed:

$$\begin{cases} u_t = \alpha u_x + \beta v, \\ v_t = \gamma u, \end{cases}$$
(2.1.4)

where  $\alpha, \beta, \gamma \in \mathbb{R}$  are constants. Without loss of generality, we may assume that  $\alpha < 0$ . The problems that we are interested in are (null) controllability and asymptotic stabilization problems under only boundary controls, and how the control properties are determined by the coefficients  $\alpha, \beta, \gamma$ .

More precisely, we first study the null controllability problem for the system (2.1.4) under the boundary control  $h \in L^2(0,T)$  for some T > 0:

$$\begin{cases} u_t = \alpha u_x + \beta v, & (t, x) \in (0, T) \times (0, 2\pi), \\ v_t = \gamma u, & (t, x) \in (0, T) \times (0, 2\pi), \\ u(t, 0) - u(t, 2\pi) = h(t), & t \in (0, T). \end{cases}$$
(2.1.5)

**Definition 2.1.1** Let T > 0, the system (2.1.5) is (null) controllable if: for any  $(u_0, v_0) \in (L^2(0, 2\pi))^2$ , there exists a function  $h \in L^2(0, T)$  such that the solution of

the mixed initial-boundary problem (2.1.5) with the following initial data

$$(u(0,x), v(0,x)) = (u_0(x), v_0(x)), \quad x \in (0,2\pi)$$
(2.1.6)

satisfies the finial condition:

$$(u(T,x), v(T,x)) = (0,0), \quad x \in (0,2\pi).$$
(2.1.7)

The system (2.1.5) is asymptotically (null) controllable if it is (null) controllable for  $T = +\infty$  in the sense that

$$\lim_{t \to +\infty} (u(t, \cdot), v(t, \cdot)) = (0, 0) \quad in \ L^2(0, 2\pi).$$
(2.1.8)

We also consider the asymptotic stabilization problem for system (2.1.4) under a feedback law

$$\begin{cases} u_t = \alpha u_x + \beta v, & (t, x) \in (0, +\infty) \times (0, 2\pi), \\ v_t = \gamma u, & (t, x) \in (0, +\infty) \times (0, 2\pi), \\ u(t, 0) = k u(t, 2\pi), & t \in (0, +\infty). \end{cases}$$
(2.1.9)

**Definition 2.1.2** The system (2.1.5) is asymptotically stabilizable if: there exits  $k \in \mathbb{R}$  such that for any  $(u_0, v_0) \in (L^2(0, 2\pi))^2$ , the solution of the closed-loop system (2.1.9) with the initial data (2.1.6) is asymptotically stable, i.e., (2.1.8) holds.

The Main results that we obtain in this chapter are the following theorems.

**Theorem 2.1.1** For any given  $0 < T < +\infty$ , the system (2.1.5) is not (null) controllable.

**Theorem 2.1.2** The system (2.1.9) is asymptotically stabilizable if and only if  $\beta \gamma < 0$ .

**Theorem 2.1.3** If  $\beta \gamma < 0$ , the system (2.1.5) is asymptotically (null) controllable.

The rest of this chapter is organized as follows. In section 2.2, we state the well-posedness of both the open-loop system (2.1.5) and the closed-loop system (2.1.9). The proofs of the Theorem 2.1.1, Theorem 2.1.2 and Theorem 2.1.3 are given in Section 2.3, Section 2.4 and Section 2.5, respectively.

# **2.2 Well-posedness of the system** (2.1.5) and the system (2.1.9)

In this section, we give the definition and well-posedness of solution to the system (2.1.5) and the system (2.1.9). The well-posedness issue is fundamental to the control problems. Here we only present the results without proof, which can be derived by classical methods, such as characteristic method [84] or by theory of semigroup of linear operators [94].

**Definition 2.2.1** Let  $0 < T \leq +\infty$ ,  $h \in L^2(0,T)$  and  $(u_0,v_0) \in (L^2(0,2\pi))^2$  be given. A solution of the initial boundary problem (2.1.5)-(2.1.6) is a function  $(u,v) \in C^0([0,T]; (L^2(0,2\pi))^2)$ , such that, for every  $\tau \in [0,T]$  (if  $T = +\infty, \tau \in [0,+\infty)$ ) and for every  $(\varphi_1,\varphi_2) \in (C^1([0,\tau] \times [0,2\pi]))^2$  such that  $\varphi_1(t,0) = \varphi_1(t,2\pi), \forall t \in [0,\tau]$ , one has

$$-\int_{0}^{\tau}\int_{0}^{2\pi} \left[(\varphi_{1t} - \alpha\varphi_{1x})u + \beta v\varphi_{1}\right] dxdt + \alpha \int_{0}^{\tau} h(t)\varphi_{1}(t, 2\pi)dt + \int_{0}^{2\pi}\varphi_{1}(\tau, x)u(\tau, x)dx - \int_{0}^{2\pi}\varphi_{1}(0, x)u_{0}(x)dx = 0$$
(2.2.1)

and

$$-\int_{0}^{\tau}\int_{0}^{2\pi} [\varphi_{2t}v + \gamma u\varphi_{2}] \, dxdt + \int_{0}^{2\pi} \varphi_{2}(\tau, x)v(\tau, x)dx - \int_{0}^{2\pi} \varphi_{2}(0, x)v_{0}(x)dx = 0.$$
(2.2.2)

**Lemma 2.2.1** Let T > 0,  $(u_0, v_0) \in (L^2(0, 2\pi))^2$  and  $u \in L^2(0, T)$  be given. Then the initial boundary problem (2.1.5) with (2.1.6) has a unique solution, which depends continuously on  $(u_0, v_0, h)$ .

**Definition 2.2.2** Let  $k \in \mathbb{R}$  and  $(u_0, v_0) \in (L^2(0, 2\pi))^2$  be given. A solution of the initial boundary problem (2.1.9) and (2.1.6) is a function  $(u, v) \in C^0([0, +\infty); (L^2(0, 2\pi))^2)$ , such that, for every  $\tau \in [0, \infty)$  and for every  $(\varphi_1, \varphi_2) \in (C^1([0, \tau] \times [0, 2\pi]))^2$  such that  $k\varphi_1(t, 0) = \varphi_1(t, 2\pi)$ ,  $\forall t \in [0, \tau]$ , one has

$$-\int_{0}^{\tau}\int_{0}^{2\pi} \left[(\varphi_{1t} - \alpha\varphi_{1x})u + \beta v\varphi_{1}\right]dxdt + \int_{0}^{2\pi}\varphi_{1}(\tau, x)u(\tau, x)dx - \int_{0}^{2\pi}\varphi_{1}(0, x)u_{0}(x)dx = 0$$

and

$$-\int_{0}^{\tau}\int_{0}^{2\pi} [\varphi_{2t}v + \gamma u\varphi_{2}] \, dx dt + \int_{0}^{2\pi} \varphi_{2}(\tau, x)v(\tau, x) dx - \int_{0}^{2\pi} \varphi_{2}(0, x)v_{0}(x) dx = 0$$

**Lemma 2.2.2** Let  $k \in \mathbb{R}$  and  $(u_0, v_0) \in (L^2(0, 2\pi))^2$  be given. Then the initial boundary problem (2.1.9) with (2.1.6) has a unique solution, which depends continuously on  $(u_0, v_0)$ .

### 2.3 Proof of Theorem 2.1.1

We divide the proof of Theorem 2.1.1 into two cases:  $\beta \gamma = 0$  and  $\beta \gamma \neq 0$ .

**Case 1:**  $\beta \gamma = 0$ . In Case 1, the result is trivial in the sense that (at least) one of the two equations in (2.1.5) is decoupled from the other one.

**Case 1.1:**  $\gamma = 0$ . Since  $\gamma = 0$ , the second equation in (2.1.5) becomes

$$v_t = 0, \quad (t, x) \in (0, T) \times (0, 2\pi),$$
(2.3.1)

which implies that  $v(t, \cdot) = v_0(\cdot) \in L^2(0, 2\pi)$  for all  $t \in [0, T]$ . Therefore, it is impossible to drive v from  $v_0 \neq 0 \in L^2(0, 2\pi)$  to zero at time T no matter what the control h is.

**Case 1.2:**  $\beta = 0$  and  $\gamma \neq 0$ . Since  $\beta = 0$ , the first equation in (2.1.5) becomes

$$u_t = \alpha u_x, \tag{2.3.2}$$

which does not depend on v. Suppose that the system (2.1.5) is (null) controllable. Let

$$(u_0(x), v_0(x)) \equiv (1, 0), \quad x \in (0, 2\pi),$$

$$(2.3.3)$$

then there exits  $h \in L^2(0, 2\pi)$  driving (u, v) from  $(u_0, v_0)$  to (0, 0) at time T. Multiplying (2.3.2) by  $e^{inx}$   $(n \in \mathbb{Z})$  and integrating on  $(0, T) \times (0, 2\pi)$ , we get from  $\gamma \neq 0$  and (2.1.5), (2.1.7) and (2.3.3) that

$$0 = \int_{0}^{T} \int_{0}^{2\pi} (u_{t} - \alpha u_{x}) e^{inx} dx dt$$
  
=  $\int_{0}^{2\pi} \left[ u e^{inx} \right]_{t=0}^{t=T} dx - \alpha \int_{0}^{T} \left[ u e^{inx} \right]_{x=0}^{x=2\pi} dt + in\alpha \int_{0}^{T} \int_{0}^{2\pi} u e^{inx} dx dt$   
=  $-\int_{0}^{2\pi} u_{0}(x) e^{inx} + \alpha \int_{0}^{T} h(t) dt + \frac{in\alpha}{\gamma} \int_{0}^{2\pi} \left[ v e^{inx} \right]_{t=0}^{t=T} dx dt$   
=  $-\int_{0}^{2\pi} e^{inx} + \alpha \int_{0}^{T} h(t) dt, \quad \forall n \in \mathbb{Z}.$  (2.3.4)

Therefore  $\int_0^T h(t) = 0$  as  $n \neq 0$ , while  $\int_0^T h(t) = -\frac{2\pi}{\alpha} \neq 0$  as n = 0. This contradiction concludes the proof of Theorem 2.1.1 for the **Case 1.2:**  $\beta \gamma = 0$ .

**Case 2:**  $\beta \gamma \neq 0$ . In this case, we will prove Theorem 2.1.1 by contradiction argument. The contradiction is derived by moment theory, with other applications on controllability in [91] and [99].

Let us assume that there exists  $h \in L^2(0, 2\pi)$  such that the solution (u, v) of (2.1.5) with (2.1.6) satisfies (2.1.7). We introduce the adjoint system of the system (2.1.5):

$$\begin{cases} p_t = \alpha p_x - \gamma q, \\ q_t = -\beta p, \\ p(t,0) = p(t,2\pi). \end{cases}$$
(2.3.5)

Then for the above solution (u, v) of the system (2.1.5) satisfying (2.1.7), it holds, by duality, that for all solution (p, q) of the adjoint system (2.3.5),

$$\int_{0}^{2\pi} u_{0}(x)p(0,x) + v_{0}(x)q(0,x)dx$$

$$= -\left[\left\langle \left(\begin{array}{c} u \\ v \end{array}\right), \left(\begin{array}{c} p \\ q \end{array}\right) \right\rangle_{L^{2}(0,2\pi)}\right]_{t=0}^{t=T}$$

$$= -\int_{0}^{T} \frac{d}{dt} \left\langle \left(\begin{array}{c} u \\ v \end{array}\right), \left(\begin{array}{c} p \\ q \end{array}\right) \right\rangle_{L^{2}(0,2\pi)} dt$$

$$= -\int_{0}^{T} \left\langle \left(\begin{array}{c} u_{t} \\ v_{t} \end{array}\right), \left(\begin{array}{c} p \\ q \end{array}\right) \right\rangle_{L^{2}(0,2\pi)} + \left\langle \left(\begin{array}{c} u \\ v \end{array}\right), \left(\begin{array}{c} p_{t} \\ q \end{array}\right) \right\rangle_{L^{2}(0,2\pi)} dt$$

$$= -\int_{0}^{T} \left[ \alpha u(t,x)p(t,x) \right]_{x=0}^{x=2\pi} dt$$

$$= \alpha \int_{0}^{T} h(t)p(t,0)dt.$$
(2.3.6)

Next we look for the solutions of the adjoint system (2.3.5). Clearly, the corresponding eigenvalue  $\lambda$  and the eigenvector  $(\xi, \eta)$  obey

$$\begin{cases} \lambda \xi = \alpha \xi' - \gamma \eta \\ \lambda \eta = -\beta \xi \\ \xi(0) = \xi(2\pi) \end{cases}$$
(2.3.7)

By canceling  $\eta$ , (2.3.7) is reduced to

$$\alpha\lambda\xi'(x) = (\lambda^2 - \beta\gamma)\xi(x) \quad \text{with} \quad \xi(0) = \xi(2\pi). \tag{2.3.8}$$

It is easy to see that  $\lambda = 0$  happens only when  $\xi(x) = \eta(x) \equiv 0$  since  $\beta \gamma \neq 0$ . Hence (2.3.7) possesses only nonzero eigenvalues.

Since  $\alpha < 0$ , the solution of (2.3.8) is given by

$$\xi(x) = e^{\frac{\lambda^2 - \beta\gamma}{\lambda\alpha}x} \xi(0), \qquad (2.3.9)$$

where the eigenvalue  $\lambda$  is determined by

$$e^{\frac{\lambda^2 - \beta\gamma}{\lambda\alpha}2\pi} = 1, \qquad (2.3.10)$$

i.e.

$$\frac{\lambda^2 - \beta\gamma}{\lambda\alpha} = \mathrm{i}n, \quad n \in \mathbb{Z}.$$
(2.3.11)

Therefore, we obtain the eigenvalues of the equations (2.3.7):

$$\lambda_n^{\pm} = \begin{cases} \frac{\mathrm{i}n\alpha \pm \sqrt{4\beta\gamma - n^2\alpha^2}}{2}, & \text{if } n^2 < \frac{4\beta\gamma}{\alpha^2}, n \in \mathbb{Z} \\ \frac{\mathrm{i}n\alpha \pm \mathrm{i}\sqrt{n^2\alpha^2 - 4\beta\gamma}}{2}, & \text{if } n^2 \ge \frac{4\beta\gamma}{\alpha^2}, n \in \mathbb{Z} \end{cases}$$
(2.3.12)

and the corresponding eigenvectors:

$$(\xi_n^{\pm}(x), \eta_n^{\pm}(x)) = e^{\operatorname{i}nx} \left(1, \frac{-\beta}{\lambda_n^{\pm}}\right), \quad n \in \mathbb{Z}.$$
(2.3.13)

Obviously for all  $n \in \mathbb{Z}$ ,  $(p_n^{\pm}(t,x), q_n^{\pm}(t,x)) = e^{\lambda_n^{\pm}t}(\xi_n^{\pm}(x), \eta_n^{\pm}(x)) = e^{\lambda_n^{\pm}t + inx} (1, \frac{-\beta}{\lambda_n^{\pm}})$ satisfies (2.3.5). Substituting  $(p_n^{\pm}(t,x), q_n^{\pm}(t,x))$  into (2.3.6) yields that the control  $h \in L^2(0,T)$  driving  $(u_0, v_0)$  to (0,0) satisfies that

$$\int_{0}^{T} h(t)e^{\lambda_{n}^{\pm}t}dt = \frac{1}{\alpha\lambda_{n}^{\pm}}\int_{0}^{2\pi} (\lambda_{n}^{\pm}u_{0}(x) - \beta v_{0}(x))e^{inx}dx, \quad \forall n \in \mathbb{Z}.$$
 (2.3.14)

Let

$$(u_0(x), v_0(x)) \equiv (c_1, c_2), \quad x \in (0, 2\pi),$$
 (2.3.15)

where  $c_1, c_2 \in \mathbb{R}$  are two constants such that

$$\lambda_0^{\pm} c_1 - \beta c_2 \neq 0. \tag{2.3.16}$$

Thus

$$\int_0^T h(t)e^{\lambda_n^{\pm}t}dt = \frac{\lambda_n^{\pm}c_1 - \beta c_2}{\alpha\lambda_n^{\pm}} \int_0^{2\pi} e^{\mathrm{i}nx}dx = 0, \quad \forall n \in \mathbb{Z} \setminus \{0\},$$
(2.3.17)

$$\int_{0}^{T} h(t)e^{\lambda_{0}^{\pm}t}dt = \frac{2\pi(\lambda_{0}^{\pm}c_{1} - \beta c_{2})}{\alpha\lambda_{0}^{\pm}} \neq 0.$$
(2.3.18)

Let

$$F(z) = \int_0^T h(t)e^{-izt}dt, \quad z \in \mathbb{C}.$$
(2.3.19)

Then it follows that  $F(i\lambda_0^{\pm}) \neq 0$  and  $F(i\lambda_n^{\pm}) = 0$  for all  $n \in \mathbb{Z} \setminus \{0\}$ . On the other hand, F is an entire function on  $\mathbb{C}$  due to Paley-Wiener Theorem [63]. It is easy to get from  $\alpha < 0$  and (2.3.12) that

$$\lim_{n \to +\infty} \lambda_n^+ = 0. \tag{2.3.20}$$

Therefore F is an entire function vanishing on the set  $\{z = i\lambda_n^{\pm} | n \in \mathbb{Z} \setminus \{0\}\}$  which has an accumulation point  $z = 0 \in \mathbb{C}$ . This simply implies  $F \equiv 0$ , however, this contradicts with  $F(i\lambda_0^+) \neq 0$ . Consequently, there exists no function  $h \in L^2(0,T)$ such that (2.3.14) holds for the initial data given by (2.3.15). This concludes the proof of Theorem 2.1.1 for the **Case 2:**  $\beta \gamma \neq 0$ .

### 2.4 Proof of Theorem 2.1.2

In this section, we will prove that  $k \in \mathbb{R}$  does exist such that the closed-loop system (2.1.9) is asymptotically stable if and only if  $\beta \gamma < 0$ . The proof relies on the Riesz basis method and a careful analysis of the spectrum.

**Proposition 2.4.1** All the the eigenvalues of the closed loop system (2.1.9) have a strictly negative real part if and only if  $\beta\gamma < 0$  and 0 < |k| < 1.

**Proof of Proposition 2.4.1.** Clearly, the corresponding eigenvalue  $\lambda$  and the eigenvector  $(\phi, \psi)$  satisfy

$$\begin{cases} \lambda(\phi,\psi)^{tr} = \mathcal{A}(\phi,\psi)^{tr} \\ \phi(0) = k\phi(2\pi) \end{cases}$$
(2.4.1)

where

$$\mathcal{A} = \begin{pmatrix} \alpha \partial_x & \beta \\ \gamma & 0 \end{pmatrix} \tag{2.4.2}$$

is the infinitesimal generator of the semigroup corresponding to the system (2.1.9). By canceling  $\psi$ , (2.4.1) is reduced to

$$\alpha \lambda \phi'(x) = (\lambda^2 - \beta \gamma) \phi(x) \quad \text{with} \quad \phi(0) = k \phi(2\pi). \tag{2.4.3}$$

It is easy to see that  $\lambda = 0$  happens only when  $\phi(x) = \psi(x) \equiv 0$  since  $\beta \gamma \neq 0$ . Hence (2.4.1) has no zero eigenvalues.

Since  $\alpha < 0$ , the solution of (2.4.3) is given by

$$\phi(x) = e^{\frac{\lambda^2 - \beta\gamma}{\lambda\alpha}x}\phi(0), \qquad (2.4.4)$$

where the eigenvalue  $\lambda$  is determined by

$$e^{\frac{\lambda^2 - \beta\gamma}{\lambda\alpha}2\pi} = \frac{1}{k}, \quad k \neq 0.$$
(2.4.5)

For every  $n \in \mathbb{Z}$ , let

$$\tilde{n} = \begin{cases} n, & \text{if } k > 0, \\ n + \frac{1}{2}, & \text{if } k < 0, \end{cases}$$
(2.4.6)

then

$$\frac{\lambda^2 - \beta\gamma}{\lambda\alpha} 2\pi = i2\tilde{n}\pi - \ln|k|, \qquad (2.4.7)$$

or equivalently,

$$\lambda^2 - \frac{\alpha}{2\pi} (i2\tilde{n}\pi - \ln|k|)\lambda - \beta\gamma = 0.$$
(2.4.8)

Therefore, for any  $n \in \mathbb{Z}$  and  $k \neq 0$ , (2.4.8) has two roots  $\lambda_{n,k}^{\pm} \in \mathbb{C}$  such that

$$\left[\lambda_{n,k}^{\pm} - \frac{\alpha}{4\pi} (i2\tilde{n}\pi - \ln|k|)\right]^2 = \beta\gamma + \frac{\alpha^2}{16\pi^2} (i2\tilde{n}\pi - \ln|k|)^2$$
(2.4.9)

Direct computations give us the real and imaginary part of  $\lambda_{n,k}^{\pm}$ :

$$\Re(\lambda_{n,k}^{\pm}) = \frac{-\alpha \ln|k|}{4\pi} \pm \sqrt{\frac{c_{n,k} + \sqrt{c_{n,k}^2 + 4d_{n,k}^2}}{2}}, \qquad (2.4.10)$$

$$\Im(\lambda_{n,k}^{\pm}) = \frac{\alpha \tilde{n}}{2} \mp \operatorname{sgn}(\tilde{n} \ln |k|) \sqrt{\frac{-c_{n,k} + \sqrt{c_{n,k}^2 + 4d_{n,k}^2}}{2}}, \qquad (2.4.11)$$

where

$$c_{n,k} = \beta \gamma + \frac{\alpha^2}{16\pi^2} (\ln^2 |k| - 4\tilde{n}^2 \pi^2), \qquad (2.4.12)$$

$$d_{n,k} = \frac{\alpha^2 \ln|k|\,\tilde{n}}{8\pi}.$$
(2.4.13)

Since  $\alpha < 0$ , it is obvious to see that if  $|k| \ge 1$ ,  $\Re(\lambda_{n,k}^+) \ge 0$ . On the other hand, if 0 < |k| < 1, we have

$$\Re(\lambda_{n,k}^{\pm}) < 0, \quad \forall n \in \mathbb{Z}$$

$$\iff \frac{\alpha^2 \ln^2 |k|}{16\pi^2} > \frac{c_{n,k} + \sqrt{c_{n,k}^2 + 4d_{n,k}^2}}{2}$$

$$\iff -\beta\gamma + \frac{\alpha^2 (\ln^2 |k| + 4\tilde{n}^2\pi^2)}{16\pi^2} > \sqrt{c_{n,k}^2 + 4d_{n,k}^2}$$

$$\iff -\frac{\beta\gamma\alpha^2 \ln^2 |k|}{4\pi^2} > 0$$

$$\iff \beta\gamma < 0. \qquad (2.4.14)$$

This ends the proof of Proposition 2.4.1.

In addition to Proposition 2.4.1, we can obtain, after careful computations to (2.4.10)-(2.4.11), the asymptotic behavior of the eigenvalues  $\lambda_{n,k}^{\pm}$  as  $n \to \infty$ :

$$\lim_{n \to \infty} \lambda_{n,k}^+ = 0 \quad \text{and} \quad \lim_{n \to \infty} |\lambda_{n,k}^-| = +\infty.$$
(2.4.15)

In view of (2.4.4) and (2.4.7), the corresponding eigenvectors can be chosen as

$$E_{n,k}^{+} := (\phi_{n,k}^{+}(x), \psi_{n,k}^{+}(x))^{tr} = e^{(\frac{-\ln|k|}{2\pi} + i\tilde{n})x} (\lambda_{n,k}^{+}, \gamma)^{tr}, \quad n \in \mathbb{Z},$$
(2.4.16)

$$E_{n,k}^{-} := (\phi_{n,k}^{-}(x), \psi_{n,k}^{-}(x))^{tr} = e^{(\frac{-\ln|k|}{2\pi} + i\tilde{n})x} (1, \frac{\gamma}{\lambda_{n,k}^{-}})^{tr}, \quad n \in \mathbb{Z}.$$
 (2.4.17)

In order to prove Theorem 2.1.2, we first prove that if  $\beta \gamma \geq 0$  and  $k \in \mathbb{R}$ , the system (2.1.9) is not asymptotically stable. According to Proposition 2.4.1, if  $\beta \gamma \geq 0$ , then for any  $k \in \mathbb{R}$ , there exists  $n \in \mathbb{Z}$ , such that  $\Re(\lambda_{n,k}^+) \geq 0$ . Let the initial data be the corresponding eigenvector :  $(u_0, v_0)^{tr} = E_{n,k}^+ \in (L^2(0, 2\pi))^2$ , then the corresponding solution of the system (2.1.9) is given by  $(u(t, \cdot), v(t, \cdot)) = e^{\lambda_{n,k}^+ t}(u_0, v_0)$ , which is obviously not stable.

Now it remains to prove that if  $\beta \gamma < 0$ , there exists suitable  $k \in \mathbb{R}$  such that the system (2.1.9) is asymptotically stable. More precisely, we apply the Riesz basis approach to prove that the solution of the system (2.1.9) satisfies (2.1.8) if  $\beta \gamma < 0$ and 0 < |k| < 1.

Let us emphasize that it is possible that some eigenvalues may coincide. Actually, the occurrence of an eigenvalue with multiplicity greater than one happens if and only if

$$\beta \gamma = -\frac{\alpha^2 \ln^2 k}{16\pi^2} < 0 \quad \text{and} \quad 0 < k < 1.$$
 (2.4.18)

Consequently, the only multiple eigenvalue is given by

$$\lambda_{0,k}^{+} = \lambda_{0,k}^{-} = \lambda_0 := -\frac{\alpha \ln k}{4\pi} < 0.$$
(2.4.19)

In this case, the dimension of the eigenspace of  $\lambda_0$  is one and the corresponding eigenvector can be chosen as

$$E_{0,k}^{+} := e^{\frac{-\ln k}{2\pi}x} (\lambda_0, \gamma)^{tr}.$$
(2.4.20)

Let  $E_{0,k}^{-}$  be the root vector corresponding to  $\lambda_0$ :

$$\mathcal{A}E_{0,k}^{-} = \lambda_0 E_{0,k}^{-} + E_{0,k}^{+} \tag{2.4.21}$$

where the operator  $\mathcal{A}$  is given by (2.4.2). Then by (2.4.18), (2.4.19), it is easy to find a typical root vector

$$E_{0,k}^{-} := e^{\frac{-\ln k}{2\pi}x} (\lambda_0 + 1, \gamma)^{tr}$$
(2.4.22)

which satisfies (2.4.21) and is linearly independent of  $E_{0,k}^+$ .

Now we claim the following proposition:

**Proposition 2.4.2** For any fixed  $\beta\gamma < 0$  and 0 < |k| < 1,  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$  forms a Riesz basis of the complex Hilbert space  $(L^2(0, 2\pi))^2$ .

**Proof of Proposition 2.4.2.** We introduce the classical orthogonal basis  $\{(e_{n,k}^+, e_{n,k}^-), n \in \mathbb{Z}\}$  of  $(L^2(0, 2\pi))^2$ , where

$$e_{n,k}^+ = e^{i\tilde{n}x}(1,0)^{tr}, \quad e_{n,k}^- = e^{i\tilde{n}x}(0,1)^{tr}, \quad n \in \mathbb{Z}.$$
 (2.4.23)

Clearly, we have

$$\langle e_{n,k}^+, e_{m,k}^- \rangle_{(L^2(0,\pi))^2} = 0, \quad \forall n, m \in \mathbb{Z},$$

$$(2.4.24)$$

$$< e_{n,k}^+, e_{m,k}^+ >_{(L^2(0,\pi))^2} = < e_{n,k}^-, e_{m,k}^- >_{(L^2(0,\pi))^2} = 2\pi\delta_{nm}, \quad \forall n, m \in \mathbb{Z},$$
 (2.4.25)

where  $\delta_{nm}$  stands for the Kronecker's delta. The proof will be divided into two cases due to the possibility of the occurrence of multiple eigenvalues.

**Case 1.** (2.4.18) **is not true.** In this case, all the eigenvalues are distinct. Since  $\lambda_{n,k}^+ \neq \lambda_{n,k}^-$  for all  $n \in \mathbb{Z}$ , the matrix  $\begin{pmatrix} \lambda_{n,k}^+ & 1 \\ \gamma & \frac{\gamma}{\lambda_{n,k}^-} \end{pmatrix}$  is invertible. Then it is easy to get from

$$(E_{n,k}^+, E_{n,k}^-) = e^{\frac{-\ln|k|}{2\pi}x} (e_{n,k}^+, e_{n,k}^-) \begin{pmatrix} \lambda_{n,k}^+ & 1\\ \gamma & \frac{\gamma}{\lambda_{n,k}^-} \end{pmatrix}.$$
 (2.4.26)

that  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$  forms a basis of the complex Hilbert space  $(L^2(0, 2\pi))^2$ .

For any  $(f,g) \in (L^2(0,2\pi))^2$ , there is a series of complex pairs  $\{(\alpha_n^+, \alpha_n^-) \in \mathbb{C}^2\}_{n \in \mathbb{Z}}$  such that

$$(f,g)^{tr} = \sum_{n \in \mathbb{Z}} (\alpha_n^+ E_{n,k}^+ + \alpha_n^- E_{n,k}^-) = \sum_{n \in \mathbb{Z}} (E_{n,k}^+, E_{n,k}^-) (\alpha_n^+, \alpha_n^-)^{tr}.$$
 (2.4.27)

By (2.4.26), it follows that

$$(f,g)^{tr} = \sum_{n \in \mathbb{Z}} (\beta_n^+ e_{n,k}^+ + \beta_n^- e_{n,k}^-) e^{-\frac{\ln|k|}{2\pi}x}, \qquad (2.4.28)$$

where

$$\begin{pmatrix} \beta_n^+ \\ \beta_n^- \end{pmatrix} = \begin{pmatrix} \lambda_{n,k}^+ & 1 \\ \gamma & \frac{\gamma}{\lambda_{n,k}^-} \end{pmatrix} \begin{pmatrix} \alpha_n^+ \\ \alpha_n^- \end{pmatrix}.$$
(2.4.29)

Notting that 0 < |k| < 1 and  $e^{\ln |k|} \le e^{\frac{\ln |k|}{2\pi}x} \le 1$ ,  $\forall x \in [0, 2\pi]$ . Then we get, from (2.4.23) and (2.4.28) that

$$2\pi \sum_{n \in \mathbb{Z}} (|\beta_n^+|^2 + |\beta_n^-|^2) \le \|(f,g)\|_{(L^2(0,2\pi))^2}^2 \le e^{-2\ln|k|} 2\pi \sum_{n \in \mathbb{Z}} (|\beta_n^+|^2 + |\beta_n^-|^2). \quad (2.4.30)$$

Moreover, it is easy to see from (2.4.29) and (2.4.15) that there exists a constant  $C_1 > 0$  such that

$$|\beta_n^+|^2 + |\beta_n^-|^2 \le C_1(|\alpha_n^+|^2 + |\alpha_n^-|^2), \quad \forall n \in \mathbb{Z}.$$
(2.4.31)

On the other hand, we can also derive from (2.4.29) that

$$\begin{pmatrix} \alpha_n^+ \\ \alpha_n^- \end{pmatrix} = \begin{pmatrix} \lambda_{n,k}^+ & 1 \\ \gamma & \frac{\gamma}{\lambda_{n,k}^-} \end{pmatrix}^{-1} \begin{pmatrix} \beta_n^+ \\ \beta_n^- \end{pmatrix}$$
$$= \begin{pmatrix} a_{n,k}^+ & b_{n,k}^+ \\ a_{n,k}^- & b_{n,k}^- \end{pmatrix} \begin{pmatrix} \beta_n^+ \\ \beta_n^- \end{pmatrix}$$
(2.4.32)

where

$$a_{n,k}^{+} = \frac{1}{\lambda_{n,k}^{+} - \lambda_{n,k}^{-}}, \quad a_{n,k}^{-} = \frac{-\lambda_{n,k}^{-}}{\lambda_{n,k}^{+} - \lambda_{n,k}^{-}},$$
 (2.4.33)

$$b_{n,k}^{+} = \frac{-\lambda_{n,k}^{-}}{\gamma(\lambda_{n,k}^{+} - \lambda_{n,k}^{-})}, \quad b_{n,k}^{-} = \frac{\lambda_{n,k}^{+}\lambda_{n,k}^{-}}{\gamma(\lambda_{n,k}^{+} - \lambda_{n,k}^{-})}.$$
(2.4.34)

Thanks to (2.4.15), we get additionally that

$$\lim_{n \to \infty} a_{n,k}^+ = 0, \quad \lim_{n \to \infty} a_{n,k}^- = 1, \quad \lim_{n \to \infty} b_{n,k}^+ = \frac{1}{\gamma}, \quad \lim_{n \to \infty} b_{n,k}^- = 0.$$
(2.4.35)

Therefore we can see that there exists a constant  $C_2 > 0$  such that

$$|\alpha_n^+|^2 + |\alpha_n^-|^2 \le C_2(|\beta_n^+|^2 + |\beta_n^-|^2), \quad \forall n \in \mathbb{Z}.$$
(2.4.36)

Obviously, (2.4.30)-(2.4.31) and (2.4.36) show that  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$  forms a Riesz basis of  $(L^2(0, 2\pi))^2$  in **Case 1**.

**Case 2.** (2.4.18) is true. In this case, we still have  $\lambda_{n,k}^+ \neq \lambda_{n,k}^-$  and (2.4.26) for all  $n \in \mathbb{Z} \setminus \{0\}$ . While for n = 0, we get from (2.4.20), (2.4.22) and (2.4.23) that

$$(E_{0,k}^+, E_{0,k}^-) = e^{\frac{-\ln k}{2\pi}x} (e_{0,k}^+, e_{0,k}^-) \begin{pmatrix} \lambda_0 & \lambda_0 + 1\\ \gamma & \gamma \end{pmatrix}.$$
 (2.4.37)

Obviously, the matrix  $\begin{pmatrix} \lambda_0 & \lambda_0 + 1 \\ \gamma & \gamma \end{pmatrix}$  is reversible since  $\gamma \neq 0$ . Then it is similar to prove that  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$  still forms a Riesz basis of  $(L^2(0, 2\pi))^2$  as in **Case 1**. This ends the proof of Proposition 2.4.2.

Thanks to Proposition 2.4.2, the solution (u, v) of the initial-boundary problem (2.1.9) can be decomposed with respect to the Riesz basis  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$ . Actually, for any given initial data  $(u_0, v_0)^{tr} \in (L^2(0, 2\pi))^2$ , there exists a series  $\{(c_{n,k}^+, c_{n,k}^-) \in \mathbb{C}^2\}_{n \in \mathbb{Z}}$  such that

$$(u_0, v_0)^{tr} = \sum_{n \in \mathbb{Z}} (c_{n,k}^+ E_{n,k}^+ + c_{n,k}^- E_{n,k}^-).$$
(2.4.38)

In order to decompose the solution in terms of the Riesz basis  $\{(E_{n,k}^+, E_{n,k}^-), n \in \mathbb{Z}\}$ , we have to discuss the various cases whether the eigenvalues are distinct or not.

**Case 1.** (2.4.18) **is not true.** In this case, all the eigenvalues are distinct. Since  $\{E_{n,k}^{\pm}\}$  is the eigenvector corresponding to the eigenvalue  $\lambda_{n,k}^{\pm}$ , it follows from (2.4.38) that

$$(u(t,\cdot),v(t,\cdot))^{tr} = \sum_{n\in\mathbb{Z}} \left( c_{n,k}^+ e^{\lambda_{n,k}^+ t} E_{n,k}^+ + c_{n,k}^- e^{\lambda_{n,k}^- t} E_{n,k}^- \right).$$
(2.4.39)

Then by Proposition 2.4.1, we get that in the case of  $\beta \gamma < 0$ ,

$$|e^{\lambda_{n,k}^{\pm}t}| < 1, \quad \forall n \in \mathbb{Z}$$

$$(2.4.40)$$

and

$$\lim_{d \to +\infty} |e^{\lambda_{n,k}^{\pm}t}| \to 0, \quad \forall n \in \mathbb{Z}.$$
(2.4.41)

Therefore, there exists  $C_3 > 0$  independent of  $(u_0, v_0)$  such that

$$\| (u(t, \cdot), v(t, \cdot)) \|_{(L^{2}(0, 2\pi))^{2}}^{2}$$

$$\leq \sum_{n \in \mathbb{Z}} \left\| c_{n,k}^{+} E_{n,k}^{+} + c_{n,k}^{-} E_{n,k}^{-} \right\|_{(L^{2}(0, 2\pi))^{2}}^{2}$$

$$\leq C_{3} \| (u_{0}, v_{0}) \|_{(L^{2}(0, 2\pi))^{2}}^{2},$$

$$(2.4.42)$$

which implies that the series on the right hand side of (2.4.39) converges uniformly and strongly in  $(L^2(0, 2\pi))^2$ . Then taking the  $(L^2(0, 2\pi))^2$  norm of (2.4.39) and letting  $t \to +\infty$ , we easily conclude by (2.4.41) that

$$\lim_{t \to +\infty} (u(t, \cdot), v(t, \cdot))^{tr} = \sum_{n \in \mathbb{Z}} \lim_{t \to +\infty} \left( c_{n,k}^+ e^{\lambda_{n,k}^+ t} E_{n,k}^+ + c_{n,k}^- e^{\lambda_{n,k}^- t} E_{n,k}^- \right) = 0.$$
(2.4.43)

This ends the proof that if  $\beta \gamma < 0$ , then the system (2.1.9) can be stabilized by taking the tuning parameter k such that 0 < |k| < 1.

**Case 2.** (2.4.18) is true. In this case, the only eigenvalue, whose multiplicity is more than one, is  $\lambda_0$  given by (2.4.19). We get from (2.4.20), (2.4.22), (2.4.23) and (2.4.38) that

$$(u(t,\cdot),v(t,\cdot))^{tr} = e^{\lambda_0 t} [(c_{0,k}^- t + c_{0,k}^+) E_{0,k}^+ + c_{0,k}^- E_{0,k}^-] + \sum_{n \in \mathbb{Z} \setminus \{0\}} \left( c_{n,k}^+ e^{\lambda_{n,k}^+ t} E_{n,k}^+ + c_{n,k}^- e^{\lambda_{n,k}^- t} E_{n,k}^- \right).$$
(2.4.44)

Again by Proposition 2.4.1, we have (2.4.41) and

$$\lim_{t \to +\infty} |e^{\lambda_0 t}t| \to 0.$$
(2.4.45)

Then we easily conclude the asymptotic stability of the closed-loop system (2.1.9), similarly as in **Case 1**. This completes the proof of Theorem 2.1.2.

### 2.5 Proof of Theorem 2.1.3

Thanks to Theorem 2.1.2, one has that if  $\beta \gamma < 0$ , there exists  $k \in \mathbb{R}$  such that the system (2.1.9) is asymptotically stable, i.e., (2.1.8) is true for all initial data  $(u_0, v_0) \in (L^2(0, 2\pi))^2$ . Let (u(t, x), v(t, x)) be the solution of the closed-loop system (2.1.9) with (2.1.6). Then according to the uniqueness of solution, it is clear that the
system (2.1.5) is asymptotically (null) controllable by taking the control as:

\_\_\_\_\_

$$h(t) = u(t,0) - u(t,2\pi).$$
(2.5.1)

This concludes the proof of Theorem 2.1.3.

# Part II

# Boundary Stabilization for Inhomogeneous Hyperbolic Systems

# Chapter 3

# Feedback Control Design of Homodirectional and General Heterodirectional Linear Coupled Hyperbolic PDEs

# Contents

3.1	Intro	oduction	76
<b>3.2</b>	$\mathbf{Syst}$	$\operatorname{em} \operatorname{description} \ldots \ldots$	78
<b>3.3</b>	Fini	te-time stabilization of heterodirectional systems	79
	3.3.1	Target system	80
	3.3.2	Backstepping transformation	82
	3.3.3	Control law and main stabilization result	84
<b>3.4</b>	Unc	ollocated observer design and output feedback con-	
	$\operatorname{troll}$	er	85
	3.4.1	Observer design	85
	3.4.2	Target system and backstepping tranformation $\ . \ . \ .$ .	86
	3.4.3	Output feedback controller	89
<b>3.5</b>	$\operatorname{Mot}$	ion planning for homodirectional systems	89
	3.5.1	Definition of the motion planning problem	89
	3.5.2	Tracking control design	90
	3.5.3	An explicit motion planning example	93
3.6	Proc	of of Theorem 3.3.1: well-posedness of the kernel	
$equations  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $			
	3.6.1	Method of characteristics	96

	3.6.2	Method of successive approximations	
	3.6.3	Convergence of the successive approximation series 101	
3.7	Con	cluding remarks	
3.8	App	Appendix: Explicit definition of the characteristic lines 106	
		-	
	3.8.1	Characteristics of the $K$ kernels	

# 3.1 Introduction

a) Background: Coupled first-order linear hyperbolic systems, typically formulated on a 1-D spatial domain normalized to the interval (0, 1), are common in modeling of traffic flow [7], heat exchangers [116], open channel flow [16, 28] or multiphase flow [29, 33, 35].

Research on stabilization of such PDEs has been dominated by the focus on pairs of counter-convecting transport PDEs with distributed local coupling. In [26] a first solution allowing actuation on only one boundary and permitting coupling coefficients of arbitrary size was presented. A recent extension [31] allows stabilization using only one control for a system containing an arbitrary number of coupled transport PDEs that convect at different speeds against the direction of the PDE whose boundary is actuated.

In this paper we present a solution to the fully general case of coupled hyperbolic PDEs. We divide such PDE systems into two categories:

- homodirectional systems of m transport PDEs, for which all the m transport velocities have the same signs, i.e., all of the PDEs convect in the same direction. Because of the finite length of the spatial domain, these are inherently stable but the coupling between states can cause undesirable transient behaviors and the trajectory planning problem is non-trivial.
- *heterodirectional* systems of n + m transport PDEs, for which there exist at least two transport velocities with opposite signs, i.e., where m PDEs convect in one direction and n PDEs convect in the opposite direction. The coupling between states traveling in opposite directions may cause instability.

In this chapter, we present control designs for the fully general case of coupled heterodirectional hyperbolic PDEs, allowing the numbers m and n of PDEs in either direction to be arbitrary, and with actuation applied on only one boundary (to all the m PDEs that convect downstream from that boundary). To solve this general problem, we solve, as a special case, the heretofore unsolved problem of control of coupled homodirectional hyperbolic linear PDEs, where multiple transport PDEs convect in the same direction, have possibly distinct speeds, and arbitrary local coupling.

Our approach is based on PDE backstepping and yields solutions to stabilization, by both full-state and observer-based output feedback, trajectory planning, and trajectory tracking problems.

b) Literature: Controllability of hyperbolic systems has first been investigated using explicit computation of the solution along the characteristic curves in the framework of  $C^1$  norm [42, 69, 98]. Later, so-called Control Lyapunov Functions methods emerged, enabling the design of dissipative boundary conditions for nonlinear hyperbolic systems in the context of both  $C^1$  norm and  $H^2$  norm [19, 21, 20]. Further, using Lyapunov functions method, sufficient boundary conditions for the exponential stability of linear [32] or nonlinear [46, 47] hyperbolic systems of balance laws have been derived. All of these results impose restrictions on the magnitude of the coupling coefficients, which are responsible for potential instabilities.

In [26], a full-state feedback control law, with actuation only on one end of the domain, which achieves  $H^2$  exponential stability of closed-loop 2–state heterodirectional linear and quasilinear hyperbolic systems is derived using a backstepping method. With a similar backstepping transformation, an output-feedback controller is designed in [31] for heterodirectional systems with m = 1 (controlled) negative velocity and n (arbitrary) positive ones. These results hold regardless of the (bounded) magnitude of the coupling coefficients. Unfortunately, the method presented in [26, 31] can not be extended to the case m > 1.

c) Contribution: The first step towards this chapter's general solution for m > 1 was presented (but not published as a paper) in [26] for m = 2 and n = 0. In conference paper [53], an extension to m = 2 and n = 1 is achieved.

The contribution of this chapter is two-fold. For (n + m)-state heterodirectional systems, we derive a stabilizing boundary feedback law that ensures finite-time convergence of all the states to zero. For homodirectional systems (for which stability is not an issue), we design a boundary control law ensuring tracking of a given reference trajectory at the uncontrolled boundary.

Both designs rely on the backstepping approach. A particular choice of the target system, featuring a cascade structure similar to [26, Section 3.5], enables the use of a classical Volterra integral transformation. Well-posedness of the system of kernel equations, which is the main technical challenge of this chapter, is proved by a method of successive approximations using a novel recursive bound.

In the case of heterodirectional systems, the approach yields a full-state feedback law that would necessitate full distributed measurements to be implemented, which is not realistic in practice. For this reason, we derive an observer relying on measurements of the states at a single boundary (the anti-controlled one). Along with the full-state feedback law, this yields an output feedback controller amenable to implementation.

d) Organization: In Section 3.2 we introduce the model equations. In Section 3.3 we present the stabilization result for heterodirectional systems: the target system is presented in Section 3.3.1 while the backstepping transformation is derived in Section 3.3.2. The design is summarized in Section 3.3.3. In Section 3.4 we present the boundary observer design. In Section 3.5 we present the motion planning result for homodirectional systems. Section 3.6 contains the main technical difficulty of the paper, i.e. the proof of well-posedness of the backstepping transformation. We conclude in Section 3.7 by discussing open problems.

# 3.2 System description

We consider the following general linear hyperbolic system

$$u_t(t,x) + \Lambda^+ u_x(t,x) = \Sigma^{++} u(t,x) + \Sigma^{+-} v(t,x)$$
(3.2.1)

$$v_t(t,x) - \Lambda^- v_x(t,x) = \Sigma^{-+} u(t,x) + \Sigma^{--} v(t,x)$$
(3.2.2)

with the following boundary conditions

$$u(t,0) = Q_0 v(t,0), v(t,1) = R_1 u(t,1) + U(t) (3.2.3)$$

where

$$u = \begin{pmatrix} u_1 & \cdots & u_n \end{pmatrix}^T, \qquad v = \begin{pmatrix} v_1 & \cdots & v_m \end{pmatrix}^T \qquad (3.2.4)$$

$$\Lambda^{+} = \begin{pmatrix} \lambda_{1} & 0 \\ & \ddots & \\ 0 & & \lambda_{n} \end{pmatrix}, \qquad \Lambda^{-} = \begin{pmatrix} \mu_{1} & 0 \\ & \ddots & \\ 0 & & \mu_{m} \end{pmatrix}$$
(3.2.5)

with

$$-\mu_1 < \dots < -\mu_m < 0 < \lambda_1 \le \dots \le \lambda_n \tag{3.2.6}$$

and

$$\Sigma^{++} = \{\sigma_{ij}^{++}\}_{1 \le i \le n, 1 \le j \le n}, \quad \Sigma^{+-} = \{\sigma_{ij}^{+-}\}_{1 \le i \le n, 1 \le j \le m},$$
  

$$\Sigma^{-+} = \{\sigma_{ij}^{-+}\}_{1 \le i \le m, 1 \le j \le n}, \quad \Sigma^{--} = \{\sigma_{ij}^{--}\}_{1 \le i \le m, 1 \le j \le m}$$
(3.2.7)

$$Q_0 = \{q_{ij}\}_{1 \le i \le n, 1 \le j \le m}, \quad R_1 = \{\rho_{ij}\}_{1 \le i \le m, 1 \le j \le n}, \quad U(t) = \begin{pmatrix} U_1(t) & \cdots & U_m(t) \end{pmatrix}^T$$
(3.2.8)

**Remark 3.2.1** We consider here constant coupling coefficients and transport velocities for the sake of readability. The method straightforwardly extends to spatially varying coefficients, with more involved technical developments.

Besides, we also make the following assumption without loss of generality

$$\forall j = 1, ..., m$$
  $\sigma_{ij}^{--} = 0,$  (3.2.9)

i.e. there are no (internal) diagonal coupling terms for v-system. Such coupling terms can be removed using a change of coordinates as presented in, e.g., [26] and [53]. This yields spatially-varying coupling terms, which is not an issue in the light of Remark 3.2.1.

**Remark 3.2.2** If two or more states have the same transport speeds (i.e.  $\mu_i = \mu_i$ for some  $i \neq j$  we refer to those states as isotachic. This case was intentionally avoided in (3.2.6). To deal with isotachic states, we consider the change of coordinates  $\bar{v}(t,x) = A(x)v(t,x)$ . The matrix A(x) is a block-diagonal matrix, with  $A_{ii} = 1$  if  $\mu_i \neq \mu_j$  for  $j \neq i$ . If there is a set of  $n_i$  isotachic states (i.e. there is i such that  $\mu_i = \mu_i$  for  $j = i + 1, \dots, i + n_i - 1$ , then there is in A(x) a corresponding block B(x) of dimension  $n_i \times n_i$  in A(x). Each of these B(x) is computed independently for each isotachic set of states. If we call  $\Sigma_{iso}$  the matrix of coupling coefficients among these isotachic states (i.e. with coefficients  $\sigma_{jk}^{--}$  for  $j, k = i, i + 1, \ldots, i + 1, \ldots, i + 1, \ldots, j +$  $n_i - 1$ ), then B(x) is computed from the initial value problem  $B(x) = 1/\mu_i B(x) \Sigma_{iso}$ ,  $B(0) = I_{n_i \times n_i}$ . It is easy to see that this transformation is invertible, since one can define a matrix C(x) from  $C(x) = 1/\mu_i \Sigma_{iso} B(x)$ ,  $C(0) = I_{n_i \times n_i}$ . One has that C(x)is the inverse of B(x) as  $B(0)C(0) = I_{n_i \times n_i}$  and  $\frac{d}{dx}B(x)C(x) = 0$ . Applying this invertible transformation eliminates the coupling coefficients between isotachic states, but results in some spatially-varying coupling terms, which is not an issue as explained in Remark 3.2.1.

# 3.3 Finite-time stabilization of heterodirectional systems

In this section, we derive a stabilizing feedback law for the general (n + m)state system. Notice that this is interesting only in the case  $n \neq 0$ , since instability arises from coupling between states traveling in opposite directions. Following the backstepping approach, we seek to map system (3.2.1)–(3.2.3) to a target system with desirable stability properties using an invertible Volterra transformation.

#### 3.3.1 Target system

#### Target system design

The strategy we used here is so-called backstepping method, which transforms the original system into a "good" (for example, exponentially stable) target system by using invertible backstepping transformation (see, for example, [19, Chapter 12.5] and [65]). Here, we want to map system (3.2.1)-(3.2.3) to the following target system

$$\alpha_t(t,x) + \Lambda^+ \alpha_x(t,x) = \Sigma^{++} \alpha(t,x) + \Sigma^{+-} \beta(t,x) + \int_0^x C^+(x,\xi) \alpha(\xi) d\xi + \int_0^x C^-(x,\xi) \beta(\xi) d\xi$$
(3.3.1)

$$\beta_t(t,x) - \Lambda^- \beta_x(t,x) = G(x)\beta(0) \tag{3.3.2}$$

with the following boundary conditions

$$\alpha(t,0) = Q_0 \beta(t,0), \qquad \beta(t,1) = 0 \qquad (3.3.3)$$

where  $C^+$  and  $C^-$  are  $L^{\infty}$  matrix functions on the domain

$$\mathcal{T} = \{ 0 \le \xi \le x \le 1 \}, \qquad (3.3.4)$$

while  $G \in L^{\infty}(0,1)$  is a lower triangular matrix with the following structures

$$G(x) = \begin{pmatrix} 0 & \cdots & \cdots & 0 \\ g_{2,1}(x) & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \vdots \\ g_{m,1}(x) & \cdots & g_{m,m-1}(x) & 0 \end{pmatrix}.$$
 (3.3.5)

The coefficients of  $C^+$ ,  $C^-$  and G will be determined in section 3.3.2.

#### Stability of the target system

The following lemma assesses the finite-time stability of the target system.

**Lemma 3.3.1** Consider system (3.3.1), (3.3.2) with boundary conditions (3.3.3). Its zero equilibrium is reached in finite time  $t = t_F$ , where

$$t_F := \frac{1}{\lambda_1} + \sum_{j=1}^m \frac{1}{\mu_j}.$$
(3.3.6)

Noting (3.3.2)-(3.3.3) and (3.3.5), we find that the  $\beta$ -system is in fact a cascade system, which allows us to explicitly solve it by recursion as follows. The explicit

solution of  $\beta_1$  is given by

$$\beta_1(t,x) = \begin{cases} \beta_1(0,x+\mu_1 t) & \text{if } t < \frac{1-x}{\mu_1}, \\ 0 & \text{if } t \ge \frac{1-x}{\mu_1}. \end{cases}$$
(3.3.7)

Notice in particular that  $\beta_1$  is identically zero for  $t \ge \mu_1^{-1}$ . From the time  $t \ge \mu_1^{-1}$  on, we have that  $\beta_2(t, x)$  satisfies the following equation

$$\beta_{2t}(t,x) - \mu_2 \beta_{2x}(t,x) = 0. \tag{3.3.8}$$

Similarly, by expressing the solution along the characteristic lines, one obtains that

$$\beta_2(t,x) \equiv 0 \quad \forall t \ge \mu_1^{-1} + \mu_2^{-1}.$$
 (3.3.9)

Thus, by mathematical induction, one can easily get that  $\beta_j (j = 1, \dots, m)$  vanishes after

$$t = \sum_{k=1}^{j} \frac{1}{\mu_k}.$$
 (3.3.10)

This yields that

$$\beta(t,x) \equiv 0, \quad t > \sum_{j=1}^{m} \frac{1}{\mu_j}.$$
 (3.3.11)

When  $t > \sum_{j=1}^{m} \frac{1}{\mu_j}$ , the  $\alpha$ -system becomes

$$\alpha_t(t,x) + \Lambda^+ \alpha_x(t,x) = \Sigma^{++} \alpha(t,x) + \int_0^x C^+(x,\xi) \alpha(\xi) d\xi$$
 (3.3.12)

with the boundary conditions

$$\alpha(t,0) = 0. \tag{3.3.13}$$

Since there are no zero eigenvalues for the  $\alpha$ -system (see (3.2.6)), we may change the status of t and x, and Equations (3.3.12) can be rewritten as

$$\alpha_x(t,x) + (\Lambda^+)^{-1}\alpha_t(t,x) = (\Lambda^+)^{-1}\Sigma^{++}\alpha(t,x) + \int_0^x (\Lambda^+)^{-1}C^+(x,\xi)\alpha(\xi)d\xi \quad (3.3.14)$$

with the initial condition (3.3.13). Then by the uniqueness of the system (3.3.13),(3.3.14), and noting the order of the transport speeds of the  $\alpha$ -system (see (3.2.6)), this yields

(see [64]) that  $\alpha$  identically vanishes for

$$t \ge \frac{1}{\lambda_1} + \sum_{j=1}^m \frac{1}{\mu_j}$$
(3.3.15)

This concludes the proof.

# 3.3.2 Backstepping transformation

To map system (3.2.1)-(3.2.3) to the target system (3.3.1)-(3.3.3), we consider the following backstepping (Volterra) transformation

$$\alpha(t,x) = u(t,x) \tag{3.3.16}$$

$$\beta(t,x) = v(t,x) - \int_0^x \left[ K(x,\xi)u(\xi) + L(x,\xi)v(\xi) \right] d\xi$$
(3.3.17)

where the kernels to be determined K and L are defined on the triangular domain  $\mathcal{T}$ . Deriving (3.3.17) with respect to space and time yields, respectively (omitting the time argument)

$$\beta_t(x) = \Lambda^- v_x(x) + \Sigma^{-+} u(x) + \Sigma^{--} v(x) + K(x, x) \Lambda^+ u(x) - K(x, 0) \Lambda^+ u(0) - \int_0^x \left[ K_{\xi}(x, \xi) \Lambda^+ + K(x, \xi) \Sigma^{++} \right] u(\xi) d\xi - \int_0^x \left[ K(x, \xi) \Sigma^{+-} \right] v(\xi) d\xi - L(x, x) \Lambda^- v(x) + L(x, 0) \Lambda^- v(0) + \int_0^x \left[ L_{\xi}(x, \xi) \Lambda^- - L(x, \xi) \Sigma^{--} \right] v(\xi) d\xi - \int_0^x \left[ L(x, \xi) \Sigma^{-+} \right] u(\xi) d\xi$$

and

$$\beta_x(x) = v_x(x) - K(x, x)u(x) - L(x, x)v(x) - \int_0^x \left[K_x(x, \xi)u(\xi) + L_x(x, \xi)v(\xi)\right] d\xi$$

Plugging into the target system equations and noting  $\beta(0) \equiv v(0)$  yields the following system of kernel equations

$$0 = K(x, x)\Lambda^{+} + \Lambda^{-}K(x, x) + \Sigma^{-+}$$
(3.3.18)

$$0 = \Lambda^{-}L(x,x) - L(x,x)\Lambda^{-} + \Sigma^{--}$$
(3.3.19)

$$L(x,0)\Lambda^{-} = K(x,0)\Lambda^{+}Q_{0} + G(x)$$
(3.3.20)

$$0 = \Lambda^{-} K_{x}(x,\xi) - K_{\xi}(x,\xi)\Lambda^{+} - K(x,\xi)\Sigma^{++} - L(x,\xi)\Sigma^{-+}$$
(3.3.21)

$$0 = \Lambda^{-} L_{x}(x,\xi) + L_{\xi}(x,\xi)\Lambda^{-} - L(x,\xi)\Sigma^{--} - K(x,\xi)\Sigma^{+-}$$
(3.3.22)

and yields the following equations for  $C^{-}(x,\xi)$  and  $C^{+}(x,\xi)$ 

$$C^{-}(x,\xi) = \Sigma^{+-}L(x,\xi) + \int_{\xi}^{x} C^{-}(x,s)L(s,\xi)d\xi \qquad (3.3.23)$$

$$C^{+}(x,\xi) = \Sigma^{+-}K(x,\xi) + \int_{\xi}^{x} C^{-}(x,s)K(s,\xi)d\xi \qquad (3.3.24)$$

**Remark 3.3.1** For each  $x \in [0,1]$ , Equation (3.3.23) is a Volterra equation of the second kind on [0, x] with  $C^{-}(x, \cdot)$  as the unknown. Besides, Equation (3.3.24) explicitly gives  $C^+(x,\xi)$  as a function of  $C^-(x,\xi)$  and  $K(x,\xi)$ . Therefore, provided the kernels K and L are well-defined and bounded, so are  $C^+$  and  $C^-$ .

Developing equations (3.3.18)–(3.3.22) leads to the following set of kernel PDEs

$$\frac{\text{for } 1 \le i \le m, \ 1 \le j \le n}{\mu_i \partial_x K_{ij}(x,\xi) - \lambda_j \partial_\xi K_{ij}(x,\xi)} = \sum_{k=1}^n \sigma_{kj}^{++} K_{ik}(x,\xi) + \sum_{p=1}^m \sigma_{pj}^{-+} L_{ip}(x,\xi) \qquad (3.3.25)$$

$$\frac{\text{for } 1 \le i \le m, \ 1 \le j \le m}{\mu_i \partial_x L_{ij}(x,\xi) + \mu_j \partial_\xi L_{ij}(x,\xi)} = \sum_{p=1}^m \sigma_{pj}^{--} L_{ip}(x,\xi) + \sum_{k=1}^n \sigma_{kj}^{+-} K_{ik}(x,\xi) \qquad (3.3.26)$$

along with the following set of boundary conditions

$$K_{ij}(x,x) = -\frac{\sigma_{ij}^{-+}}{\mu_i + \lambda_j} \stackrel{\Delta}{=} k_{ij} \qquad \text{for } 1 \le i \le m, \quad 1 \le j \le n \qquad (3.3.27)$$

$$L_{ij}(x,x) = -\frac{\sigma_{ij}^{--}}{\mu_i - \mu_j} \stackrel{\Delta}{=} l_{ij} \qquad \text{for } 1 \le i, j \le m, \quad i \ne j \qquad (3.3.28)$$

$$\mu_j L_{ij}(x,0) = \sum_{k=1}^n \lambda_k K_{ik}(x,0) q_{k,j} \quad \text{for } 1 \le i \le j \le m.$$
(3.3.29)

To ensure well-posedness of the kernel equations, we add the following artificial boundary conditions for  $L_{ij}(i > j)$ 

$$L_{ij}(1,\xi) = l_{ij}, \text{ for } 1 \le j < i \le m$$
 (3.3.30)

k=1

While the  $g_{ij}$ , for  $1 \le j < i \le n$ , are given by

$$g_{ij}(x) = \mu_j L_{ij}(x,0) - \sum_{p=1}^n \lambda_p q_{pj} K_{ip}(x,0)$$
(3.3.31)

provided the K and L kernels are properly defined by (3.3.25)-(3.3.30), which we prove in the next section.

**Remark 3.3.2** The choice of imposing (3.3.30) as the boundary condition for  $L_{ij}(1 \leq j < i \leq m)$ , on the boundary x = 1 is arbitrary and was designed to ensure continuity of some of the kernels. This degree of freedom in the control design had never appeared in previous backstepping designs for hyperbolic system [26, 31]. The impact of the boundary values of  $L_{ij}$ ,  $1 \leq j < i \leq m$  on the transient behavior of the closed-loop system remains an open question, out of the scope of this chapter.

**Remark 3.3.3** If there are isotachic states, and the transformation explained in Remark 3.2.2 is applied, then the  $L_{ij}$  kernels for i, j corresponding to isotachic states  $(\mu_i = \mu_j)$  have all boundary conditions of the type (3.3.29) instead of (3.3.28)—which would become singular—or (3.3.30). The results that follow do not change, but we have omitted the case for the sake of brevity.

The well-posedness of the target system equations is assessed in the following Theorem.

**Theorem 3.3.1** Consider system (3.3.25)–(3.3.30). There exists a unique solution K and L in  $L^{\infty}(\mathcal{T})$ . Moreover, all the boundary traces for the K-kernel and L-kernel are functions of  $L^{\infty}(0, 1)$ .

The proof of this Theorem is the main technical difficulty of the chapter and is presented in Section 3.6.

#### 3.3.3 Control law and main stabilization result

We are now ready to state the main stabilization result as follows.

**Theorem 3.3.2** Consider system (3.2.1)-(3.2.2) with boundary conditions (3.2.3) and the following feedback control law (3.3.32)

$$U(t) = \int_0^1 \left[ K(1,\xi)u(\xi) + L(1,\xi)v(\xi) \right] d\xi - R_1 u(t,1), \qquad (3.3.32)$$

where K and L are the solutions of system (3.3.25)-(3.3.30), the existence of which are guaranteed by the Theorem 3.3.1. For any initial condition  $(u_0, v_0) \in (L^{\infty}(0, 1))^{(n+m)\times(n+m)}$ , the zero equilibrium is reached in finite time  $t = t_F$ , where  $t_F$  is given by (3.3.6).

First, notice that evaluating transformation (3.3.17) at x = 1 yields (3.3.32). Besides, rewriting transformation (3.3.17) as follows

$$\begin{pmatrix} \alpha(t,x)\\ \beta(t,x) \end{pmatrix} = \begin{pmatrix} u(t,x)\\ v(t,x) \end{pmatrix} - \int_0^x \begin{pmatrix} 0 & 0\\ K(x,\xi) & L(x,\xi) \end{pmatrix} \begin{pmatrix} u(t,\xi)\\ v(t,\xi) \end{pmatrix} d\xi. \quad (3.3.33)$$

one notices that it is a classical Volterra equation of the second kind. One can check from, e.g., [50] and [108] that there exists a unique matrix function  $\mathcal{R} \in (L^{\infty}(\mathcal{T}))^{(n+m)\times(n+m)}$  such that

$$\begin{pmatrix} u(t,x)\\v(t,x) \end{pmatrix} = \begin{pmatrix} \alpha(t,x)\\\beta(t,x) \end{pmatrix} - \int_0^x \mathcal{R}(x,\xi) \begin{pmatrix} \alpha(t,\xi)\\\beta(t,\xi) \end{pmatrix} d\xi.$$
(3.3.34)

Applying Lemma 3.3.1 implies that  $(\alpha, \beta)$  go to zero in finite time  $t = t_F$ , therefore, by (3.3.34), (u, v) also converge to zero in finite time.

# 3.4 Uncollocated observer design and output feedback controller

In this section, we derive an observer that relies on the measurement of the v states at the left boundary, i.e.

$$y(t) = v(t,0) \tag{3.4.1}$$

Then, using the estimates from the observer along with the control law (3.3.32), we derive an output feedback controller.

### 3.4.1 Observer design

The observer equations read as follows

$$\hat{u}_t(t,x) + \Lambda^+ \hat{u}_x(t,x) = \Sigma^{++} \hat{u}(t,x) + \Sigma^{+-} \hat{v}(t,x) - P^+(x)(\hat{v}(t,0) - v(t,0)) \quad (3.4.2)$$

$$\hat{v}_t(t,x) - \Lambda^- \hat{v}_x(t,x) = \Sigma^{-+} \hat{u}(t,x) + \Sigma^{--} \hat{v}(t,x) - P^-(x)(\hat{v}(t,0) - v(t,0)) \quad (3.4.3)$$

with the following boundary conditions

$$\hat{u}(t,0) = Q_0 v(t,0),$$
  $\hat{v}(t,1) = R_1 \hat{u}(t,1) + \hat{u}(t)$  (3.4.4)

where  $P^+(\cdot)$  and  $P^-(\cdot)$  have yet to be designed. This yields the following error system

$$\tilde{u}_t(t,x) + \Lambda^+ \tilde{u}_x(t,x) = \Sigma^{++} \tilde{u}(t,x) + \Sigma^{+-} \tilde{v}(t,x) - P^+(x)\tilde{v}(t,0)$$
(3.4.5)

$$\tilde{v}_t(t,x) - \Lambda^- \tilde{v}_x(t,x) = \Sigma^{-+} \tilde{u}(t,x) + \Sigma^{--} \tilde{v}(t,x) - P^-(x)\tilde{v}(t,0)$$
(3.4.6)

with the following boundary conditions

$$\tilde{u}(t,0) = 0,$$
  $\tilde{v}(t,1) = R_1 \tilde{u}(t,1)$  (3.4.7)

**Remark 3.4.1** One should notice that the output is directly injected at the left boundary, which means potential sensor noise is only filtered throughout the spatial domain. Combining the approach of [31] and the cascade structure of (3.3.1)–(3.3.3), we now derive a target system and backstepping transformation to design observer gains  $P^+(\cdot)$  and  $P^-(\cdot)$  that yield finite-time stability of the error system (3.4.5)– (3.4.7).

#### 3.4.2 Target system and backstepping tranformation

We map system (3.4.5)–(3.4.7) to the following target system

$$\tilde{\alpha}_t(t,x) + \Lambda^+ \tilde{\alpha}_x(t,x) = \Sigma^{++} \tilde{\alpha}(t,x) + \int_0^x D^+(x,\xi) \tilde{\alpha}(\xi) d\xi \qquad (3.4.8)$$

$$\tilde{\beta}_t(t,x) - \Lambda^- \tilde{\beta}_x(t,x) = \Sigma^{-+} \tilde{\alpha}(t,x) + \int_0^x D^-(x,\xi) \tilde{\alpha}(\xi) d\xi \qquad (3.4.9)$$

with the following boundary conditions

$$\tilde{\alpha}(t,0) = 0, \qquad \qquad \tilde{\beta}(t,1) = R_1 \tilde{\alpha}(t,1) - \int_0^1 H(\xi) \tilde{\beta}(\xi) d\xi \qquad (3.4.10)$$

where  $D^+$  and  $D^-$  are  $L^{\infty}$  matrix functions on the domain  $\mathcal{T}$  and  $H \in L^{\infty}(0, 1)$  is an upper triangular matrix with the following structure

$$H(x) = \begin{pmatrix} 0 & h_{1,2}(x) & \cdots & h_{1,m}(x) \\ \vdots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & h_{m-1,m}(x) \\ 0 & \cdots & \cdots & 0 \end{pmatrix}$$
(3.4.11)

all of which have yet to be determined.

**Proposition 3.4.1 (Stability of the observer target system)** The solutions of system (3.4.8)–(3.3.3) converge to zero in finite time. More precisely, one has

$$\forall t \ge t_F, \quad \tilde{\alpha} \equiv \tilde{\beta} \equiv 0 \tag{3.4.12}$$

where  $t_F$  is defined by (3.3.6).

The system consists in a cascade of the  $\tilde{\alpha}$ -system (that has zero input at the left boundary) into the  $\tilde{\beta}$ -system. Further, the  $\tilde{\beta}$  is a cascade of its slow states into its fast states. The rigorous proof follows the same steps that the proof of Lemma 3.3.1 and is therefore omitted here. To map system (3.4.5)-(3.4.7) to the target system (3.4.8)-(3.4.10), we consider the following backstepping (Volterra) transformation

$$\tilde{u}(t,x) = \tilde{\alpha}(t,x) + \int_0^x M(x,\xi)\tilde{\beta}(\xi)d\xi \qquad (3.4.13)$$

$$\tilde{v}(t,x) = \tilde{\beta}(t,x) + \int_0^x N(x,\xi)\tilde{\beta}(\xi)d\xi \qquad (3.4.14)$$

where the kernels to be determined M and N are defined on the triangular domain  $\mathcal{T}$ . Deriving (3.4.13),(3.4.14) with respect to space and time yields (omitting the time argument)

$$\begin{aligned} 0 &= \tilde{u}_t(x) + \Lambda^+ \tilde{u}_x(x) - \Sigma^{++} \tilde{u}(x) - \Sigma^{+-} \tilde{v}(x) + P^+(x) \tilde{v}(0) \\ &= \left[ -P^+(x) - M(x,0)\Lambda^- \right] \tilde{\beta}(0) + \left[ \Lambda^+ M(x,x) + M(x,x)\Lambda^- - \Sigma^{+-} \right] \tilde{\beta}(x) \\ &+ \int_0^x \left[ D^+(x,\xi) + M(x,\xi)\Sigma^{-+} + \int_{\xi}^x M(x,s)D^-(s,\xi)ds \right] \tilde{\alpha}(\xi)d\xi \\ &+ \int_0^x \left[ \Lambda^+ M_x(x,\xi) - M_{\xi}(x,\xi)\Lambda^- - \Sigma^{+-}N(x,\xi) - \Sigma^{++}M(x,\xi) \right] \tilde{\beta}(\xi)d\xi \end{aligned}$$

and

$$\begin{aligned} 0 &= \tilde{v}_t(x) + \Lambda^+ \tilde{v}_x(x) - \Sigma^{-+} \tilde{u}(x) - \Sigma^{--} \tilde{v}(x) + P^-(x) \tilde{v}(0) \\ &= \left[ -P^-(x) + N(x,0)\Lambda^- \right] \tilde{\beta}(0) + \left[ \Lambda^- N(x,x) - N(x,x)\Lambda^- \right] \tilde{\beta}(t,x) \\ &+ \int_0^x \left[ D^-(x,\xi) + N(x,\xi)\Sigma^{-+} + \int_{\xi}^x N(x,s)D^-(s,\xi)ds \right] \alpha(\xi)d\xi \\ &- \int_0^x \left[ \Lambda^- N_x(x,\xi) + N_{\xi}(x,\xi)\Lambda^- + \Sigma^{-+}M(x,\xi) + \Sigma^{--}N(x,\xi) \right] \beta(\xi)d\xi \end{aligned}$$

which yields

$$\frac{\text{for } 1 \le i \le n, \ 1 \le j \le m}{\lambda_i \partial_x M_{ij}(x,\xi) - \mu_j \partial_\xi M_{ij}(x,\xi)} = \sum_{k=1}^n \sigma_{ik}^{++} M_{kj}(x,\xi) + \sum_{p=1}^m \sigma_{ip}^{+-} N_{pj}(x,\xi) \qquad (3.4.15)$$
  
for  $1 \le i \le m, \ 1 \le j \le m$ 

$$\mu_i \overline{\partial_x N_{ij}(x,\xi) + \mu_j \partial_\xi N_{ij}(x,\xi)} = \sum_{k=1}^n \sigma_{ik}^{-+} M_{kj}(x,\xi) + \sum_{p=1}^m \sigma_{ip}^{--} N_{pj}(x,\xi) \qquad (3.4.16)$$

along with the following set of boundary conditions

$$M_{ij}(x,x) = \frac{\sigma_{ij}^{+-}}{\mu_i + \lambda_j} \stackrel{\Delta}{=} m_{ij} \qquad \text{for } 1 \le i \le m, \quad 1 \le j \le n \qquad (3.4.17)$$

$$N_{ij}(x,x) = 0$$
 for  $1 \le i, j \le m, \quad i \ne j$  (3.4.18)

besides, evaluating (3.4.13), (3.4.14) at x = 1 yields

$$N_{ij}(1,x) = \sum_{k=1}^{n} \rho_{ik} M_{kj}(1,x) \qquad \text{for } 1 \le j \le i \le m.$$
(3.4.19)

To ensure well-posedness of the kernel equations, we add the following artificial boundary conditions for  $N_{ij}(i < j)$ 

$$N_{ij}(x,0) = 0,$$
 for  $1 \le i < j \le m$  (3.4.20)

while the  $d_{ij}^+$ ,  $d_{ij}^-$  and  $h_{ij}$  are given by

$$h_{ij}(x) = N_{ij}(1, x) - \sum_{k=1}^{n} \rho_{ik} M_{kj}(1, x)$$
(3.4.21)

$$d_{ij}^{+}(x,\xi) = -\sum_{k=1}^{m} M_{ik}(x,\xi)\sigma_{kj}^{-+} + \int_{\xi}^{x} \sum_{k=1}^{m} M_{ik}(x,s)d_{kj}^{-}(s,\xi)ds$$
(3.4.22)

$$d_{ij}^{-}(x,\xi) = -\sum_{k=1}^{m} N_{ik}(x,\xi)\sigma_{kj}^{-+} + \int_{\xi}^{x} \sum_{k=1}^{m} N_{ik}(x,s)d_{kj}^{-}(s,\xi)ds$$
(3.4.23)

provided the M and N kernels are properly defined. Interestingly, the well-posedness of the system of kernel equations of the observer (3.4.15)-(3.4.20) is equivalent to that of the controller kernels (3.3.25)-(3.3.30). Indeed, considering the following alternate variables

$$\bar{M}_{ij}(\chi, y) = M_{ij}(1 - y, 1 - \chi) = M_{ij}(x, \xi), \quad \bar{N}_{ij}(\chi, y) = N_{ij}(1 - y, 1 - \chi) = N_{ij}(x, \xi)$$
(3.4.24)

yields

$$\frac{\text{for } 1 \le i \le n, \ 1 \le j \le m}{\mu_j \partial_\chi \bar{M}_{ij}(\chi, y) - \lambda_i \partial_y \bar{M}_{ij}(\chi, y)} = -\sum_{k=1}^n \sigma_{ik}^{++} \bar{M}_{kj}(\chi, y) - \sum_{p=1}^m \sigma_{ip}^{+-} \bar{N}_{pj}(\chi, y) \quad (3.4.25)$$

$$\frac{\text{for } 1 \le i \le m, \ 1 \le j \le m}{\mu_j \partial_\chi \bar{N}_{ij}(\chi, y) + \mu_i \partial_y \bar{N}_{ij}(\chi, y)} = -\sum_{k=1}^n \sigma_{ik}^{-+} \bar{M}_{kj}(\chi, y) - \sum_{p=1}^m \sigma_{ip}^{--} \bar{N}_{pj}(\chi, y) \quad (3.4.26)$$

along with the following set of boundary conditions

$$\bar{M}_{ij}(\chi,\chi) = \frac{\sigma_{ij}^{+-}}{\mu_i + \lambda_j} \stackrel{\Delta}{=} m_{ij} \quad \text{for } 1 \le i \le m, \quad 1 \le j \le n \quad (3.4.27)$$

$$\bar{N}_{ij}(\chi,\chi) = 0 \quad \text{for } 1 \le i,j \le m, \quad i \ne j \quad (3.4.28)$$

$$\bar{N}_{ij}(\chi,0) = \sum_{k=1}^{n} \rho_{ik} \bar{M}_{kj}(\chi,0) \quad \text{for } 1 \le j \le i \le m. \quad (3.4.29)$$

$$\bar{N}_{ij}(1,y) = 0,$$
 for  $1 \le i < j \le m$  (3.4.30)

which has the exact same structure as the controller kernel system, the well-posedness of which is assessed in Theorem 3.3.1.

## 3.4.3 Output feedback controller

The estimates can be used in an observer-controller scheme to derive an output feedback law yielding finite-time stability of the zero equilibrium. More precisely, we have the following Lemma.

**Lemma 3.4.1** Consider the system composed of the original (3.2.1)–(3.2.3) and target systems (3.4.2)–(3.4.4) with the following control law

$$U(t) = \int_0^1 \left[ K(1,\xi)\hat{u}(\xi) + L(1,\xi)\hat{v}(\xi) \right] d\xi - R_1\hat{u}(t,1)$$
(3.4.31)

where K and L are defined by (3.3.25)-(3.3.30). Its solutions  $(u, v, \hat{u}, \hat{v})$  converge in finite time to zero.

Proposition 3.4.1 along with the existence of the observer backstepping transformation (3.4.13),(3.4.14) yields convergence of the observer error states  $\tilde{u}$ ,  $\tilde{v}$  defined by (3.4.5)–(3.4.7) to zero for  $t \geq t_F^*$ . Therefore, for  $t \geq t_F$ , one has  $v(t,0) = \hat{v}(t,0)$ and Theorem 3.3.2 applies to the observer system (3.4.2)–(3.4.4). Therefore, for  $t \geq$  $2t_F$ , one has  $(\tilde{u}, \tilde{v}, \hat{u}, \hat{v}) \equiv 0$  which also yields  $(u, v) \equiv 0$ .

# **3.5** Motion planning for homodirectional systems

## 3.5.1 Definition of the motion planning problem

Consider now the case n = 0. Then system (3.2.1)–(3.2.3) reduces to

$$v_t(t,x) - \Lambda^- v_x(t,x) = \Sigma^{--} v(t,x), \qquad (3.5.1)$$

<sup>\*</sup>the proof of this claim follows the exact same steps as in the controller case, see Section 3.3.3

where coefficients  $\Lambda^-$  and  $\Sigma^{--}$  defined as in (3.2.5) and (3.2.7), with boundary conditions

$$v(t,1) = U(t). \tag{3.5.2}$$

For simplicity in this section we drop the super-indices in the coefficients.

Equation (3.5.1) represents a system of m states moving in the same direction (in this case, from right to left). We call such a system *homodirectional* (in oposition with *heterodirectional* systems, whose states move in different directions, such as (3.2.1)–(3.2.3) with  $n, m \neq 0$ ). Homodirectional systems are inherently finite-time stable. Physically, this is due to the fact that they are transport equations with information flowing only in one direction; thus, setting U(t) to zero in (3.5.2) and solving the equations with the method of characteristics, we obtain  $u(t, x) \equiv 0$  for  $t \geq \frac{1}{\mu_m}$  (the slowest transport time in (3.5.1)).

For (3.5.1)–(3.5.2) we consider the following *motion planning* problem. Given  $\Phi(t)$ , a known function defined as

$$\Phi(t) = \begin{pmatrix} \Phi_1(t) & \cdots & \Phi_n(t) \end{pmatrix}^T, \qquad (3.5.3)$$

find the value of U(t) so that  $v(t,0) \equiv \Phi(t)$  for  $t \ge t_M$ , for some  $t_M > 0$ .

**Remark 3.5.1** Even though the plant (3.5.1) is finite-time stable, and a formula for the states can be written by using the method of characteristics, the motion planning problem is not trivial to solve. The entanglement of different states moving with different speeds severely complicates finding a solution. This design difficulty will be explicitly shown with an example in Section 3.5.3.

#### 3.5.2 Tracking control design

The following result solves the motion planning problem.

**Theorem 3.5.1** Consider system (3.5.1) with boundary conditions (3.5.2), initial condition  $v_0 \in (L^2(0,1))^m$ , and feedback control law

$$U_{i}(t) = \Phi_{i}\left(t + \frac{1}{\mu_{i}}\right) - \sum_{j=1}^{i-1} \int_{0}^{1} \frac{\mu_{j}}{\mu_{i}} L_{ij}(\xi, 0) \Phi_{j}\left(t + \frac{1-\xi}{\mu_{i}}\right) d\xi + \sum_{j=1}^{i=m} \int_{0}^{1} L_{ij}(1,\xi) v_{j}(\xi) d\xi$$
(3.5.4)

Then,  $v(t,0) \equiv \Phi(t)$  if  $t \ge t_M$ , for  $t_M = \sum_{j=1}^m \frac{1}{\mu_j}$ .

**Remark 3.5.2** The motion planning problem has been solved for the homodirectional case for the sake of clarity. However, it can be formulated for the full heterodirectional system (3.2.1)–(3.2.3) with only minor modifications. Noting  $u(t, 0) = Q_0 v(t, 0)$ , the values of some  $u_i$ 's could be chosen as part of the output instead of some of the  $v_i$ 's, for a total of m states. The only condition would be that all the rows of the output vector (written in terms of the  $v_i$ 's) are linearly independent.

We start by using the backstepping transformation (3.3.17)—where the kernels K are zero due to n being zero—to map (3.5.1)–(3.5.2) into the target system

$$\beta_t(t,x) - \Lambda \beta_x(t,x) = G(x)\beta(0), \qquad (3.5.5)$$

where G(x) was defined in Section 3.3.2 as a function of the kernels, with the following boundary conditions

$$\beta(t,1) = B(t), \tag{3.5.6}$$

where B(t) in (3.5.6) is a function defined as

$$B(t) = \begin{pmatrix} B_1(t) & \cdots & B_n(t) \end{pmatrix}^T, \qquad (3.5.7)$$

with components to be determined. B represents an extra degree of freedom that did not appear in the target system for the homodirectional control problem (Equation 3.3.3). It will be used to solve the motion planning problem. The presence of B(t) in the boundary conditions does not change the backstepping transformation; however it modifies the feedback control law to

$$U(t) = B(t) + \int_0^1 L(1,\xi)v(\xi)d\xi.$$
 (3.5.8)

Now, noticing that if one sets x = 0 in the transformation (3.3.17) one obtains  $v_i(t,0) = \beta_i(t,0)$ , it is clear that we only need to solve the motion planning problem for the target  $\beta$  system by using B(t). The next steps of the proof are devoted to finding the value of B(t).

Using the method of characteristics, the explicit solution for each state  $\beta_i(t, x)$  of (3.5.5) with boundary condition (3.5.6) at time  $t \ge \frac{1-x}{\mu_i}$  is

$$\beta_i(t,x) = B_i\left(t + \frac{x-1}{\mu_i}\right) + \sum_{j=1}^m \frac{1}{\mu_i} \int_x^1 G_{ij}(\xi)\beta_j\left(t + \frac{x-\xi}{\mu_i}, 0\right) d\xi, \quad (3.5.9)$$

Using (3.3.5) and (3.3.31) in (3.5.9), we obtain

$$\beta_i(t,x) = B_i\left(t + \frac{x-1}{\mu_i}\right) + \sum_{j=1}^{i-1} \int_x^1 \frac{\mu_j}{\mu_i} L_{ij}(\xi,0)\beta_j\left(t + \frac{x-\xi}{\mu_i},0\right) d\xi.$$
(3.5.10)

To solve now the motion planning problem, consider first (3.5.10) for i = 1 and x = 0, for  $t \ge \frac{1}{\mu_1}$ . Imposing  $\beta_1(t, 0) = \Phi_1(t)$ , we obtain:

$$\Phi_1(t) = B_1\left(t - \frac{1}{\mu_1}\right),\tag{3.5.11}$$

thus, setting  $B_1(t) = \Phi_1\left(t + \frac{1}{\mu_1}\right)$  for  $t \ge 0$ , we obtain the desired behavior for  $\beta_1(t,0)$  for  $t \ge \frac{1}{\mu_1}$ . Now consider (3.5.10) for i = 2 and x = 0, for  $t \ge \frac{1}{\mu_2}$ . Imposing  $\beta_2(t,0) = \Phi_2(t)$ , we obtain:

$$\Phi_2(t) = B_2\left(t - \frac{1}{\mu_2}\right) + \int_0^1 \frac{\mu_1}{\mu_2} L_{21}(\xi, 0)\beta_1\left(t - \frac{\xi}{\mu_2}, 0\right) d\xi.$$
(3.5.12)

Solving for  $B_2$  as before

$$B_2(t) = \Phi_2\left(t + \frac{1}{\mu_2}\right) - \int_0^1 \frac{\mu_1}{\mu_2} L_{21}(\xi, 0)\beta_1\left(t + \frac{1-\xi}{\mu_2}, 0\right) d\xi.$$
(3.5.13)

To be able to substitute  $\beta_1(t,0)$  for  $\Phi_1(t)$  in the whole domain of the integral in (3.5.13) we need to wait until  $t = \frac{1}{\mu_1}$ . Thus choosing

$$B_2(t) = \Phi_2\left(t + \frac{1}{\mu_2}\right) - \int_0^1 \frac{\mu_1}{\mu_2} L_{21}(\xi, 0) \Phi_1\left(t + \frac{1-\xi}{\mu_2}, 0\right) d\xi, \qquad (3.5.14)$$

we get that  $\beta_2(t,0) = \Phi_2(t)$  for  $t \ge \frac{1}{\mu_1} + \frac{1}{\mu_2}$  (as we have to wait an extra  $\frac{1}{\mu_2}$  time for (3.5.14) to propagate). It is clear that this procedure can be continued for  $i = 3, \ldots, m$ . Thus we obtain that

$$B_i(t) = \Phi_i\left(t + \frac{1}{\mu_i}\right) - \sum_{j=1}^{i-1} \int_0^1 \frac{\mu_j}{\mu_i} L_{ij}(\xi, 0) \Phi_j\left(t + \frac{1-\xi}{\mu_i}\right) d\xi$$
(3.5.15)

solves the motion problem for  $\beta_i$  for  $t \ge \sum_{j=1}^i \frac{1}{\mu_j}$ . Applying (3.5.15) for  $i = 1, \ldots, m$ and substituting in (3.5.8) produces the feedback law (3.5.4), thus solving the motion planning problem in time  $t_M = \sum_{j=1}^m \frac{1}{\mu_j}$ .

**Remark 3.5.3** Theorem 3.5.1 gives in fact tracking (in finite-time) of the desired output signal, a result stronger than pure motion planning. To obtain a pure motion planning result, one should take (3.5.10)—the explicit solutions of the target system obtained in the proof of the theorem—and substitute the values of  $B_i$  found in (3.5.15), so that the  $\beta_i$ 's are explicit functions of the  $\Phi_i$ 's. Then, using the inverse backstepping transformation (3.3.34), find the  $v_i$ 's as explicit functions of the  $\Phi_i$ 's and substitute them in the control law (3.5.4), which would then be an exclusive function of the outputs. We omit this result for the sake of brevity purposes.

## 3.5.3 An explicit motion planning example

Next we present an specific example of a motion planning problem for m = 2. Consider the plant

$$v_{1t}(t,x) - \mu_1 v_{1x}(t,x) = \sigma_{12} v_2(t,x), \qquad (3.5.16)$$

$$v_{2t}(t,x) - \mu_2 v_{2x}(t,x) = \sigma_{21} v_1(t,x), \qquad (3.5.17)$$

with boundary conditions

$$v_1(t,1) = U_1(t), \quad v_2(t,1) = U_2(t).$$
 (3.5.18)

The objective is to design  $U_1(t)$  and  $U_2(t)$  so that  $v_1(t,0) = \Phi_1(t)$  and  $v_2(t,0) = \Phi_2(t)$ for some functions  $\Phi_1, \Phi_2$  for  $t \ge t_M$ . Notice that since (3.5.16)–(3.5.18) is explicitly solvable, one might think that the inputs can be directly designed. Using the method of characteristics to explicitly write a solution of the system, one gets, after time  $t = \frac{1}{\mu_2}$ ,

$$v_1(t,0) = U_1\left(t - \frac{1}{\mu_1}\right) + \frac{1}{\mu_1} \int_0^1 \sigma_{12} v_2\left(t - \frac{\xi}{\mu_1}, \xi\right) d\xi, \qquad (3.5.19)$$

$$v_2(t,0) = U_2\left(t - \frac{1}{\mu_2}\right) + \frac{1}{\mu_2} \int_0^1 \sigma_{21} v_1\left(t - \frac{\xi}{\mu_2}, \xi\right) d\xi.$$
(3.5.20)

However, if one tries to proceed as in the proof of Theorem 3.5.1, by plugging in  $\Phi_1(t)$  in (3.5.19) and  $\Phi_2(t)$  in (3.5.20), and then solve for  $U_1(t)$  and  $U_2(t)$ , one ends up with a feedback law that requires knowing *future* values of  $v_1$  and  $v_2$ , i.e., a *non-causal* (and therefore not implementable) feedback law. Thus, a direct approach does not work even for the m = 2 case. To solve the motion planning problem, we resort to Theorem 3.5.1; in this particular case, the motion planning problem is solved by the inputs

$$U_1(t) = \Phi_1\left(t + \frac{1}{\mu_1}\right) + \int_0^1 L_{11}(1,\xi)v_1(\xi)d\xi + \int_0^1 L_{12}(1,\xi)v_2(\xi)d\xi, \qquad (3.5.21)$$

$$U_{2}(t) = \Phi_{2}\left(t + \frac{1}{\mu_{2}}\right) - \int_{0}^{1} \frac{\mu_{1}}{\mu_{2}} L_{21}(\xi, 0) \Phi_{1}\left(t + \frac{1-\xi}{\mu_{2}}\right) d\xi + \int_{0}^{1} L_{21}(1,\xi) v_{1}(\xi) d\xi + \int_{0}^{1} L_{22}(1,\xi) v_{2}(\xi) d\xi,$$

$$(3.5.22)$$

where the kernels  $L_{11}$ ,  $L_{12}$ ,  $L_{21}$  and  $L_{22}$  verify

$$\mu_1 \partial_x L_{11}(x,\xi) + \mu_1 \partial_\xi L_{11}(x,\xi) = \sigma_{21} L_{12}(x,\xi)$$
(3.5.23)

$$\mu_1 \partial_x L_{12}(x,\xi) + \mu_2 \partial_\xi L_{12}(x,\xi) = \sigma_{12} L_{11}(x,\xi), \qquad (3.5.24)$$

$$\mu_2 \partial_x L_{21}(x,\xi) + \mu_1 \partial_\xi L_{21}(x,\xi) = \sigma_{21} L_{22}(x,\xi)$$
(3.5.25)

$$\mu_2 \partial_x L_{22}(x,\xi) + \mu_2 \partial_\xi L_{22}(x,\xi) = \sigma_{12} L_{21}(x,\xi), \qquad (3.5.26)$$

with boundary conditions

$$L_{11}(x,0) = L_{12}(x,0) = L_{22}(x,0) = 0,$$
  

$$L_{12}(x,x) = \frac{\sigma_{12}}{\mu_2 - \mu_1}, L_{21}(x,x) = \frac{\sigma_{21}}{\mu_1 - \mu_2},$$
(3.5.27)

plus the artificial boundary condition  $L_{21}(1,\xi) = l_{21}(\xi)$ , where the function  $l_{21}$  is arbitrary. These kernel PDEs can be explicitly solved using techniques akin to those used in [107]. The resulting kernels (whose validity can be verified by substitution in the kernel equations) are

$$\begin{split} L_{11}(x,\xi) &= \begin{cases} \frac{\sqrt{\sigma_{12}\sigma_{21}}}{\mu_2 - \mu_1} \sqrt{\frac{\mu_1\xi - \mu_2 x}{\mu_1 (x - \xi)}} I_1 \left(\frac{2}{\mu_1 - \mu_2} \sqrt{\frac{\sigma_{12}\sigma_{21} (x - \xi)(\mu_1\xi - \mu_2 x)}{\mu_1}}\right), &\xi \ge \frac{\mu_2}{\mu_1} x\\ \xi < \frac{\mu_2}{\mu_1} x \end{cases} \\ L_{12}(x,\xi) &= \begin{cases} \frac{\sigma_{21}}{\mu_2 - \mu_1} I_0 \left(\frac{2}{\mu_1 - \mu_2} \sqrt{\frac{\sigma_{12}\sigma_{21} (x - \xi)(\mu_1\xi - \mu_2 x)}{\mu_1}}\right), &\xi \ge \frac{\mu_2}{\mu_1} x\\ \xi < \frac{\mu_2}{\mu_1} x \end{cases} \\ L_{21}(x,\xi) &= \frac{\sigma_{21}\xi}{\mu_1 x - \mu_2 \xi} J_0 \left(\frac{2}{\mu_1 - \mu_2} \sqrt{\frac{\sigma_{12}\sigma_{21} (x - \xi)(\mu_1 x - \mu_2 \xi)}{\mu_2}}\right) \\ &+ \mu_1 \sqrt{\frac{\sigma_{21}\mu_2 (x - \xi)}{\sigma_{12} (\mu_1 x - \mu_2 \xi)^3}} J_1 \left(\frac{2}{\mu_1 - \mu_2} \sqrt{\frac{\sigma_{12}\sigma_{21} (x - \xi)(\mu_1 x - \mu_2 \xi)}{\mu_2}}\right), \end{cases} \\ L_{22}(x,\xi) &= \xi \sqrt{\frac{\sigma_{12}\sigma_{21}}{\mu_2 (x - \xi)(\mu_1 x - \mu_2 \xi)}} J_1 \left(\frac{2}{\mu_1 - \mu_2} \sqrt{\frac{\sigma_{12}\sigma_{21} (x - \xi)(\mu_1 x - \mu_2 \xi)}{\mu_2}}\right) \end{split}$$

where  $I_0$  and  $I_1$  are the modified Bessel functions of order 0 and 1, and  $J_0$  and  $J_1$  are the (regular) Bessel functions of order 0 and 1, respectively.

The kernels appearing in (3.5.19)-(3.5.20) are depicted in Fig 1 for the case  $\mu_1 = 1$ ,  $\mu_2 = 0.2$  and  $\sigma_{12} = 2$ ,  $\sigma_{21} = 5$ . It can be seen that  $L_{11}(1,\xi)$  and  $L_{12}(1,\xi)$  have a monotone behaviour (they are always negative or zero), whereas  $L_{21}(1,\xi)$ ,  $L_{21}(\xi,0)$ , and  $L_{22}(1,\xi)$  are oscillatory. Fig. 2 shows  $L_{11}$  and  $L_{12}$  in the whole domain  $\mathcal{T}$ ; notice that  $L_{12}(x,\xi)$  is discontinuous along the line  $\xi = \frac{\mu_2}{\mu_1}$  (which is the lower domain on Figure 4c), whereas  $L_{11}(x,\xi)$  is not discontinuous. On the other hand, it is evident that  $l_{21}(\xi) = L_{21}(1,\xi)$  is rather non-trivial. In fact, the procedure that was followed to find these explicit solutions was not setting a value of  $l_{21}$  a priori, but rather *extending* the domain shown in Figure 4a up to  $x = \frac{\mu_1}{\mu_1 - \mu_2}$ , so that boundary condition (3.5.27) can be used to actually find the value of  $l_{21}$ .



Figure 1: Motion planning kernels (n = 0, m = 2). Solid:  $L_{11}(1,\xi)$  and  $L_{21}(\xi,0)$ . Dash-dotted:  $L_{12}(1,\xi)$ . Dotted:  $L_{21}(1,\xi)$ . Dashed:  $L_{22}(1,\xi)$ .



Figure 2: Motion planning kernels  $L_{11}(x,\xi)$  and  $L_{12}(x,\xi)$  (n = 0, m = 2).

# 3.6 Proof of Theorem 3.3.1: well-posedness of the kernel equations

To prove well-posedness of the kernel equations, we classically transform them into integral equations and use the method of successive approximations.

**Remark 3.6.1** Similar proofs have been derived for less general systems, e.g. in [26] or [31]. The proof is more involved here due to the existence of homodirectional controlled states, which lead to the homodirectional kernel PDEs (3.3.26).

### **3.6.1** Method of characteristics

#### Characteristics of the K kernels

For each  $1 \leq i \leq m, 1 \leq j \leq n$ , and  $(x,\xi) \in \mathcal{T}$ , we define the following characteristic lines  $(x_{ij}(x,\xi;\cdot),\xi_{ij}(x,\xi;\cdot))$  corresponding to Equations (3.3.25)

$$\begin{cases} \frac{dx_{ij}}{ds}(x,\xi;s) &= -\mu_i, \quad s \in \left[0, s_{ij}^F(x,\xi)\right] \\ x_{ij}(x,\xi;0) &= x, \; x_{ij}(x,\xi;s_{ij}^F(x,\xi)) = x_{ij}^F(x,\xi) \; ' \\ \begin{cases} \frac{d\xi_{ij}}{ds}(x,\xi;s) &= \lambda_j, \quad s \in \left[0, s_{ij}^F(x,\xi)\right] \\ \xi_{ij}(x,\xi;0) &= \xi, \; \xi_{ij}(x,\xi;s_{ij}^F(x,\xi)) = x_{ij}^F(x,\xi) \end{cases}$$
(3.6.1)

These lines, depicted on Figure 3, originate at the point  $(x,\xi)$  and terminate on the hypothenuse at the point  $(x_{ij}^F(x,\xi), x_{ij}^F(x,\xi))$ . The expressions of  $x_{ij}(x,\xi;s)$ ,  $\xi_{ij}(x,\xi;s) s_{ij}^F(x,\xi)$  and  $x_{ij}^F(x,\xi)$  are given in Appendix 3.8.1. Integrating (3.3.25)



Figure 3: Characteristic lines of the K kernels

along these characteristic lines and plugging in the boundary condition (3.3.27) yields

$$K_{ij}(x,\xi) = k_{ij} + \int_0^{s_{ij}^F(x,\xi)} \left[ \sum_{k=1}^n \sigma_{kj}^{++} K_{ik}(x_{ij}(x,\xi;s),\xi_{ij}(x,\xi;s)) + \sum_{p=1}^m \sigma_{pj}^{-+} L_{ip}(x_{ij}(x,\xi;s),\xi_{ij}(x,\xi;s)) \right] ds \quad (3.6.2)$$

#### Characteristics of the L kernels

For each  $1 \leq i \leq m, 1 \leq j \leq m$ , and  $(x,\xi) \in \mathcal{T}$ , we define the following characteristic lines  $(\chi_{ij}(x,\xi;\cdot), \zeta_{ij}(x,\xi;\cdot))$  corresponding to Equations (3.3.26)

$$\begin{cases} \frac{d\chi_{ij}}{d\nu}(x,\xi;\nu) &= \varepsilon_{ij}\mu_i, \quad \nu \in \left[0,\nu_{ij}^F(x,\xi)\right] \\ \chi_{ij}(x,\xi;0) &= x, \; \chi_{ij}(x,\xi;\nu_{ij}^F(x,\xi)) = \chi_{ij}^F(x,\xi) \;, \\ \begin{cases} \frac{d\zeta_{ij}}{d\nu}(x,\xi;\nu) &= \varepsilon_{ij}\mu_j, \quad \nu \in \left[0,\nu_{ij}^F(x,\xi)\right] \\ \zeta_{ij}(x,\xi;0) &= \xi, \; \zeta_{ij}(x,\xi;\nu_{ij}^F(x,\xi)) = \zeta_{ij}^F(x,\xi) \end{cases}$$
(3.6.3)

where  $\varepsilon_{ij}$  is defined by

$$\varepsilon_{ij}(x,\xi) = \begin{cases} 1 & \text{if } i > j \\ -1 & \text{otherwise} \end{cases}$$
(3.6.4)

These lines all originate at  $(x, \xi)$  and terminate on  $\partial \mathcal{T}$  at the point  $(\chi_{ij}^F(x,\xi), \zeta_{ij}^F(x,\xi))$ . They are depicted on Figure 4 in the three distinct cases i < j, i = j and i > j. The detailed expressions of  $\chi_{ij}(x,\xi;s)$ ,  $\zeta_{ij}(x,\xi;s)$   $\nu_{ij}^F(x,\xi)$ ,  $\chi_{ij}^F(x,\xi)$  and  $\zeta_{ij}^F(x,\xi)$  are given in Appendix 3.8.2. Integrating (3.3.26) along these characteristics and plugging in the boundary conditions (3.3.28),(3.3.29) and (3.3.30) yields

$$L_{ij}(x,\xi) = \delta_{ij}(x,\xi)l_{ij} + (1 - \delta_{ij}(x,\xi))\frac{1}{\mu_j}\sum_{r=1}^n \lambda_r q_{rj}K_{ir}(\chi_{ij}^F(x,\xi),0) -\varepsilon_{ij}\int_0^{\nu_{ij}^F(x,\xi)} \left[\sum_{p=1}^m \sigma_{pj}^{--}L_{ip}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right) + \sum_{k=1}^n \sigma_{kj}^{+-}K_{ik}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right)\right]d\nu$$
(3.6.5)

where the coefficient  $\delta_{ij}(x,\xi)$ , defined by

$$\delta_{ij}(x,\xi) = \begin{cases} 0 & \text{if } i = j \\ 0 & \text{if } i < j \text{ and } \mu_i \xi - \mu_j x \le 0 , \\ 1 & \text{otherwise} \end{cases}$$
(3.6.6)

reflects the fact that some characteristics terminate on the  $\xi = 0$  boundary of  $\mathcal{T}$ , while others terminate on the hypotenuse or on the x = 1 boundary of  $\mathcal{T}$ . Plugging in (3.6.2) evaluated at  $(\chi_{ij}^F(x,\xi), 0)$  yields



Figure 4: Characteristic lines of the L kernels

$$L_{ij}(x,\xi) = \delta_{ij}(x,\xi)l_{ij} + (1 - \delta_{ij}(x,\xi))\frac{1}{\mu_j}\sum_{r=1}^n \lambda_r q_{rj}k_{ir}$$

$$+ \frac{1 - \delta_{ij}(x,\xi)}{\mu_j}\sum_{r=1}^n \lambda_r q_{rj} \int_0^{s_{ir}^F(\chi_{ij}^F(x,\xi),0)} \left[\sum_{k=1}^n \sigma_{kr}^{++} K_{ik}(x_{ir}(\chi_{ij}^F(x,\xi),0;s),\xi_{ir}(\chi_{ij}^F(x,\xi),0;s))\right] ds$$

$$+ \sum_{p=1}^m \sigma_{pr}^{-+} L_{ip}(x_{ir}(\chi_{ij}^F(x,\xi),0;s),\xi_{ir}(\chi_{ij}^F(x,\xi),0;s))] ds$$

$$- \varepsilon_{ij} \int_0^{\nu_{ij}^F(x,\xi)} \left[\sum_{p=1}^m \sigma_{pj}^{--} L_{ip}(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)) + \sum_{k=1}^n \sigma_{kj}^{+-} K_{ik}(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu))\right] d\nu$$
(3.6.7)

# 3.6.2 Method of successive approximations

We now use the method of successive approximations to solve equations (3.6.2), (3.6.7). Define first

$$\varphi_{ij}(x,\xi) = k_{ij}, \qquad 1 \le i \le m, 1 \le j \le n$$

$$(3.6.8)$$

$$\psi_{ij}(x,\xi) = \delta_{ij}(x,\xi)l_{ij} + (1 - \delta_{ij}(x,\xi))\frac{1}{\mu_j}\sum_{r=1}^n \lambda_r q_{rj}k_{ir}, \quad 1 \le i \le m, 1 \le j \le m.$$

$$(3.6.9)$$

Besides, we define  ${\bf H}$  as the vector containing all the kernels, reordered line by line and stacked up as follows

$$\mathbf{H} = \begin{pmatrix} H_{1} \\ \vdots \\ H_{n} \\ H_{n+1} \\ \vdots \\ H_{nm} \\ H_{nm+1} \\ \vdots \\ H_{nm+m^{2}} \end{pmatrix} = \begin{pmatrix} K_{11} \\ \vdots \\ K_{1n} \\ K_{21} \\ \vdots \\ K_{mn} \\ L_{11} \\ \vdots \\ L_{mm} \end{pmatrix}, \quad \text{and similarly} \quad \boldsymbol{\phi} = \begin{pmatrix} \phi_{1} \\ \vdots \\ \phi_{n} \\ \phi_{n+1} \\ \vdots \\ \phi_{nm} \\ \phi_{nm+1} \\ \vdots \\ \phi_{nm} \\ \phi_{nm+1} \\ \vdots \\ \phi_{nm} \\ \psi_{11} \\ \vdots \\ \psi_{mm} \end{pmatrix}$$
(3.6.10)

we consider the following linear operators acting on **H**, for  $1 \le i \le m, 1 \le j \le n$ 

$$\Phi_{ij}[\mathbf{H}](x,\xi) = \int_0^{s_{ij}^F(x,\xi)} \left[ \sum_{k=1}^n \sigma_{kj}^{++} K_{ik}(x_{ij}(x,\xi;s), \xi_{ij}(x,\xi;s)) + \sum_{p=1}^m \sigma_{pj}^{-+} L_{ip}(x_{ij}(x,\xi;s), \xi_{ij}(x,\xi;s)) \right] ds \quad (3.6.11)$$

and for  $1 \leq i \leq m, 1 \leq j \leq m$ 

$$\Psi_{ij}[\mathbf{H}](x,\xi) = (1 - \delta_{ij}(x,\xi)) \frac{1}{\mu_j} \sum_{r=1}^n \lambda_r q_{rj} \int_0^{s_{ir}^F(\chi_{ij}^F(x,\xi),0)} \left[ \sum_{k=1}^n \sigma_{kr}^{++} K_{ik}(x_{ir}(\chi_{ij}^F(x,\xi),0;s),\xi_{ir}(\chi_{ij}^F(x,\xi),0;s)) \right]$$

$$+\sum_{p=1}^{m} \sigma_{pr}^{-+} L_{ip}(x_{ir}(\chi_{ij}^{F}(x,\xi),0;s),\xi_{ir}(\chi_{ij}^{F}(x,\xi),0;s)) \bigg] ds$$
$$-\varepsilon_{ij} \int_{0}^{\nu_{ij}^{F}(x,\xi)} \left[ \sum_{p=1}^{m} \sigma_{pj}^{--} L_{ip}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right) + \sum_{k=1}^{n} \sigma_{kj}^{+-} K_{ik}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right) \bigg] d\nu. \quad (3.6.12)$$

Define then the following sequence

$$\mathbf{H}^{0}(x,\xi) = 0, \qquad (3.6.13) \\
\mathbf{H}^{q}(x,\xi) = \boldsymbol{\phi}(x,\xi) + \boldsymbol{\Phi}[\mathbf{H}^{q-1}](x,\xi) \qquad (3.6.14) \\
= \begin{pmatrix} \varphi_{11}(x,\xi) + \Phi_{11}[\mathbf{H}^{q-1}](x,\xi) \\ \vdots \\ \varphi_{1n}(x,\xi) + \Phi_{1n}[\mathbf{H}^{q-1}](x,\xi) \\ \varphi_{21}(x,\xi) + \Phi_{21}[\mathbf{H}^{q-1}](x,\xi) \\ \vdots \\ \varphi_{mn}(x,\xi) + \Phi_{mn}[\mathbf{H}^{q-1}](x,\xi) \\ \vdots \\ \psi_{11}(x,\xi) + \Psi_{11}[\mathbf{H}^{q-1}](x,\xi) \end{pmatrix} \qquad (3.6.15)$$

One should notice that if the limit exists, then  $\mathbf{H} = \lim_{q \to +\infty} \mathbf{H}^q(x,\xi)$  is a solution of the integral equations, and thus solves the original hyperbolic system. Besides, define for  $q \ge 1$  the increment  $\Delta \mathbf{H}^q = \mathbf{H}^q - \mathbf{H}^{q-1}$ , with  $\Delta \mathbf{H}^0 = \boldsymbol{\phi}$  by definition. Since the functional  $\boldsymbol{\Phi}$  is linear, the following equation  $\Delta \mathbf{H}^q(x,\xi) = \boldsymbol{\Phi}[\mathbf{H}^{q-1}](x,\xi)$  holds. Using the definition of  $\Delta \mathbf{H}^q$ , it follows that if the sum  $\sum_{q=0}^{+\infty} \Delta \mathbf{H}^q(x,\xi)$  is finite, then

$$\mathbf{H}(x,\xi) = \sum_{q=0}^{+\infty} \Delta \mathbf{H}^q(x,\xi)$$
(3.6.16)

In the next section, we prove convergence of the series in  $L^{\infty}$ .

## 3.6.3 Convergence of the successive approximation series

To prove convergence of the series, we look for a recursive upper bound, similarly to, e.g. [64]. More precisely, let  $\varepsilon$  be such that

$$0 < \varepsilon < 1 - \max_{1 \le j < i \le m} \left\{ \frac{\mu_i}{\mu_j} \right\}.$$
(3.6.17)

Then, the following result holds

**Proposition 3.6.1** For  $q \ge 1$ , assume that

$$\forall (x,\xi) \in \mathcal{T}, \ \forall i = 1, ..., nm + m^2 \quad |\Delta H_i^q(x,\xi)| \le \bar{\phi} \frac{M^q (x - (1-\varepsilon)\xi)^q}{q!}$$
 (3.6.18)

where  $\Delta H_i^q(x,\xi)$  denotes the *i*-th  $(i = 1, \dots, mn + m^2)$  component of  $\Delta H^q(x,\xi)$ , then it follows that

$$\forall (x,\xi) \in \mathcal{T}, \ \forall i = 1, ..., m \forall j = 1, ..., n$$
$$|\Phi_{ij}[\Delta \mathbf{H}^q](x,\xi)| \le \bar{\phi} \frac{M^{q+1} (x - (1-\varepsilon)\xi)^{q+1}}{(q+1)!}$$
(3.6.19)

and

$$\forall (x,\xi) \in \mathcal{T}, \ \forall i = 1, ..., m \forall j = 1, ..., m \\ |\Psi_{ij}[\Delta \mathbf{H}^q](x,\xi)| \le \bar{\phi} \frac{M^{q+1} (x - (1 - \varepsilon)\xi)^{q+1}}{(q+1)!}$$
(3.6.20)

The proof of this proposition relies on the following Lemma, which is crucial and different with previous works.

**Lemma 3.6.1** For  $q \in \mathbb{N}$ ,  $(x,\xi) \in \mathcal{T}$ , and  $s_{ij}^F(x,\xi)$ ,  $\nu_{ij}^F(x,\xi)$ ,  $\chi_{ij}(x,\xi,\cdot)$ ,  $\eta_{ij}(x,\xi,\cdot)$  defined as in (3.8.2),(3.8.4),(3.6.1),(3.6.3), respectively, the following inequalities holds

$$\int_{0}^{s_{ij}^{F}(x,\xi)} (x_{ij}(x,\xi;s) - (1-\varepsilon)\xi_{ij}(x,\xi;s))^{q} ds \leq M_{\lambda} \frac{(x-(1-\varepsilon)\xi)^{q+1}}{q+1}, \quad 1 \leq i \leq m, 1 \leq j \leq m$$
(3.6.21)
$$\int_{0}^{\nu_{ij}^{F}(x,\xi)} (\chi_{ij}(x,\xi;\nu) - (1-\varepsilon)\zeta_{ij}(x,\xi;\nu))^{q} d\nu \leq M_{\lambda} \frac{(x-(1-\varepsilon)\xi)^{q+1}}{q+1}, \quad 1 \leq i,j \leq m$$
(3.6.22)

where

$$M_{\lambda} = \max_{i,s=1,\dots,m,j=1,\dots,n} \left\{ \frac{1}{\mu_i + (1-\varepsilon)\lambda_j}, \frac{1}{-\varepsilon_{ij} \left(\mu_i - (1-\varepsilon)\mu_s\right)} \right\}$$
(3.6.23)

Consider the following change of variables, noting (3.6.1),

$$\tau = x_{ij}(x,\xi;s) - (1-\varepsilon)\xi_{ij}(x,\xi;s), \quad d\tau = \left[\frac{dx_{ij}}{ds}(x,\xi;s) - (1-\varepsilon)\frac{d\xi_{ij}}{ds}(x,\xi;s)\right]ds$$

$$(3.6.24)$$

$$= (-\mu_i - (1-\varepsilon)\lambda_j)ds$$

$$(3.6.25)$$

The left-hand-side of (3.6.21) becomes

$$\int_{0}^{s_{ij}^{F}(x,\xi)} (x_{ij}(x,\xi;s) - (1-\varepsilon)\xi_{ij}(x,\xi;s))^{q} ds 
= \int_{x-(1-\varepsilon)\xi}^{x_{ij}^{F}(x,\xi) - (1-\varepsilon)\xi_{ij}^{F}(x,\xi)} \frac{-\tau^{q}}{\mu_{i} + (1-\varepsilon)\lambda_{j}} d\tau 
= \frac{(x - (1-\varepsilon)\xi)^{q+1} - (x_{ij}^{F}(x,\xi) - (1-\varepsilon)\xi_{ij}^{F}(x,\xi))^{q+1}}{(\mu_{i} + (1-\varepsilon)\lambda_{j})(q+1)} 
\leq M_{\lambda} \frac{(x - (1-\varepsilon)\xi)^{q+1}}{q+1}$$
(3.6.26)

where we have used the fact that for all  $1 \le i \le m, 1 \le j \le n$ , one has

$$x_{ij}^F(x,\xi) - (1-\varepsilon)\xi_{ij}^F(x,\xi) \ge 0$$
(3.6.27)

which is trivially satisfied since  $(x_{ij}^F(x,\xi),\xi_{ij}^F(x,\xi)) \in \partial \mathcal{T}$  and  $\varepsilon > 0$ . Consider now the following change of variables

$$\tau = \chi_{ij}(x,\xi;s) - (1-\varepsilon)\zeta_{ij}(x,\xi;s), \quad d\tau = \left[\frac{d\chi_{ij}}{ds}(x,\xi;s) - (1-\varepsilon)\frac{d\zeta_{ij}}{ds}(x,\xi;s)\right]ds$$

$$(3.6.28)$$

$$= \varepsilon_{ij}\left(\mu_i - (1-\varepsilon)\mu_j\right)ds$$

$$(3.6.29)$$

Thus, left-hand-side of (3.6.22) becomes

$$\int_{0}^{\nu_{ij}^{F}(x,\xi)} (\chi_{ij}(x,\xi;\nu) - (1-\varepsilon)\zeta_{ij}(x,\xi;\nu))^{q} d\nu 
= \int_{x-(1-\varepsilon)\xi}^{\chi_{ij}^{F}(x,\xi) - (1-\varepsilon)\zeta_{ij}^{F}(x,\xi)} \frac{\tau^{q}}{\varepsilon_{ij} (\mu_{i} - (1-\varepsilon)\mu_{j})} d\tau 
= \frac{(x - (1-\varepsilon)\xi)^{q+1} - (\chi_{ij}^{F}(x,\xi) - (1-\varepsilon)\zeta_{ij}^{F}(x,\xi))^{q+1}}{-\varepsilon_{ij} (\mu_{i} - (1-\varepsilon)\mu_{j}) (q+1)}$$
(3.6.30)

Given the definition of  $\varepsilon_{ij}$  given by (3.6.4), one has

$$-\varepsilon_{ij}\left(\mu_i - (1-\varepsilon)\mu_j\right) = \begin{cases} \mu_i - (1-\varepsilon)\mu_j & \text{if } i \le j\\ (1-\varepsilon)\mu_j - \mu_i & \text{if } i > j \end{cases}$$
(3.6.31)

Therefore, given the definition of  $\varepsilon$  (Equation (3.6.17)) in the case i > j and the ordering of the  $\mu_i$  in the case  $i \leq j$ , one has

$$-\varepsilon_{ij}\left(\mu_i - (1 - \varepsilon)\mu_j\right) > 0 \tag{3.6.32}$$

Besides, since  $(\chi_{ij}^F(x,\xi),\zeta_{ij}^F(x,\xi)) \in \mathcal{T}$ , one has  $(\chi_{ij}^F(x,\xi) - (1-\varepsilon)\zeta_{ij}^F(x,\xi)) > 0$ and (3.6.30) becomes

$$\int_{0}^{\nu_{ij}^{F}(x,\xi)} \left(\chi_{ij}(x,\xi;\nu) - (1-\varepsilon)\zeta_{ij}(x,\xi;\nu)\right)^{q} d\nu \le M_{\lambda} \frac{\left(x - (1-\varepsilon)\xi\right)^{q+1}}{q+1} \qquad (3.6.33)$$

which concludes the proof.

**Remark 3.6.2** Notice that (3.6.32) also implies that, for any  $(x, \xi) \in \mathcal{T}$  and  $1 \leq i \leq m, 1 \leq j \leq n$  the function

$$\nu \in [0, \nu_{ij}^F(x,\xi)] \mapsto \chi_{ij}(x,\xi;\nu) - (1-\varepsilon)\zeta_{ij}(x,\xi;\nu)$$
(3.6.34)

is strictly decreasing, in particular the following inequality holds

$$0 \le \chi_{ij}^F(x,\xi) - (1-\varepsilon)\zeta_{ij}^F(x,\xi) \le x - (1-\varepsilon)\xi$$
(3.6.35)

which will be useful in the proof of Proposition 3.6.1.

#### Proof of Proposition 3.6.1 Define

$$\bar{\lambda} = \max\left\{\lambda_n, \mu_1\right\}, \qquad \underline{\lambda} = \max\left\{\frac{1}{\lambda_1}, \frac{1}{\mu_n}\right\},$$
(3.6.36)

$$\bar{\sigma} = \max_{i,j} \left\{ \sigma^{++}, \sigma^{-+}, \sigma^{+-}, \sigma^{--} \right\}, \qquad \bar{q} = \max_{i,j} \{ q_{ij} \}$$
(3.6.37)

$$M = \left(n\lambda\underline{\lambda}\bar{q}+1\right)\left(n+m\right)\bar{\sigma}M_{\lambda}, \qquad \phi = \max_{i,j} \max_{(x,\xi)\in\mathcal{T}}\left\{\left|\varphi_{i,j}(x,\xi)\right|, \ \left|\psi_{i,j}(x,\xi)\right|\right\}$$
(3.6.38)

Let now  $q \in \mathbb{N}$  and assume that

$$\forall (x,\xi) \in \mathcal{T}, \ \forall i = 1, ..., nm + m^2 \quad |\Delta H_i^q(x,\xi)| \le \bar{\phi} \frac{M^q (x - (1 - \xi))^q}{q!}$$
 (3.6.39)

Then, for  $1 \le i \le m$ ,  $1 \le j \le n$ ,  $(x, \xi) \in \mathcal{T}$  one has

$$\begin{aligned} |\Phi_{ij}[\Delta \mathbf{H}^{q}](x,\xi)| &\leq \int_{0}^{s_{ij}^{F}(x,\xi)} \left| \sum_{k=1}^{n} \sigma_{kj}^{++} K_{ik}(x_{ij}(x,\xi;s), \xi_{ij}(x,\xi;s)) + \sum_{p=1}^{m} \sigma_{pj}^{-+} L_{ip}(x_{ij}(x,\xi;s), \xi_{ij}(x,\xi;s)) \right| ds \end{aligned}$$
(3.6.40)

using (3.6.21) and (3.6.39), this yields

$$\begin{split} |\Phi_{ij}[\Delta \mathbf{H}^{q}](x,\xi)| &\leq (n+m)\bar{\sigma} \int_{0}^{s_{ij}^{F}(x,\xi)} \bar{\phi} \frac{M^{q} \left(x_{ij}(x,\xi;s) - (1-\varepsilon)\xi_{ij}(x,\xi;s)\right)^{q}}{q!} ds \\ &\leq (n+m)\bar{\sigma} \frac{\bar{\phi}M^{q}}{q!} M_{\lambda} \frac{(x-(1-\xi)^{q+1}}{q+1} \\ &\leq \bar{\phi} \frac{M^{q+1}(x-(1-\varepsilon)\xi)^{q+1}}{(q+1)!} \end{split}$$
(3.6.41)

Similarly, for  $1 \leq i, j \leq m$ , one gets that

$$\begin{aligned} |\Psi_{ij}[\Delta \mathbf{H}^{q}](x,\xi)| &\leq \\ (1-\delta_{ij}(x,\xi)) \frac{1}{\mu_{j}} \sum_{r=1}^{n} \lambda_{r} q_{rj} \int_{0}^{s_{ir}^{F}(\chi_{ij}^{F}(x,\xi),0)} \left| \sum_{k=1}^{n} \sigma_{kr}^{++} K_{ik}(x_{ir}(\chi_{ij}^{F}(x,\xi),0;s),\xi_{ir}(\chi_{ij}^{F}(x,\xi),0;s)) + \sum_{p=1}^{m} \sigma_{pr}^{-+} L_{ip}(x_{ir}(\chi_{ij}^{F}(x,\xi),0;s),\xi_{ir}(\chi_{ij}^{F}(x,\xi),0;s)) \right| ds \\ + \int_{0}^{\nu_{ij}^{F}(x,\xi)} \left| \sum_{p=1}^{m} \sigma_{pj}^{--} L_{ip}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right) + \sum_{k=1}^{n} \sigma_{kj}^{+-} K_{ik}\left(\chi_{ij}(x,\xi;\nu),\zeta_{ij}(x,\xi;\nu)\right) \right| d\nu \\ (3.6.42) \end{aligned}$$

Using (3.6.39) yields

$$\begin{aligned} |\Psi_{ij}[\Delta \mathbf{H}^{q}](x,\xi)| &\leq \\ \bar{\lambda}\underline{\lambda}\bar{q}(n+m)\bar{\sigma}\sum_{r=1}^{n}\int_{0}^{s_{ir}^{F}(\chi_{ij}^{F}(x,\xi),0)}\bar{\phi}\frac{M^{q}\left(x_{ir}(\chi_{ij}^{F}(x,\xi),0;s)-(1-\varepsilon)\xi_{ir}(\chi_{ij}^{F}(x,\xi),0;s)\right)^{q}}{q!}ds \\ &+ (n+m)\bar{\sigma}\int_{0}^{\nu_{ij}^{F}(x,\xi)}\bar{\phi}\frac{M^{q}\left(\chi_{ij}(x,\xi;\nu)-(1-\varepsilon)\zeta_{ij}(x,\xi;\nu)\right)^{q}}{q!}d\nu \quad (3.6.43)\end{aligned}$$

Then, using (3.6.21) at  $(x,\xi) = (\chi_{ij}^F(x,\xi), 0)$  and (3.6.22) yields

$$|\Psi_{ij}[\Delta \mathbf{H}^{q}](x,\xi)| \leq \bar{\lambda}\underline{\lambda}\bar{q}(n+m)\bar{\sigma}n\bar{\phi}M_{\lambda}M^{q}\frac{\left(\chi_{ij}^{F}(x,\xi)-(1-\varepsilon)\zeta_{ij}^{F}(x,\xi)\right)^{q+1}}{(q+1)!} + (n+m)\bar{\sigma}\bar{\phi}\frac{M^{q}M_{\lambda}(x-(1-\varepsilon)\xi)^{q+1}}{(q+1)!} \quad (3.6.44)$$

Inequality (3.6.35) yields

$$|\Psi_{ij}[\Delta \mathbf{H}^q](x,\xi)| \le \left(n\bar{\lambda}\underline{\lambda}\bar{q}+1\right)(n+m)\bar{\sigma}\bar{\phi}M_{\lambda}\frac{M^q(x-(1-\varepsilon)\xi)^{q+1}}{(q+1)!} \tag{3.6.45}$$

$$\leq \bar{\phi} \frac{M^{q+1} (x - (1 - \varepsilon)\xi)^{q+1}}{(q+1)!} \tag{3.6.46}$$

which concludes the proof.

Proposition 3.6.1 directly leads to Theorem 3.3.1, since by the same procedures presented in [26] and [31], one has that (3.6.16) converges and

$$|\mathbf{H}(x,\xi)| = \left|\sum_{q=0}^{+\infty} \Delta \mathbf{H}^q(x,\xi)\right| \le \bar{\phi} \mathrm{e}^{M(x-(1-\varepsilon)\xi)}.$$
(3.6.47)

# 3.7 Concluding remarks

We have presented boundary control designs for a general class of linear first-order hyperbolic systems: an output-feedback law for stabilization of heterodirectional systems and a tracking controller for motion planning for homodirectional systems.

These results bridge the gap with the results of, e.g. [77], where the null (or weak) controllability of (n+m)-state heterodirectional states is proved but no explicit design is given.

Our results open the door for a large number of related problems to be solved, e.g. collocated observer design, disturbance rejection, similarly to [1], parameter identification as in [30], stabilization of quasilinear systems as in [26].

Another important question concerns the degree of freedom given by Equation (3.3.30) in the control design. The effect of the boundary value of the kernels on the transient performances of the closed-loop system is non-trivial, yet crucial for applications.

# **3.8** Appendix: Explicit definition of the characteristic lines

In this appendix, we give the detailed expressions of the characteristic lines of the kernel equations (3.3.25), (3.3.26).

# **3.8.1** Characteristics of the *K* kernels

The characteristic lines corresponding to Equations (3.3.25), for  $1 \leq i \leq m$ ,  $1 \leq j \leq n$  are given by

$$x_{ij}(x,\xi;s) = x - \mu_i s, \qquad \xi_{ij}(x,\xi;s) = \xi + \lambda_j s, \qquad s \in \left[0, s_{ij}^{F}(x,\xi)\right]$$
(3.8.1)

These lines, depicted on Figure 3 originate at  $(x, \xi)$  and arrive on the hypotenuse at  $(x_{ij}^F, x_{ij}^F)$ , with

$$s_{ij}^F(x,\xi) = \frac{x-\xi}{\mu_i + \lambda_j}, \qquad \qquad x_{ij}^F(x,\xi) = \frac{\lambda_j x + \mu_i \xi}{\lambda_j + \mu_i} \tag{3.8.2}$$

## **3.8.2** Characteristics of the *L* kernels

The characteristic lines corresponding to Equations (3.3.26), for  $1 \leq i \leq m$ ,  $1 \leq j \leq m$  are given by

$$\chi_{ij}(x,\xi;\nu) = x + \varepsilon_{ij}\mu_i\nu, \quad \zeta_{ij}(x,\xi;\nu) = \xi + \varepsilon_{ij}\mu_j\nu, \quad \nu \in \left[0,\nu_{ij}^F(x,\xi)\right] \quad (3.8.3)$$

where  $\varepsilon_{ij}(x,\xi)$  are given by (3.6.4) and

$$\nu_{ij}^{F}(x,\xi) = \begin{cases} \frac{\xi}{\mu_{j}} & \text{if } i = j \\ \frac{\xi}{\mu_{j}} & \text{if } i < j \text{ and } \mu_{i}\xi - \mu_{j}x \leq 0 \\ \frac{x-\xi}{\mu_{i}-\mu_{j}} & \text{if } (i < j \text{ and } \mu_{i}\xi - \mu_{j}x \geq 0) \text{ or } (i > j \text{ and } \mu_{i}\xi - \mu_{j}x \geq \mu_{i} - \mu_{j}) \\ \frac{1-x}{\mu_{i}} & \text{if } i > j \text{ and } \mu_{i}\xi - \mu_{j}x \leq \mu_{i} - \mu_{j}. \end{cases}$$
(3.8.4)

Therefore the terminating points  $(\chi_{ij}^F(x,\xi),\zeta_{ij}^F(x,\xi)) \in \partial \mathcal{T}$  of the characteristics are given by

$$\chi_{ij}^{F}(x,\xi) = \begin{cases} x - \xi & \text{if } i = j \\ x - \frac{\mu_{i}}{\mu_{j}} \xi & \text{if } i < j \text{ and } \mu_{i}\xi - \mu_{j}x \leq 0 \\ \frac{\mu_{i}\xi - \mu_{j}x}{\mu_{i} - \mu_{j}} & \text{if } i < j \text{ and } \mu_{i}\xi - \mu_{j}x \geq 0 \\ \frac{\mu_{j}x - \mu_{i}\xi}{\mu_{i} - \mu_{j}} & \text{if } i > j \text{ and } \mu_{i}\xi - \mu_{j}x \geq \mu_{i} - \mu_{j} \\ 1 & \text{if } i > j \text{ and } \mu_{i}\xi - \mu_{j}x \leq \mu_{i} - \mu_{j} \end{cases}$$
(3.8.5)

and

$$\zeta_{ij}^{F}(x,\xi) = \begin{cases} 0 & \text{if } i = j \\ 0 & \text{if } i < j \text{ and } \mu_{i}\xi - \mu_{j}x \leq 0 \\ \frac{\mu_{i}\xi - \mu_{j}x}{\mu_{i} - \mu_{j}} & \text{if } i < j \text{ and } \mu_{i}\xi - \mu_{j}x \geq 0 \\ \frac{\mu_{j}x - \mu_{i}\xi}{\mu_{i} - \mu_{j}} & \text{if } i > j \text{ and } \mu_{i}\xi - \mu_{j}x \geq \mu_{i} - \mu_{j} \\ \xi + \mu_{j}\frac{1 - x}{\mu_{i}} & \text{if } i > j \text{ and } \mu_{i}\xi - \mu_{j}x \leq \mu_{i} - \mu_{j}. \end{cases}$$
(3.8.6)
# Chapter 4

# Boundary Exponential Stabilization of 1-D Inhomogeneous Quasilinear Hyperbolic Systems

### Contents

4.1	4.1 Introduction and Main Result		
4.2 Preliminaries–Linear Case			
4	4.2.1	Target System	
4	4.2.2	Backstepping transformation and Kernel Equations 119	
4	4.2.3	The inverse transformation and stabilization for linear system $120$	
4.3 Backstepping boundary control design for nonlinear sys-			
	$\mathbf{tem}$		
4.4 Proof of Theorem 4.1.1			
4	4.4.1	Definitions $\dots \dots \dots$	
4	4.4.2	Analyzing the growth of $\ \gamma\ _{L^2}$	
4	4.4.3	Analyzing the growth of $\ \gamma_t\ _{L^2}$	
4	1.4.4	Analyzing the growth of $\ \gamma_{tt}\ _{L^2}$	
4	4.4.5	Proof of the $H^2$ stability for $\gamma$	
4.5 Appendix A			
4.6 Appendix B			

## 4.1 Introduction and Main Result

Consider the following 1-D  $n \times n$  inhomogeneous quasilinear hyperbolic system

$$\frac{\partial u}{\partial t} + A(x, u)\frac{\partial u}{\partial x} = F(x, u), \ x \in [0, 1], \ t \in [0, +\infty),$$
(4.1.1)

where,  $u = (u_1, \ldots, u_n)^T$  is an unknown vector function of (t, x), A(x, u) is an  $n \times n$ matrix with  $C^2$  entries  $a_{ij}(x, u)(i, j = 1, \cdots, n)$ ,  $F : [0, 1] \times \mathbb{R}^n \to \mathbb{R}^n$  is a vector valued function with  $C^2$  components  $f_i(x, u)(i = 1, \cdots, n)$  with respect to u and

$$F(x,0) \equiv 0. \tag{4.1.2}$$

Denote

$$\frac{\partial F}{\partial u}(x,0) := (f_{ij}(x))_{n \times n}, \qquad (4.1.3)$$

we assume that  $f_{ij} \in C^2([0,1])$ 

By the definition of hyperbolicity, we assume that A(x,0) is a diagonal matrix with distinct and nonzero eigenvalues  $A(x,0) = \text{diag}(\Lambda_1(x), \dots, \Lambda_n(x))$ , which are, without loss of generality, ordered as follows:

$$\Lambda_1(x) < \Lambda_2(x) < \dots < \Lambda_m(x) < 0 < \Lambda_{m+1}(x) < \dots < \Lambda_n(x), \forall x \in [0,1].$$
(4.1.4)

Here and in what follows,  $\operatorname{diag}(\Lambda_1(x), \dots, \Lambda_n(x))$  denotes the diagonal matrix whose i-th element on the diagonal is  $\Lambda_i(x)$ .

Under the assumption (4.1.4), a general kind of boundary conditions which guarantee the well- posedness of the forward problem on the domain  $\{(t, x)|t \ge 0, 0 \le x \le 1\}$  can be written as (see [69]):

$$x = 0: u_s = G_s(u_1, \cdots, u_m), \quad s = m + 1, \cdots, n,$$
(4.1.5)

$$x = 1: u_r = h_r(t), \quad r = 1, \cdots, m,$$
(4.1.6)

where  $G_s$  are  $C^2$  functions, and we assume that they vanish at the origin, i.e.

$$G_s(0, \cdots, 0) \equiv 0, s = m + 1, \cdots, n.$$
 (4.1.7)

while,  $H = (h_1, \dots, h_m)^T$  are boundary controls. Our concern, in this paper, is to design a feedback control law for H(t) in order to ensure that the closed-loop system is locally exponentially stable in the  $H^2$  norm.

In other words, we are interested in the following stabilization problem for the system (4.1.1) and (4.1.5)-(4.1.6):

**Problem (ES).** For any given  $\lambda > 0$ . Suppose that  $C^1$  compatibility conditions

are satisfied at the point (t, x) = (0, 0). Does there exist a linear feedback control  $\mathcal{B}: (H^2(0, 1))^n \to \mathbb{R}^m$ , verifying the  $C^1$  compatibility conditions at the point (t, x) = (0, 1), such that for some  $\varepsilon > 0$ , the mixed initial-boundary value problem (4.1.1), (4.1.5)-(4.1.6) and the initial conditions

$$t = 0: u(0, x) := \phi(x) = (\phi_1(x), \cdots, \phi_n(x)), \tag{4.1.8}$$

with  $H(t) = \mathcal{B}(u(t, \cdot))$  admits a unique  $C^0([0, \infty); (H^2(0, 1))^n)$  solution u = u(t, x), which satisfies

$$\|u(t,\cdot)\|_{H^2(0,L)} \le Ce^{-\lambda t} \|\phi(\cdot)\|_{H^2(0,L)},\tag{4.1.9}$$

for some C > 0, provided that  $\|\phi(\cdot)\|_{H^2(0,L)} \leq \varepsilon$ ?

The boundary stabilization problem for linear and nonlinear hyperbolic system has been widely studied in the last three decades or so. During this time, three parallel mathematical approaches have emerged. The first one is the so-called "Characteristic method", i.e. computing corresponding bounds by using explicit evolution of the solution along the characteristic curves. With this method, Problem (ES) has been previously investigated by Greenberg and Li (see [42]) for  $2 \times 2$  systems and Li and Qin (see [69, 98]) for a generalization to  $n \times n$  homogeneous systems in the framework of  $C^1$  norm. Also, this method was developed by Li and Rao [75] to study the exact boundary controllability for general inhomogeneous quasilinear hyperbolic systems.

The second method is the "Control Lyapunov Functions method", which is a useful tool to analyze the asymptotic behavior of dynamical systems. This method was first used by Coron et.al. to design dissipative boundary conditions for nonlinear homogeneous hyperbolic systems in the context of both  $C^1$  and  $H^2$  norm [20, 21, 22]. More recently, it has been shown in [25] that the exponential stability strongly depends on the considered norm, i.e. a previously known sufficient condition for exponential stability with respect to the  $H^2$  norm is not sufficient in the framework of  $C^1$  norm. Although the Control Lyapunov Functions method has been introduced to study exponential stability for hyperbolic systems of balance laws, however, finding a "good" Lyapunov Function is the main difficulty, especially when the "natural" control Lyapunov functions do not lead to arbitrarily large exponential decay rate to the original system (see [10], [19, Pages 314 and 361–371]). This phenomenon indeed happens when we deal with Problem (ES) for the inhomogeneous hyperbolic systems (see [21] and [22]).

The third one is the "Backstepping method", which is now a popular mathematical tool to stabilize the finite dimensional and infinite dimensional dynamic systems (see [64, 65, 103, 105, 106]). In [26], a full-state feedback control law, with actuation on only one end of the domain, which achieves  $H^2$  exponential stability of the closed-loop  $2 \times 2$  linear and quasilinear hyperbolic system is derived using a backstepping method. Moreover, this method ensures that the linear hyperbolic system vanishes in finite

time. Unfortunately, the method presented in [26] can not be directly extended to  $n \times n$  cases, especially when several states convecting in the same direction are controlled (see also [31]). In [53], a first step towards generalization to  $3 \times 3$  linear hyperbolic systems is addressed, in the case where two controlled states are considered. With a similar Volterra transformation, designing an appropriate form of the target system, Hu et.al.[54] adopt a classical backstepping controller to handle the Problem (ES) for general  $n \times n$  linear hyperbolic systems. Well-posedness of the system of kernel equations, which is the main technical challenge, is shown there by an improved successive approximation method.

In this paper, based on the results for the linear case [54], we will use the linearized feedback control to stabilize the nonlinear system as it is mentioned in [26]. Although the target system is a little different from the one in [26] with a linear term involved in the equations, thanks to its special structure, we show that all the procedures to handle nonlinearities in [26] can be also adapted in this paper with more technical developments. Let us recall some definitions and statements [26]. Define the norms

$$\begin{aligned} \|u(t,\cdot)\|_{H^1} &= \|u(t,\cdot)\|_{L^2} + \|u_x(t,\cdot)\|_{L^2}, \\ \|u(t,\cdot)\|_{H^2} &= \|u(t,\cdot)\|_{H^1} + \|u_{xx}(t,\cdot)\|_{L^2} \end{aligned}$$

Our main result is given by

**Theorem 4.1.1** Under the assumptions in §4.1, suppose furthermore that  $C^1$  compatibility conditions are satisfied at the point (t,x) = (0,0), there exists a continuous linear feedback control laws  $\mathcal{B} : (H^2(0,1))^n \to \mathbb{R}^m$ , satisfying the  $C^1$  compatibility conditions at the points (t,x) = (0,1), then for every  $\lambda > 0$ , there exist  $\delta > 0$  and c > 0, such that the mixed initial-boundary value problem (4.1.1), (4.1.5), (4.1.6) and (4.1.8) with  $H(t) = \mathcal{B}(u(t, \cdot))$  admits a unique  $C^0([0,\infty), (H^2(0,1))^n)$  solution u = u(t,x), which verifies

$$\|u(t,\cdot)\|_{H^2} \le ce^{-\lambda t} \|\phi\|_{H^2}, \tag{4.1.10}$$

provided that  $\|\phi\|_{H^2} \leq \delta$ .

**Remark 4.1.1** The  $C^1$  compatibility conditions at the point (t, x) = (0, 0) are given by

$$\phi_{s}(0) = G_{s}(\phi_{1}(0), \cdots, \phi_{m}(0)) \ s = m + 1, \cdots, n,$$

$$f_{s}(0, \phi(0)) - \sum_{j=1}^{n} a_{sj}(0, \phi(0))\phi'_{j}(0) =$$

$$\sum_{r=1}^{m} \frac{\partial G_{s}}{\partial u_{r}}(\phi_{1}(0), \cdots, \phi_{m}(0)) \cdot \left(f_{r}(0, \phi(0)) - \sum_{j=1}^{n} a_{rj}(0, \phi(0))\phi'_{j}(0)\right) \ s = m + 1, \cdots, n.$$

$$(4.1.12)$$

The  $C^1$  compatibility conditions at the point (t, x) = (0, 1) are similar.

**Remark 4.1.2** For convenience, we always assume that the feedback controls  $H(t) = \mathcal{B}(u(t, \cdot))$  satisfy the  $C^1$  compatibility conditions at the point (t, x) = (0, 1). However, if this property fails, one can add some dynamic terms to the controllers (see also Remark 4.3.1 and [26, Section 4]).

The rest of this paper is organized as follows. In §4.2, we review a former result on the boundary backstepping controls for  $n \times n$  linear hyperbolic system. Besides, we design a Lyapunov function to stabilize the linear system in the  $L^2$  norm. In §4.3, we input the corresponding linearized closed-loop control to the original nonlinear system and give the feedback control design. In §4.4, we prove exponential stability of zero equilibrium for the quasilinear system by using the Control Lyapunov Function method. We finally include two appendices with some technical details.

## 4.2 Preliminaries–Linear Case

In this section, we review the results on stabilization of  $n \times n$  hyperbolic linear system by using the backstepping method. Similar to the situation in [26], this procedure can be applied to locally stabilize the original nonlinear system. Consider the following  $n \times n$  hyperbolic systems

$$w_t(t,x) + \Lambda(x)w_x(t,x) = \Sigma(x)w(t,x)$$
(4.2.1)

where,  $w = (w_1, \dots, w_n)^T$  is a vector function of (t, x),  $\Lambda: [0, 1] \to \mathcal{M}_{n,n}(\mathbb{R})$  is an  $n \times n \ C^2$  diagonal matrix, i.e.

$$\Lambda(x) = \begin{pmatrix} \Lambda_{-}(x) & 0\\ 0 & \Lambda_{+}(x) \end{pmatrix}$$
(4.2.2)

in which  $\Lambda_{-}(x) := \operatorname{diag}(\lambda_{1}(x), \dots, \lambda_{m}(x))$  and  $\Lambda_{+}(x) := \operatorname{diag}(\lambda_{m+1}(x), \dots, \lambda_{n}(x))$ are diagonal submatrices, without loss of generality, satisfying

$$\lambda_1(x) < \dots < \lambda_m(x) < 0 < \lambda_{m+1}(x) < \dots < \lambda_n(x), \forall x \in [0, 1].$$

$$(4.2.3)$$

On the other hand,  $\Sigma : [0,1] \to \mathcal{M}_{n,n}(\mathbb{R})$  is a  $n \times n$  matrix with

$$\Sigma(x) = \begin{pmatrix} \Sigma^{--}(x) & \Sigma^{-+}(x) \\ \Sigma^{+-}(x) & \Sigma^{++}(x) \end{pmatrix}$$
(4.2.4)

in which  $\Sigma^{--} \in \mathcal{M}_{m,m}(\mathbb{R}), \Sigma^{-+} \in \mathcal{M}_{m,n-m}(\mathbb{R}), \Sigma^{+-} \in \mathcal{M}_{n-m,m}(\mathbb{R})$  and  $\Sigma^{--} \in \mathcal{M}_{n-m,n-m}(\mathbb{R})$  are all  $C^2$  submatrices with respect to x. Moreover, for any  $i = C^2$ 

 $1, \cdots, n$ , we assume that

$$\Sigma_{ii}(x) \equiv 0, \quad \forall x \in [0, 1]. \tag{4.2.5}$$

The boundary conditions for the linear hyperbolic system (4.2.1) are given by

$$x = 0: w_{+}(t, 0) = Qw_{-}(t, 0),$$
 (4.2.6)

and

$$x = 1: w_{-}(t, 1) = U(t).$$
(4.2.7)

where  $w_{-} \in \mathbb{R}^{m}, w_{+} \in \mathbb{R}^{n-m}$  are defined by requiring that  $w := (w_{-}, w_{+})^{T}, U = (U_{1}, \cdots, U_{m})^{T}$  are boundary feedback controls,  $Q \in \mathcal{M}_{n-m,m}$  is a constant matrix. Our purpose in this section is to find a full-state feedback control law for U(t) to ensure that the closed-loop system (4.2.1), (4.2.6)-(4.2.7) is globally asymptotically stable in the  $L^{2}$  norm, which is defined by  $||w(t, \cdot)||_{L^{2}} = \sqrt{\sum_{i=1}^{n} \int_{0}^{1} w_{i}^{2}(t, x) dx}$ .

### 4.2.1 Target System

In Section 4.2.2, it will be shown that we can transform the system (4.2.1), (4.2.6)-(4.2.7) into the following cascade system

$$\gamma_t(t,x) + \Lambda(x)\gamma_x(t,x) = G(x)\gamma(t,0) \tag{4.2.8}$$

with the boundary conditions

$$x = 0: \quad \gamma_+(t,0) = Q\gamma_-(t,0) \tag{4.2.9}$$

and

$$x = 1: \gamma_{-}(t, 1) = 0, \tag{4.2.10}$$

where  $\gamma_{-} \in \mathbb{R}^{m}, \gamma_{+} \in \mathbb{R}^{n-m}$  are defined by requiring that  $\gamma := (\gamma_{-}, \gamma_{+})^{T}$ , G is a lower triangular matrix with following structure

$$G(x) = \begin{pmatrix} \mathcal{G}_1(x) & 0\\ \mathcal{G}_2(x) & 0 \end{pmatrix}$$
(4.2.11)

in which  $\mathcal{G}_1 \in \mathcal{M}_{m,m}(\mathbb{R})$  is a lower triangular matrix, i.e.

$$\mathcal{G}_{1}(x) = \begin{pmatrix} 0 & \cdots & \cdots & 0 \\ g_{2,1}(x) & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \vdots \\ g_{m,1}(x) & \cdots & g_{m,m-1}(x) & 0 \end{pmatrix},$$
(4.2.12)

and  $\mathcal{G}_2(x) \in \mathcal{M}_{n-m,m}(\mathbb{R})$ . The coefficients of both  $\mathcal{G}_1$  and  $\mathcal{G}_2$  are to be determined in §2.2. Next, we prove that the cascade system (4.2.8)-(4.2.10) verifies the following proposition.

**Proposition 4.2.1** For any given matrix function  $G(\cdot) \in C^1(0,1)$ , the mixed initial-boundary value problem (4.2.8)-(4.2.10) with initial condition

$$t = 0: \gamma(0, x) = \gamma_0(x), \tag{4.2.13}$$

where  $\gamma_0 \in (L^2(0,1))^n$  admits a  $C^0([0,\infty); (L^2(0,1))^n)$  solution  $\gamma = \gamma(t,x)$ , which is globally exponentially stable in the  $L^2$  norm, i.e. for every  $\lambda > 0$ , there exists c > 0 such that

$$\|\gamma(t,\cdot)\|_{L^2} \le c e^{-\lambda t} \|\gamma_0\|_{L^2}.$$
(4.2.14)

In fact, this solution vanishes in finite time  $t > t_F$ , where  $t_F$  is given by

$$t_F = \int_0^1 \frac{1}{\lambda_{m+1}(s)} + \sum_{r=1}^m \frac{1}{|\lambda_r(s)|} ds.$$
(4.2.15)

**Proof.** Equations (4.2.8) can be rewritten as

$$\partial_t \gamma_-(t,x) + \Lambda_-(x) \partial_x \gamma_-(t,x) = \mathcal{G}_1(x) \gamma_-(t,0), \partial_t \gamma_+(t,x) + \Lambda_+(x) \partial_x \gamma_+(t,x) = \mathcal{G}_2(x) \gamma_-(t,0),$$
(4.2.16)

then consider the following Lyapunov functional

$$V_0(t) = \int_0^1 e^{-\delta x} \gamma_+(t,x)^T \left(\Lambda_+(x)\right)^{-1} \gamma_+(t,x) dx - \int_0^1 e^{\delta x} \gamma_-(t,x)^T B \left(\Lambda_-(x)\right)^{-1} \gamma_-(t,x) dx$$
(4.2.17)

where  $\delta > 0$  is a parameter,  $B = \text{diag}(b_1, \dots, b_m)$  (with  $b_r > 0, r = 1, \dots, m$ ) whose coefficients are to be determined. Obviously,  $\sqrt{V_0}$  is a norm equivalent to  $\|\gamma(t, \cdot)\|_{L^2}$ . Differentiating  $V_0$  with respect to t and integrating by parts yields

$$V_0(t) = I + II + III + IV$$

with

$$I = \left[ -e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) + e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) \right]_{0}^{1}$$
  

$$II = -\int_{0}^{1} \delta e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx - \int_{0}^{1} \delta e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) dx$$
  

$$III = 2\int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} (\Lambda_{+}(x))^{-1} \mathcal{G}_{2}(x) \gamma_{-}(t,0) dx$$

$$IV = -2\int_0^1 e^{\delta x} \gamma_-(t,x)^T B \left(\Lambda_-(x)\right)^{-1} \mathcal{G}_1(x) \gamma_-(t,0) dx$$

Noting the boundary conditions (4.2.9)-(4.2.10), we have that

$$I = -e^{\delta} \gamma_{+}(t,1)^{T} \gamma_{+}(t,1) - \gamma_{-}(t,0)^{T} \Big( B - Q^{T} Q \Big) \gamma_{-}(t,0)$$

$$III \leq \int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx + \gamma_{-}(t,0)^{T} \int_{0}^{1} e^{-\delta x} \mathcal{G}_{2}^{T}(x) \left(\Lambda_{+}(x)\right)^{-2} \mathcal{G}_{2}(x) dx \gamma_{-}(t,0)$$

$$\leq \int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx + \gamma_{-}(t,0)^{T} \int_{0}^{1} \mathcal{G}_{2}^{T}(x) \left(\Lambda_{+}(x)\right)^{-2} \mathcal{G}_{2}(x) dx \gamma_{-}(t,0)$$

$$(4.2.19)$$

$$\begin{split} IV &= -2\int_{0}^{1} e^{\delta x} \sum_{m \ge i > j \ge 1} \gamma_{i}(t, x) \frac{b_{i}}{\Lambda_{i}(x)} g_{ij}(x) \gamma_{j}(t, 0) dx \\ &\leq -M \int_{0}^{1} e^{\delta x} \sum_{m \ge i > j \ge 1} \frac{b_{i}}{\Lambda_{i}(x)} \gamma_{i}^{2}(t, x) dx - M \int_{0}^{1} e^{\delta x} \sum_{m \ge i > j \ge 1} \frac{b_{i}}{\Lambda_{i}(x)} \gamma_{j}^{2}(t, 0) dx \\ &\leq -M \int_{0}^{1} e^{\delta x} \sum_{m \ge i > j \ge 1} \frac{b_{i}}{\Lambda_{i}(x)} \gamma_{i}^{2}(t, x) dx + M \mu e^{\delta} \gamma_{-}(t, 0)^{T} \mathcal{C} \gamma_{-}(t, 0) \\ &\leq -mM \int_{0}^{1} e^{\delta x} \sum_{i=2}^{m} \frac{b_{i}}{\Lambda_{i}(x)} \gamma_{i}^{2}(t, x) dx + M \mu e^{\delta} \gamma_{-}(t, 0)^{T} \mathcal{C} \gamma_{-}(t, 0) \\ &\leq -mM \int_{0}^{1} e^{\delta x} \gamma_{-}(t, x)^{T} B \left(\Lambda_{-}(x)\right)^{-1} \gamma_{-}(t, x) dx + M \mu e^{\delta} \gamma_{-}(t, 0)^{T} \mathcal{C} \gamma_{-}(t, 0) \end{split}$$

$$(4.2.20)$$

in which

$$M := \|G\|_{L^{\infty}}, \quad \mathcal{C} := \operatorname{diag}(\mathcal{C}_1, \cdots, \mathcal{C}_m)$$
(4.2.21)

with

$$C_r := \begin{cases} \sum_{j=r+1}^m b_j, & 1 \le r \le m-1\\ 0, & r = m. \end{cases}$$
(4.2.22)

and

$$\mu := \max_{i} \left\{ \frac{1}{\|\Lambda_i\|_{C^0}} \right\}.$$
(4.2.23)

Let

$$P = Q^{T}Q + \int_{0}^{1} \mathcal{G}_{2}^{T}(x) \left(\Lambda_{-}(x)\right)^{-2} \mathcal{G}_{2}(x) dx.$$
(4.2.24)

There exists a diagonal matrix  $S = \text{diag}(s_1, \cdots, s_m)$  with  $s_r > 0$   $(r = 1, \cdots, m)$ being large enough, such that

$$P \prec S. \tag{4.2.25}$$

where  $P \prec S$  denotes that S - P is a positive-definite matrix. This yields

$$\dot{V}_{0}(t) \leq -\gamma_{-}(t,0)^{T} \Big( B - S - M\mu e^{\delta} \mathcal{C} \Big) \gamma_{-}(t,0) - (\delta - 1) \int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx - (\delta - mM\mu) \int_{0}^{1} e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) dx.$$

Thus, for any given  $\lambda > 0$ , picking

$$\delta > \max\left\{\lambda\mu + mM\mu, \lambda\mu + 1\right\} \tag{4.2.26}$$

$$b_r > \begin{cases} M \mu e^{\delta} \sum_{j=r+1}^m b_j + s_r, & 1 \le r \le m-1 \\ s_m, & r = m. \end{cases}$$
(4.2.27)

we have

$$\dot{V}_0 \le -\lambda V_0 \tag{4.2.28}$$

where  $\lambda$  can be chosen as large as desired. It is easy to see that Parameter matrix B does exist, since one can easily check (4.2.27) by induction. This shows exponential stability of  $\gamma$  system.

To show finite-time convergence to the origin, one can find the explicit solution of (4.2.8)-(4.2.10) as follows. Define

$$\phi_i(x) = \int_0^x \frac{1}{|\lambda_i(\xi)|} d\xi, \quad 1 \le i \le n.$$
(4.2.29)

Notice that every  $\phi_i (1 \le i \le n)$  is monotonically increasing  $C^2$  functions of x, and thus invertible. With the same statement in [26] and noting (4.2.8)-(4.2.12), one can express the explicit solution of  $\gamma_1$  by

$$\gamma_1(t,x) = \begin{cases} \gamma_1(0,\phi_1^{-1}(\phi_1(x)+t)) & \text{if } t < \phi_1(1) - \phi_1(x), \\ 0 & \text{if } t \ge \phi_1(1) - \phi_1(x) \end{cases}$$
(4.2.30)

Notice in particular that  $\gamma_1$  is identically zero for  $t \ge \phi_1(1)$ . From (4.2.8) and (4.2.12), we obtain that  $\gamma_2(t, x)$  satisfies the following equation for  $t \ge \phi_1(1)$ 

$$\partial_t \gamma_2(t, x) + \lambda_2(x) \partial_x \gamma_2(t, x) = 0, \qquad (4.2.31)$$

with

$$\gamma_2(t,1) = 0, \tag{4.2.32}$$

which ensures the explicit expression of  $\gamma_2(t, x)$  to be

$$\gamma_2(t,x) = \begin{cases} \gamma_2(\phi_1(1), \phi_2^{-1}(\phi_2(x) + t)) & \text{if } \phi_1(1) < t < \phi_1(1) + \phi_2(1) - \phi_2(x), \\ 0 & \text{if } t \ge \phi_1(1) + \phi_2(1) - \phi_2(x). \end{cases}$$

$$(4.2.33)$$

Therefore, by induction, one has that  $\gamma_r(t, x)(2 \leq r \leq m)$  satisfies the following equations, for  $t > \sum_{k=1}^{r-1} \phi_k(1)$ ,

$$\partial_t \gamma_r(t, x) + \lambda_r(x) \partial_x \gamma_r(t, x) = 0, \qquad (4.2.34)$$

with the boundary condition

$$\gamma_r(t,1) = 0. \tag{4.2.35}$$

Thus, when  $t > \sum_{k=1}^{r-1} \phi_k(1)$ , we have  $\gamma_r(t, x) = \begin{cases} \gamma_r(\sum_{k=1}^{r-1} \phi_k(1), \phi_r^{-1}(\phi_r(x) + t)) & \text{if } \sum_{k=1}^{r-1} \phi_k(1) < t < \sum_{k=1}^r \phi_k(1) - \phi_r(x), \\ 0 & \text{if } t \ge \sum_{k=1}^r \phi_k(1) - \phi_r(x). \end{cases}$ (4.2.36)

This yields that  $\gamma_{-}(t,x) \equiv 0$   $\left(t > \sum_{k=1}^{m} \phi_{k}(1)\right)$ . From the time  $t = \sum_{k=1}^{m} \phi_{k}(1)$  on, we find  $\gamma_{+}$  becomes the solution of the following system

$$\partial_t \gamma_+(t,x) + \Lambda_+(x) \partial_x \gamma_+(t,x) = 0 \qquad (4.2.37)$$

with

$$x = 0: \gamma_+(t,0) \equiv 0. \tag{4.2.38}$$

Since (4.2.37)-(4.2.38) is a completely decoupled system, by the characteristic method, after  $t = t_F$ , where

$$t_F = \phi_{m+1}(1) + \sum_{r=1}^m \phi_r(1) = \int_0^1 \frac{1}{\lambda_{m+1}(s)} + \sum_{r=1}^m \frac{1}{|\lambda_r(s)|} ds, \qquad (4.2.39)$$

one can see that  $\gamma_+(t,x) \equiv 0 (t \geq t_F)$ , which concludes the Proof of Proposition 4.2.1.

#### 4.2.2**Backstepping transformation and Kernel Equations**

To map the original system (4.2.1) into the target system (4.2.8), we use the following Volterra transformation of the second kind, which is similar to the one in [26] and [31]:

$$\gamma(t,x) = w(t,x) - \int_0^x K(x,\xi)w(t,\xi)d\xi.$$
(4.2.40)

We point out here that this transformation yields that  $w(t,0) \equiv \gamma(t,0) \ (\forall t > 0)$ , which is crucial to design our feedback law.

Utilizing (4.2.1) and straightforward computations, one can show that

$$\gamma_t + \Lambda(x)\gamma_x = -\int_0^x \left( K_{\xi}(x,\xi)\Lambda(\xi) + \Lambda(x)K_x(x,\xi) + K(x,\xi)\Sigma(\xi) + K(x,\xi)\Lambda_{\xi}(\xi) \right) w(t,\xi)d\xi + \left( \Sigma(x) + K(x,x)\Lambda(x) - \Lambda(x)K(x,x) \right) w(t,x) - K(x,0)\Lambda(0) \begin{pmatrix} I & 0 \\ Q & 0 \\ (4.2.41) \end{pmatrix} w(t,0) (4.2.41)$$

The original system (4.2.1) is mapped into the target system (4.2.8) if one has the following kernel equations:

$$\Lambda(x)K_x(x,\xi) + K_\xi(x,\xi)\Lambda(\xi) + K(x,\xi)\Sigma(\xi) + K(x,\xi)\Lambda_\xi(\xi) = 0$$
(4.2.42)

$$\Sigma(x) + K(x,x)\Lambda(x) - \Lambda(x)K(x,x) = 0$$
(4.2.43)

$$G(x) = -K(x,0)\Lambda(0) \begin{pmatrix} I & 0\\ Q & 0 \end{pmatrix}$$
(4.2.44)

Developing equations (4.2.42)-(4.2.44) leads to the following set of kernel PDEs

$$\lambda_i(x)\partial_x K_{ij}(x,\xi) + \lambda_j(\xi)\partial_\xi K_{ij}(x,\xi) = -\sum_{k=1}^n \left(\sigma_{kj}(\xi) + \delta_{kj}\lambda'_j(\xi)\right)K_{ik}(x,\xi) \quad (4.2.45)$$

along with the following set of boundary conditions

$$K_{ij}(x,x) = \frac{\sigma_{ij}(x)}{\lambda_i(x) - \lambda_j(x)} \stackrel{\Delta}{=} k_{ij}(x) \qquad \text{for } 1 \le i, j \le n (i \ne j), \quad (4.2.46)$$

$$K_{ij}(x,0) = -\frac{1}{\lambda_j(0)} \sum_{k=1}^{n} \lambda_{m+k}(0) K_{i,m+k}(x,0) q_{k,j} \qquad \text{for } 1 \le i \le j \le m. \quad (4.2.47)$$

To ensure well-posedness of the kernel equations, we add the following artificial boundary conditions for  $K_{ij}$   $(m \ge i > j \ge 1, n \ge j > i \ge m+1)$  on x = 1:

$$K_{ij}(1,\xi) = k_{ij}^{(1)}(\xi), \text{ for } 1 \le j < i \le m \cup m+1 \le i < j \le n$$
 (4.2.48)

and the boundary conditions for  $K_{ij}$   $(n \ge i \ge j \ge m+1)$  on  $\xi = 0$ :

$$K_{ij}(x,0) = k_{ij}^{(2)}(x), \text{ for } m+1 \le j \le i \le n.$$
 (4.2.49)

where  $k_{ij}^{(1)}$  and  $k_{ij}^{(2)}$  are chosen as functions of  $C^{\infty}[0, 1]$  satisfying the  $C^1$  compatibility conditions at the point  $(x, \xi) = (1, 1)$  (see Remark 4.2.1). The equations evolve in the triangular domain  $\mathcal{T} = \{(x, \xi) : 0 \leq \xi \leq x \leq 1\}$ . By Theorem 4.5.1, one finds that there exists a unique piecewise  $C^2(\mathcal{T})$  solution  $K(x, \xi)$  to (4.2.45)-(4.2.49)with  $K(x, x), K(x, 0) \in C^1(0, 1)$ , provided that  $\sigma_{ij}(x)$  are  $C^2[0, 1], \lambda_i(x)$  are  $C^2[0, 1]$ . While  $G(x) \in C^1$  (with bounded  $C^1$  norm) is given by (4.2.44) under the wellposedness of K(x, 0), which is proved in Theorem 4.5.1.

**Remark 4.2.1** The  $C^1$  compatibility conditions at the point  $(x,\xi) = (1,1)$  are given by

$$k_{ij}(1) = k_{ij}^{(1)}(1), \text{ for } 1 \le j < i \le m \cup m + 1 \le i < j \le n,$$
 (4.2.50)

$$\dot{k}_{ij}^{(1)}(1) = \frac{\lambda_i(1)k'_{ij}(1) + \sum_{k=1}^{\infty} \left(\sigma_{kj}(1) + \delta_{kj}\lambda'_j(1)\right)k_{ik}(1)}{\lambda_i(1) - \lambda_j(1)},$$

$$(4.2.51)$$

$$for \ 1 \le j < i \le m \cup m + 1 \le i < j \le n.$$

# 4.2.3 The inverse transformation and stabilization for linear system

Transformation (4.2.40) is a classical Volterra equation of the second kind, one can check from Theorem 4.5.2 that there exists a unique piecewise  $C^2(\mathcal{T})$  matrix function  $L(x,\xi)$  such that

$$w(t,x) = \gamma(t,x) + \int_0^x L(x,\xi)\gamma(t,\xi)d\xi.$$
 (4.2.52)

From the transformation (4.2.40) evaluated at x = 1, one gets the following feedback control laws

$$U_i(t) = \int_0^1 \sum_{j=1}^n K_{ij}(1,\xi) w_j(t,\xi) d\xi, \quad (i = 1, \cdots, m),$$
(4.2.53)

which immediately leads to our feedback stabilization result for the linear system as follows:

**Theorem 4.2.1** The mixed initial-boundary value problem (4.2.1) with the boundary conditions (4.2.6), the feedback control law (4.2.53) and initial condition

$$t = 0: w(0, x) = w_0(x), \tag{4.2.54}$$

in which  $w_0 \in (L^2(0,1))^n$ , admits a  $(L^2(0,1))^n$  solution w = w(t,x). Moreover, for every  $\eta > 0$ , there exists c > 0 such that

$$\|w(\cdot,t)\|_{L^2} \le ce^{-\eta t} \|w_0\|_{L^2}. \tag{4.2.55}$$

In fact, w vanishes in finite time  $t > t_F$ , where  $t_F$  is given by (4.2.15).

**Remark 4.2.2** If we focus on the linear problem,  $\Lambda$  and  $\Sigma$  can be assumed to be  $C^1([0,1])$  and  $C^0([0,1])$  functions. The corresponding kernels K and L are then both functions of  $L^{\infty}(\mathcal{T})$ .

## 4.3 Backstepping boundary control design for nonlinear system

As mentioned in [26], we wish the linear controller (4.2.53) designed by backstepping method to work locally for the corresponding nonlinear system. Let us show that this is indeed the case. Introduce

$$\varphi_i(x) := \exp\left(-\int_0^x \frac{f_{ii}(s)}{\Lambda_i(s)} ds\right) \quad i = 1, \cdots, n.$$
(4.3.1)

One can make the following coordinates transformation

$$w(t,x) = \begin{pmatrix} \varphi_1(x) & & \\ & \ddots & \\ & & \varphi_n(x) \end{pmatrix} u(t,x) = \Phi(x)u(t,x).$$
(4.3.2)

Then the original control system u is transformed into the following system expressed in the new coordinates:

$$w_t(t,x) + \overline{A}(x,w)w_x(t,x) = \widetilde{F}(x,w), \qquad (4.3.3)$$

in which

$$\overline{A}(x,w) = \Phi(x)A(x,\Phi^{-1}(x)w)\Phi^{-1}(x),$$
(4.3.4)

$$\widetilde{F}(x,w) = \Phi(x)F(x,\Phi^{-1}(x)w) - \overline{A}(x,w) \begin{pmatrix} \frac{f(1)(x)}{\Lambda_1(x)} & & \\ & \ddots & \\ & & \frac{f_{nn}(x)}{\Lambda_n(x)} \end{pmatrix} w.$$
(4.3.5)

Obviously, one can check that

$$\widetilde{F}(x,0) = 0, \tag{4.3.6}$$

$$\overline{A}(x,0) = \Phi(x)A(x,0)\Phi^{-1}(x) = A(x,0).$$
(4.3.7)

Moreover, define

$$\Sigma(x) = \frac{\partial \widetilde{F}(x,w)}{\partial w}\Big|_{w=0},$$
(4.3.8)

we have that

$$\Sigma_{ij}(x) = \begin{cases} \frac{\varphi_i(x)}{\varphi_j(x)} f_{ij}(x), i \neq j, \\ 0, i = j. \end{cases}$$
(4.3.9)

Therefore, we may rewrite (4.3.3) as a linear system with the same structure as (4.2.1) plus nonlinear terms:

$$w_t(t,x) + \Lambda(x)w_x(t,x) = \Sigma(x)w(t,x) + \Lambda_{NL}(x,w)w_x(t,x) + f_{NL}(x,w), \quad (4.3.10)$$

where

$$\Lambda(x) = A(x,0), \tag{4.3.11}$$

and

$$\Lambda_{NL}(x,w) = \Lambda(x) - \overline{A}(x,w), \ f_{NL}(x,w) = \widetilde{F}(x,w) - \Sigma(x)w(t,x).$$
(4.3.12)

For the boundary conditions of the system (4.3.10), defining

$$Q = \left(\frac{\partial G_s}{\partial u_r}\right)_{(n-m) \times m} \bigg|_{u=0} \text{ and } G_{NL}(w_-(t,0)) = G(w_-(t,0)) - Qw_-(t,0), \quad (4.3.13)$$

one obtains that

$$x = 0: w_{+}(t, 0) = Qw_{-}(t, 0) + G_{NL}(w_{-}(t, 0))$$
(4.3.14)

and

$$x = 1: w_{-}(t, 1) = U(t).$$
 (4.3.15)

where

$$U(t) = \begin{pmatrix} \varphi_1(1) & & \\ & \ddots & \\ & & \varphi_m(1) \end{pmatrix} H(t) = \widetilde{\Phi}(1)H(t).$$
(4.3.16)

It is easily verified that

$$\Lambda(x,0) = 0, \quad f_{NL}(x,0) = \frac{\partial f_{NL}}{\partial w}(x,0) = 0$$
(4.3.17)

and

$$G_{NL}(0) = \frac{\partial G_{NL}}{\partial w}(0) = 0.$$
(4.3.18)

Thus, the feedback control law can be chosen as

$$h_r(t) = \widetilde{\Phi}_{rr}^{-1}(1)U_r(t) = \widetilde{\Phi}_{rr}^{-1}(1)\int_0^1 \sum_{j=1}^n K_{rj}(1,\xi)\widetilde{\Phi}_{jj}(\xi)u_j(t,\xi)d\xi, \quad r = 1, \cdots, m$$
(4.3.19)

where the kernels are computed from (4.2.45)–(4.2.49) with the coefficients  $\Sigma(x)$  and  $\Lambda(x)$  obtained from (4.3.9) and (4.3.11). One easily verifies that under the assumptions of §4.1, both  $\Sigma$  and  $\Lambda$  are functions of  $C^2$ .

**Remark 4.3.1** The  $C^1$  compatibility conditions at the point (t, x) = (0, 1) for system (4.1.1) with boundary conditions (4.3.15) should be

$$\phi_r(1) = \sum_{j=1}^n \int_0^1 \tilde{k}_{rj}(\xi) \phi_j(\xi) d\xi \quad r = 1, \cdots, m,$$

$$f_r(1, \phi(1)) - \sum_{j=1}^n a_{rj}(1, \phi(1)) \phi'_j(1) =$$

$$\sum_{k=1}^n \int_0^1 \tilde{k}_{rk}(\xi) \Big( f_k(1, \phi(1)) - \sum_{j=1}^n a_{kj}(1, \phi(1)) \phi'_j(1) \Big) \quad r = 1, \cdots, m,$$

$$(4.3.20)$$

where  $\tilde{k}_{rk}(\xi)$  are the elements of the matrix  $\tilde{K}(\xi)$  with

$$\widetilde{K}(\xi) = \widetilde{\Phi}^{-1}(1)K(1,\xi)\widetilde{\Phi}(\xi).$$
(4.3.22)

Notice that (4.3.20)-(4.3.21) depend on the feedback control design, however, there are no physical reasons that the initial data should satisfy them. In order to guarantee the initial conditions independent of these artificial conditions, we, following [26], modify the boundary controls on x = 1 as

$$x = 1: \quad u_r = h_r(t) + a_r(t) + b_r(t) \quad r = 1, \cdots, m,$$
 (4.3.23)

where  $a_r$  and  $b_r$  are the state of the following dynamic systems

$$\dot{a}_r(t) = -d_r a_r(t), \quad \dot{b}_r(t) = -\tilde{d}_r b_r(t) \quad r = 1, \cdots, m$$
(4.3.24)

with  $d_r > 0$ ,  $\tilde{d}_r > 0$  and  $d_r \neq \tilde{d}_r$ ,  $r = 1, \dots, m$ . By the modified control designs (4.3.23), the compatibility conditions on x = 1 are rewritten by

$$\phi_r(1) = \sum_{j=1}^n \int_0^1 \tilde{k}_{rj}(\xi) \phi_j(\xi) d\xi + a_r(0) + b_r(0) \quad r = 1, \cdots, m,$$

$$f_r(1, \phi(1)) - \sum_{j=1}^n a_{rj}(1, \phi(1)) \phi'_j(1) =$$

$$\sum_{k=1}^n \int_0^1 \tilde{k}_{rk}(\xi) \Big( f_k(1, \phi(1)) - \sum_{j=1}^n a_{kj}(1, \phi(1)) \phi'_j(1) \Big) - d_r a_r(0) - \tilde{d}_r b_r(0) \quad r = 1, \cdots, m$$

$$(4.3.26)$$

For any  $1 \leq r \leq m$ , call

$$\mathcal{P}_{r}(\phi) = \phi_{r}(1) - \sum_{j=1}^{n} \int_{0}^{1} \tilde{k}_{rj}(\xi) \phi_{j}(\xi) d\xi$$
(4.3.27)

$$\mathcal{M}_{r}(\phi) = f_{r}(1,\phi(1)) - \sum_{j=1}^{n} a_{rj}(1,\phi(1))\phi_{j}'(1) - \sum_{k=1}^{n} \int_{0}^{1} \tilde{k}_{rk}(\xi) \left(f_{k}(1,\phi(1)) - \sum_{j=1}^{n} a_{kj}(1,\phi(1))\phi_{j}'(1)\right)$$

$$(4.3.28)$$

Picking

$$a_r(0) = -\frac{\mathcal{M}_r(\phi) + \tilde{d}_r \mathcal{P}_r(\phi)}{d_r - \tilde{d}_r}, b_r(0) = \frac{d_r \mathcal{P}_r(\phi) + \mathcal{M}_r(\phi)}{d_r - \tilde{d}_r}$$
(4.3.29)

the compatibility conditions are automatically verified. Similar stabilization results as Theorem 4.1.1 are still valid for the closed-loop system (4.1.1), (4.1.5) and (4.3.23) (see [26, Theorem 4.1]). In fact, this dynamic extension is designed to avoid restriction for artificial boundary conditions due to the compatibility conditions at the points (t, x) = (0, 1), and it has been introduced in [17] to deal with the stabilization of the Euler equations of incompressible fluids (see also [108]).

## 4.4 Proof of Theorem 4.1.1

In this section, we will prove the exponential stability for the system (4.1.1), (4.1.5) and (4.1.6) under the boundary feedback controls (4.3.19) by Control Lyapunov Function method. The whole proof is divided into the following steps.

### 4.4.1 Definitions

We first define some notations (omitting the time argument):

$$\|\gamma\|_{\infty} := \underset{x \in [0,1]}{\operatorname{ess}} \sup_{|\gamma(x)|} |\gamma(x)|, \quad \|\gamma\|_{L^p} := \left(\int_0^1 |\gamma(\xi)|^p d\xi\right)^{\frac{1}{p}}, \ 1 \le p < +\infty.$$
(4.4.1)

For a  $n \times n$  matrix, denote

$$|M| := \max\{ \|M\gamma\|_{L^{\infty}} : \gamma \in \mathbb{R}^n, |\gamma| = 1 \}.$$
 (4.4.2)

For a piecewise kernel matrix  $K(x,\xi)$ , which is a continuous function on each domain  $D_i(i = 1, \dots, S)$ , respectively, with

$$\mathcal{T} = \bigcup_{i=1}^{\mathcal{S}} D_i, \tag{4.4.3}$$

$$D_i \cap D_j = \emptyset, (i \neq j). \tag{4.4.4}$$

Let

$$||K||_{\infty} := \max_{i} \sup_{(x,\xi) \in D_{i}} |K(x,\xi)|.$$
(4.4.5)

As before, we recall the following symbols of [26] for simplicity:

$$\mathcal{K}[\gamma](t,x) = \gamma(t,x) - \int_0^x K(x,\xi)\gamma(t,\xi)d\xi, \qquad (4.4.6)$$

$$\mathcal{L}[\gamma](t,x) = \gamma(t,x) + \int_0^x L(x,\xi)\gamma(t,\xi)d\xi, \qquad (4.4.7)$$

$$\mathcal{K}_1[\gamma](t,x) = -K(x,x)\gamma(t,x) + \int_0^x K_{\xi}(x,\xi)\gamma(t,\xi)d\xi, \qquad (4.4.8)$$

$$\mathcal{K}_2[\gamma](t,x) = -K(x,x)\gamma(t,x) - \int_0^x K_x(x,\xi)\gamma(t,\xi)d\xi, \qquad (4.4.9)$$

$$\mathcal{L}_1[\gamma](t,x) = L(x,x)\gamma(t,x) + \int_0^\infty L_x(x,\xi)\gamma(t,\xi)d\xi.$$
(4.4.10)

Define  $F_1[\gamma]$  and  $F_2[\gamma]$  as

$$F_1[\gamma] := \Lambda_{NL}(x, \mathcal{L}[\gamma]), \quad F_2[\gamma] := f_{NL}(x, \mathcal{L}[\gamma]). \tag{4.4.11}$$

To prove our result, we notice that if we apply the (inverse) backstepping transformation (4.2.40) to the nonlinear system (4.3.10), we obtain the following transformed system

$$\gamma_t(t, x) + \Lambda(x)\gamma_x(t, x) - G(x)\gamma(t, 0)$$
  
= $\mathcal{K}[\Lambda_{NL}(x, w)w_x] + \mathcal{K}[f_{NL}(x, w)]$   
= $\mathcal{K}[\Lambda_{NL}(x, w)\gamma_x] + \mathcal{K}[\Lambda_{NL}(x, w)\mathcal{L}_1[\gamma]] + \mathcal{K}[f_{NL}(x, w)]$   
= $F_3[\gamma, \gamma_x] + F_4[\gamma]$  (4.4.12)

where

$$F_3 = \mathcal{K}[F_1[\gamma]\gamma_x],$$
  

$$F_4 = \mathcal{K}[F_1[\gamma]\mathcal{L}_1[\gamma] + F_2[\gamma]]$$

The boundary conditions are

$$x = 0: \gamma_{+}(t,0) = Q\gamma_{-}(t,0) + G_{NL}(\gamma_{-}(t,0))$$
(4.4.13)

and

$$x = 1: \gamma_{-}(t, 1) = 0. \tag{4.4.14}$$

Notice that here we may lose the regularity on the point (0,0) for the kernels Kand L, which leads both of them to be discontinuous (see [54]). However, by the assumptions on the coefficients and applying Theorem 4.5.1 and Theorem 4.5.2, the direct and inverse transformations (4.2.40) and (4.2.52) have  $C^2$  piecewise kernels functions. Fortunately, differentiating twice with respect to x in these transformations, by the similar argument in [26] and [108, Proposition 3.1] as well as the additive property of the integral, it can be shown that the  $H^2$  norm of  $\gamma$  is equivalent to the  $H^2$  norm of w. Thus, if we show  $H^2$  local stability of the origin for (4.4.12)-(4.4.14), the same holds for w i.e. u.

In order to get the desired  $H^2$  estimation for  $\gamma$ , the things left are just estimating

the growth of  $\|\gamma\|_{L^2}$ ,  $\|\gamma_t\|_{L^2}$  and  $\|\gamma_{tt}\|_{L^2}$ , respectively.

#### Analyzing the growth of $\|\gamma\|_{L^2}$ 4.4.2

Let

$$F_{3}[\gamma,\gamma_{x}] = \left(F_{3}^{-}[\gamma,\gamma_{x}], F_{3}^{+}[\gamma,\gamma_{x}]\right)^{T}, \quad F_{4}[\gamma] = \left(F_{4}^{-}[\gamma], F_{4}^{+}[\gamma]\right)^{T}.$$
(4.4.15)

where  $F_3^-$  and  $F_4^- \in \mathbb{R}^m$ ,  $F_3^+$  and  $F_4^+ \in \mathbb{R}^{n-m}$ .

Define

$$V_{1}(t) = \int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \left(\Lambda_{+}(x)\right)^{-1} \gamma_{+}(t,x) dx - \int_{0}^{1} e^{\delta x} \gamma_{-}(t,x)^{T} B \left(\Lambda_{-}(x)\right)^{-1} \gamma_{-}(t,x) dx$$
(4.4.16)

Differentiating  $V_1$  with respect to time and integrating by parts yields

$$\dot{V}_1(t) = V + VI + VII + VIII + IX + X$$

with

$$\begin{split} V &= \left[ -e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) + e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) \right]_{0}^{1} \\ VI &= -\int_{0}^{1} \delta e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx - \int_{0}^{1} \delta e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) dx \\ VIII &= 2\int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \left(\Lambda_{+}(x)\right)^{-1} \mathcal{G}_{2}(x) \gamma_{-}(t,0) dx \\ VIII &= -2\int_{0}^{1} e^{\delta x} \gamma_{-}(t,x)^{T} B \left(\Lambda_{-}(x)\right)^{-1} \mathcal{G}_{1}(x) \gamma_{-}(t,0) dx \\ IX &= 2\int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \left(\Lambda_{+}(x)\right)^{-1} \left(F_{3}^{+}[\gamma,\gamma_{x}] + F_{4}^{+}[\gamma]\right) dx \\ X &= -2\int_{0}^{1} e^{\delta x} \gamma_{-}(t,x)^{T} B \left(\Lambda_{-}(x)\right)^{-1} \left(F_{3}^{-}[\gamma,\gamma_{x}] + F_{4}^{-}[\gamma]\right) dx \end{split}$$

By the same argument in [26] and noting Lemma 4.6.2, we have

$$IX + X \le K_1 \int_0^1 |\gamma| (|F_3[\gamma, \gamma_x]| + |F_4[\gamma]|) dx$$
  
$$\le K_2 (\|\gamma_x\|_{\infty} V_1 + V_1^{\frac{3}{2}}).$$
(4.4.17)

Moreover, for 
$$\|\gamma\|_{\infty} \leq \delta$$
,  $|G_{NL}(\gamma_{-}(t,0))| \leq K_{3}|\gamma_{-}(t,0)|$ , then  

$$V = -e^{-\delta}\gamma_{+}(t,1)^{T}\gamma_{+}(t,1) + e^{\delta}\gamma_{-}(t,1)^{T}B\gamma_{-}(t,1) + \gamma_{+}(t,0)^{T}\gamma_{+}(t,0) - \gamma_{-}(t,0)^{T}B\gamma_{-}(t,0)$$

$$\leq -\gamma_{-}(t,0)^{T} \Big(B - Q^{T}Q - K_{3}^{2}I_{m}\Big)\gamma_{-}(t,0)$$
(4.4.18)

By (4.2.19) and (4.2.20), one immediately obtains

$$\dot{V}_{1}(t) \leq -\gamma_{-}(t,0)^{T} \Big( B - \widetilde{S} - M\mu e^{\delta} \mathcal{C} \Big) \gamma_{-}(t,0) - (\delta - 1) \int_{0}^{1} e^{-\delta x} \gamma_{+}(t,x)^{T} \gamma_{+}(t,x) dx - (\delta - mM\mu) \int_{0}^{1} e^{\delta x} \gamma_{-}(t,x)^{T} B \gamma_{-}(t,x) dx + K_{2} \Big( V_{1}^{\frac{3}{2}} + \|\gamma_{x}\|_{\infty} V_{1} \Big)$$

where M, C,  $\mu$  are given by (4.2.21) and (4.2.23),  $\tilde{S} := S + K_3^2 I_m$  with S stated in (4.2.25). Thus, for any given  $\lambda_1 > 0$ , picking

$$\delta > \max\left\{\lambda_1 \mu + mM\mu, \lambda_1 \mu + 1\right\}$$
(4.4.19)

$$b_r := \begin{cases} M \mu e^{\delta} \sum_{j=r+1}^m b_j + \tilde{s}_r, & 1 \le r \le m-1\\ \tilde{s}_m, & r = m. \end{cases}$$
(4.4.20)

we have the following

**Proposition 4.4.1** For any given  $\lambda_1 > 0$ , there exists  $\delta_1 > 0$  and  $K_2 > 0$ , such that

$$\dot{V}_1 \le -\lambda_1 V_1 + K_2 \left( V_1^{\frac{3}{2}} + \|\gamma_x\|_{\infty} V_1 \right), \tag{4.4.21}$$

provided  $\|\gamma\|_{\infty} \leq \delta_1$ .

## 4.4.3 Analyzing the growth of $\|\gamma_t\|_{L^2}$

Let  $\zeta = \gamma_t$ . Taking the partial derivative with t in (4.4.12) yields:

$$\zeta_t(t,x) + (\Lambda(x) - F_1[\gamma])\zeta_x(t,x) - G(x)\zeta(t,0) = F_5[\gamma,\gamma_x,\zeta] + F_6[\gamma,\zeta], \quad (4.4.22)$$

where

$$F_{5} = \mathcal{K}_{1}[F_{1}[\gamma]\zeta] + \int_{0}^{x} K(x,\xi)F_{12}[\gamma,\gamma_{x}]\zeta(\xi)d\xi + K(x,0)\Lambda_{NL}(0,\gamma(0))\zeta(0) + \mathcal{K}[F_{11}[\gamma,\zeta]\gamma_{x}],$$

$$(4.4.23)$$

$$F_{6} = \mathcal{K}[F_{11}[\gamma,\zeta]\mathcal{L}_{1}[\gamma]] + \mathcal{K}[F_{1}[\gamma]\mathcal{L}_{1}[\zeta]] + \mathcal{K}[F_{21}[\gamma,\zeta]],$$

$$(4.4.24)$$

with

$$F_{11} = \frac{\partial \Lambda_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]) \mathcal{L}[\zeta],$$
  

$$F_{12} = \frac{\partial \Lambda_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]) (\gamma_x + \mathcal{L}_1[\gamma]) + \frac{\partial \Lambda_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]),$$
(4.4.25)

and

$$F_{21} = \frac{\partial f_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]) \mathcal{L}[\zeta].$$
(4.4.26)

The boundary conditions are given by

$$x = 0: \zeta_{+}(t,0) = Q\zeta_{-}(t,0) + \frac{\partial G_{NL}}{\partial \gamma_{-}} \Big(\gamma_{-}(t,0)\Big)\zeta_{-}(t,0)$$
(4.4.27)

and

$$x = 1: \zeta_{-}(t, 1) = 0, \tag{4.4.28}$$

in which  $\zeta_{-} \in \mathbb{R}^{m}, \zeta_{+} \in \mathbb{R}^{n-m}$  are defined by requiring that  $\zeta := (\zeta_{-}, \zeta_{+})^{T}$ .

Similarly as in [26], we need the following lemma in order to find a Lyapunov function for  $\zeta(t, x)$ :

**Lemma 4.4.1** There exists  $\delta > 0$  such that, for any  $\|\gamma\|_{\infty} \leq \delta$ , there exists a symmetric matrix  $R[\gamma]$  satisfying the identity

$$R[\gamma](\Lambda(x) - F_1[\gamma]) - (\Lambda(x) - F_1[\gamma])^T R[\gamma] = 0.$$
(4.4.29)

Moreover, we have that

$$R[\gamma](x)| \le c_1 + c_2 \|\gamma\|_{\infty}, \tag{4.4.30}$$

$$\left| \left( (R[\gamma] - D(x))\Lambda(x) \right)_x \right| \le c_2(\|\gamma\|_\infty + \|\gamma_x\|_\infty) \tag{4.4.31}$$

$$|(R[\gamma])_t| \le c_3(|\zeta| + ||\zeta||_{L^1}), \tag{4.4.32}$$

where  $c_1$ ,  $c_2$  and  $c_3$  are positive constants, and

$$D(x) = \begin{pmatrix} -e^{\delta x} B(\Lambda_{-}(x))^{-1} & 0\\ 0 & e^{-\delta x} (\Lambda_{+}(x))^{-1} \end{pmatrix}.$$
 (4.4.33)

**Proof:** Denote  $\mathcal{D}_n(x)$  as the set of  $n \times n$  diagonal matrices with  $C^1$  elements. Let  $\Lambda(x) := \operatorname{diag}(\Lambda_1(x), \dots, \Lambda_n(x)) \in \mathcal{D}_n(x)$  be such that  $\Lambda_i(x) \neq \Lambda_j(x) (i \neq j \forall x \in [0,1])$  holds. Notice that  $D \in \mathcal{D}_n(x)$ . Based on the proof in [21, Lemma 4.1], one can easily see that there exist a positive real number  $\eta$  and a map  $\mathcal{N} : \{M \in \mathcal{N}\}$ 

 $\mathcal{M}_{n,n}(\mathbb{R};x); \|M(x) - \Lambda(x)\|_{C^1} < \eta\} \to \mathcal{S}_n$  of class  $C^{\infty}$  such that

$$\mathcal{N}(\Lambda(x)) = D(x), \tag{4.4.34}$$

and

$$\mathcal{N}(M)M - M^T \mathcal{N}(M) = 0 \quad \forall M \in \mathcal{M}_{n,n}(\mathbb{R}; x), \quad \|M(x) - \Lambda(x)\|_{C^1} < \eta.$$
(4.4.35)

It then suffices to define  $R[\gamma]$  by

$$R[\gamma] = \mathcal{N}(\Lambda(x) - F_1[\gamma]). \tag{4.4.36}$$

Moreover, by the regularity of  $\mathcal{N}$  and Lemma 4.6.2–4.6.3, one can show that

$$|R[\gamma]| \le |D(x)| + |R[\gamma] - D(x)| \le c_4 + c_5 |F_1[\gamma]| \le c_4 + c_6 ||\gamma||_{\infty},$$
(4.4.37)

$$\left| \left( (R[\gamma] - D(x))\Lambda(x) \right)_x \right| \le |(R[\gamma] - D(x))_x \Lambda(x)| + |(R[\gamma] - D(x))\Lambda_x(x)| \\ \le c_7 |F_{12}| + c_8 |F_1| \\ \le c_9 (\|\gamma\|_{\infty} + \|\gamma_x\|_{\infty})$$
(4.4.38)

and

$$|R[\gamma]_t| \le c_{10} \left| \frac{\partial F_1[\gamma]}{\partial t} \right| \tag{4.4.39}$$

$$\leq c_{10}|F_{11}[\gamma,\zeta]| \tag{4.4.40}$$

$$\leq c_{11}(|\zeta| + \|\zeta\|_{L^1}). \tag{4.4.41}$$

This concludes the proof of Lemma 4.1.

Define

$$V_2(t) = \int_0^1 \zeta^T(t, x) R[\gamma] \zeta(t, x) dx.$$
(4.4.42)

Using (4.4.29) and straightforward computations, one can show that

$$\dot{V}_2(t) = XI + XII + XIII + XIV + XV$$

with

$$XI = \int_0^1 \zeta^T(t, x) (R[\gamma](\Lambda(x) - F_1[\gamma]))_x \zeta(t, x) dx$$

Chapter 4: Boundary Exponential Stabilization of 1-D Inhomogeneous Quasilinear Hyperbolic Systems 131

$$\begin{split} XII &= -\left[\zeta^T(t,x)R[\gamma](\Lambda(x) - F_1[\gamma])\zeta(t,x)\right]_{x=0}^{x=1} \\ XIII &= \int_0^1 \zeta(t,x)(R[\gamma])_t \zeta(t,x) dx \\ XIV &= 2\int_0^1 \zeta^T(t,x)R[\gamma]F_5[\gamma,\gamma_x,\zeta,\zeta_x] dx + 2\int_0^1 \zeta^T(t,x)R[\gamma]F_6[\gamma,\zeta] dx \\ XV &= 2\int_0^1 \zeta^T(t,x)R[\gamma]G(x)\zeta(t,0) dx. \end{split}$$

For XII and XV, by the boundary conditions (4.4.27)-(4.4.28), we have

$$\begin{split} XII + XV &= -[\zeta^{T}(t,x)(D(x) + \Theta[\gamma])(\Lambda(x) - F_{1}[\gamma])\zeta(t,x)]_{x=0}^{x=1} \\ &+ 2\int_{0}^{1}\zeta^{T}(t,x)(D(x) + \Theta[\gamma])G(x)\zeta(t,0)dx \\ &= -[\zeta^{T}(t,x)(D(x)\Lambda(x) + \Theta[\gamma]\Lambda(x) - D(x)F_{1}[\gamma] - \Theta[\gamma]F_{1}[\gamma])\zeta(t,x)]_{x=0}^{x=1} \\ &+ 2\int_{0}^{1}\zeta^{T}(t,x)D(x)G(x)\zeta(t,0)dx + 2\int_{0}^{1}\zeta^{T}(t,x)\Theta[\gamma]G(x)\zeta(t,0)dx \\ &\leq -\zeta_{-}(t,0)^{T}\Big(B - \widetilde{S} - M\mu e^{\delta}\mathcal{C} - K_{3}\|\gamma\|_{\infty}I_{m}\Big)\zeta_{-}(t,0) \\ &+ \int_{0}^{1}e^{-\delta x}\zeta_{+}(t,x)^{T}\zeta_{+}(t,x)dx + mM\mu\int_{0}^{1}e^{\delta x}\zeta_{-}(t,x)^{T}B\zeta_{-}(t,x)dx \\ &+ K_{4}\|\gamma\|_{\infty}V_{2} \end{split}$$

As stated in [26], we obtain

$$XI \le -\lambda_2 V_2 + K_4 \|\zeta\|_{L^2}^2 (\|\gamma\|_{\infty} + \|\gamma_x\|_{\infty}), \tag{4.4.43}$$

$$XIII \le K_5 \|\zeta\|_{L^2}^2 \|\zeta\|_{\infty}, \tag{4.4.44}$$

$$XIV \le K_6 \Big( \|\zeta\|_{L^2}^2 (\|\gamma\|_{\infty} + \|\gamma_x\|_{\infty}) + \|\zeta\|_{L^2} |\zeta(t,0)| |\gamma(t,0)| \Big).$$
(4.4.45)

Following Lemma 4.6.5, we are in the position to conclude that

**Proposition 4.4.2** For any given  $\lambda_2 > 0$ , there exists  $\delta_2 > 0$  and  $K_7 > 0$ , such that

$$\dot{V}_2 \le -\lambda_2 V_2 + K_7 \big( \|\zeta\|_{\infty} + \|\gamma\|_{\infty} \big) V_2, \tag{4.4.46}$$

provided that  $\|\gamma\|_{\infty} \leq \delta_2$ .

## 4.4.4 Analyzing the growth of $\|\gamma_{tt}\|_{L^2}$

We next deal with  $\|\gamma_{tt}\|_{L^2}$ . Define  $\theta = \gamma_{tt}$ . Taking a partial derivative with respect to t for (4.4.22), one obtains an equation of  $\theta$ :

$$\theta_t + [\Lambda(x) - F_1[\gamma]]\theta_x = G(x)\theta(t,0) + F_7[\gamma,\gamma_x,\zeta,\zeta_x,\theta] + F_8[\gamma,\zeta,\theta], \qquad (4.4.47)$$

where

$$F_{7} = \mathcal{K}_{1}[F_{11}[\gamma,\zeta]\zeta] + \int_{0}^{x} K(x,\xi)F_{12}[\gamma,\gamma_{x}]\theta(\xi)d\xi + \mathcal{K}_{1}[F_{1}[\gamma]\theta] + \int_{0}^{x} K(x,\xi)F_{14}[\gamma,\gamma_{x},\zeta,\zeta_{x}]\zeta(\xi)d\xi + K(x,0)\frac{\partial\Lambda_{NL}}{\partial\gamma}(0,\gamma(0))\zeta(0)\zeta(0) \qquad (4.4.48) + K(x,0)\Lambda_{NL}(0,\gamma(0))\theta(0) + \mathcal{K}[F_{11}[\gamma,\zeta]\zeta_{x}] + \mathcal{K}[F_{13}[\gamma,\zeta,\theta]\gamma_{x}], F_{8} = 2\mathcal{K}[F_{11}[\gamma,\zeta]\mathcal{L}_{1}[\zeta]] + \mathcal{K}[F_{1}[\gamma]\mathcal{L}_{1}[\theta]] + \mathcal{K}[F_{13}[\gamma,\zeta,\theta]\mathcal{L}_{1}[\gamma]] + \mathcal{K}[F_{22}[\gamma,\zeta,\theta]] (4.4.49)$$

with

$$F_{13} = \frac{\partial \Lambda_{NL}^2}{\partial \gamma^2} (x, \mathcal{L}[\gamma]) \mathcal{L}[\zeta] \mathcal{L}[\zeta] + \frac{\partial \Lambda_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]) \mathcal{L}[\theta], \qquad (4.4.50)$$

$$F_{14} = \frac{\partial \Lambda_{NL}^2}{\partial \gamma^2} (x, \mathcal{L}[\gamma]) \mathcal{L}[\zeta](\gamma_x + \mathcal{L}_1[\gamma]) + \frac{\partial \Lambda_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma])(\zeta_x + \mathcal{L}_1[\zeta])$$

$$\xrightarrow{\partial^2 \Lambda} (4.4.51)$$

$$+\frac{\partial^2 \Lambda_{NL}}{\partial x \partial \gamma}(x, \mathcal{L}[\gamma])\mathcal{L}[\zeta],$$

$$F_{22} = \frac{\partial^2 f_{NL}}{\partial \gamma^2} (x, \mathcal{L}[\gamma]) \mathcal{L}[\zeta] \mathcal{L}[\zeta] + \frac{\partial f_{NL}}{\partial \gamma} (x, \mathcal{L}[\gamma]) \mathcal{L}[\theta].$$
(4.4.52)

The boundary conditions of  $\theta$  are given by

$$x = 0: \theta_{+}(t,0) = Q\theta_{-}(t,0) + \frac{\partial G_{NL}}{\partial \gamma_{-}} \Big(\gamma_{-}(t,0)\Big)\theta_{-}(t,0) + \frac{\partial^{2} G_{NL}}{\partial \gamma_{-}^{2}} \Big(\gamma_{-}(t,0)\Big)\zeta_{-}(t,0)\zeta_{-}(t,0)$$
(4.4.53)

and

$$x = 1: \theta_{-}(t, 1) = 0. \tag{4.4.54}$$

where  $\theta_{-} \in \mathbb{R}^{m}, \theta_{+} \in \mathbb{R}^{n-m}$  are defined by requiring that  $\theta := (\theta_{-}, \theta_{+})^{T}$ .

In order to control  $\|\theta\|_{L^2}$ , we introduce

$$V_3(t) = \int_0^1 \theta^T(t, x) R[\gamma] \theta(t, x) dx, \qquad (4.4.55)$$

then it is easy to see that

$$\dot{V}_3(t) = XVI + XVII + XVIII + XIX + XX$$
(4.4.56)

with

$$\begin{aligned} XVI &= \int_0^1 \theta^T(t,x) (R[\gamma](\Lambda(x) - F_1[\gamma]))_x \theta(t,x) dx \\ XVII &= -\left[\theta^T(t,x) R[\gamma](x) (\Lambda(x) - F_1[\gamma](x)) \theta(t,x)\right]_{x=0}^{x=1} \\ XVIII &= + \int_0^1 \theta^T(t,x) (R[\gamma])_t \theta(t,x) dx \\ XIX &= + 2 \int_0^1 \theta^T(t,x) R[\gamma] F_7[\gamma, \gamma_x, \zeta, \zeta_x, \theta] dx + 2 \int_0^1 \theta^T(t,x) R[\gamma] F_8[\gamma, \zeta, \theta] dx \\ XX &= + 2 \int_0^1 \theta^T(t,x) R[\gamma] G(x) \theta(t,0) dx. \end{aligned}$$

Let us first look at the second and the last term of (4.4.56) (i.e. XVII and XX), by some straight computations, one gets

$$XVII + XX \leq -\theta_{-}(t,0)^{T} \Big( B - \widetilde{S} - M\mu e^{\delta} \mathcal{C} - K_{8} \|\gamma\|_{\infty} I_{m} \Big) \theta_{-}(t,0)$$
  
+ 
$$\int_{0}^{1} e^{-\delta x} \theta_{+}(t,x)^{T} \theta_{+}(t,x) dx + mM\mu \int_{0}^{1} e^{\delta x} \theta_{-}(t,x)^{T} B \theta_{-}(t,x) dx$$
  
+ 
$$K_{9} \|\gamma\|_{\infty} V_{3}$$
(4.4.57)

Then by the same procedures in [26], we have the following

**Proposition 4.4.3** For any given  $\lambda_3 > 0$ , there exists  $\delta_3 > 0$  and positive constants  $K_{10}$ ,  $K_{11}$ ,  $K_{12}$ ,  $K_{13}$  and  $K_{14}$ , such that

$$\dot{V}_{3} \leq -\lambda_{3}V_{3} + K_{10} \|\gamma\|_{\infty} V_{3} + K_{11}V_{3}V_{2}^{\frac{1}{2}} + K_{12}V_{2}V_{3}^{\frac{1}{2}} + K_{13}V_{3}^{\frac{3}{2}} + K_{14} \|\zeta\|_{\infty}^{3}, \quad (4.4.58)$$

provided that  $\|\gamma\|_{\infty} + \|\zeta\|_{\infty} \leq \delta_3$ .

## 4.4.5 Proof of the $H^2$ stability for $\gamma$

Denote  $W = V_1 + V_2 + V_3$ , by Proposition 4.4.1, 4.4.2 and 4.4.3 as well as Lemma 4.6.7, one can show that for any given  $\lambda > 0$ , there exists  $\delta > 0$  and  $K_{15} > 0$ , such that

$$\dot{W} \le -\lambda W + K_{15} W^{\frac{3}{2}},\tag{4.4.59}$$

provided that  $\|\gamma\|_{\infty} + \|\zeta\|_{\infty} \leq \delta$ . This concludes the whole proof of Theorem 4.1.1.

### 4.5 Appendix A

In this section, we will show the well-posedness and piecewise smoothness of the Kernel K and L which are given by the following Theorems.

**Theorem 4.5.1** Let  $N \in \mathbb{N}^+$ . Under the assumption that  $\sigma_{ij} \in C^N[0,1]$ ,  $\lambda_i \in C^N[0,1](i, j = 1, \dots, n)$ , there exists a unique piecewise  $C^N(\mathcal{T})$  solution K to the hyperbolic system (4.2.45)-(4.2.49). Moreover, if the  $C^{N-1}$  compatibility conditions at the point  $(x, \xi) = (1, 1)$  are satisfied, then  $K(\cdot, \cdot) \in C^{N-1}(0, 1)$ ,  $K(\cdot, 0) \in C^{N-1}(0, 1)$  with bounded  $C^{N-1}$  norm.

**Proof.** We divided the proof into two parts. For the first part, we prove the regularity of the kernels. For this, we only prove the case N = 1. For  $N \ge 1$ , the results can be obtained by induction. In the case N = 1, one can, in fact, refer [54] and Remark 4.5.1 to find there exists a piecewise  $C^0$  kernel K for the boundary problem (4.2.45)-(4.2.49), where though only constant coupling coefficients and transport velocities are considered. However, the method in [54] straightforwardly extends to spatially varying coefficients with more involved technical developments. Next, we will improve the regulality of K. Let  $\mathcal{H}_{ij} = \partial_x K_{ij}(x,\xi)$  and  $\mathcal{Y}_{ij} = \partial_{\xi} K_{ij}(x,\xi)$ . By differentiating with respect to x in (4.2.45), one can show that

$$\lambda_i(x)\partial_x \mathcal{H}_{ij}(x,\xi) + \lambda_j(\xi)\partial_\xi \mathcal{H}_{ij}(x,\xi) = -\sum_{k=1}^n \left(\sigma_{kj}(\xi) + \delta_{kj}\lambda'_j(\xi)\right) \mathcal{H}_{ik}(x,\xi) - \lambda'_i(x)\mathcal{H}_{ij}(x,\xi)$$

$$(4.5.1)$$

Differentiating the boundary conditions in (4.2.46) and (4.2.47), we have

$$\mathcal{H}_{ij}(x,x) + \mathcal{Y}_{ij}(x,x) = k'_{ij}(x) \qquad \text{for } 1 \le i,j \le n(i \ne j), \quad (4.5.2)$$

$$\mathcal{H}_{ij}(x,0) = -\frac{1}{\lambda_j(0)} \sum_{k=1}^{n-m} \lambda_{m+k}(0) \mathcal{H}_{i,m+k}(x,0) q_{k,j} \qquad \text{for } 1 \le i \le j \le m.$$
(4.5.3)

Next, differentiating the boundary conditions in (??)–(3.3.30), we have

$$\mathcal{Y}_{ij}(1,\xi) = \dot{k}_{ij}^{(1)}(\xi), \text{ for } 1 \le j < i \le m \cup m+1 \le i < j \le n$$
 (4.5.4)

and the boundary conditions for  $\mathcal{H}_{ij}$   $(n \ge i \ge j \ge m+1)$  on  $\xi = 0$ :

$$\mathcal{H}_{ij}(x,0) = \dot{k}_{ij}^{(2)}(x), \text{ for } m+1 \le j \le i \le n.$$
 (4.5.5)

In view of the equations (4.2.45), it is easy to see that

$$\lambda_i(x)\mathcal{H}_{ij}(x,x) + \lambda_j(x)\mathcal{Y}_{ij}(x,x) = -\sum_{k=1}^n \left(\sigma_{kj}(x) + \delta_{kj}\lambda'_j(x)\right)K_{ik}(x,x)$$
(4.5.6)

$$\lambda_i(1)\mathcal{H}_{ij}(1,\xi) + \lambda_j(\xi)\mathcal{Y}_{ij}(1,\xi) = -\sum_{k=1}^n \left(\sigma_{kj}(\xi) + \delta_{kj}\lambda'_j(\xi)\right)K_{ik}(1,\xi)$$
(4.5.7)

Combining (4.5.2) and (4.5.6), we have

$$\mathcal{H}_{ij}(x,x) = \frac{\lambda_j(x)k'_{ij}(x) + \sum_{k=1}^n \left(\sigma_{kj}(x) + \delta_{kj}\lambda'_j(x)\right)K_{ik}(x,x)}{\lambda_j(x) - \lambda_i(x)} \quad \text{for } 1 \le i,j \le n(i \ne j).$$
(4.5.8)

Similarly, plugging (4.5.7) into (4.5.4), one immediately obtains, for  $1 \le j < i \le m \cup m+1 \le i < j \le n$ , we have

$$\mathcal{H}_{ij}(1,\xi) = -\frac{1}{\lambda_i(1)} \left( \sum_{k=1}^n \left( \sigma_{kj}(\xi) + \delta_{kj} \lambda'_j(\xi) \right) K_{ik}(1,\xi) + \lambda_j(\xi) \dot{k}_{ij}^{(1)}(\xi) \right), \quad (4.5.9)$$

which are piecewise  $C^0(0, 1)$  function. By the theory in [54], we can prove that there exists a unique piecewise  $\mathcal{H} \in C^0(\mathcal{T})$  for the boundary value problem (4.5.1), (4.5.3), (4.5.5) and (4.5.8)–(4.5.9). Noting the equations (4.2.45), we know that  $\mathcal{Y}$  shares the same regularity as  $\mathcal{H}$ .

Next, we prove the regularity of  $K(\cdot, 0)$ . Obviously, for N = 1, by the theory in [54] and Remark 4.5.1, one can prove that both  $K(\cdot, \cdot)$  and  $K(\cdot, 0) \in C^0(0, 1)$ with bounded  $C^0$  norm, provided that the  $C^0$  compatibility conditions (4.2.50) are satisfied at the the point  $(x, \xi) = (1, 1)$ . Next, we prove the case N = 2. Taking an  $\xi$ -derivative in (4.2.45) yields

$$\lambda_{i}(x)\partial_{x}\mathcal{Y}_{ij}(x,\xi) + \lambda_{j}(\xi)\partial_{\xi}\mathcal{Y}_{ij}(x,\xi) = -\sum_{k=1}^{n} \left(\sigma_{kj}(\xi) + \delta_{kj}\lambda_{j}'(\xi)\right)\mathcal{Y}_{ik}(x,\xi) - \lambda_{j}'(\xi)\mathcal{Y}_{ij}(x,\xi) - \sum_{k=1}^{n} \left(\sigma_{kj}'(\xi) + \delta_{kj}\lambda_{j}''(\xi)\right)\mathcal{K}_{ik}(x,\xi)$$
(4.5.10)

Combining (4.5.2) and (4.5.6), we have

$$\mathcal{Y}_{ij}(x,x) = \frac{\lambda_i(x)k'_{ij}(x) + \sum_{k=1}^n \left(\sigma_{kj}(x) + \delta_{kj}\lambda'_j(x)\right)K_{ik}(x,x)}{\lambda_i(x) - \lambda_j(x)} \quad \text{for } 1 \le i,j \le n(i \ne j).$$
(4.5.11)

Since

$$\lambda_i(x)\mathcal{H}_{ij}(x,0) + \lambda_j(0)\mathcal{Y}_{ij}(x,0) = -\sum_{k=1}^n \left(\sigma_{kj}(0) + \delta_{kj}\lambda'_j(0)\right)K_{ik}(x,0)$$
(4.5.12)

Plugging (4.5.3) and (4.5.5), respectively, one obtains

$$\mathcal{Y}_{ij}(x,0) = -\frac{1}{\lambda_j(0)} \bigg( \lambda_i(x) \dot{k}_{ij}^{(2)}(x) + \sum_{k=1}^n \big( \sigma_{kj}(0) + \delta_{kj} \lambda_j'(0) \big) K_{ik}(x,0) \bigg), \quad (4.5.13)$$
  
for  $m+1 \le j \le i \le n$ 

and

$$\mathcal{Y}_{ij}(x,0) = -\frac{1}{\lambda_j(0)} \sum_{k=1}^n \left( \sigma_{kj}(0) + \delta_{kj} \lambda_j'(0) \right) K_{ik}(x,0) + \frac{1}{\lambda_j^2(0)} \sum_{k=1}^{n-m} \lambda_{m+k}^2(0) q_{k,j} \mathcal{Y}_{i,m+k}(x,0) + \frac{1}{\lambda_j^2(0)} \sum_{k=1}^{n-m} \sum_{s=1}^n \lambda_{m+k}(0) q_{k,j} \left( \sigma_{s,m+k}(0) + \delta_{s,m+k} \lambda_{m+k}'(0) \right) K_{is}(x,0)$$
for  $1 \le i \le j \le m$ .
$$(4.5.14)$$

Noting (4.5.4), (4.5.11), (4.5.13) and  $K(\cdot, 0) \in C^0$ , we know that  $\mathcal{Y}_{ij}(\cdot, \cdot) \in C^0(0, 1) (i \neq j)$ .  $\mathcal{Y}_{ij}(1, \cdot) \in C^0(0, 1) (\text{for } 1 \leq j < i \leq m \cup m + 1 \leq i < j \leq n)$  and  $\mathcal{Y}_{ij}(\cdot, 0) \in C^0(0, 1) (\text{for } m + 1 \leq j \leq i \leq n)$ . By the  $C^1$  compatibility conditions (4.2.51) at the point  $(x, \xi) = (1, 1)$  and using the theory in [54] and Remark 4.5.1, we can prove that there exists a unique piecewise  $C^0$  function  $\mathcal{Y} = \mathcal{Y}(x, \xi)$  for the boundary value problem (4.5.10), (4.5.11), (4.5.13), (4.5.4) and (4.5.14), which satisfies  $\mathcal{Y}(\cdot, \cdot), \ \mathcal{Y}(\cdot, 0) \in C^0(0, 1)$ . Noting (4.5.12) and (4.5.6), we know that  $\mathcal{H}(\cdot, \cdot), \ \mathcal{H}(\cdot, 0) \in C^0(0, 1)$ . This finishes the proof.

**Remark 4.5.1** It is worthy of mentioning that in [54], we only prove  $K \in L^{\infty}(\mathcal{T})$ and do not clarify the regularity of the kernel because of brevity purposes. However, with the same procedure in [26, Section A.3] and [31], one can prove that K is a piecewise  $C^0$  function with  $K(\cdot, \cdot)$ ,  $K(\cdot, 0) \in C^0(0, 1)$  and  $K(1, \cdot)$  being a function of piecewise  $C^0(0, 1)$  for the boundary problem (4.2.45)-(4.2.49), provided  $\sigma_{ij} \in C^0[0, 1], \lambda_i \in C^1[0, 1] (i, j = 1, \dots, n)$  and the  $C^0$  compatibility conditions (4.2.50) are satisfied at the the point  $(x, \xi) = (1, 1)$ .

**Theorem 4.5.2** Under the assumptions of Theorem 4.5.1, For any  $N \in \mathbb{N}$ , there exists a unique piecewise  $C^{N}(\mathcal{T})$  kernel L to the inverse transformation (4.2.52). Moreover, L(x, x),  $L(x, 0) \in C^{N-1}(0, 1)$ .

**Proof.** Substituting (4.2.40) for (4.2.52), it is easy to see that L is the solution of the following Volterra equations

$$L(x,\xi) = K(x,\xi) + \int_{\xi}^{x} K(x,s)L(s,\xi)ds$$
(4.5.15)

which yields that

$$L(x,x) = K(x,x) \in C^{N-1}(0,1).$$
(4.5.16)

Noting (4.2.43), we have

$$\Sigma(x) + L(x,x)\Lambda(x) - \Lambda(x)L(x,x) = 0.$$
(4.5.17)

Next, Taking a partial derivative in x and  $\xi$  in (4.5.15), respectively, one obtains

$$L_x(x,\xi) = K_x(x,\xi) + K(x,x)L(x,\xi) + \int_{\xi}^{x} K_x(x,\xi)L(s,\xi)ds, \qquad (4.5.18)$$

$$L_{\xi}(x,\xi) = K_{\xi}(x,\xi) - K(x,\xi)L(\xi,\xi) + \int_{\xi}^{x} K(x,s)L_{\xi}(s,\xi)ds.$$
(4.5.19)

Substituting (4.5.18) and (4.5.19) for (4.2.42) and using integration by parts, one has

$$\Lambda(x)L_x(x,\xi) + L_{\xi}(x,\xi)\Lambda(\xi) = \left(\Sigma(x) - \Lambda_{\xi}(\xi)\right)L(x,\xi)$$
(4.5.20)

Again by (4.5.15), we have

$$L(x,0) = K(x,0) + \int_0^x K(x,s)L(s,0)ds$$
(4.5.21)

since both K(x, 0) and K(x, x) are  $C^{N-1}$  continuous functions, by a suitable iteration procedure (see [87, Theorem 3.2, Pages 32–34]), it easy to see that there exists  $L(x, 0) = l(x) \in C^{N-1}(0, 1)$  for the Volterra equation of the second kind (4.5.21).

On the other hand, substituting (4.2.52) for (4.2.40), one gets

$$L(x,\xi) = K(x,\xi) + \int_{\xi}^{x} L(x,s)K(s,\xi)ds$$
(4.5.22)

then

$$L(1,\xi) = K(1,\xi) + \int_{\xi}^{1} L(1,s)K(s,\xi)ds$$
(4.5.23)

With the same argument above, we can see that  $L_{ij}(1,\xi) = \tilde{l}_{ij}(\xi) (m \ge i > j \ge 1, n \ge j > i \ge m+1)$  on x = 1 are functions of piecewise  $C^N(0,1)$ . Then, For the boundary

problem (4.5.20) with the boundary conditions (4.5.17), and

$$L_{ij}(1,\xi) = l_{ij}(\xi), \text{ for } 1 \le j < i \le m \ \cup \ m+1 \le i < j \le n \tag{4.5.24}$$

$$L_{ij}(n,\xi) = l_{ij}(\xi), \text{ for } 1 \le i \le i \le m \ \cup \ m+1 \le i \le i \le n \tag{4.5.25}$$

$$L_{ij}(x,0) = l_{ij}(x), \text{ for } 1 \le i \le j \le m \cup m+1 \le j \le i \le n.$$
 (4.5.25)

by Theorem 4.5.1, one immediately gets Theorem 4.5.2.

#### Appendix B 4.6

In this appendix, we first sketch out four useful lemmas (the details can be found in [26]).

**Lemma 4.6.1** There exists a positive real number  $c_1$ , such that

$$|\mathcal{K}[\gamma]| + |\mathcal{L}[\gamma]| + |\mathcal{K}_1[\gamma]| + |\mathcal{K}_2[\gamma]| + |\mathcal{L}_1[\gamma]| \le c_1(|\gamma| + ||\gamma||_{L^1}).$$
(4.6.1)

**Lemma 4.6.2** Suppose  $\|\gamma\|_{\infty}$  is suitable small, one can see that

$$F_1| \le c_2(|\gamma| + ||\gamma||_{L^1}), \tag{4.6.2}$$

$$F_2| \le c_3(|\gamma|^2 + ||\gamma||_{L^1}^2), \tag{4.6.3}$$

$$|F_3| \le c_4(|\gamma| + ||\gamma||_{L^1})(||\gamma_x||_{L^2} + |\gamma_x|), \qquad (4.6.4)$$

$$|F_4| \le c_5(|\gamma|^2 + \|\gamma\|_{L^1}^2). \tag{4.6.5}$$

### Lemma 4.6.3

 $|F_{11}| \le c_6(|\zeta| + \|\zeta\|_{L^1}),$ (4.6.6) $|F_{12}| \le c_7(|\gamma_x| + |\gamma| + ||\gamma||_{L^1}),$ (4.6.7) $|F_{21}| \le c_8(|\gamma| + \|\gamma\|_{L^1})(|\zeta| + \|\zeta\|_{L^1}),$ (4.6.8) $|F_5| \le c_9(|\zeta| + \|\zeta\|_{L^2})(|\gamma| + \|\gamma\|_{L^2}) + c_{10}(|\zeta| + \|\zeta\|_{L^2})((|\gamma_x| + \|\gamma_x\|_{L^2})) + c_{11}|\gamma(0)||\zeta(0)|,$ (4.6.9) $F_{c} < c_{10}(|\gamma| + ||\gamma||_{L^{2}})(|\zeta| + ||\zeta||_{L^{2}})$ (4.6.10)

$$|F_6| \le c_{12}(|\gamma| + ||\gamma||_{L^2})(|\zeta| + ||\zeta||_{L^2})$$

### Lemma 4.6.4

$$\begin{aligned} |F_{13}| &\leq c_{13}(|\zeta|^2 + \|\zeta\|_{L^2}^2) + c_{14}(|\theta| + \|\theta\|_{L^1}), \end{aligned} \tag{4.6.11} \\ |F_{14}| &\leq c_{14}(|\zeta| + \|\zeta\|_{L^1})(1 + |\gamma_x| + |\gamma| + \|\gamma\|_{L^1}) + c_{15}(|\zeta| + |\zeta_x| + \|\zeta\|_{L^1}), \end{aligned} \tag{4.6.12} \\ |F_{22}| &\leq c_{16}(|\gamma| + \|\gamma\|_{L^1})(|\theta| + \|\theta\|_{L^1}) + c_{17}(|\zeta|^2 + \|\zeta\|_{L^2}^2), \end{aligned} \tag{4.6.13} \\ |F_7| &\leq c_{18}(|\zeta|^2 + \|\zeta\|_{L^2}^2)(1 + |\gamma| + \|\gamma_x\|) \\ &+ c_{19}(|\zeta| + \|\zeta\|_{L^2})(|\zeta_x| + \|\zeta\|_{L^2}) \end{aligned}$$

Chapter 4: Boundary Exponential Stabilization of 1-D Inhomogeneous Quasilinear Hyperbolic Systems 139

$$+ c_{20}(|\gamma| + ||\gamma||_{L^{2}} + |\gamma_{x}|)(|\theta| + ||\theta||_{L^{2}}) + c_{21}(|\zeta(0)|^{2} + |\gamma(0)||\theta(0)|), |F_{8}| \leq c_{22}(|\zeta|^{2} + ||\zeta||_{L^{2}}^{2})(1 + ||\gamma||_{\infty}) + c_{23}(|\gamma| + ||\gamma||_{L^{2}})(|\theta| + ||\theta||_{L^{2}}).$$
(4.6.15)

Next, we show the following proposition which is also mentioned in [26], however here more technical developments are involved.

**Proposition 4.6.1** There exists  $\delta > 0$  such that for any  $|\gamma| + |\zeta| \leq \delta$ , one has

$$\|\theta\|_{\infty} \le C_1(\|\gamma_{xx}\|_{\infty} + \|\gamma_x\|_{\infty} + \|\gamma\|_{\infty}), \tag{4.6.16}$$

$$\|\theta\|_{L^2} \le C_2(\|\gamma_{xx}\|_{L^2} + \|\gamma_x\|_{L^2} + \|\gamma\|_{L^2}), \qquad (4.6.17)$$

$$\|\gamma_{xx}\|_{\infty} \le C_3(\|\theta\|_{\infty} + \|\zeta\|_{\infty} + \|\gamma\|_{\infty}), \tag{4.6.18}$$

$$\|\gamma_{xx}\|_{L^2} \le C_4(\|\theta\|_{L^2} + \|\zeta\|_{L^2} + \|\gamma\|_{L^2}), \tag{4.6.19}$$

where  $C_1$ ,  $C_2$ ,  $C_3$  and  $C_4$  are positive constants.

**Proof.** We prove the next three lemmas to get Proposition 4.6.1.

**Lemma 4.6.5** There exists  $\delta$  such that, if  $|\gamma| \leq \delta$ , then the following inequalities hold:

$$\|\zeta\|_{\infty} \le c_1(\|\gamma_x\|_{\infty} + \|\gamma\|_{\infty}) \tag{4.6.20}$$

$$\|\zeta\|_{L^2} \le c_2(\|\gamma_x\|_{L^2} + \|\gamma\|_{L^2}), \tag{4.6.21}$$

$$\|\gamma_x\|_{\infty} \le c_3(\|\zeta\|_{\infty} + \|\gamma\|_{\infty}), \tag{4.6.22}$$

$$\|\gamma_x\|_{L^2} \le c_4(\|\zeta\|_{L^2} + \|\gamma\|_{L^2}) \tag{4.6.23}$$

**Proof.** Noting (4.4.12), one can easily see that

÷

:

$$\zeta(t,x) + \Lambda(x)\gamma_x(t,x) - G(x)\gamma(t,0) = F_3[\gamma,\gamma_x] + F_4[\gamma]$$

$$(4.6.24)$$

The difference between our proof and the proof in [26, Lemma B.6] is the appearance of the term  $G(x)\gamma(t,0)$  in (4.6.24). Noting (4.2.44) and Theorem 4.5.1, we have  $G(\cdot) \in C^1(0,1)$  with bounded  $C^1$  norm. Then since one can show that

$$\|G(\cdot)\gamma(t,0)\|_{L^2} \le c_5 \|G(\cdot)\gamma(t,0)\|_{\infty} \le c_6 \|\gamma\|_{\infty} \le c_7 (\|\gamma_x\|_{L^2} + \|\gamma\|_{L^2}), \qquad (4.6.25)$$

which yields, by the same argument in [26, Lemma B.6], (4.6.20)-(4.6.22).

On the other hand, by the special structure of G(x), we have

$$\|\partial_x \gamma_1\|_{L^2} \le c_8(\|\zeta\|_{L^2} + \|\gamma_x\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}\|\gamma\|_{\infty}), \tag{4.6.26}$$

$$\|\partial_x \gamma_2\|_{L^2} \le c_9(\|\zeta\|_{L^2} + \|\gamma_1\|_{\infty} + \|\gamma_x\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}\|\gamma\|_{\infty}), \qquad (4.6.27)$$

.

÷

$$\|\partial_x \gamma_m\|_{L^2} \le c_{m+7}(\|\zeta\|_{L^2} + \sum_{r=1}^{m-1} \|\gamma_r\|_{\infty} + \|\gamma_x\|_{L^2} \|\gamma\|_{\infty} + \|\gamma\|_{L^2} \|\gamma\|_{\infty}), \qquad (4.6.28)$$

$$\|\partial_x \gamma_s\|_{L^2} \le c_{s+7}(\|\zeta\|_{L^2} + \sum_{r=1}^m \|\gamma_r\|_\infty + \|\gamma_x\|_{L^2} \|\gamma\|_\infty + \|\gamma\|_{L^2} \|\gamma\|_\infty)$$
(4.6.29)

in which  $s = m + 1, \dots, n$ . Noting the classical Sobolev's inequality

$$\|\gamma\|_{L^{\infty}} \le \widetilde{C}\Big(\|\gamma\|_{L^2} + \|\gamma_x\|_{L^2}\Big) \le \widetilde{\overline{C}}\|\gamma\|_{H^1}, \qquad (4.6.30)$$

one gets that

$$\|\partial_x \gamma_1\|_{L^2} \le C_1(\|\zeta\|_{L^2} + \|\gamma_x\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}\|\gamma\|_{\infty}), \tag{4.6.31}$$

$$\begin{aligned} \|\partial_x \gamma_2\|_{L^2} &\leq C_2(\|\zeta\|_{L^2} + \|\gamma\|_{L^2} + \|\partial_x \gamma_1\|_{L^2} + \|\gamma_x\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}\|\gamma\|_{\infty}), \quad (4.6.32) \\ \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \end{aligned}$$

$$\|\partial_x \gamma_m\|_{L^2} \le C_m(\|\zeta\|_{L^2} + \|\gamma\|_{L^2} + \sum_{r=1}^{m-1} \|\gamma_r\|_{L^2} + \|\gamma_x\|_{L^2} \|\gamma\|_{\infty} + \|\gamma\|_{L^2} \|\gamma\|_{\infty}), \quad (4.6.33)$$

$$\|\partial_x \gamma_s\|_{L^2} \le C_s(\|\zeta\|_{L^2} + \|\gamma\|_{L^2} + \sum_{r=1}^m \|\gamma_r\|_{L^2} + \|\gamma_x\|_{L^2} \|\gamma\|_{L^\infty} + \|\gamma\|_{L^2} \|\gamma\|_{\infty}) \quad (4.6.34)$$

where  $s = m + 1, \dots, n$ . Then, we can easily obtain by induction that

$$\|\gamma_x\|_{L^2} \le \tilde{c}_1(\|\zeta\|_{L^2} + \|\gamma_x\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}\|\gamma\|_{\infty} + \|\gamma\|_{L^2}), \qquad (4.6.35)$$

which concludes (4.6.23), under the assumption that  $\|\gamma\|_{\infty}$  is small enough.

Combining the same technical approach as in [26, Lemma B.7 and Lemma B.8] and an analogous argument used in the proof of Lemma 4.6.5 and noting  $G \in C^1$ , the details of which we omit, one can show the next two lemmas.

**Lemma 4.6.6** There exists  $\delta$  such that, if  $\|\gamma\|_{\infty} \leq \delta$ , then the following inequalities hold:

$$\|\gamma_{xx}\|_{\infty} \le c_1(\|\zeta_x\|_{\infty} + \|\zeta\|_{\infty} + \|\gamma\|_{\infty}) \tag{4.6.36}$$

$$\|\gamma_{xx}\|_{L^2} \le c_2(\|\zeta_x\|_{L^2} + \|\zeta\|_{L^2} + \|\gamma\|_{L^2})$$
(4.6.37)

$$\|\zeta_x\|_{\infty} \le c_3(\|\gamma_{xx}\|_{\infty} + \|\zeta\|_{\infty} + \|\gamma\|_{\infty})$$
(4.6.38)

$$\|\zeta_x\|_{L^2} \le c_4(\|\gamma_{xx}\|_{L^2} + \|\zeta\|_{L^2} + \|\gamma\|_{L^2}), \qquad (4.6.39)$$

where  $c_1$ ,  $c_2$ ,  $c_3$  and  $c_4$  are positive constants.

and

**Lemma 4.6.7** There exists  $\delta$  such that, if  $\|\gamma\|_{\infty} + \|\zeta\|_{\infty} \leq \delta$ , then the following inequalities hold:

$$\|\theta\|_{\infty} \le c_1(\|\zeta_x\|_{\infty} + \|\zeta\|_{\infty} + \|\gamma\|_{\infty}) \tag{4.6.40}$$

$$\|\theta\|_{L^2} \le c_2(\|\zeta_x\|_{L^2} + \|\zeta\|_{L^2} + \|\gamma\|_{L^2}) \tag{4.6.41}$$

$$\|\theta\|_{L^{2}} \leq c_{2}(\|\zeta_{x}\|_{L^{2}} + \|\zeta\|_{L^{2}} + \|\gamma\|_{L^{2}})$$

$$\|\zeta_{x}\|_{\infty} \leq c_{3}(\|\theta\|_{\infty} + \|\zeta\|_{\infty} + \|\gamma\|_{\infty})$$

$$(4.6.42)$$

$$\|\zeta_x\|_{L^2} \le c_4(\|\theta\|_{L^2} + \|\zeta\|_{L^2} + \|\gamma\|_{L^2}), \tag{4.6.43}$$

where  $c_1, c_2, c_3$  and  $c_4$  are positive constants.

The above three Lemma 4.6.5–4.6.7 immediately yield Proposition 4.6.1. 

# Part III

# Exact Boundary Synchronization for a Coupled System of Wave Equations
# Chapter 5

# Exact Boundary Controllability and Observability for a Coupled System of Quasilinear Wave Equations

5.1	Introduction	145
5.2	Existence and Uniqueness of Semi-global $\mathbb{C}^2$ Solution .	148
5.3	Local Exact Boundary Controllability	153
<b>5.4</b>	Local Exact Boundary Observability	155

### 5.1 Introduction

There are many publications concerning the exact controllability for linear hyperbolic systems (see [88]-[100] and the references therein). As a special form of second order hyperbolic equations, the exact boundary controllability for linear wave equation was obtained in a complete manner (see [88]-[89]). Using the Hilbert Uniqueness Method (HUM) suggested in [88]-[89], Zuazua [122]-[123], Emanuilov [37] and Lasiecka & Triggiani [66] subsequently established the exact boundary controllability for some semilinear wave equations. In the quasilinear case, based on the result on the semi-global  $C^2$  solution, by a direct constructive method, Li and Yu established the local exact boundary controllability for single 1-D quasilinear wave equation with boundary conditions of various types (see [70]-[83]). Later, this result has been applied to get the exact boundary controllability of nodal profile and on a

tree-like network for quasilinear wave equations respectively (see [43], [110]). For the following second order quasilinear hyperbolic system

$$u_{tt} + A(u, u_x, u_t)u_{tx} + B(u, u_x, u_t)u_{xx} = C(u, u_x, u_t)$$
(5.1.1)

under some different hypothesis on matrices A, B and C, the corresponding local exact boundary controllability was obtained by Yu [118] and Wang [109] respectively.

On the other hand, as a dual problem of controllability, the exact boundary observability for wave equations has been widely studied (see [27], [67], [88]-[89]). In fact, The essence of J.-L. Lions's HUM is to use the duality to get the controllability by a corresponding observability inequality. Based on the result on semi-global classical solutions for quasilinear hyperbolic systems, by a constructive method, the exact boundary observability for single quasilinear wave equation was established by Li (see [72],[71]) and Guo(see [49]) respectively, and some implicit dualities have been shown between the exact boundary controllability and the exact boundary observability in the quasilinear case. As for the second order quasilinear hyperbolic system (5.1.1), the corresponding local exact boundary observability was obtained by Yu [119] under some hypothesis on matrices A, B and C.

In this chapter, we consider a kind of coupled system of 1-D quasilinear wave equations, which can be rewritten in the form of second order quasilinear hyperbolic systems discussed in [109] and [119], but the systems in [109] and [119] were too general, which is not convenient to get the results we want. Therefore, for the coupled system of quasilinear wave equations under consideration, we restudy its controllability and observability and get brief results. In addition, the corresponding discussion of exact boundary null controllability and weak observability are added. Based on the existence and uniqueness of semi-global  $C^1$  solution and the local exact boundary controllability for first order quasilinear hyperbolic systems, by a constructive method developed by Li (see [74] and [71] et.al.), we can obtain the local exact boundary (null) controllability and (weak) observability for a coupled system of quasilinear wave equations with boundary conditions of different types. The conclusions we get provide basis for studying the exact boundary synchronization for the coupled system of wave equations (see [81]).

Consider the following the coupled system of 1-D quasilinear wave equations

$$\frac{\partial^2 w_i}{\partial t^2} - a_i^2(w) \frac{\partial^2 w_i}{\partial x^2} + \sum_{j=1}^n a_{ij}(w) w_j = 0, \qquad (5.1.2)$$

where  $w = (w_1, \ldots, w_n)^T$  is the unknown vector function of (t, x),  $a_i(w)$  and  $a_{ij}(w)$   $(i, j = 1, \cdots, n)$  are all  $C^1$  functions of w on the domain under consideration, satisfying

$$a_i(0) > 0, i = 1, \cdots, n.$$
 (5.1.3)

On one end x = 0, we prescribe any one of the following Dirichlet type, Neumann type, Coupled third type and Coupled dissipative type boundary conditions:

$$x = 0: w_i = h_i(t) \ (i = 1, \cdots, n),$$
 (5.1.4a)

$$x = 0: \frac{\partial w_i}{\partial x} = h_i(t) \ (i = 1, \cdots, n), \tag{5.1.4b}$$

$$x = 0: \frac{\partial w_i}{\partial x} - \sum_{j=1}^n b_{ij}(w)w_j = h_i(t) \ (i = 1, \cdots, n),$$
(5.1.4c)

$$x = 0: \frac{\partial w_i}{\partial x} - \sum_{j=1}^n c_{ij}(w) \frac{\partial w_j}{\partial t} = h_i(t) \ (i = 1, \cdots, n),$$
(5.1.4d)

where  $b_{ij} = b_{ij}(w)$  and  $c_{ij} = c_{ij}(w)$  are  $C^1$  functions of w,  $h_i(t)$  are  $C^2$  functions (in case(5.1.4a)) or  $C^1$  functions (in case(5.1.4b)-(5.1.4d)).

Similarly, on another end x = L, the boundary conditions are

$$x = L : w_i = \overline{h}_i(t) \ (i = 1, \cdots, n), \tag{5.1.5a}$$

$$x = L : \frac{\partial w_i}{\partial x} = \overline{h}_i(t) \ (i = 1, \cdots, n),$$
 (5.1.5b)

$$x = L : \frac{\partial w_i}{\partial x} + \sum_{j=1}^n \overline{b}_{ij}(w)w_j = \overline{h}_i(t) \ (i = 1, \cdots, n),$$
(5.1.5c)

$$x = L : \frac{\partial w_i}{\partial x} + \sum_{j=1}^n \overline{c}_{ij}(w) \frac{\partial w_j}{\partial t} = \overline{h}_i(t) \ (i = 1, \cdots, n),$$
(5.1.5d)

where  $\overline{b}_{ij} = \overline{b}_{ij}(w)$  and  $\overline{c}_{ij} = \overline{c}_{ij}(w)$  are  $C^1$  functions of w,  $\overline{h}_i(t)$  are  $C^2$  functions (in case(5.1.5a)) or  $C^1$  functions (in case(5.1.5b)-(5.1.5d)).

The initial conditions are given by

$$t = 0: (w, w_t) = (\varphi(x), \psi(x)), \ 0 \le x \le L,$$
(5.1.6)

where  $\varphi = (\varphi_1, \dots, \varphi_n)^T$  is a  $C^2$  vector function on [0, L],  $\psi = (\psi_1, \dots, \psi_n)^T$  is a  $C^1$  vector function on [0, L], they are all with small  $C^2$  norm or  $C^1$  norm, such that the conditions of  $C^2$  compatibility at the points (t, x) = (0, 0) and (0, L) are satisfied respectively.

Obviously, w = 0 is an equilibrium of system (5.1.2). Based on the theory of semiglobal  $C^2$  solution, by a constructive method (see [74] and [71]), we will establish local exact boundary controllability and observability around w = 0.

This chapter is organized as follows. The existence and uniqueness of semi-global  $C^2$  solution to the coupled system of quasilinear wave equations (5.1.2) with boundary conditions (5.1.4) and (5.1.5) of different types will be presented in § 5.2. Based on

this, in § 5.3, we obtain the corresponding local exact boundary (null) controllability with boundary controls on one end or on two ends and in § 5.4, we obtain the corresponding local exact boundary (weak) observability with observed values on one end or on two ends.

## 5.2 Existence and Uniqueness of Semi-global $C^2$ Solution

For the purpose of getting the local exact boundary controllability and observability for system (5.1.2) with boundary conditions (5.1.4)-(5.1.5), we should first prove the existence and uniqueness of semi-global  $C^2$  solution for mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6). In order to get it in a unified way, the best way is to reduce the equations to a first order quasilinear hyperbolic system and use the corresponding results of semi-global  $C^1$  solution.

Setting

$$u_i = \frac{\partial w_i}{\partial x}, v_i = \frac{\partial w_i}{\partial t} (i = 1, \cdots, n), \qquad (5.2.1)$$

$$u = (u_1, \cdots, u_n)^T, v = (v_1, \cdots, v_n)^T,$$
 (5.2.2)

equations (5.1.2) can be reduced to the following first order quasilinear system

$$\begin{cases} \frac{\partial w}{\partial t} = v, \\ \frac{\partial u}{\partial t} - \frac{\partial v}{\partial x} = 0, \\ \frac{\partial v}{\partial t} - \Lambda^2(w)\frac{\partial u}{\partial x} = -A(w)w, \end{cases}$$
(5.2.3)

where  $\Lambda(w) = diag\{a_1(w), \dots, a_n(w)\}$  and  $A(w) = (a_{ij}(w))_{n \times n}$ . Its equivalent matrix form can be written as

$$\frac{\partial}{\partial t} \begin{pmatrix} w \\ u \\ v \end{pmatrix} + \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -I_n \\ 0 & -\Lambda^2(w) & 0 \end{pmatrix} \frac{\partial}{\partial x} \begin{pmatrix} w \\ u \\ v \end{pmatrix} = \begin{pmatrix} v \\ 0 \\ -A(w)w \end{pmatrix}.$$
 (5.2.4)

The characteristic equation of (5.2.3) or (5.2.4) is

$$det \begin{pmatrix} \lambda I_n & 0 & 0\\ 0 & \lambda I_n & I_n\\ 0 & \Lambda^2(w) & \lambda I_n \end{pmatrix} = \lambda^n |\lambda^2 I_n - \Lambda^2(w)| = 0, \qquad (5.2.5)$$

whose solutions, the real eigenvalues of system (5.2.3) or (5.2.4), are

$$\lambda_i^- = -a_i(w), \quad \lambda_i^0 = 0, \quad \lambda_i^+ = a_i(w) \quad (i = 1, \cdots, n),$$
 (5.2.6)

the corresponding left eigenvectors, which constitute a complete set, can be chosen as

$$l_i^- = (\mathbf{0}, a_i(w)\mathbf{e}_i, \mathbf{e}_i), l_i^0 = (\mathbf{e}_i, \mathbf{0}, \mathbf{0}), l_i^+ = (\mathbf{0}, -a_i(w)\mathbf{e}_i, \mathbf{e}_i), \quad (i = 1, \cdots, n), \quad (5.2.7)$$

in which  $\mathbf{0} = (0, \dots, 0)$  is the n - D zero vector,  $\mathbf{e}_i = (0, \dots, \overset{(i)}{1}, \dots, 0)$  is a n - D row vector. Thus, (5.2.3) or (5.2.4) reduced from the system (5.1.2) is a first order quasilinear hyperbolic system.

Let

$$U = (w, u, v)^T.$$
 (5.2.8)

Setting diagonal variables as

$$\begin{cases} V_i^- = l_i^- U = a_i(w)u_i + v_i, \\ V_i^0 = l_i^0 U = w_i, \\ V_i^+ = l_i^+ U = -a_i(w)u_i + v_i, \end{cases} (i = 1, \cdots, n), \qquad (5.2.9)$$

namely,

$$\begin{cases} V^{-} = \Lambda(w)u + v, \\ V^{0} = w, \\ V^{+} = -\Lambda(w)u + v, \end{cases}$$
(5.2.10)

we have

$$\begin{cases} w = V^{0}, \\ u = \frac{1}{2}\Lambda^{-1}(V^{0})(V^{-} - V^{+}), \\ v = \frac{1}{2}(V^{-} + V^{+}). \end{cases}$$
(5.2.11)

When equations (5.1.2) are reduced to system (5.2.3) or (5.2.4), the boundary conditions (5.1.4) will be correspondingly replaced by

$$x = 0: v = \dot{H}(t),$$
 (5.2.12a)

$$x = 0: u = H(t),$$
 (5.2.12b)

$$x = 0: u - B(w)w = H(t),$$
 (5.2.12c)

$$x = 0: u - C(w)v = H(t),$$
 (5.2.12d)

in which  $H(t) = (h_1(t), \cdots, h_n(t))^T$ ,  $B(w) = (b_{ij}(w))_{n \times n}$ ,  $C(w) = (c_{ij}(w))_{n \times n}$ .

Noting (5.2.11), if

$$det(\Lambda^{-1}(0) + C(0)) \neq 0, \tag{5.2.13}$$

it is easy to see that, at least in a neighborhood of U = 0, the boundary conditions

(5.2.12) (namely, (5.1.4)) on x = 0 can be rewritten as

$$x = 0:$$
  $V^+ = -V^- + 2\dot{H}(t),$  (5.2.14a)

$$x = 0:$$
  $V^+ = V^- - 2\Lambda(V^0)H(t),$  (5.2.14b)

$$x = 0: V^+ = V^- - 2\Lambda(V^0)B(V^0)V^0 - 2\Lambda(V^0)H(t), (5.2.14c)$$

$$x = 0: V^{+} = \left(\Lambda^{-1}(V^{0}) + C(V^{0})\right)^{-1} \left(\Lambda^{-1}(V^{0}) - C(V^{0})\right) V^{-} - 2\left(\Lambda^{-1}(V^{0}) + C(V^{0})\right)^{-1} H(t), (5.2.14d)$$

which can be uniformly presented as

$$x = 0: V^+ = G(t, V^-, V^0) + \widetilde{H}(t),$$
 (5.2.15)

where G and  $\widetilde{H}$  are  $C^1$  functions with respect to their arguments, and without loss of generality, we may assume that

$$G(t,0,0) \equiv 0. \tag{5.2.16}$$

Similarly, if

$$det(\Lambda^{-1}(0) + \overline{C}(0)) \neq 0,$$
 (5.2.17)

then at least in a neighborhood of U = 0, the boundary conditions (5.1.5) on x = L are all of the form

$$x = L: V^{-} = \overline{G}(t, V^{0}, V^{+}) + \overline{H}(t),$$
 (5.2.18)

in which  $\overline{G}$  and  $\overline{H}$  are  $C^1$  functions with respect to their arguments, and without loss of generality, we may assume that

$$\overline{G}(t,0,0) \equiv 0.$$
 (5.2.19)

Meanwhile, the corresponding initial condition (5.1.6) can be reduced as

$$t = 0 : U = (\varphi(x), \varphi'(x), \psi(x))^T, \ 0 \le x \le L.$$
 (5.2.20)

Together with the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) and (0, L) for the coupled system of wave equations (5.1.2) with the boundary conditions (5.1.4)-(5.1.5) on x = 0 and x = L respectively and the initial condition(5.1.6), it is easy to see that the conditions of  $C^1$  compatibility at these two points for the mixed initial-boundary value problem (5.2.3), (5.2.15), (5.2.18) and (5.2.20) are also satisfied.

For the convenience of statement, we denote in the whole chapter that

$$d = \begin{cases} 2 & \text{for}(5.1.4a), \\ 1 & \text{for}(5.1.4b) - (5.1.4d) \end{cases}$$
(5.2.21)

and

$$\bar{d} = \begin{cases} 2 & \text{for}(5.1.5a), \\ 1 & \text{for}(5.1.5b) - (5.1.5d), \end{cases}$$
(5.2.22)

Based on the theory of semi-global  $C^1$  solution for first order quasilinear hyperbolic system with zero eigenvalues (see [71], [73], [70] and [117]), it is easy to get the following

**Lemma 5.2.1** Under the hypotheses given in §5.1, suppose that the conditions of  $C^2$  compatibility are satisfied at the points (t,x) = (0,0) and (0,L), respectively. Suppose furthermore that (5.2.13) and (5.2.17) hold. For any given and possibly quite large T > 0, if  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$ ,  $\|h_i\|_{C^d[0,T]}$  and  $\|\overline{h}_i\|_{C^{\overline{d}}[0,T]}$  ( $i = 1, \dots, n$ ) are suitably small, then the forward mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6) admits a unique semi-global  $C^2$  solution w = w(t,x) on the domain  $R(T) = \{(t,x)| 0 \le t \le T, 0 \le x \le L\}$ , and

$$\|w\|_{C^{2}[R(T)]} \leq C\left(\sum_{i=1}^{n} \|(\varphi_{i}, \psi_{i})\|_{C^{2}[0,L] \times C^{1}[0,L]} + \sum_{i=1}^{n} \|(h_{i}, \bar{h}_{i})\|_{C^{d}[0,L] \times C^{\bar{d}}[0,L]}\right), \quad (5.2.23)$$

where C is a positive constant possibly depending on T.

**Corollary 5.2.1** Under the hypotheses given in §5.1, if  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  ( $i = 1, \dots, n$ ) are suitably small, then Cauchy problem (5.1.2) and (5.1.6) admits a unique global  $C^2$  solution w = w(t, x) on the whole maximum determinate domain (see, for example, [84, Remark 4.2 on Chapter 1 pages 72-73])and

$$\|w\|_{C^2} \le C \sum_{i=1}^n \|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]},$$
(5.2.24)

where C is a positive constant.

As for the backward mixed initial-boundary value problem (5.1.2), (5.1.4)-(5.1.5) with final condition

$$t = T : (w, w_t) = (\Phi(x), \Psi(x)), \ 0 \le x \le L.$$
(5.2.25)

in which  $\Phi = (\Phi_1, \dots, \Phi_n)^T$  is a  $C^2$  vector function on [0, L],  $\Psi = (\Psi_1, \dots, \Psi_n)^T$  is a  $C^1$  vector function on [0, L], they are all with small  $C^2$  norm or  $C^1$  norm, such that the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L) are satisfied respectively, similar results can be obtained as follows

**Lemma 5.2.2** Under the hypotheses given in §5.1, suppose that the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (T, 0) and (T, L), respectively. Suppose furthermore that

$$det(\Lambda^{-1}(0) - C(0)) \neq 0, \tag{5.2.26}$$

and

$$det(\Lambda^{-1}(0) - \overline{C}(0)) \neq 0, \tag{5.2.27}$$

hold. For any given and possibly quite large T > 0, if  $\|(\Phi_i, \Psi_i)\|_{C^2[0,L] \times C^1[0,L]}$ ,  $\|h_i\|_{C^d[0,T]}$ and  $\|\overline{h}_i\|_{C^{\overline{d}}[0,T]}$   $(i = 1, \dots, n)$  are suitably small, then the backward mixed initialboundary value problem (5.1.2), (5.2.25) and (5.1.4)-(5.1.5) admits a unique semiglobal  $C^2$  solution w = w(t,x) on the domain  $R(T) = \{(t,x)| 0 \le t \le T, 0 \le x \le L\}$ and

$$\|w\|_{C^{2}[R(T)]} \leq C\left(\sum_{i=1}^{n} \|(\Phi_{i}, \Psi_{i})\|_{C^{2}[0,L] \times C^{1}[0,L]} + \sum_{i=1}^{n} \|(h_{i}, \bar{h}_{i})\|_{C^{d}[0,L] \times C^{\bar{d}}[0,L]}\right),$$
(5.2.28)

where C is a positive constant possibly depending on T.

**Remark 5.2.1** If  $a_i, a_{ij}, b_{ij}, \overline{b}_{ij}, c_{ij}, \overline{c}_{ij}$  are all independent of w, then the problem is linear and it is not necessary to assume the small date for initial (resp. finial) value as well as boundary functions in Lemma 5.2.1 (resp. Lemma 5.2.2). Moreover, the  $C^2$  solution obtained is global.

**Remark 5.2.2** Suppose  $a_i(w) \equiv a > 0 (i = 1, \dots, n)$  are the same constant, the conditions (5.2.13) and (5.2.17) in Lemma 5.2.1 mean -a is not the eigenvalue of both matrices C(0) and  $\overline{C}(0)$ , while the conditions (5.2.26) and (5.2.27) in Lemma 5.2.2 mean a is not the eigenvalue of both matrices C(0) and  $\overline{C}(0)$ .

**Remark 5.2.3** By Lemma 5.2.1, conditions (5.2.13) and (5.2.17) for matrices C and  $\overline{C}$  in the coupled dissipative type boundary conditions (5.1.4d) and (5.1.5d) are imposed for guaranteing the well-posedness of the corresponding forward mixed initialboundary value problem, however, by Lemma 5.2.2, conditions (5.2.26) and (5.2.27) for matrices C and  $\overline{C}$  in the coupled dissipative type boundary conditions (5.1.4d) and (5.1.4d) and (5.1.5d) are imposed for guaranteing the well-posedness of the corresponding backward mixed initial-boundary value problem. Other coupled matrices A, B and  $\overline{B}$ , however, can be completely arbitrary.

**Remark 5.2.4** Lemma 5.2.1, Corollary 5.2.1 and Lemma 5.2.2 are still valid for the following equations

$$\frac{\partial^2 w_i}{\partial t^2} - a_i^2(w) \frac{\partial^2 w_i}{\partial x^2} + \sum_{j=1}^n \overline{a}_{ij}(w) \frac{\partial w_j}{\partial t} = 0, \qquad (5.2.29)$$

or

$$\frac{\partial^2 w_i}{\partial t^2} - a_i^2(w) \frac{\partial^2 w_i}{\partial x^2} + \sum_{j=1}^n a_{ij}(w) w_j + \sum_{j=1}^n \overline{a}_{ij}(w) \frac{\partial w_j}{\partial t} = 0, \qquad (5.2.30)$$

in which  $a_i(w)$  and  $a_{ij}(w)$ ,  $\overline{a}_{ij}(w)$  are all  $C^1$  functions with respect to their arguments on the domain under consideration, satisfying (5.1.3).

## 5.3 Local Exact Boundary Controllability

**Theorem 5.3.1** (*Two-sided control*) Under the hypotheses given in  $\S5.1$ , suppose furthermore that (5.2.13) and (5.2.17) hold. Let

$$T > L \max_{i=1,\cdots,n} \left(\frac{1}{a_i(0)}\right).$$
 (5.3.1)

For any given initial data  $(\varphi, \psi)$  and final data  $(\Phi, \Psi)$  with small norms  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$ ,  $\|(\Phi_i, \Psi_i)\|_{C^2[0,L] \times C^1[0,L]}$   $(i = 1, \dots, n)$ , there exist boundary controls  $H = (h_1, \dots, h_n)$ and  $\overline{H} = (\overline{h}_1, \dots, \overline{h}_n)$  with small norms  $\|h_i\|_{C^d[0,T]}$  and  $\|\overline{h}_i\|_{C^{\overline{d}}[0,T]}$   $(i = 1, \dots, n)$ , such that the mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6) admits a unique semi-global  $C^2$  solution w = w(t, x) with small  $C^2$  norm on the domain  $R(T) = \{(t, x)|0 \le t \le T, 0 \le x \le L\}$ , which exactly satisfies the final condition (5.2.25).

In order to prove Theorem 5.3.1, it suffices to use the constructive method suggested in [74] and [83] to prove the following lemma, the details of whose proofs are skipped here.

**Lemma 5.3.1** Under the assumptions of Theorem 5.3.1, for any given initial data  $(\varphi, \psi)$  and final data  $(\Phi, \Psi)$  with small norms $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  and  $\|(\Phi_i, \Psi_i)\|_{C^2[0,L] \times C^1[0,L]}$   $(i = 1, \dots, n)$ , the coupled system of quasilinear wave equations (5.1.2) admits a  $C^2$  solution w = w(t, x) with small  $C^2$  norm on the domain  $R(T) = \{(t, x)| 0 \le t \le T, 0 \le x \le L\}$ , which satisfies simultaneously the initial condition (5.1.6) and the final condition (5.2.25).

**Remark 5.3.1** If  $a_i, a_{ij}, b_{ij}, \overline{b}_{ij}, c_{ij}, \overline{c}_{ij}$  are all independent of w, then the problem is linear, Theorem 5.3.1 implies the corresponding global exact boundary controllability.

**Remark 5.3.2** The exact controllability time (5.3.1) given in Theorem 5.3.1 is sharp.

**Theorem 5.3.2** (One-sided control) Under the hypotheses given in  $\S5.1$ , suppose furthermore that (5.2.13), (5.2.17) and (5.2.26) hold. Let

$$T > 2L \max_{i=1,\dots,n} \left(\frac{1}{a_i(0)}\right).$$
 (5.3.2)

For any given initial data  $(\varphi, \psi)$  and final data  $(\Phi, \Psi)$  with small norms  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}, \|(\Phi_i, \Psi_i)\|_{C^2[0,L] \times C^1[0,L]}$   $(i = 1, \dots, n)$  and for any given boundary functions  $H = (h_1, \dots, h_n)$  on x = 0 with small norms  $\|h_i\|_{C^d[0,T]}$   $(i = 1, \dots, n)$ , such that the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, 0)and (T, 0) respectively, there exist boundary controls  $\overline{H} = (\overline{h}_1, \dots, \overline{h}_n)$  on x = L with small norms  $\|\overline{h}_i\|_{C^{\overline{d}}[0,T]}$   $(i = 1, \dots, n)$ , such that the mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6) admits a unique semi-global  $C^2$  solution w = w(t, x)with small  $C^2$  norm on the domain  $R(T) = \{(t, x)|0 \le t \le T, 0 \le x \le L\}$ , which exactly satisfies the final condition (5.2.25).

In order to get Theorem 5.3.2, similarly, it suffices to prove the following

**Lemma 5.3.2** Under the assumptions of Theorem 5.3.2, for any given initial data  $(\varphi, \psi)$  and final data  $(\Phi, \Psi)$  with small norms  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  and  $\|(\Phi_i, \Psi_i)\|_{C^2[0,L] \times C^1[0,L]}$   $(i = 1, \dots, n)$ , for any given boundary functions  $H = (h_1, \dots, h_n)$  on x = 0 with small norm  $\|h_i\|_{C^d[0,T]}(i = 1, \dots, n)$ , such that the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, 0) and (T, 0) respectively, the coupled system of quasilinear wave equations (5.1.2) with the boundary condition (5.1.4) on x = 0 admits a  $C^2$  solution w = w(t, x) with small  $C^2$  norm on the domain  $R(T) = \{(t, x)|0 \le t \le T, 0 \le x \le L\}$ , which satisfies simultaneously the initial condition (5.1.6) and the final condition (5.2.25).

**Remark 5.3.3** Similar results hold if we take the boundary controls H(t)  $(0 \le t \le T)$  at x = 0 and hypothesis (5.2.26) is replaced by (5.2.27).

**Remark 5.3.4** If  $a_i, a_{ij}, b_{ij}, \overline{b}_{ij}, c_{ij}, \overline{c}_{ij}$  are all independent of w, then the problem is linear, Theorem 5.3.2 implies the corresponding global exact boundary controllability.

**Remark 5.3.5** The exact controllability time (5.3.2) given in Theorem 5.3.2 is sharp.

If we only consider the corresponding null controllability (see [77]), for which the final data (5.2.25) are specially taken as

$$t = T : w = 0, w_t = 0, \ 0 \le x \le L, \tag{5.3.3}$$

then for the coupled dissipative type boundary conditions (5.1.4d) (resp. (5.1.5d)), the condition (5.2.26) (resp. (5.2.27)) is not necessary. In fact, in this situation, similar to [77], we have

**Theorem 5.3.3** (One-sided null control) Let T > 0 satisfies (5.3.2). Suppose that (5.2.13) and (5.2.17) hold. Suppose furthermore that

$$H(t) \equiv 0. \tag{5.3.4}$$

at x = 0. For any given initial data  $(\varphi, \psi)$ , where  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}(i = 1, \dots, n)$ are small enough, such that the conditions of  $C^2$  compatibility are satisfied at the point (t, x) = (0, 0), there exists boundary controls  $\overline{H} = (\overline{h}_1, \dots, \overline{h}_n)$  on x = L with small norms  $\|\overline{h}_i\|_{C^{\overline{d}}[0,T]}(i = 1, \dots, n)$ , such that the mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6) admits a unique semi-global  $C^2$  solution w = w(t, x) with small  $C^2$  norm on the domain  $R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\}$ , which exactly satisfies the final condition (5.3.3).

**Remark 5.3.6** Similar results hold if we take the boundary controls  $H(t)(0 \le t \le T)$  on x = 0, and hypothesis (5.3.4) is replaced by

$$\overline{H}(t) \equiv 0. \tag{5.3.5}$$

**Remark 5.3.7** Theorem 5.3.1, Theorem 5.3.2 and Theorem 5.3.3 are still valid for the coupled system (5.2.29) or (5.2.30), which satisfies (5.1.3).

#### 5.4 Local Exact Boundary Observability

We now consider the exact boundary observability for the mixed initial-boundary value problem (5.1.2) and (5.1.4)-(5.1.6), in which the boundary functions  $h_i(t)$  and  $\bar{h}_i(t)(i = 1, \dots, n)$  are given.

The principle of choosing the observed values on the boundary is that the observed values together with the boundary condition can uniquely determine the values  $(w, w_x)$  on the boundary. Hence, the observed values at x = 0 can be taken as

1.  $w_{ix} = k_i(t)(i = 1, \dots, n)$ , for the *Dirichlet* type boundary conditions (5.1.4a), then

$$x = 0: (w_i, w_{ix}) = (h_i(t), k_i(t));$$
 (5.4.1a)

2.  $w_i = k_i(t)(i = 1, \dots, n)$ , for the Neumann type boundary conditions (5.1.4b), then

$$x = 0: (w_i, w_{ix}) = (k_i(t), h_i(t));$$
 (5.4.1b)

3.  $w_i = k_i(t)(i = 1, \dots, n)$ , for the Coupled third type boundary conditions (5.1.4c), then

$$x = 0: \quad (w_i, w_{ix}) = (k_i(t), \sum_{j=1}^n b_{ij}k_j(t) + h_i(t)); \tag{5.4.1c}$$

4.  $w_i = k_i(t)(i = 1, \dots, n)$ , for the Coupled dissipative type boundary conditions (5.1.4d), then

$$x = 0: \quad (w_i, w_{ix}) = \left(k_i(t), \sum_{j=1}^n c_{ij}k'_j(t) + h_i(t)\right). \tag{5.4.1d}$$

where  $k_i(t)(i = 1, \dots, n) \in C^1$  for (5.1.4a) or  $\in C^2$  for (5.1.4b)-(5.1.4d). Thus, by means of the observed values at x = 0, we have

$$x = 0: (w, w_x) = (a(t), b(t)),$$
(5.4.2)

and for any given T > 0,

$$\|(a,b)\|_{C^{2}[0,T] \times C^{1}[0,T]} \leq C \Big(\sum_{i=1}^{n} \|k_{i}\|_{C^{l}[0,T]} + \sum_{i=1}^{n} \|h_{i}\|_{C^{d}[0,T]}\Big),$$
(5.4.3)

where C is a positive constant, d is given by (5.2.21) and

$$l = \begin{cases} 1 & \text{for}(5.1.4a), \\ 2 & \text{for}(5.1.4b) - (5.1.4d), \end{cases}$$
(5.4.4)

The corresponding observed values  $\bar{k}_i(t)(i = 1, \dots, n)$  at x = L can be similarly taken, then we have

$$x = 0: (w, w_x) = (\bar{a}(t), b(t)), \tag{5.4.5}$$

and for any given T > 0,

$$\|(\bar{a},\bar{b})\|_{C^{2}[0,T]\times C^{1}[0,T]} \leq C\Big(\sum_{i=1}^{n} \|\bar{k}_{i}\|_{C^{\bar{l}}[0,T]} + \sum_{i=1}^{n} \|\bar{h}_{i}\|_{C^{\bar{d}}[0,T]}\Big),$$
(5.4.6)

where C is a positive constant,  $\bar{d}$  is given by (5.2.22) and

$$\bar{l} = \begin{cases} 1 & \text{for}(5.1.5a), \\ 2 & \text{for}(5.1.5b) - (5.1.5d). \end{cases}$$
(5.4.7)

By the constructive method suggested in [49],[77],[72] and [71], we can prove following theorems.

**Theorem 5.4.1** (Two-sided observation) Under the hypotheses given in §5.1, suppose furthermore that (5.2.13) and (5.2.17) hold. Let T > 0 satisfies (5.3.1), for any given initial condition  $(\varphi, \psi)$ , such that  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  are suitably small and the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, 0) and (0, L) respectively. If we have the observed values  $k_i(t)(i = 1, \dots, n)$  at x = 0 and  $\bar{k}_i(t)(i = 1, \dots, n)$  at x = L on the interval [0, T], then the initial data  $(\varphi, \psi)$  can be uniquely determined and the following observability inequality holds:

$$\|(\varphi,\psi)\|_{C^{2}[0,L]\times C^{1}[0,L]} \leq C\Big(\sum_{i=1}^{n} \|(k_{i},\bar{k}_{i})\|_{C^{l}[0,T]\times C^{\bar{l}}[0,T]} + \sum_{i=1}^{n} \|(h_{i},\bar{h}_{i})\|_{C^{d}[0,T]\times C^{\bar{d}}[0,T]}\Big),$$
(5.4.8)

where C is a positive constant.

**Remark 5.4.1** If  $a_i, a_{ij}, b_{ij}, \overline{b}_{ij}, c_{ij}, \overline{c}_{ij}$  are all independent of w, then the problem is linear, Theorem 5.4.1 implies the corresponding global exact boundary observability.

**Remark 5.4.2** The exact observability time (5.3.1) given in Theorem 5.4.1 is sharp.

**Theorem 5.4.2** (One-sided observability) Under the hypotheses given in §5.1, suppose furthermore that (5.2.13), (5.2.17) and (5.2.27) hold. Let T > 0 satisfies (5.3.2), for any given initial condition  $(\varphi, \psi)$ , such that  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  are suitably small and the conditions of  $C^2$  compatibility are satisfied at the points (t, x) =(0,0) and (0,L) respectively. If we have the observed values  $k_i(t)(i = 1, \dots, n)$  at x = 0 on the interval [0,T], then the initial data  $(\varphi, \psi)$  can be uniquely determined and the following observability inequality holds:

$$\|(\varphi,\psi)\|_{C^{2}[0,L]\times C^{1}[0,L]} \leq C\Big(\sum_{i=1}^{n} \|k_{i}\|_{C^{l}[0,T]} + \sum_{i=1}^{n} \|(h_{i},\bar{h}_{i})\|_{C^{d}[0,T]\times C^{\bar{d}}[0,T]}\Big), \quad (5.4.9)$$

where C is a positive constant.

**Remark 5.4.3** Similar results hold if we take observed values  $\bar{k}_i(t)(i = 1, \dots, n)$ at x = L and hypothesis (5.2.27) is replaced by (5.2.26).

**Remark 5.4.4** If  $a_i, a_{ij}, b_{ij}, \overline{b}_{ij}, c_{ij}, \overline{c}_{ij}$  are all independent of w, then the problem is linear, Theorem 5.4.2 implies the corresponding global exact boundary observability.

**Remark 5.4.5** The exact observability time (5.3.2) given in Theorem 5.4.2 is sharp.

If we only consider the corresponding weak observability (see [77]), for which the final condition (5.2.25) can be uniquely determined by the observed values  $k_i(t)(i = 1, \dots, n)$  at x = 0, then for the coupled dissipative type boundary conditions (5.1.4d) (resp.(5.1.5d)), the condition (5.2.26) (resp.(5.2.27)) is not necessary. In fact, in this situation, we have

**Theorem 5.4.3** (One-sided weak observability) Under the hypotheses given in §5.1, suppose furthermore that (5.2.13), (5.2.17) hold. Let T > 0 satisfies (5.3.2), for any given initial condition  $(\varphi, \psi)$ , such that  $\|(\varphi_i, \psi_i)\|_{C^2[0,L] \times C^1[0,L]}$  are suitably small and the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, 0)and (0, L) respectively. If we have the observed values  $k_i(t)(i = 1, \dots, n)$  at x = 0on the interval [0,T], then the final data  $(\Phi, \Psi)$  can be uniquely determined by the observed values  $k_i(t)(i = 1, \dots, n)$  at x = 0 and the boundary conditions  $h_i(t)$  and  $\bar{h}_i(t)$  on the interval [0,T], in addition, the following weak observability inequality holds:

$$\|(\Phi,\Psi)\|_{C^{2}[0,L]\times C^{1}[0,L]} \leq C\Big(\sum_{i=1}^{n} \|k_{i}\|_{C^{l}[0,T]} + \sum_{i=1}^{n} \|(h_{i},\bar{h}_{i})\|_{C^{d}[0,T]\times C^{\bar{d}}[0,T]}\Big), \quad (5.4.10)$$

where C is a positive constant.

**Remark 5.4.6** Similar results hold if we take observed values  $k_i(t)(i = 1, \dots, n)$ at x = 0.

**Remark 5.4.7** Theorem 5.4.1, Theorem 5.4.2 and Theorem 5.4.3 are still valid for the coupled system (5.2.29) or (5.2.30), which satisfies (5.1.3).

# Chapter 6

# Exact Boundary Synchronization for a Coupled System of Linear Wave Equations

#### Contents

6.1	Introduction
6.2	Exact boundary synchronization
6.3	Exact boundary synchronization by groups 168
6.4	Exact boundary null controllability and synchronization by groups
6.5	Generalized exact boundary synchronization 178
6.6	Remarks

## 6.1 Introduction

Synchronization is a widespread natural phenomenon. Thousands of fireflies may twinkle at the same time; audiences in the theater can applaud with a rhythmic beat; pacemaker cells of the heart function simultaneously; and field crickets give out a unanimous cry—all these are phenomena of synchronization [104], [115].

In principle, synchronization happens when different individuals possess likeness in nature, that is, they conform essentially to the same governing equation, and meanwhile, the individuals should bear a certain coupled relation.

The phenomenon of synchronization was first observed by Huygens in 1665 [60]. The theoretical research on synchronization phenomena dates back to H. Fujisaka and T. Yamada's study of synchronization for coupled equations in 1983 [39]. The previous studies focused on systems described by ODEs, such as

$$\frac{\mathrm{d}X_i}{\mathrm{d}t} = f(X_i, t) + \sum_{j=1}^N A_{ij} X_j \qquad (i = 1, \cdots, N),$$
(6.1.1)

where  $X_i(i = 1, \dots, N)$  are *n*-dimensional vectors,  $A_{ij}(i, j = 1, \dots, N)$  are  $n \times n$ matrices, and f(X, t) is an n-dimensional vector function independent of *i*. The right-hand side of (6.1.1) shows that every  $X_i(i = 1, \dots, N)$  possesses two basic features: satisfying a fundamental governing equation and bearing a coupled relation among one another.

Our goal is to synchronize a hyperbolic system through boundary controls. Different from the ODE situation, the coupling of systems can then be fulfilled by coupling of the equations or (and) the boundary conditions, which has richer research implications, moreover, boundary controls can be used to realize our goal in a finite time.

Roughly speaking, we want to find a T > 0 and some boundary controls on [0, T], such that from the time t = T on, the system states tend to the same. That is to say, we hope to achieve the synchronization of the system states not only at the moment t = T under the action of boundary controls on [0, T], but also when  $t \ge T$  and withdrawing all the controls. This is forever, instead of short-lived, synchronization, as is desired in many actual applications.

Obviously, if the system has the exact boundary null controllability ([77], [88], [89], [100]), it must have the exact boundary synchronization, but this is a trivial situation that should be excluded beforehand. The exact boundary null controllability here also means that the system states remain null not only at the moment t = T under the action of boundary controls on [0, T], but also when  $t \ge T$  and withdrawing all the controls.

In this chapter, we consider the exact boundary synchronization for a coupled system of 1-D wave equations with various boundary condition in the framework of classical solutions. The same problem for a coupled system of *n*-dimensional wave equations with Dirichlet boundary controls in the framework of weak solutions can be found in [79]. We point out that in 1-D case, thanks to the theory on semi-global classical solutions(see [72] and [73]), there is a unified theory on the exact boundary (null) controllability for a coupled system of wave equations with various kinds of coupled boundary conditions (see [55], [72], [77], [109]), which enables us to get a complete result on the exact boundary synchronization. However, in the framework of weak solutions in n-dimensional case, for getting the corresponding exact boundary synchronization, we have to study the exact boundary (null) controllability for a coupled system of wave equations with of boundary conditions (see [35], [30]), there is a unified the corresponding exact boundary (null) controllability for a coupled system of study the exact boundary (null) controllability for a coupled system of yeak solutions in n-dimensional case, for getting the corresponding exact boundary ary synchronization, we have to study the exact boundary (null) controllability for a coupled system of wave equations with every prescribed kind of boundary conditions separately, and, up to now, only the Dirichlet type of boundary conditions

was discussed in [79]. In addition, the generalized exact boundary synchronization is introduced and discussed in this chapter.

Precisely speaking, we will consider the following coupled system of wave equations

$$\frac{\partial^2 U}{\partial t^2} - \frac{\partial^2 U}{\partial x^2} + AU = 0, \qquad (6.1.2)$$

where

$$U = (u_1, \cdots, u_N)^T \tag{6.1.3}$$

is an unknown vector function of (t, x),  $A = (a_{ij})$  is a  $N \times N$  coupling matrix with constant elements. (6.1.2) can be written as

$$\frac{\partial^2 u_i}{\partial t^2} - \frac{\partial^2 u_i}{\partial x^2} + \sum_{j=1}^N a_{ij} u_j = 0 \quad (i = 1, \cdots, N).$$
(6.1.4)

At the end x = 0, we prescribe anyone of the following boundary conditions:

$$x = 0$$
:  $U = H(t)$  (Dirichlet type), (6.1.5a)

$$x = 0: U_x = H(t)$$
 (Neumann type), (6.1.5b)

$$x = 0: U_x - BU = H(t)$$
 (Coupled third type), (6.1.5c)

where  $B = (b_{ij})$  is an  $N \times N$  coupling matrix with constant elements, and

$$H(t) = (h_1(t), \cdots, h_N(t))^T,$$
 (6.1.6)

the components of which will be totally or partially taken as boundary controls.

Similarly, at the end x = L, since no boundary controls are concerned for onesided control, we prescribe anyone of the following homogeneous boundary conditions:

$$x = L: U = 0,$$
 (6.1.7a)

$$x = L: U_x = 0,$$
 (6.1.7b)

$$x = L: \ U_x + \overline{B}U = 0, \tag{6.1.7c}$$

where  $\overline{B} = (\overline{b}_{ij})$  is an  $N \times N$  coupling matrix with constant elements.

The initial condition is given by

$$t = 0: U = \Phi(x), \quad U_t = \Psi(x),$$
 (6.1.8)

where  $\Phi(x) = (\varphi_1(x), \cdots, \varphi_N(x))^T$  and  $\Psi(x) = (\psi_1(x), \cdots, \psi_N(x))^T$ .

For the forward mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) of the coupled system of wave equations, by means of its reduction to the mixed initial-boundary value problem of a corresponding first order hyperbolic system and using the theory of global  $C^1$  solutions to first order linear hyperbolic systems, we have (see [55], [72])

**Lemma 6.1.1** For any given T > 0, assume that  $\Phi \in (C^2[0, L])^N$ ,  $\Psi \in (C^1[0, L])^N$ and  $H \in (C^2[0, T])^N$  (case (6.1.5a)) or  $(C^1[0, T])^N$ (cases (6.1.5b)-(6.1.5c)) satisfy the conditions of  $C^2$  compatibility at the points (t, x) = (0, 0) and (0, L), respectively (see Remark 6.1.2). For any given coupling matrices A, B and  $\overline{B}$ , the forward mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) admits a unique  $C^2$  solution U = U(t, x) on the domain  $R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\}$ .

**Remark 6.1.1** In what follows, (6.1.5) always means anyone of (6.1.5a), (6.1.5b) and (6.1.5c), and it is similar for (6.1.7).

**Remark 6.1.2** The conditions of  $C^2$  compatibility at the point (t, x) = (0, L) can be precisely written as

$$\begin{cases} \Phi(L) = 0, \\ \Psi(L) = 0, & for \ (6.1.7a); \\ \Phi''(L) = 0, \end{cases}$$
(6.1.9a)

$$\begin{cases} \Phi'(L) = 0, \\ \Psi'(L) = 0, \end{cases} \quad for \quad (6.1.7b); \tag{6.1.9b}$$

or

$$\begin{cases} \Phi'(L) + \overline{B}\Phi(L) = 0, \\ \Psi'(L) + \overline{B}\Psi(L) = 0, \end{cases} \quad for \quad (6.1.7c). \tag{6.1.9c}$$

The conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) can be similarly obtained.

**Remark 6.1.3** By Lemma 6.1.1, in order to guarantee the well-posedness of the forward mixed problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8), no assumptions on the coupling matrices A, B and  $\overline{B}$  are needed.

**Remark 6.1.4** For the backward mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) with the final condition

$$t = T : U = \overline{\Phi}(x), \quad U_t = \overline{\Psi}(x).$$
 (6.1.10)

where  $\overline{\Phi}(x) = (\overline{\varphi}_1(x), \cdots, \overline{\varphi}_N(x))^T$  and  $\overline{\Psi}(x) = (\overline{\psi}_1(x), \cdots, \overline{\psi}_N(x))^T$ , the conclusion of Lemma 6.1.1 is still valid.

By Lemma 6.1.1 and using a constructive method, we have the following result on the exact boundary null controllability (also see [55], [109], [118]).

Lemma 6.1.2 Let

$$T > 2L.$$
 (6.1.11)

For any given initial state  $(\Phi, \Psi) \in (C^2[0, L])^N \times (C^1[0, L])^N$ , satisfying the conditions of  $C^2$  compatibility at the point (t,x) = (0,L), there exist boundary controls  $H \in (C^2[0,T])^N$  (case (6.1.5a)) or  $(C^1[0,T])^N$  (cases (6.1.5b)–(6.1.5c)) with support on [0,T] at the end x = 0, such that the corresponding mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) admits a unique  $C^2$  solution U = U(t, x)on  $t \geq 0$ , and as  $t \geq T$  we have

$$U(t,x) \equiv 0, \quad 0 \le x \le L.$$
 (6.1.12)

In what follows, we will use Lemma 6.1.1 and Lemma 6.1.2 to consider the exact boundary synchronization, the exact boundary synchronization by groups, and the exact boundary null controllability and synchronization by groups in  $\S6.2$ ,  $\S6.3$  and  $\S6.4$ , respectively, and the generalized exact boundary synchronization in  $\S6.5$ , for the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7). Moreover, several remarks are given in  $\S6.6$ .

#### 6.2 Exact boundary synchronization

**Definition 6.2.1** If there exists T > 0 such that for any given initial state  $(\Phi(x), \Psi(x)),$  we can find some boundary controls with support on [0, T] in H(t), such that the corresponding mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \cdots, u_N(t, x))^T$  on  $t \geq 0$ , and as  $t \geq T$  we have

$$u_1(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L,$$
(6.2.1)

then the coupled system (6.1.2), (6.1.5) and (6.1.7) is said to possess the exact boundary synchronization or is said to be exactly synchronizable, and  $\tilde{u} =$  $\widetilde{u}(t,x)$  is called the corresponding synchronizable state.

Obviously, if the system possesses the exact boundary null controllability, then it must be exactly synchronizable. However, this trivial situation should be excluded in the discussion.

**Lemma 6.2.1** If the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7) is exactly synchronizable, but not exactly null controllable, then the coupling matrices A, B and  $\overline{B}$  should satisfy the following necessary conditions: The values of the sum of every row of these matrices,  $\sum_{j=1}^{N} a_{ij}$ ,  $\sum_{j=1}^{N} b_{ij}$  and  $\sum_{j=1}^{N} \overline{b}_{ij}$  are independent of  $i = 1, \dots, N$ , namely,

$$\sum_{j=1}^{N} a_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}} \quad (i = 1, \cdots, N), \tag{6.2.2}$$

$$\sum_{j=1}^{N} b_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{b}} \quad (i = 1, \cdots, N), \tag{6.2.3}$$

$$\sum_{j=1}^{N} \bar{b}_{ij} \stackrel{\text{def.}}{=} \widetilde{\tilde{b}} \quad (i = 1, \cdots, N), \tag{6.2.4}$$

where  $\tilde{\tilde{a}}$ ,  $\tilde{\tilde{b}}$  and  $\tilde{\tilde{b}}$  are constants independent of  $i = 1, \dots, N$ .

**Proof.** By synchronization, there exists T > 0 and a synchronizable state  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t, x)$  such that (6.2.1) holds. Then, it follows from (6.1.4) that for  $t \geq T$  we have

$$\frac{\partial^2 \widetilde{\widetilde{u}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}}{\partial x^2} + \left(\sum_{j=1}^N a_{ij}\right) \widetilde{\widetilde{u}} = 0, \quad 0 \le x \le L, \ i = 1, \cdots, N.$$
(6.2.5)

In particular, for  $t \ge T$  we get

$$\left(\sum_{j=1}^{N} a_{ij}\right)\widetilde{\widetilde{u}} = \left(\sum_{j=1}^{N} a_{kj}\right)\widetilde{\widetilde{u}}, \quad 0 \le x \le L, \ i, k = 1, \cdots, N.$$

By the non exact null controllability, there exists at least an initial data  $(\Phi, \Psi)$  for which the corresponding solution U, or equivalently  $\tilde{\tilde{u}}$ , does not identically vanish for  $t \geq T$  whatever boundary controls H are chosen. This leads to (6.2.2). In the meantime, noting (6.2.5), the synchronizable state  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t, x)$  satisfies the following wave equation

$$\frac{\partial^2 \widetilde{\widetilde{u}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}}{\partial x^2} + \widetilde{\widetilde{a}} \widetilde{\widetilde{u}} = 0, \qquad (6.2.6)$$

where  $\widetilde{\widetilde{a}}$  is given by (6.2.2).

We now prove (6.2.3). (6.2.4) can be similarly obtained.

Similarly to (6.2.5), it follows from (6.1.5c) that for  $t \ge T$  we have

$$x = 0: \ \frac{\partial \widetilde{\widetilde{u}}}{\partial x} - \left(\sum_{j=1}^{N} b_{ij}\right) \widetilde{\widetilde{u}} = 0, \quad i = 1, \cdots, N,$$
(6.2.7)

then for  $t \geq T$  we get

$$x = 0: \left(\sum_{j=1}^{N} b_{ij}\right) \widetilde{\widetilde{u}} = \left(\sum_{j=1}^{N} b_{kj}\right) \widetilde{\widetilde{u}}, \quad i, k = 1, \cdots, N.$$

Since the system is not exactly null controllable, we claim that at least for an initial data  $(\Phi, \Psi)$ ,  $\tilde{\tilde{u}}$  does not identically vanish at x = 0 for  $t \ge T$ , then (6.2.3) holds. Otherwise, noting (6.2.7) we have

$$x = 0: \ \widetilde{\widetilde{u}} \equiv \frac{\partial \widetilde{\widetilde{u}}}{\partial x} \equiv 0, \quad t \ge T,$$

then, by the exact boundary observability ([71], [72]) for the wave equation (6.2.6),  $\tilde{\tilde{u}}$  should identically vanish for  $t \geq T$  and  $0 \leq x \leq L$ , which gives a contradiction to the non exact null controllability.

This completes the proof of Lemma 6.2.1.  $\Box$ 

**Remark 6.2.1** Conditions (6.2.2)–(6.2.4) mean that all the matrices A, B and  $\overline{B}$  have a common right eigenvector  $(1, 1, \dots, 1)^T$  with the corresponding eigenvalues  $\widetilde{\widetilde{a}}, \ \widetilde{\widetilde{b}} \ and \ \widetilde{\widetilde{b}}, \ respectively.$ 

**Theorem 6.2.1** Suppose that (6.2.2)-(6.2.4) hold and

$$T > 2L. \tag{6.2.8}$$

For any given initial state  $(\Phi, \Psi) \in (C^2[0, L])^N \times (C^1[0, L])^N$ , satisfying the conditions of  $C^2$  compatibility at the point (t, x) = (0, L), there exist  $(N - 1) C^2(case(6.1.5a))$ or  $C^1$  (cases (6.1.5b)-(6.1.5c)) boundary controls with support on [0, T] in H(t) (for example, take  $h_2(t), \dots, h_N(t)$  with  $h_1(t) \equiv 0$ ), such that the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the exact boundary synchronization.

**Proof** Let

$$w_i = u_i - u_{i+1}$$
  $(i = 1, \cdots, N-1).$  (6.2.9)

It is easily shown (see [79]) that by (6.2.2), the original system (6.1.2) for the variable U can be reduced to a self-closing system of the same kind for the variable W =

 $(w_1, \cdots, w_{N-1})^T$  as follows:

$$\frac{\partial^2 W}{\partial t^2} - \frac{\partial^2 W}{\partial x^2} + \widetilde{A}W = 0, \qquad (6.2.10)$$

where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-1) \times (N-1)$  matrix with the entries

$$\widetilde{a}_{ij} = -\sum_{p=j+1}^{N} (a_{ip} - a_{i+1,p}) = -\sum_{p=1}^{j} (a_{i+1,p} - a_{ip}), \quad i, j = 1, \cdots, N-1.$$
(6.2.11)

Similarly, by (6.2.3)–(6.2.4), the original boundary conditions (6.1.5) and (6.1.7) for the variable U can be reduced, respectively, to self-closing boundary conditions of the same kind for the variable W as follows:

$$x = 0: \quad W = \widetilde{H}(t), \tag{6.2.12a}$$

$$x = 0: \quad W_x = \tilde{H}(t), \tag{6.2.12b}$$

$$x = 0: W_x - BW = H(t)$$
 (6.2.12c)

and

$$x = L: W = 0,$$
 (6.2.13a)

$$x = L: W_x = 0, \tag{6.2.13b}$$

$$x = L: W_x + \overline{B}W = 0, \qquad (6.2.13c)$$

where  $\widetilde{B} = (\widetilde{b}_{ij})$  and  $\widetilde{\overline{B}} = (\widetilde{\overline{b}}_{ij})$  are  $(N-1) \times (N-1)$  coupling matrices reduced by  $B = (b_{ij})$  and  $\overline{B} = (\overline{b}_{ij})$  in the following way:

$$\widetilde{b}_{ij} = -\sum_{p=j+1}^{N} (b_{ip} - b_{i+1,p}) = -\sum_{p=1}^{j} (b_{i+1,p} - b_{ip}), \quad i, j = 1, \cdots, N-1$$
(6.2.14)

etc. Moreover,

$$\widetilde{H}(t) = (\widetilde{h}_1(t), \cdots, \widetilde{h}_{N-1}(t))^T$$
(6.2.15)

with

$$\widetilde{h}_i(t) = h_i(t) - h_{i+1}(t), \quad i = 1, \cdots, N - 1.$$
(6.2.16)

The initial condition of W is given by

$$t = 0: W = \widetilde{\Phi}(x), \quad W_t = \widetilde{\Psi}(x)$$
 (6.2.17)

Chapter 6: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations 167

with

$$\begin{cases} \widetilde{\Phi}(x) = (\varphi_1(x) - \varphi_2(x), \varphi_2(x) - \varphi_3(x), \cdots, \varphi_{N-1}(x) - \varphi_N(x))^T, \\ \widetilde{\Psi}(x) = (\psi_1(x) - \psi_2(x), \psi_2(x) - \psi_3(x), \cdots, \psi_{N-1}(x) - \psi_N(x))^T. \end{cases}$$
(6.2.18)

Thus, according to Lemma 6.1.2, by means of  $(N-1) C^2$  (case(6.2.12a)) or  $C^1$ (cases (6.2.12b)-(6.2.12c)) boundary controls  $\widetilde{H}(t)$  (for instance, take  $h_2(t), \dots, h_n(t)$ with  $h_1(t) \equiv 0$  at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly synchronizable. This finished the proof.

Remark 6.2.2 Noting Remark 6.1.3, by Lemma 6.1.1 and Lemma 6.1.2, the reduced mixed initial-boundary value problem (6.2.10), (6.2.12)-(6.2.13) and (6.2.17)for the variable W is always well-posed and exactly null controllable.

**Remark 6.2.3** As  $t \ge T$ , the synchronizable state  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  defined by (6.2.1) satisfies the wave equation (6.2.6) with the following boundary conditions:

$$x = 0: \quad \widetilde{\widetilde{u}} = 0, \tag{6.2.19a}$$

$$x = 0: \quad \widetilde{\widetilde{u}}_x = 0, \tag{6.2.19b}$$

$$x = 0: \quad \widetilde{\widetilde{u}}_x - \widetilde{\widetilde{b}}\widetilde{\widetilde{u}} = 0 \tag{6.2.19c}$$

and

$$x = L: \quad \widetilde{\widetilde{u}} = 0, \tag{6.2.20a}$$

$$x = L: \quad \widetilde{\widetilde{u}}_x = 0, \tag{6.2.20b}$$

$$x = L: \quad \widetilde{\widetilde{u}}_x + \widetilde{\widetilde{b}}\widetilde{\widetilde{u}} = 0, \qquad (6.2.20c)$$

where  $\tilde{\tilde{a}}, \tilde{\tilde{b}}$  and  $\tilde{\tilde{b}}$  are given by (6.2.2)–(6.2.4), respectively.

Noting Remark 6.1.3, the forward mixed initial-boundary value problem for (6.2.6) and (6.2.19)-(6.2.20) is always well-posed. Hence, if we know the initial state of  $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t, x)$  at the moment t = T:

$$t = T$$
:  $\widetilde{\widetilde{u}} = \varphi(x), \ \widetilde{\widetilde{u}}_t = \psi(x), \quad 0 \le x \le L,$  (6.2.21)

we can completely determine the evolution of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  with respect to t.

**Remark 6.2.4** Under assumptions (6.2.2)-(6.2.4), in order to realize the exact boundary synchronization, the (N-1) boundary controls in H(t) can be chosen in infinitely many ways, therefore, the initial state  $(\varphi, \psi)$  in (6.2.21) will be not unique.

Moreover, it can be shown that the attainable set of all possible initial data of synchronizable state is the whole space  $C^2[0, L] \times C^1[0, L]$  associated with the conditions of  $C^2$  compatibility at (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (6.2.19) and (6.2.20).

To illustrate this, for any given  $(\varphi, \psi) \in C^2[0, L] \times [C^1[0, L]$  satisfying the conditions of  $C^2$  compatibility with the boundary conditions (6.2.19)-(6.2.20) at (t, x) = (T, 0) and (T, L) respectively, we can solve the backward mixed initial-boundary value problem (6.2.6) and (6.2.19)-(6.2.21) to get a unique  $C^2$  solution u = u(t, x) and its initial value at t = 0

$$t = 0: \quad u = \widehat{\varphi}(x), \quad u_t = \psi(x). \tag{6.2.22}$$

Then, by assumptions (6.2.2)–(6.2.4), it is easy to see that

$$U(t,x) = (u, u, \cdots, u)^{T}(t,x)$$
(6.2.23)

is the  $C^2$  solution to the original problem (6.1.2), (6.1.5) and (6.1.7) with the null control  $H \equiv 0$  and the initial condition

$$t = 0: U = \widehat{\Phi}(x) = (\widehat{\varphi}(x), \widehat{\varphi}(x), \cdots, \widehat{\varphi}(x))^{T}, U_{t} = \widehat{\Psi}(x) = (\widehat{\psi}(x), \widehat{\psi}(x), \cdots, \widehat{\psi}(x))^{T}.$$
(6.2.24)

Then, by solving the mixed problem (6.1.2), (6.1.5), (6.1.7) and (6.2.24) with null boundary controls, we can reach any given synchronizable state  $(\varphi, \psi)$  at the moment t = T.

#### 6.3 Exact boundary synchronization by groups

The concept of exact boundary synchronization can be generalized to the exact boundary synchronization by groups. Without loss of generality, we consider here the exact boundary synchronization by 2-groups. For this purpose, setting  $U = \binom{U^{(1)}}{U^{(2)}}$ with  $U^{(1)} = (u_1, \dots, u_m)^T$  and  $U^{(2)} = (u_{m+1}, \dots, u_N)^T$ , we look for some boundary controls in H(t), such that  $U^{(1)}$  and  $U^{(2)}$  are independently synchronizable.

**Definition 6.3.1** If there exists T > 0 such that for any given initial state  $(\Phi(x), \Psi(x))$ , we can find some boundary controls with support on [0, T] in H(t), such that the corresponding mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  on  $t \ge 0$ , and for  $t \ge T$  we have

$$u_1(t,x) \equiv \dots \equiv u_m(t,x) \stackrel{\text{def.}}{=} \widetilde{u}_1(t,x), \quad 0 \le x \le L, \tag{6.3.1}$$

$$u_{m+1}(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{u}_2(t,x), \quad 0 \le x \le L, \tag{6.3.2}$$

1 6 ~

Chapter 6: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations 169

then the coupled system (6.1.2), (6.1.5) and (6.1.7) is said to possess the exact boundary synchronization by 2-groups or is said to be exactly synchronizable by 2-groups, and  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t,x) = \begin{pmatrix} \widetilde{\widetilde{u}}_1(t,x) \\ \widetilde{\widetilde{u}}_2(t,x) \end{pmatrix}$  is called the corresponding synchronizable state by 2-groups.

Obviously, if the coupled system possesses the exact boundary synchronization, then it must be exactly synchronizable by 2-groups. However, this trivial situation should be excluded in advance. For this purpose we need to impose suitable assumptions on the synchronizable states by 2-groups  $\tilde{\widetilde{u}}_1(t,x)$  and  $\tilde{\widetilde{u}}_2(t,x)$ . We have

**Lemma 6.3.1** Suppose that the coupled system (6.1.2), (6.1.5) and (6.1.7) is exactly synchronizable by 2-groups. Suppose furthermore that at least for an initial state  $(\Phi(x), \Psi(x))$  the synchronizable states by 2-groups  $\tilde{u}_1(t, x)$  and  $\tilde{u}_2(t, x)$  are linearly independent. Then the coupling matrix  $A = (a_{ij})$  in system (6.1.2) of wave equations should satisfy the following necessary conditions:

$$\sum_{j=1}^{m} a_{ij} \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\widetilde{a}}_{11} & (i=1,\cdots,m), \\ \widetilde{\widetilde{a}}_{21} & (i=m+1,\cdots,N), \end{cases}$$
(6.3.3)

$$\sum_{j=m+1}^{N} a_{ij} \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\widetilde{a}}_{12} & (i=1,\cdots,m), \\ \widetilde{\widetilde{a}}_{22} & (i=m+1,\cdots,N), \end{cases}$$
(6.3.4)

where  $\tilde{\tilde{a}}_{11}$  and  $\tilde{\tilde{a}}_{12}$  are independent of  $i = 1, \cdots, m$ , while,  $\tilde{\tilde{a}}_{21}$  and  $\tilde{\tilde{a}}_{22}$  are independent of  $i = m + 1, \cdots, N$ .

**Proof.** By Definition 6.3.1, for  $t \ge T$  we have

$$\frac{\partial^2 \widetilde{\widetilde{u}}_1}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}_1}{\partial x^2} + \left(\sum_{p=1}^m a_{ip}\right) \widetilde{\widetilde{u}}_1 + \left(\sum_{p=m+1}^N a_{ip}\right) \widetilde{\widetilde{u}}_2 = 0, \quad 0 \le x \le L, \ i = 1, \cdots, m,$$
(6.3.5)

$$\frac{\partial^2 \widetilde{\widetilde{u}}_2}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}_2}{\partial x^2} + \left(\sum_{p=1}^m a_{ip}\right) \widetilde{\widetilde{u}}_1 + \left(\sum_{p=m+1}^N a_{ip}\right) \widetilde{\widetilde{u}}_2 = 0, \quad 0 \le x \le L, \ i = m+1, \cdots, N.$$
(6.3.6)

Hence we get

$$t \ge T: \ \left(\sum_{p=1}^{m} a_{ip} - \sum_{p=1}^{m} a_{kp}\right) \widetilde{\widetilde{u}}_{1} + \left(\sum_{p=m+1}^{N} a_{ip} - \sum_{p=m+1}^{N} a_{kp}\right) \widetilde{\widetilde{u}}_{2} = 0, \quad 0 \le x \le L$$

for  $i, k = 1, \dots, m$  and  $i, k = m + 1, \dots, N$ , respectively. Since at least for an initial data  $(\Phi(x), \Psi(x))$ , the synchronizable states by 2-groups,  $\tilde{u}_1$  and  $\tilde{u}_2$  are linearly

independent, we get (6.3.3)-(6.3.4) immediately.

For the coupling matrices B and  $\overline{B}$ , we can not get the corresponding necessary conditions under the assumptions of Lemma 6.3.1 (unless we assume that at least for an initial data  $(\Phi(x), \Psi(x))$ ,  $\tilde{\tilde{u}}_1(t, x)$  and  $\tilde{\tilde{u}}_2(t, x)$  are linearly independent at x = 0and at x = L, respectively). However, based on Lemma 6.3.1, in what follows we impose the following similar assumptions:

$$\sum_{j=1}^{m} b_{ij}(\text{resp. } \bar{b}_{ij}) \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\tilde{b}}_{11}(\text{resp. } \widetilde{\tilde{b}}_{11}) & (i = 1, \cdots, m), \\ \widetilde{\tilde{b}}_{21}(\text{resp. } \widetilde{\tilde{b}}_{21}) & (i = m + 1, \cdots, N), \end{cases}$$
(6.3.7)

$$\sum_{j=m+1}^{N} b_{ij}(\text{resp. } \bar{b}_{ij}) \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\tilde{b}}_{12}(\text{resp. } \widetilde{\tilde{b}}_{12}) & (i=1,\cdots,m), \\ \widetilde{\tilde{b}}_{22}(\text{resp. } \widetilde{\tilde{b}}_{22}) & (i=m+1,\cdots,N), \end{cases}$$
(6.3.8)

where  $\tilde{\tilde{b}}_{11}$  (resp.  $\tilde{\tilde{b}}_{11}$ ) and  $\tilde{\tilde{b}}_{12}$  (resp.  $\tilde{\tilde{b}}_{12}$ ) are independent of  $i = 1, \dots, m$ , while,  $\tilde{\tilde{b}}_{21}$  (resp.  $\tilde{\tilde{b}}_{21}$ ) and  $\tilde{\tilde{b}}_{22}$  (resp.  $\tilde{\tilde{b}}_{22}$ ) are independent of  $i = m + 1, \dots, N$ .

**Theorem 6.3.1** Suppose that (6.3.3)-(6.3.4) and (6.3.7)-(6.3.8) hold and T > 0satisfies (6.2.8). For any given initial state  $(\Phi(x), \Psi(x))$ , there exist  $(N-2) C^2$ (case(6.1.5a)) or  $C^1$  (cases (6.1.5b)-(6.1.5c)) boundary controls with support on [0,T]in H(t) (for instance,  $h_2(t), \dots, h_m(t), h_{m+2}(t), \dots, h_N(t)$  with  $h_1(t) \equiv h_{m+1}(t) \equiv 0$ ), such that the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the exact boundary synchronization by 2-groups.

**Proof** Let

$$\begin{cases} w_i = u_i - u_{i+1} \quad (i = 1, \cdots, m - 1), \\ w_i = u_{i+1} - u_{i+2} \quad (i = m, \cdots, N - 2). \end{cases}$$
(6.3.9)

Similarly to the proof of Theorem 6.2.1 (cf. [79]), under assumptions (6.3.3)–(6.3.4), the original system (6.1.2) for the variable U can be reduced to a self-closing system (6.2.10) of the same kind for the variable  $W = (w_1, \dots, w_{N-2})^T$ , where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-2) \times (N-2)$  matrix with the entries

$$\widetilde{a}_{ij} = \begin{cases} -\sum_{p=j+1}^{m} (a_{ip} - a_{i+1,p}) = -\sum_{p=1}^{j} (a_{i+1,p} - a_{ip}), & i = 1, \cdots, N-2; \ j = 1, \cdots, m-1, \\ -\sum_{p=j+2}^{N} (a_{ip} - a_{i+1,p}) = -\sum_{p=m+1}^{j+1} (a_{i+1,p} - a_{ip}), & i = 1, \cdots, N-2; \ j = m, \cdots, N-2 \end{cases}$$

$$(6.3.10)$$

Similarly, under assumptions (6.3.7)–(6.3.8), the original boundary conditions (6.1.5) and (6.1.7) for the variable U can be reduced to self-closing boundary condi-

Chapter 6: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations 171

tions (6.2.12) and (6.2.13) of the same kind for the variable W, in which the coupling matrices  $\widetilde{B} = (\widetilde{b}_{ij})$  and  $\overline{B} = (\overline{b}_{ij})$  are  $(N-2) \times (N-2)$  matrices reduced by  $B = (b_{ij})$ and  $\overline{B} = (\overline{b}_{ij})$  in the following way:

$$\widetilde{b}_{ij} = \begin{cases} -\sum_{p=j+1}^{m} (b_{ip} - b_{i+1,p}) = -\sum_{p=1}^{j} (b_{i+1,p} - b_{ip}), & i = 1, \cdots, N-2; \ j = 1, \cdots, m-1, \\ -\sum_{p=j+2}^{N} (b_{ip} - b_{i+1,p}) = -\sum_{p=m+1}^{j+1} (b_{i+1,p} - b_{ip}), & i = 1, \cdots, N-2; \ j = m, \cdots, N-2 \end{cases}$$
(6.3.11)

etc. Moreover,

$$\widetilde{H}(t) = (\widetilde{h}_1(t), \cdots, \widetilde{h}_{N-2}(t))$$
(6.3.12)

with

$$\widetilde{h}_{i}(t) = \begin{cases} h_{i}(t) - h_{i+1}(t), & i = 1, \cdots, m - 1, \\ h_{i+1}(t) - h_{i+2}(t), & i = m, \cdots, N - 2. \end{cases}$$
(6.3.13)

The initial condition (6.2.17) of W is given by

$$\widetilde{\Phi}(x) = (\widetilde{\varphi}_1(x), \cdots, \widetilde{\varphi}_{N-2}(x))^T, \quad \widetilde{\Psi}(x) = (\widetilde{\psi}_1(x), \cdots, \widetilde{\psi}_{N-2}(x))^T$$
(6.3.14)

with

$$\widetilde{\varphi}_{i}(x) = \begin{cases} \varphi_{i}(x) - \varphi_{i+1}(x), & i = 1, \cdots, m-1, \\ \varphi_{i+1}(x) - \varphi_{i+2}(x), & i = m, \cdots, N-2, \end{cases}$$

$$\widetilde{\Psi}_{i}(x) = \begin{cases} \psi_{i}(x) - \psi_{i+1}(x), & i = 1, \cdots, m-1, \\ \psi_{i+1}(x) - \psi_{i+2}(x), & i = m, \cdots, N-2. \end{cases}$$
(6.3.15)

Thus, according to Lemma 6.1.2, by means of (N-2)  $C^2$  (case(6.2.12a)) or  $C^1$ (cases (6.2.12b)-(6.2.12c)) boundary controls H(t) (for instance, take  $h_2(t), \dots, h_m(t)$ ,  $h_{m+2}(t), \cdots, h_N(t)$  with  $h_1(t) \equiv h_{m+1}(t) \equiv 0$  at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly synchronizable by 2-groups. This completes the proof.

Remark 6.3.1 Lemma 6.1.1 and Lemma 6.1.2 are always available for the reduced mixed initial-boundary value problem (6.2.10), (6.2.12)–(6.2.13) and (6.3.14)for the variable W.

**Remark 6.3.2** As  $t \ge T$ , the synchronizable state by 2-groups  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t,x) =$  $\begin{pmatrix} \tilde{\tilde{u}}_1(t,x) \\ \tilde{\tilde{u}}_2(t,x) \end{pmatrix}$  satisfies the following coupled system of wave equations

$$\frac{\partial^2 \widetilde{\widetilde{U}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{U}}}{\partial x^2} + \widetilde{\widetilde{A}} \widetilde{\widetilde{U}} = 0, \qquad (6.3.16)$$

where

$$\widetilde{\widetilde{A}} = \begin{pmatrix} \widetilde{\widetilde{a}}_{11} & \widetilde{\widetilde{a}}_{12} \\ \widetilde{\widetilde{a}}_{21} & \widetilde{\widetilde{a}}_{22} \end{pmatrix},$$
(6.3.17)

and  $\tilde{\tilde{a}}_{11}$ ,  $\tilde{\tilde{a}}_{12}$ ,  $\tilde{\tilde{a}}_{21}$  and  $\tilde{\tilde{a}}_{21}$  are given by (6.3.3)–(6.3.4). Moreover,  $\tilde{\tilde{U}} = \tilde{\tilde{U}}(t,x)$  satisfies the following boundary conditions:

$$x = 0: \quad \widetilde{\widetilde{U}} = 0, \tag{6.3.18a}$$

$$x = 0: \quad \widetilde{U}_x = 0, \tag{6.3.18b}$$

$$x = 0: \quad \widetilde{\widetilde{U}}_x - \widetilde{\widetilde{B}}\widetilde{\widetilde{U}} = 0 \tag{6.3.18c}$$

and

$$x = 0: \quad \widetilde{\widetilde{U}} = 0, \tag{6.3.19a}$$

$$x = 0: \ \widetilde{U}_x = 0,$$
 (6.3.19b)

$$x = 0: \quad \widetilde{\widetilde{U}}_x + \frac{\widetilde{\widetilde{B}}}{\widetilde{\widetilde{D}}} = 0, \quad (6.3.19c)$$

where

$$\widetilde{\widetilde{B}} = \begin{pmatrix} \widetilde{\widetilde{b}}_{11} & \widetilde{\widetilde{b}}_{12} \\ \widetilde{\widetilde{b}}_{21} & \widetilde{\widetilde{b}}_{22} \end{pmatrix}, \quad \widetilde{\overline{B}} = \begin{pmatrix} \widetilde{\widetilde{b}}_{11} & \widetilde{\widetilde{b}}_{12} \\ \widetilde{\widetilde{b}}_{21} & \widetilde{\widetilde{b}}_{22} \end{pmatrix}, \quad (6.3.20)$$

the elements of which are provided by (6.3.7)-(6.3.8).

Hence, by Lemma 6.1.1, if we know the initial state of  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t,x)$  at the moment t = T:

$$t = T: \quad \widetilde{\widetilde{U}} = (\varphi^{(1)}(x), \varphi^{(2)}(x))^T, \quad \widetilde{\widetilde{U}}_t = (\psi^{(1)}(x), \psi^{(2)}(x))^T, \quad 0 \le x \le L, \quad (6.3.21)$$

then the evolution of  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t,x)$  with respect to t can be completely determined.

**Remark 6.3.3** Under assumptions (6.3.3)-(6.3.4) and (6.3.7)-(6.3.8), in order to realize the exact boundary synchronization by 2-groups, one has infinitively many ways to choose (N-2) boundary controls at the end x = 0. Moreover, as in Remark 6.2.4, any given state  $(\varphi^{(1)}, \varphi^{(2)})^T$  and  $(\psi^{(1)}, \psi^{(2)})^T$  in  $(C^2[0, L])^2 \times (C^1[0, L])^2$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (6.3.18) and (6.3.19), belongs to the attainable set of the initial data of synchronizable state by 2-groups.

Thus, at least for some initial states  $(\Phi(x), \Psi(x))$ , the synchronizable states by

2-groups  $\tilde{\widetilde{u}}_1(t,x)$  and  $\tilde{\widetilde{u}}_2(t,x)$  are linearly independent on t = T then for  $t \geq T$ . It shows that the corresponding requirement on  $\tilde{\widetilde{u}}_1$  and  $\tilde{\widetilde{u}}_2$ , given in Lemma 6.3.1, is reasonable.

**Remark 6.3.4** In the special case that m = N - 1,  $U^{(2)}(t, x)$  is composed of only one component  $u_N(t, x)$ , then assumptions (6.3.4) and (6.3.8) reduce to

$$a_{iN} = \widetilde{\widetilde{a}}_{12} \quad (i = 1, \cdots, N - 1),$$
 (6.3.22)

and

$$b_{iN} \text{ (resp. } \overline{b}_{iN}) = \widetilde{\widetilde{b}}_{12} \text{ (resp. } \widetilde{\widetilde{b}}_{12}) \quad (i = 1, \cdots, N-1),$$
 (6.3.23)

where  $\tilde{\tilde{a}}_{12}$  and  $\tilde{\tilde{b}}_{12}$  (resp.  $\tilde{\tilde{b}}_{12}$ ) are independent of  $i = 1, \dots, N-1$ , namely, the first (N-1) components of the last column of the coupling matrices A and B (resp.  $\overline{B}$ ) are the same, respectively.

**Remark 6.3.5** The exact boundary synchronization by k-groups can be treated in a similar way.

In order to realize the exact boundary synchronization by 2-groups, the number of boundary controls is equal to (N-2), while, in order to realize the exact boundary synchronization by k-groups, the number of boundary controls is equal to (N-k).

## 6.4 Exact boundary null controllability and synchronization by groups

For the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7), by Lemma 6.1.2, we can use N boundary controls to realize the exact boundary null controllability. When the number of boundary controls is less than N, however, generically speaking, it is impossible to realize the same requirement. A natural question is whether we can get the exact boundary null controllability for a part of state variables when the number of boundary controls is less than N. For instance, is it possible or not to realize the exact boundary null controllability for (N - 1) state variables by means of only (N - 1) boundary controls? Since all the state variables are coupled each other, generally speaking, this requirement is impossible to be realized. However, following the idea given in §6.3, under certain assumptions on the coupled system, if we divide the state variables into two groups  $U^{(1)} = (u_1, \dots, u_m)^T$  and  $U^{(2)} = (u_{m+1}, \dots, u_N)^T$  such that  $U^{(1)}$  is exactly null controllable, while  $U^{(2)}$  is exactly synchronizable, then we can realize the previous requirement in some sense. It leads to the following

**Definition 6.4.1** If there exists T > 0 such that for any given initial state  $(\Phi(x), \Psi(x))$ , we can find some boundary controls with support on [0, T] in H(t), such that the corresponding mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  on  $t \ge 0$ , and as  $t \ge T$  we have

$$u_1(t,x) \equiv \dots \equiv u_m(t,x) \equiv 0, \quad 0 \le x \le L, \tag{6.4.1}$$

$$u_{m+1}(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L, \tag{6.4.2}$$

then the coupled system (6.1.2), (6.1.5) and (6.1.7) is said to possess the exact boundary null controllability and synchronization by 2-groups or is said to be exactly null controllable and synchronizable by 2-groups, and  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t, x)$ is called to be the partially synchronizable state.

Obviously, if the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the exact null controllability, then it must satisfy the requirement of Definition 6.4.1. This situation should be excluded in advance.

**Lemma 6.4.1** Suppose that the coupled system (6.1.2), (6.1.5) and (6.1.7) is exactly null controllable and synchronizable by 2-groups, but not exactly null controllable, then the coupling matrix  $A = (a_{ij})$  in system (6.1.2) must satisfy the following necessary conditions:

$$\sum_{j=m+1}^{N} a_{ij} = 0 \quad (i = 1, \cdots, m), \tag{6.4.3}$$

$$\sum_{j=m+1}^{N} a_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}} \quad (i=m+1,\cdots,N), \tag{6.4.4}$$

where  $\tilde{\tilde{a}}$  is a constant independent of  $i = m + 1, \cdots, N$ .

For the coupling matrices  $B = (b_{ij})$  and  $\overline{B} = (\overline{b}_{ij})$  in the boundary conditions (6.1.5) and (6.1.7), similar necessary conditions hold, namely,

$$\sum_{j=m+1}^{N} b_{ij}(\text{resp. } \bar{b}_{ij}) = 0 \quad (i = 1, \cdots, m),$$
(6.4.5)

$$\sum_{j=m+1}^{N} b_{ij}(\text{resp. } \bar{b}_{ij}) \stackrel{\text{def.}}{=} \widetilde{\tilde{b}}(\text{resp. } \widetilde{\tilde{b}}) \quad (i=m+1,\cdots,N),$$
(6.4.6)

where  $\tilde{\tilde{b}}$  and  $\tilde{\tilde{b}}$  are constants independent of  $i = m + 1, \cdots, N$ .

Chapter 6: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations 175

**Proof** By Definition 6.4.1, it follows from (6.1.4) that for  $t \ge T$  we have

$$\left(\sum_{j=m+1}^{N} a_{ij}\right)\widetilde{\widetilde{u}} = 0, \quad 0 \le x \le L, \ i = 1, \cdots, m,$$
(6.4.7)

$$\frac{\partial^2 \widetilde{\widetilde{u}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}}{\partial x^2} + \Big(\sum_{j=m+1}^N a_{ij}\Big)\widetilde{\widetilde{u}} = 0, \quad 0 \le x \le L, \ i = m+1, \cdots, N.$$
(6.4.8)

In particular, for  $t \ge T$  we get from (6.4.8) that

$$\left(\sum_{j=m+1}^{N} a_{ij}\right)\widetilde{\widetilde{u}} = \left(\sum_{j=m+1}^{N} a_{kj}\right)\widetilde{\widetilde{u}}, \quad 0 \le x \le L, \ i,k = m+1,\cdots,N.$$
(6.4.9)

Then, (6.4.3)-(6.4.4) follows directly from the non exact null controllability.

Similarly to the proof of Lemma 6.2.1, we get (6.4.5)–(6.4.6).

**Theorem 6.4.1** Suppose that (6.4.3)-(6.4.6) hold and T > 0 satisfies (6.2.8). For any given initial state  $(\Phi(x), \Psi(x))$ , we can find  $(N-1) C^2$  (case (6.1.5a)) or  $C^1$ (cases(6.1.5b)-(6.1.5c)) boundary controls with support on [0,T] in H(t) (for instance, take  $h_1(t), \dots, h_m(t), h_{m+2}(t), \dots, h_N(t)$  with  $h_{m+1}(t) \equiv 0$ , such that the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the exact boundary null controllability and synchronization by 2-groups.

**Proof** Let

$$\begin{cases} w_i = u_i \quad (i = 1, \cdots, m), \\ w_i = u_i - u_{i+1} \quad (i = m + 1, \cdots, N - 1). \end{cases}$$
(6.4.10)

Similarly to the proof of Theorem 6.2.1 (cf.[79]), under assumptions (6.4.3)-(6.4.4), the original system (6.1.2) for the variable U can be reduced to a selfclosing system (6.2.10) of the same kind for the variable  $W = (w_1, \cdots, w_{N-1})^T$ , where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-1) \times (N-1)$  matrix with

$$\widetilde{a}_{ij} = \begin{cases}
a_{ij}, \quad j = 1, \cdots, m, \\
-\sum_{p=j+1}^{N} a_{ip}, \quad j = m+1, \cdots, N-1, \\
a_{ij} = \begin{cases}
a_{ij} - a_{i+1,j}, \quad j = 1, \cdots, m, \\
-\sum_{p=j+1}^{N} (a_{ip} - a_{i+1,p}) = -\sum_{p=m+1}^{j} (a_{i+1,p} - a_{ip}), \quad j = m+1, \cdots, N-1, \\
(6.4.12)
\end{cases}$$

Similarly, by (6.4.5)–(6.4.6), the original boundary conditions (6.1.5) and (6.1.7) for the variable U can be reduced, respectively, to a self-closing boundary conditions (6.2.12) and (6.2.13) of the same kind for the variable W, in which the coupling matrices  $\tilde{B} = (\tilde{b}_{ij})$  and  $\tilde{\overline{B}} = (\tilde{\overline{b}}_{ij})$  are  $(N-1) \times (N-1)$  matrices reduced by  $B = (b_{ij})$  and  $\overline{B} = (\bar{b}_{ij})$  in the following way:

$$\widetilde{b}_{ij} = \begin{cases} b_{ij}, \quad j = 1, \cdots, m, \\ -\sum_{p=j+1}^{N} b_{ip}, \quad j = m+1, \cdots, N-1, \end{cases} \text{ for } i = 1, \cdots, m, \qquad (6.4.13)$$
$$\widetilde{b}_{ij} = \begin{cases} b_{ij} - b_{i+1,j}, \quad j = 1, \cdots, m, \\ -\sum_{p=j+1}^{N} (b_{ip} - b_{i+1,p}) = -\sum_{p=1}^{j} (b_{i+1,p} - b_{ip}), \quad j = m+1, \cdots, N-1, \end{cases}$$

etc. Moreover,  $\widetilde{H}(t)$  is given by (6.2.15) with

$$\widetilde{h}_{i}(t) = \begin{cases} h_{i}(t), & i = 1, \cdots, m, \\ h_{i}(t) - h_{i+1}(t), & i = m+1, \cdots, N-1. \end{cases}$$
(6.4.15)

The initial condition (6.2.17) of W is given by

$$\widetilde{\varphi}_{i}(x) = \begin{cases} \varphi_{i}(x), & i = 1, \cdots, m, \\ \varphi_{i}(x) - \varphi_{i+1}(x), & i = m+1, \cdots, N-1, \end{cases}$$

$$\widetilde{\psi}_{i}(x) = \begin{cases} \psi_{i}(x), & i = 1, \cdots, m, \\ \psi_{i}(x) - \psi_{i+1}(x), & i = m+1, \cdots, N-1. \end{cases}$$
(6.4.16)

Thus, according to Lemma 6.1.2, by means of (N-1)  $C^2$  (case (6.2.12a)) or  $C^1$  (cases(6.2.12b)-(6.2.12c)) boundary controls  $\widetilde{H}(t)$  (for instance, take  $h_1(t), \dots, h_m(t)$ ,  $h_{m+1}(t), \dots, h_N(t)$  with  $h_{m+1}(t) \equiv 0$ ) at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly null controllable and synchronizable by 2-groups. This proves Theorem 6.4.1.

**Remark 6.4.1** Lemma 6.1.1 and Lemma 6.1.2 are always available for the reduced initial-boundary value problem for the variable W.

**Remark 6.4.2** As  $t \ge T$ , the partially synchronizable state  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  satisfies wave equation (6.2.6) and boundary conditions (6.2.19) and (6.2.20), where  $\tilde{\tilde{a}}, \tilde{\tilde{b}}$  and  $\tilde{\tilde{b}}$  are given by (6.4.4) and (6.4.6). Lemma 6.1.1 is always available for this system (6.2.6) and (6.2.19)-(6.2.20). Chapter 6: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations 177

Hence, if we know this initial condition (6.2.21) of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  at the moment t = T, then the evolution of the partially synchronizable state  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t, x)$  with respect to t can be completely determined.

Moreover, any given state  $(\varphi, \psi)$  in  $C^2[0, L] \times C^1[0, L]$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (6.2.19) and (6.2.20), belongs to the attainable set of the initial data of partially synchronizable state.

**Remark 6.4.3** In the special case that m = N - 1,  $U^{(2)}(t, x)$  is composed of only one component  $u_N(t,x)$ , then assumptions (6.4.3) and (6.4.5) become

$$a_{iN} = 0 \quad (i = 1, \cdots, N - 1),$$
 (6.4.17)

$$b_{iN}$$
 (resp.  $b_{iN}$ ) = 0 ( $i = 1, \cdots, N - 1$ ), (6.4.18)

namely, the first (N-1) components of the last column of the coupling matrices A and B (resp.  $\overline{B}$ ) are all zero. In this case, by Theorem 6.4.1, (N-1) boundary controls can be used to realize the exact boundary null controllability for the first (N-1)state variables of U, however, since the first (N-1) wave equations and boundary conditions constitute a self-closing system, this result is obvious.

Now, taking m = N-2 in Theorem 6.4.1, assumptions (6.4.3) and (6.4.6) become

$$\sum_{j=N-1}^{N} a_{ij} = 0 \quad (i = 1, \cdots, N-2), \tag{6.4.19}$$

$$\sum_{j=N-1}^{N} a_{ij} = \tilde{\tilde{a}} \quad (i = N - 1, N)$$
(6.4.20)

and

$$\sum_{j=N-1}^{N} b_{ij} \text{ (resp. } \bar{b}_{ij}) = 0, \quad (i = 1, \cdots, N-2), \tag{6.4.21}$$

$$\sum_{j=N-1}^{N} b_{ij} \text{ (resp. } \overline{b}_{ij}) = \widetilde{\widetilde{b}} \text{ (resp. } \widetilde{\widetilde{b}}) \quad (i = N - 1, N).$$
 (6.4.22)

Thus, we can use (N-1) (instead of (N-2)!) boundary controls to realize the exact boundary null controllability for (N-2) state variables in U.

## 6.5 Generalized exact boundary synchronization

We now consider the problem of synchronization from a more general mathematical point of view.

For the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7), based on the previous discussions, we can define the corresponding generalized exact boundary synchronization.

Let

$$w_i = \sum_{j=1}^{N} \theta_{ij} u_j \quad (i = 1, \cdots, M)$$
 (6.5.1)

or

$$W = \bigoplus U, \tag{6.5.2}$$

where  $W = (w_1, \dots, w_M)^T$ , 0 < M < N and  $(H) = (\theta_{ij})_{M \times N}$  is of full row-rank.

Suppose that for any given initial state  $(\Phi(x), \Psi(x))$ , the mixed initial-boundary value problem (6.1.2), (6.1.5), (6.1.7) and (6.1.8) for the variable U can be reduced to a self-closing mixed initial-boundary values problem of the same kind for the variable W, and we can use Lemma 6.1.2 to find  $M C^2$  (case (6.1.5a)) or  $C^1$  (cases(6.1.5b)-(6.1.5c)) boundary controls in H(t) with support on [0, T] with T > 2L, such that as  $t \geq T$  we have

$$W(t,x) \equiv 0, \quad 0 \le x \le L, \tag{6.5.3}$$

i.e.,

$$(\mathbf{H})U(t,x) \equiv 0, \quad 0 \le x \le L. \tag{6.5.4}$$

Thus, the coupled system (6.1.2), (6.1.5) and (6.1.7) is said to have the **generalized** exact boundary synchronization associated with  $\bigoplus$  and U = U(t, x) is the corresponding generalized synchronizable state.

It is easily seen that this definition covers all the cases discussed in the previous sections.

(1) If M = N - 1 and we take

$$w_i = u_i - u_{i+1}$$
  $(i = 1, \cdots, N-1),$  (6.5.5)

i.e.,

$$\left( \mathbf{H} \right) = \left( \begin{array}{cccc} 1 & -1 & & \\ & \ddots & \ddots & \\ & & 1 & -1 \end{array} \right)_{(N-1) \times N},$$
(6.5.6)

then the generalized exact synchronization leads to the exact synchronization given in  $\S 6.2$ .

(2) If M = N - 1 and we take

$$\begin{cases} w_i = u_i \quad (i = 1, \cdots, m), \\ w_i = u_i - u_{i+1} \quad (i = m+1, \cdots, N-1), \end{cases}$$
(6.5.7)

i.e.,

$$\left( \mathbf{H} \right) = \left( \begin{array}{ccc} \mathbf{I}_{m} & & \\ & \left( \begin{array}{ccc} 1 & -1 & \\ & \ddots & \ddots & \\ & & 1 & -1 \end{array} \right)_{(N-m-1)\times(N-m)} \right)_{(N-1)\times N}, \quad (6.5.8)$$

where  $I_m$  is the unit matrix of order m, then the generalized exact synchronization leads to the exact null controllability and synchronization by 2-groups discussed in §6.4.

(3) If M = N - 2 and we take

$$\begin{cases} w_i = u_i - u_{i+1} \quad (i = 1, \cdots, m - 1), \\ w_i = u_{i+1} - u_{i+2} \quad (i = m, \cdots, N - 2), \end{cases}$$
(6.5.9)

,

i.e.,

then the generalized exact synchronization leads to the exact synchronization by 2-groups discussed in §6.3.
A natural question is under what conditions the original coupled system for the variable U can be reduced to a self-closing coupled system of the same type for the variable W.

Multiplying the matrix (H) to the system (6.1.2), (6.1.5) and (6.1.7) from the left, it is easy to get

**Theorem 6.5.1** For any given coupling  $N \times N$  matrices A, B and  $\overline{B}$ , if there exist a full row-rank matrix  $(\mathbb{H}) = (\theta_{ij})_{M \times N}$ , and  $M \times M$  matrices  $\widetilde{A}$ ,  $\widetilde{B}$  and  $\widetilde{\overline{B}}$  such that

$$(\mathbf{H})A = \widetilde{A}(\mathbf{H}), \tag{6.5.11}$$

$$(\mathbf{H}) = \widetilde{B}(\mathbf{H}), \qquad (0.5.11)$$

$$(\mathbf{H}) = \widetilde{B}(\mathbf{H}), \qquad (6.5.12)$$

$$(\mathbf{H}) = \widetilde{B}(\mathbf{H}), \qquad (6.5.12)$$

$$(\underline{H})\overline{B} = \overline{B}(\underline{H}), \qquad (6.5.13)$$

then we can get a self-closing coupled system (6.2.10) and (6.2.12)-(6.2.13) with H(t) = (H) H(t) for the variable W, so that the generalized exact boundary synchronization associated with (H) for the variable U can be realized by means of M boundary control functions.

Remark 6.5.1 Lemma 6.1.1 and Lemma 6.1.2 are always available for the reduced coupled system for the variable W.

The next question is for any given coupling  $N \times N$  matrices A, B and  $\overline{B}$ , how to guarantee the existence of such  $M \times N$  matrix (H) and  $M \times M$  matrices A, B and  $\overline{B}$  in Theorem 6.5.1.

We first consider the case M = N - 1.

**Theorem 6.5.2** Suppose that M = N - 1. For the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7), if the coupling matrices A, B and  $\overline{B}$  possess a common real right eigenvector  $X \in \mathbb{R}^N \setminus \{0\}$ :

$$AX = \widetilde{\widetilde{a}}X,\tag{6.5.14}$$

$$BX = \tilde{b}X, \tag{6.5.15}$$

$$\overline{B}X = \widetilde{\overline{b}}X, \tag{6.5.16}$$

where  $\tilde{\widetilde{a}}, \tilde{\widetilde{b}}$  and  $\tilde{\widetilde{b}}$  are corresponding eigenvalues, respectively, then there exist an (N - 1)1) × N full row-rank matrix  $(\underline{H}) = (\theta_{ij})$  and  $(N-1) \times (N-1)$  matrices  $\widetilde{A}$ ,  $\widetilde{B}$  and  $\widetilde{\overline{B}}$  such that (6.5.11)-(6.5.13) hold. Moreover, X constitutes a system of basic solutions to the linear algebraic system

(H)
$$Y = 0.$$
 (6.5.17)

**Proof** There exist (N-1) linearly independent column vectors  $Y_1, \dots, Y_{N-1}$  perpendicular to X;

$$\langle X, Y_j \rangle = 0, \quad j = 1, \cdots, N - 1.$$
 (6.5.18)

Thus, noting (6.5.14) we have

$$\langle X, A^T Y_j \rangle = \langle AX, Y_j \rangle = 0, \quad j = 1, \cdots, N-1.$$
 (6.5.19)

Then,

$$A^{T}Y_{j} = \sum_{k=1}^{N-1} \widehat{a}_{jk}Y_{k}, \quad j = 1, \cdots, N-1,$$
(6.5.20)

where  $\hat{a}_{jk}$  are constants. Let

$$\widetilde{A} = (\widehat{a}_{jk})^T = (\widetilde{a}_{jk})_{(N-1)\times(N-1)}.$$
 (6.5.21)

It follows from (6.5.20) that

$$A^{T}(Y_{1}, \cdots, Y_{N-1}) = (Y_{1}, \cdots, Y_{N-1})\tilde{A}^{T}, \qquad (6.5.22)$$

then, taking

$$(\mathbf{H}) = (Y_1, \cdots, Y_{N-1})^T, \tag{6.5.23}$$

we get (6.5.11). Moreover, it is easy to see that X constitutes a system of basic solutions to the linear algebraic system (6.5.17).

The proof of (6.5.12)–(6.5.13) is completely similar.

**Remark 6.5.2** In case (1), M = N - 1, noting (6.5.6),  $X = (1, \dots, 1)^T$  is a system of basic solutions to (6.5.17). The requirement that the coupling matrices A, B and  $\overline{B}$  possess a common right eigenvector X implies (6.2.2)–(6.2.4). Thus, Theorem 6.5.2 leads to the conclusion given in §6.2.

**Remark 6.5.3** In case (2), M = N-1, noting (6.5.8),  $X = (0, \dots, 0, 1, \dots, 1)^T$ is a system of basic solutions to (6.5.17). The requirement that the coupled matrices  $A, B \text{ and } \overline{B} \text{ possess a common right eigenvector } X \text{ implies } (6.4.3)-(6.4.6).$  Thus, the conclusion given in §6.4 follows from Theorem 6.5.2.

On the other hand, we have

**Theorem 6.5.3** Suppose that M = N - 1. If the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the generalized exact boundary synchronization associated with  $(\mathbb{H})$ , but is not exactly boundary null controllable, then the coupling matrices A, B and  $\overline{B}$  must have a common real right eigenvector X, and X constitutes a system of basic solutions to the linear algebraic system (6.5.17).

**Proof** Since the rank of (H) is (N-1), the linear algebraic system (6.5.17) possesses a basic solution  $X = (\alpha_1, \dots, \alpha_N)^T \in \mathbb{R}^N \setminus \{0\}$ . Thus, when  $t \ge T$ , it follows from (6.5.4) that the generalized synchronizable state  $U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  can be written as

$$u_i(t,x) = \alpha_i \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L, \tag{6.5.24}$$

where  $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t, x)$  is a  $C^2$  function.

Substituting (6.5.24) into (6.1.4), when  $t \ge T$  we have

$$\alpha_i \left( \frac{\partial^2 \widetilde{\widetilde{u}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}}{\partial x^2} \right) + \left( \sum_{j=1}^N a_{ij} \alpha_j \right) \widetilde{\widetilde{u}} = 0, \quad i = 1, \cdots, N.$$
 (6.5.25)

Let

$$\aleph = \{ i \mid \alpha_i \neq 0, \ i = 1, \cdots, N \}.$$
(6.5.26)

When the number of  $\aleph$  is bigger than or equal to 2, it is easy to see form (6.5.25) that for any given  $i, k \in \aleph$ ,

$$\frac{1}{\alpha_i} \Big( \sum_{j=1}^N a_{ij} \alpha_j \Big) \widetilde{\widetilde{u}} = \frac{1}{\alpha_k} \Big( \sum_{j=1}^N a_{kj} \alpha_j \Big) \widetilde{\widetilde{u}},$$

then by the non exact null controllability, we get

$$\frac{1}{\alpha_i} \Big( \sum_{j=1}^N a_{ij} \alpha_j \Big) = \frac{1}{\alpha_k} \Big( \sum_{j=1}^N a_{kj} \alpha_j \Big) \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}}, \quad \forall i, k \in \aleph,$$

namely, for  $i \in \aleph$  we have

$$\sum_{j=1}^{N} a_{ij} \alpha_j = \widetilde{\widetilde{a}} \alpha_i, \tag{6.5.27}$$

where  $\widetilde{\widetilde{a}}$  is a constant independent of  $i \in \aleph$ . On the other hand, for  $i \notin \aleph$ , since  $\alpha_i = 0$ , by (6.5.25) and the non exact null controllability, (6.5.27) is still valid.

When the number of  $\aleph$  is equal to 1, without loss of generality, we may suppose that  $\aleph = \{1\}$ , namely,  $\alpha_1 \neq 0$  and  $\alpha_2 = \cdots \alpha_N = 0$ . Taking  $\tilde{a} = a_{11}$ , we still have (6.5.27) for i = 1. Moreover, as in the previous situation, (6.5.27) is still valid for  $i \notin \aleph$ .

Thus, for  $i = 1 \cdots, N$  we always have (6.5.27), where  $\tilde{\tilde{a}}$  is independent of i. This proves (6.5.14). Moreover, it follows from (6.5.25) and (6.5.14) that  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t, x)$  satisfies the wave equation (6.2.6) for  $t \geq T$ .

We next prove (6.5.15). The proof of (6.5.16) is similar.

Substituting (6.5.24) into the boundary condition (6.1.5c), for  $t \ge T$  we have

$$x = 0: \ \alpha_i \frac{\partial \widetilde{\widetilde{u}}}{\partial x} - \Big(\sum_{j=1}^N b_{ij} \alpha_j\Big) \widetilde{\widetilde{u}} = 0, \quad i = 1, \cdots, N.$$
(6.5.28)

As in the proof of Lemma 6.2.1, we may suppose that  $\tilde{\widetilde{u}}(t,0) \neq 0$  for  $t \geq T$ , then similarly we can get

$$\sum_{j=1}^{N} b_{ij} \alpha_j = \widetilde{\widetilde{b}} \alpha_i, \quad i = 1, \cdots, N,$$

where  $\tilde{b}$  is a constant independent of *i*. This proves (6.5.15).

**Remark 6.5.4** As  $t \geq T$ , for the generalized synchronizable state  $U(t,x) = \widetilde{\widetilde{u}}(t,x)X$ , where  $X = (\alpha_1, \dots, \alpha_N)^T$  is a basic solution of (6.5.17),  $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t,x)$  satisfies the wave equation (6.2.6) and the boundary conditions (6.2.19) and (6.2.20). Then, if we know the initial state (6.2.21) of  $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t,x)$  at the moment t = T, we can completely determine the evolution of  $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t,x)$  with respect to t.

Moreover, as in Remark 6.2.4, the set of all possible initial data  $(\varphi, \psi)$  of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  is the whole space  $C^2[0,L] \times C^1[0,L]$  associated with the conditions of  $C^2$  compatibility at the points (t,x) = (T,0) and (T,L), respectively, with boundary conditions (6.2.19) and (6.2.20).

For the case M = N - 2, we only consider the special situation that

$$\left( \begin{array}{cc} \left( \begin{array}{c} \left( \begin{array}{c} H \right)_1 \\ \end{array} \right)_{(N-2) \times N} \end{array} \right),$$
(6.5.29)

where  $(\mathbb{H})_1$  is a full row-rank  $(N_1 - 1) \times N_1$  matrix, while  $(\mathbb{H})_2$  is a full row-rank  $(N_2 - 1) \times N_2$  matrix with  $N_1 + N_2 = N$ .

Let

$$A = \begin{pmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{pmatrix}, \quad \widetilde{A} = \begin{pmatrix} \widetilde{A}_{11} & \widetilde{A}_{12} \\ \widetilde{A}_{21} & \widetilde{A}_{22} \end{pmatrix} \text{ etc.}, \quad (6.5.30)$$

where  $A_{11}$  (resp.  $\widetilde{A}_{11}$ ) is an  $N_1 \times N_1$  (resp.  $(N_1-1) \times (N_1-1)$ ) matrix,  $A_{21}$  (resp.  $\widetilde{A}_{21}$ ) is an  $N_2 \times N_1$  (resp.  $(N_2-1) \times (N_1-1)$ ) matrix,  $A_{12}$  (resp.  $\widetilde{A}_{12}$ ) is an  $N_1 \times N_2$  (resp.

 $(N_1-1) \times (N_2-1)$ ) matrix,  $A_{22}$  (resp.  $\widetilde{A}_{22}$ ) is an  $N_2 \times N_2$  (resp.  $(N_2-1) \times (N_2-1)$ ) matrix etc. Thus, (6.5.11)–(6.5.13) in Theorem 6.5.1 can be equivalently written as

$$(\mathbb{H})_i A_{ij} = \widetilde{A}_{ij} (\mathbb{H})_j, \quad i, j = 1, 2,$$
(6.5.31)

$$\left(\mathbf{H}\right)_{i}B_{ij} = \widetilde{B}_{ij}\left(\mathbf{H}\right)_{j}, \quad i, j = 1, 2, \tag{6.5.32}$$

$$\left(\mathbf{H}\right)_{i}B_{ij} = \overline{B}_{ij}\left(\mathbf{H}\right)_{j}, \quad i, j = 1, 2.$$

$$(6.5.33)$$

**Theorem 6.5.4** Suppose that M = N - 2. For the coupled system of wave equations (6.1.2), (6.1.5) and (6.1.7), if there exist real vectors  $X_1 \in \mathbb{R}^{N_1} \setminus \{0\}$  and  $X_2 \in \mathbb{R}^{N_2} \setminus \{0\}$  and real numbers  $\tilde{\tilde{a}}_{ij}$ ,  $\tilde{\tilde{b}}_{ij}$  and  $\tilde{\tilde{b}}_{ij}$  such that

$$A_{ij}X_j = \widetilde{\widetilde{a}}_{ij}X_i, \quad i, j = 1, 2, \tag{6.5.34}$$

$$B_{ij}X_j = \tilde{b}_{ij}X_i, \quad i, j = 1, 2,$$
 (6.5.35)

$$\overline{B}_{ij}X_j = \widetilde{\overline{b}}_{ij}X_i, \quad i, j = 1, 2,$$
(6.5.36)

then we can find a  $(N-2) \times N$  full row-rank matrix  $\bigoplus = \begin{pmatrix} (\mathbb{H})_1 & \\ (\mathbb{H})_2 \end{pmatrix}$  and  $(N-2) \times (N-2)$  matrices  $\widetilde{A}$ ,  $\widetilde{B}$  and  $\overline{\widetilde{B}}$  such that (6.5.31)-(6.5.33) hold, moreover,  $X_1$  (resp.  $X_2$ ) constitutes a system of basic solutions to the linear algebraic system

$$(\mathbb{H})_1 Y_1 = 0 \text{ (resp. } (\mathbb{H})_2 Y_2 = 0).$$
 (6.5.37)

**Proof** Specially taking i = j in (6.5.34)–(6.5.36), we get

$$A_{ii}X_i = \widetilde{\widetilde{a}}_{ii}X_i, \quad i = 1, 2, \tag{6.5.38}$$

$$B_{ii}X_i = \tilde{b}_{ii}X_i, \quad i = 1, 2,$$
 (6.5.39)

$$\overline{B}_{ii}X_i = \overline{\tilde{b}}_{ii}X_i, \quad i = 1, 2.$$
(6.5.40)

Then, by the proof of Theorem 6.5.2, there exist  $N_1 - 1$  (resp.  $N_2 - 1$ ) linearly independent column vectors  $Y_1, \dots, Y_{N_1-1} \in \mathbb{R}^{N_1} \setminus \{0\}$  (resp.  $Y_{N_1}, \dots, Y_{N-2} \in \mathbb{R}^{N_2} \setminus \{0\}$ ) perpendicular to  $X_1$  (resp.  $X_2$ ) such that (6.5.31)–(6.5.33) hold for i = j with some matrices  $\widetilde{A}_{ii}$ ,  $\widetilde{B}_{ii}$  and  $\widetilde{\overline{B}}_{ii}$  and

$$(\mathbb{H})_1 = (Y_1, \cdots, Y_{N_1-1})^T, \quad (\mathbb{H})_2 = (Y_{N_1}, \cdots, Y_{N-2})^T.$$
 (6.5.41)

We now prove that there exist matrices  $\widetilde{A}_{ij}$ ,  $\widetilde{B}_{ij}$  and  $\widetilde{\overline{B}}_{ij}$  for  $i \neq j$  such that (6.5.31)-(6.5.33) are still valid for  $i \neq j$ , in which  $(\mathbb{H}_1 \text{ and } (\mathbb{H}_2 \text{ are given by } (6.5.41))$ . For fixing the data, in what follows we only prove (6.5.31) for i = 1 and j = 2. Noting (6.5.34), we have

$$\langle X_2, A_{12}^T Y_j \rangle = \langle A_{12} X_2, Y_j \rangle = \langle \widetilde{\widetilde{a}}_{12} X_1, Y_j \rangle = 0, \quad j = 1, \cdots, N_1 - 1.$$
 (6.5.42)

Then

$$A_{12}^T Y_j = \sum_{k=N_1}^{N-2} \widehat{a}_{jk} Y_k, \quad j = 1, \cdots, N_1 - 1.$$
 (6.5.43)

where  $\hat{a}_{jk}$  are constants. Let

$$\widetilde{A}_{12} = (\widehat{a}_{jk})^T = (\widetilde{a}_{jk}). \tag{6.5.44}$$

By (6.5.43), we have

$$A_{12}^T(Y_1, \cdots, Y_{N_1-1}) = (Y_1, \cdots, Y_{N_1-1})\widetilde{A}_{12}^T, \qquad (6.5.45)$$

which together with (6.5.41) leads to (6.5.31) for i = 1 and j = 2.

**Remark 6.5.5** In case (3), M = N-2, noting (6.5.10), we have  $X_1 = (1, \dots, 1)^T$ and  $X_2 = (1, \dots, 1)^T$ . The requirements (6.5.34)–(6.5.36) imply (6.3.3)–(6.3.4) and (6.3.7)–(6.3.8), then Theorem 6.5.4 leads to the conclusion given in §6.3.

By (6.5.29) and noting that the rank of both  $(\mathbb{H}_1 \text{ and } (\mathbb{H}_2 \text{ is } (N-1), \text{ linear algebraic system (6.5.37) possesses a basic solution <math>X_1 = (\alpha_1, \cdots, \alpha_{N_1})^T \in \mathbb{R}^{N_1} \setminus \{0\}$  (resp.  $X_2 = (\alpha_{N_1+1}, \cdots, \alpha_N)^T \in \mathbb{R}^{N_2} \setminus \{0\}$ ). Thus, noting (6.5.4), as  $t \geq T$  the generalized synchronizable state  $U(t, x) = (u_1(t, x), \cdots, u_N(t, x))^T$  can be written as

 $u_p(t,x) = \alpha_p \widetilde{\widetilde{u}}_1(t,x), \quad p = 1, \cdots, N_1, \quad 0 \le x \le L,$  (6.5.46)

$$u_q(t,x) = \alpha_q \widetilde{u}_2(t,x), \quad q = N_1 + 1, \cdots, N, \quad 0 \le x \le L.$$
 (6.5.47)

Similarly to Theorem 6.5.3, we have

**Theorem 6.5.5** Suppose that M = N - 2. If the coupled system (6.1.2), (6.1.5) and (6.1.7) possesses the generalized exact boundary synchronization associated with (6.5.29), and at least for an initial state  $(\Phi(x), \Psi(x))$ ,  $\tilde{\tilde{u}}_1(t, x)$  and  $\tilde{\tilde{u}}_2(t, x)$  given by (6.5.46)–(6.5.47) for the generalized synchronizable state are linearly independent. Then there exist real vectors  $X_1 \in \mathbb{R}^{N_1} \setminus \{0\}$  and  $X_2 \in \mathbb{R}^{N_2} \setminus \{0\}$  and real numbers  $\tilde{\tilde{a}}_{ij}$  (i, j = 1, 2) such that (6.5.34) holds. **Proof** Substituting (6.5.46)–(6.5.47) into (6.1.4), we have

$$\alpha_i \left( \frac{\partial^2 \widetilde{\widetilde{u}}_1}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}_1}{\partial x^2} \right) + \left( \sum_{p=1}^{N_1} a_{ip} \alpha_p \right) \widetilde{\widetilde{u}}_1 + \left( \sum_{q=N_1+1}^N a_{iq} \alpha_q \right) \widetilde{\widetilde{u}}_2 = 0, \quad i = 1, \cdots, N_1,$$
(6.5.48)

$$\alpha_i \left( \frac{\partial^2 \widetilde{\widetilde{u}}_2}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{u}}_2}{\partial x^2} \right) + \left( \sum_{p=1}^{N_1} a_{ip} \alpha_p \right) \widetilde{\widetilde{u}}_1 + \left( \sum_{q=N_1+1}^N a_{iq} \alpha_q \right) \widetilde{\widetilde{u}}_2 = 0, \quad i = N_1 + 1, \cdots, N.$$

$$(6.5.49)$$

Let

 $\aleph_1 = \{ p \mid \alpha_p \neq 0, \ p = 1, \cdots, N_1 \}.$ (6.5.50)

When the number  $\aleph_1$  is bigger than or equal to 2, it is easy to see from (6.5.48) that for any given  $i, k \in \aleph_1$ 

$$\frac{1}{\alpha_i} \Big[ \Big( \sum_{p=1}^{N_1} a_{ip} \alpha_p \Big) \widetilde{\widetilde{u}}_1 + \Big( \sum_{q=N_1+1}^N a_{iq} \alpha_q \Big) \widetilde{\widetilde{u}}_2 \Big] \\= \frac{1}{\alpha_k} \Big[ \Big( \sum_{p=1}^{N_1} a_{kp} \alpha_p \Big) \widetilde{\widetilde{u}}_1 + \Big( \sum_{q=N_1+1}^N a_{kq} \alpha_q \Big) \widetilde{\widetilde{u}}_2 \Big],$$

then, since we may assume that  $\tilde{\widetilde{u}}_1$  and  $\tilde{\widetilde{u}}_2$  are linearly independent, we get

$$\frac{1}{\alpha_i} \left( \sum_{p=1}^{N_1} a_{ip} \alpha_p \right) = \frac{1}{\alpha_k} \left( \sum_{p=1}^{N_1} a_{kp} \alpha_p \right) \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}}_{11}, \quad \forall i, k \in \aleph_1,$$
$$\frac{1}{\alpha_i} \left( \sum_{q=N_1+1}^{N} a_{iq} \alpha_q \right) = \frac{1}{\alpha_k} \left( \sum_{q=N_1+1}^{N} a_{kq} \alpha_q \right) \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}}_{12}, \quad \forall i, k \in \aleph_1,$$

namely, for  $i \in \aleph_1$ , we have

$$\sum_{p=1}^{N_1} a_{ip} \alpha_p = \widetilde{\widetilde{a}}_{11} \alpha_i, \qquad (6.5.51)$$

$$\sum_{q=N_1+1}^{N} a_{iq} \alpha_q = \widetilde{\widetilde{a}}_{12} \alpha_i, \qquad (6.5.52)$$

where  $\widetilde{\widetilde{a}}_{11}$  and  $\widetilde{\widetilde{a}}_{12}$  are constants in dependent of  $i \in \aleph_1$ . On the other hand, for  $i \notin \aleph_1$ , since  $\alpha_i = 0$ , by (6.5.48) and noting that  $\widetilde{\widetilde{u}}_1$  and  $\widetilde{\widetilde{u}}_2$  are linearly independent, (6.5.51)–(6.5.52) are still valid.

When the number of  $\aleph_1$  is equal to 1, without loss of generality, we may suppose that  $\aleph_1 = \{1\}$ , namely,  $\alpha_1 \neq 0$  and  $\alpha_2 = \cdots = \alpha_{N_1} = 0$ . Taking  $\tilde{\tilde{a}}_{11} = a_{11}$  and  $\tilde{\tilde{a}}_{12} = \frac{1}{\alpha_1} \left( \sum_{q=N_1+1}^N a_{1q} \alpha_q \right)$ , we still have (6.5.51)–(6.5.52) for i = 1. Moreover, as in the previous situation, (6.5.51)–(6.5.52) are still valid for  $i \notin \aleph_1$ .

Thus, for  $i = 1, \dots, N$ , we always have (6.5.51)–(6.5.52), where  $\tilde{\tilde{a}}_{11}$  and  $\tilde{\tilde{a}}_{12}$  are independent of i. This proves (6.5.34) for i = j = 1 and i = 1, j = 2.

Similarly, by (6.5.49), we get (6.5.34) for i = 2, j = 1 and i = j = 2.

**Remark 6.5.6** As  $t \ge T$ , for the generalized sychronizable state  $U(t, x) = \begin{pmatrix} \tilde{\tilde{u}}_{1}(t, x)X_{1} \\ \tilde{\tilde{u}}_{2}(t, x)X_{2} \end{pmatrix}$ given by (6.5.46)-(6.5.47), where  $X_{1} = (\alpha_{1}, \dots, \alpha_{N_{1}})^{T}$  (resp.  $X_{2} = (\alpha_{N_{1}+1}, \dots, \alpha_{N})^{T}$ ) is a basic solution to (6.5.37),  $\tilde{\widetilde{U}}(t, x) = \begin{pmatrix} \tilde{\tilde{u}}_{1}(t, x) \\ \tilde{\tilde{u}}_{2}(t, x) \end{pmatrix}$  satisfies the coupled system (6.3.16) of wave equations with (6.3.17), in which  $\tilde{\tilde{a}}_{ij}$  (i, j = 1, 2) are given by (6.5.34), and the boundary conditions (6.3.18) and (6.3.19) with (6.3.20), in which  $\tilde{\tilde{b}}_{ij}$  and  $\tilde{\tilde{b}}_{ij}$  (i, j = 1, 2) are given by (6.5.35)-(6.5.36). Hence, if we know the initial state (6.3.21) of  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t, x)$  at the moment t = T, then the evolution of  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t, x)$ with respect to t can be completely determined.

Moreover, any given state  $(\varphi^{(1)}(x), \varphi^{(2)}(x))^T$  and  $(\psi^{(1)}(x), \psi^{(2)}(x))^T$  in  $(C^2[0, L])^2 \times (C^1[0, L])^2$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0)and (T, L), respectively, with boundary conditions (6.3.18) and (6.3.19), can be the initial data of  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t, x)$ .

Thus, at least for some initial states  $(\Phi(x), \Psi(x))$ ,  $\tilde{\tilde{u}}_1(t, x)$  and  $\tilde{\tilde{u}}_2(t, x)$  are linearly independent on t = T then for  $t \ge T$ . This shows that the corresponding requirement given in Theorem 6.5.5 is reasonable.

**Remark 6.5.7** The situation M = N - k (k > 2) can be similarly discussed.

#### 6.6 Remarks

**Remark 6.6.1** The whole previous discussion is still valid for the case of twosided controls, provided that (6.2.8) is replaced by

$$T > L \tag{6.6.1}$$

and there is an inhomogeneous term  $\overline{H}(t)$  on the right-hand side of (6.1.7).

**Remark 6.6.2** It is possible that, different from Lemma 6.1.2, in the case of one-sided controls, the exact boundary (null) controllability may be realized (perhaps

in a weaker sense) by means of M(< N) boundary controls for some special coupled systems of N wave equations (cf.[2],[3], [6],[78]). Thus, the exact boundary synchronization (perhaps in a weaker sense) could be realized by only P(< M) boundary controls.

**Remark 6.6.3** The problem of synchronization might be also considered as an optimal control problem, where an objective function of the type

$$\sum_{i=1}^{N-1} \|u_i - u_{i+1}\|^2$$

is minimized. This kind of optimal control is similar to the optimal control of partial differential equations on networks (cf.[45] for an application in flood management) and should be studied in the framework of weak solutions.

## Chapter 7

# Exact Boundary Synchronization for a Coupled System of Linear Wave Equations with Coupled Boundary Conditions of Dissipative Type

#### Contents

7.1	Introduction
7.2	Exact boundary synchronization
7.3	Exact boundary synchronization by groups 197
7.4	Exact boundary null controllability and synchronization by groups
7.5	Generalized exact boundary synchronization 208

## 7.1 Introduction

Several kinds of exact synchronizations for a coupled system of 1-D wave equations with boundary conditions of Dirichlet type, Neumann type and coupled third type are discussed in [81] in the framework of classical solutions. A similar result for a coupled system of n-dimensional wave equations with Dirichlet boundary controls in the framework of weak solutions can be found in [79]. In this chapter, we will consider the corresponding exact synchronizations for the coupled boundary conditions of dissipative type.

More precisely, we consider the following coupled system of 1-D wave equations with coupled boundary conditions of dissipative type:

$$\begin{cases} \frac{\partial^2 U}{\partial t^2} - \frac{\partial^2 U}{\partial x^2} + AU = 0, \qquad (7.1.1)\\ x = 0: U_x - CU_t = H(t), \qquad (7.1.2)\\ U_x = U_x + \overline{Q}U_x = 0, \qquad (7.1.2) \end{cases}$$

$$x = 0: U_x - CU_t = H(t), (7.1.2)$$

$$x = L : U_x + \overline{C}U_t = 0, (7.1.3)$$

where  $U = (u_1, \dots, u_N)^T$  is the unknown vector function of  $(t, x), A = (a_{ij}), C =$  $(c_{ij})$  and  $\overline{C} = (\overline{c}_{ij})$  are  $N \times N$  coupling matrices with constant elements, H(t) = $(h_1(t), \dots, h_N(t))^T$  is a  $C^1$  vector function, the components of which will be totally or partially taken as boundary controls. Our goal is to find a control time T > 0and some boundary controls in H(t) supported on [0, T], such that from the time t = T on, the system states tend to the same. That is to say, we hope to achieve the synchronization of system states not only at the moment t = T under the action of boundary controls on [0, T], but also when  $t \geq T$  and withdrawing all the controls(see also [79]-[81]).

Obviously, if the system has the exact boundary null controllability ([55], [77], [88], [89] and [100]), it must have the exact boundary synchronization, but this is a trivial situation that should be excluded beforehand. The exact boundary null controllability here also means that the system states remain null not only at the moment t = T under the action of boundary controls on [0, T], but also when t > Tand withdrawing all the controls.

For the forward mixed initial-boundary value problem (7.1.1)-(7.1.3) with the following initial condition

$$t = 0: U = \Phi(x), U_t = \Psi(x), \tag{7.1.4}$$

we have (see [55])

**Lemma 7.1.1** For any given T > 0, assume that  $\Phi \in (C^2[0,L])^N$  and  $\Psi \in$  $(C^{1}[0,L])^{N}$  satisfy the conditions of  $C^{2}$  compatibility at the points (t,x) = (0,0) and (0, L), respectively (see Remark 7.1.1). For any given coupling matrices A, C and C, if

$$\det(C+I_N) \neq 0 \tag{7.1.5}$$

and

$$\det(\overline{C} + I_N) \neq 0, \tag{7.1.6}$$

where  $I_N$  denotes the unit matrix of order N, namely, -1 is not an eigenvalue of both C and  $\overline{C}$ , then the forward mixed initial-boundary value problem (7.1.1)-(7.1.4) admits a unique  $C^2$  solution U = U(t,x) on the domain  $R(T) = \{(t,x) \mid 0 \leq t \leq t \leq t \}$ 

 $T, \ 0 \le x \le L\}.$ 

**Remark 7.1.1** The conditions of  $C^2$  compatibility at the point (t, x) = (0, L) can be written as

$$\begin{cases} \Phi'(L) + \overline{C}\Psi(L) = 0, \\ \Psi'(L) + \overline{C}(\Phi''(L) - A\Phi(L)) = 0. \end{cases}$$

$$(7.1.7)$$

The conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) can be similarly obtained.

**Remark 7.1.2** For the backward mixed initial-boundary value problem (7.1.1)-(7.1.3) with the final condition

$$t = T: U = \overline{\Phi}(x), \quad U_t = \overline{\Psi}(x),$$
(7.1.8)

where  $\overline{\Phi} \in (C^2[0,L])^N$ ,  $\overline{\Psi} \in (C^1[0,L])^N$  and the conditions of  $C^2$  compatibility are satisfied at the points (t,x) = (T,0) and (T,L), respectively. The conclusion of Lemma 7.1.1 is still valid, provided that assumptions (7.1.5)-(7.1.6) are replaced by

 $\det(C - I_N) \neq 0 \tag{7.1.9}$ 

and

$$\det(\overline{C} - I_N) \neq 0, \tag{7.1.10}$$

namely, 1 is not an eigenvalue of both C and  $\overline{C}$ .

By Lemma 7.1.1 and using a constructive method, we have the following result on the exact boundary null controllability (see [55], [72], [83], [109], [118]).

#### Lemma 7.1.2 Let

$$T > 2L.$$
 (7.1.11)

Under assumptions (7.1.5)-(7.1.6), for any given initial data  $(\Phi, \Psi) \in (C^2[0, L])^N \times (C^1[0, L])^N$ , satisfying the conditions of  $C^2$  compatibility at the point (t, x) = (0, L), there exist N C<sup>1</sup> boundary controls  $H(t) = (h_1(t), \dots, h_N(t))^T$  with support on [0, T] at the end x = 0, such that the corresponding mixed initial-boundary value problem (7.1.1)-(7.1.4) admits a unique  $C^2$  solution U = U(t, x) on  $t \ge 0$ , and as  $t \ge T$  we have

$$U(t,x) \equiv 0, \quad 0 \le x \le L.$$
 (7.1.12)

**Remark 7.1.3** Different from the situation discussed in [81], in the case of coupled boundary conditions of dissipative type, in order to guarantee the well-posedness and the exact boundary null controllability for the forward mixed initial-boundary value problem, we should impose assumptions (7.1.5)-(7.1.6) for matrices C and  $\overline{C}$ .

In what follows, we will use Lemma 7.1.1 and Lemma 7.1.2 to consider the exact boundary synchronization, the exact boundary synchronization by groups, and the exact boundary null controllability and synchronization by groups for the coupled system of wave equations (7.1.1)-(7.1.3) in §7.2, §7.3 and §7.4, respectively, and then the generalized exact boundary synchronization in §7.5. The whole forthcoming discussion is based on the situation of one-sided controls, similar results are still valid for the case of two-sided controls (see [81]).

### 7.2 Exact boundary synchronization

**Definition 7.2.1** If there exists T > 0 such that for any given initial state  $(\Phi, \Psi)$ , we can find some boundary controls with support on [0,T] in H(t), such that the corresponding mixed initial-boundary value problem (7.1.1)-(7.1.4) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  on  $t \ge 0$ , and as  $t \ge T$  we have

$$u_1(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L,$$
(7.2.1)

then the coupled system (7.1.1)-(7.1.3) is said to possess the exact boundary synchronization or is said to be exactly synchronizable, and  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  is called the corresponding synchronizable state.

Obviously, if the system possesses the exact boundary null controllability, then it must be exactly synchronizable. However, this trivial situation should be excluded in the discussion. Similarly to [81], we have

**Lemma 7.2.1** Under assumptions (7.1.5) and (7.1.6), if the coupled system of wave equations (7.1.1)-(7.1.3) is exactly synchronizable, but not exactly null controllable, then the coupling matrix A should satisfy the following necessary conditions: The values of the sum of every row of A are independent of  $i = 1, \dots, N$ , namely,

$$\sum_{j=1}^{N} a_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}} \quad (i = 1, \cdots, N),$$
(7.2.2)

where  $\tilde{\tilde{a}}$  is a constant independent of  $i = 1, \cdots, N$ .

Different from the situation discussed in [81], Lemma 7.2.1 is no longer valid in general for the coupling matrices C and  $\overline{C}$  in the coupled dissipative boundary conditions (7.1.2)-(7.1.3). However, based on Lemma 7.2.1, for matrices C and  $\overline{C}$  we impose the following similar assumptions:

$$\sum_{j=1}^{N} c_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{c}} \quad (i = 1, \cdots, N),$$
(7.2.3)

$$\sum_{j=1}^{N} \bar{c}_{ij} \stackrel{\text{def.}}{=} \widetilde{\tilde{c}} \quad (i = 1, \cdots, N),$$
(7.2.4)

where  $\tilde{\tilde{c}}$  and  $\tilde{\bar{c}}$  are constants independent of  $i = 1, \cdots, N$ .

**Remark 7.2.1** Suppose furthermore that the constant  $\tilde{a}$  given by (7.2.2) is positive, it is easy to get the necessity of (7.2.3)-(7.2.4). In fact, for a coupled system of n-dimensional wave equations with Neumann boundary conditions, in order to guarantee the uniqueness of solution, one often assumes that A is a positive definite matrix, then the hypothesis  $\tilde{a} > 0$  is reasonable since  $\tilde{a}$  is an eigenvalue of matrix A. However, in more general cases, we do not know the necessity of (7.2.3)-(7.2.4). It is a very interesting problem if we can realize the boundary synchronization for the coupled system (7.1.1)-(7.1.3) without additional assumptions (7.2.3)-(7.2.4).

**Remark 7.2.2** Conditions (7.2.2)-(7.2.4) mean that all the matrices A, C and  $\overline{C}$  have a common right eigenvector  $(1, 1, \dots, 1)^T$  with the corresponding eigenvalues  $\widetilde{\widetilde{a}}, \widetilde{\widetilde{c}}$  and  $\widetilde{\widetilde{c}}$ , respectively.

**Theorem 7.2.1** Under assumptions (7.1.5)-(7.1.6), suppose furthermore that (7.2.2) – (7.2.4) hold. Let

$$T > 2L.$$
 (7.2.5)

For any given initial state  $(\Phi, \Psi) \in (C^2[0, L])^N \times (C^1[0, L])^N$ , satisfying the conditions of  $C^2$  compatibility at the point (t, x) = (0, L), there exist  $(N - 1) C^1$  boundary controls with support on [0, T] in H(t) (for example, take  $h_2(t), \dots, h_N(t)$  with  $h_1(t) \equiv 0$ ), such that the coupled system (7.1.1)-(7.1.3) possesses the exact boundary synchronization.

#### $\mathbf{Proof} \ \ \mathrm{Let}$

$$w_i = u_{i+1} - u_i \quad (i = 1, \cdots, N - 1).$$
 (7.2.6)

It is easily shown (see [79]-[81]) that by (7.2.2), the original system (7.1.1) for the variable U can be reduced to a self-closing system of the same kind for the variable

 $W = (w_1, \cdots, w_{N-1})^T$  as follows:

$$\frac{\partial^2 W}{\partial t^2} - \frac{\partial^2 W}{\partial x^2} + \widetilde{A}W = 0, \qquad (7.2.7)$$

where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-1) \times (N-1)$  matrix with the entries

$$\widetilde{a}_{ij} = \sum_{p=j+1}^{N} (a_{i+1,p} - a_{ip}) = \sum_{p=1}^{j} (a_{ip} - a_{i+1,p}), \quad i, j = 1, \cdots, N-1.$$
(7.2.8)

Similarly, by (7.2.3)-(7.2.4), the original boundary conditions (7.1.2)-(7.1.3) for the variable U can be reduced, respectively, to self-closing boundary conditions of the same kind for the variable W as follows:

$$x = 0: \quad W_x - \widetilde{C}W_t = \widetilde{H}(t), \tag{7.2.9}$$

$$x = L: \quad W_x + \overline{C}W_t = 0, \tag{7.2.10}$$

where  $\widetilde{C} = (\widetilde{c}_{ij})$  and  $\widetilde{\overline{C}} = (\widetilde{\overline{c}}_{ij})$  are  $(N-1) \times (N-1)$  coupling matrices reduced from  $C = (c_{ij})$  and  $\overline{C} = (\overline{c}_{ij})$  in the following way:

$$\widetilde{c}_{ij} = \sum_{p=j+1}^{N} (c_{i+1,p} - c_{ip}) = \sum_{p=1}^{j} (c_{ip} - c_{i+1,p}), \quad i, j = 1, \cdots, N-1,$$
(7.2.11)

$$\widetilde{\bar{c}}_{ij} = \sum_{p=j+1}^{N} (\bar{c}_{i+1,p} - \bar{c}_{ip}) = \sum_{p=1}^{j} (\bar{c}_{ip} - \bar{c}_{i+1,p}), \quad i, j = 1, \cdots, N-1.$$
(7.2.12)

Moreover,

$$\widetilde{H}(t) = (\widetilde{h}_1(t), \cdots, \widetilde{h}_{N-1}(t))^T$$
(7.2.13)

with

$$\widetilde{h}_i(t) = h_{i+1}(t) - h_i(t), \quad i = 1, \cdots, N-1.$$
(7.2.14)

Meanwhile, the initial condition of W is given by

$$t = 0$$
:  $W = \widetilde{\Phi}(x), \quad W_t = \widetilde{\Psi}(x)$  (7.2.15)

with

$$\begin{cases} \widetilde{\Phi}(x) = (\varphi_2(x) - \varphi_1(x), \varphi_3(x) - \varphi_2(x), \cdots, \varphi_N(x) - \varphi_{N-1}(x))^T, \\ \widetilde{\Psi}(x) = (\psi_2(x) - \psi_1(x), \psi_3(x) - \psi_2(x), \cdots, \psi_N(x) - \psi_{N-1}(x))^T. \end{cases}$$
(7.2.16)

Chapter 7: Exact Boundary Synchronization for a Coupled System of Linear Wave Equations with Coupled Boundary Conditions of Dissipative Type 195

Different from the discussion in [79]-[81], in the case of coupled dissipative boundary conditions, we should verify the corresponding conditions (7.1.5)-(7.1.6) in Lemma 7.1.1 and Lemma 7.1.2 for the reduced system (7.2.7) and (7.2.9)-(7.2.10) in order to guarantee the well-posedness and the exact boundary null controllability of the reduced mixed initial-boundary value problem. For this purpose, under assumption (7.2.3) we will prove

$$det(C+I_N) = (\tilde{\tilde{c}}+1)det(\tilde{C}+I_{N-1}).$$
(7.2.17)

In fact, let  $e_i = (0, \dots, \stackrel{(i)}{1}, \dots, 0)^T$ ,  $E_j = \sum_{i=1}^j e_i$ ,  $(i, j = 1, \dots, N)$ . Noting (7.2.11), it is easy to see that

$$\widetilde{c}_{ij} = (e_i - e_j)^T C E_j, \quad 1 \le i, j \le N - 1.$$
(7.2.18)

Moreover, by (7.2.3), we have

$$(e_i - e_{i+1})^T C E_N = 0, \quad 1 \le i \le N - 1$$
 (7.2.19)

and

$$e_N^T C E_j = \sum_{i=1}^j C_{Ni} \stackrel{\text{def.}}{=} x_j, \ 1 \le j \le N$$
 (7.2.20)

with  $x_N = \widetilde{\widetilde{c}}$ . Therefore, by elementary transformation we have

$$X_N C Y_N = \begin{pmatrix} \widetilde{C} & \mathbf{0} \\ x & \widetilde{\tilde{c}} \end{pmatrix}, \qquad (7.2.21)$$

where

$$X_{N} = \begin{pmatrix} (e_{1} - e_{2})^{T} \\ (e_{2} - e_{3})^{T} \\ \vdots \\ (e_{N-1} - e_{N})^{T} \\ e_{N}^{T} \end{pmatrix} = \begin{pmatrix} 1 & -1 & & \\ & 1 & \ddots & \\ & & \ddots & \ddots & \\ & & & 1 & -1 \\ & & & & 1 \end{pmatrix}_{N \times N}, \quad (7.2.22)$$

$$Y_N = (E_1, \cdots, E_N) = \begin{pmatrix} 1 & 1 & \cdots & \cdots & 1 \\ & 1 & \cdots & \cdots & 1 \\ & & \ddots & \ddots & \vdots \\ & & & \ddots & \vdots \\ & & & & 1 \end{pmatrix}_{N \times N},$$
(7.2.23)

 $x = (x_1, \dots, x_{N-1})$  is a row vector of order N - 1, and **0** is the zero vector of order N - 1. Obviously, both  $X_N$  and  $Y_N$  are invertible matrices, and

$$X_N Y_N = I_N. (7.2.24)$$

Then

$$X_N(C+I_N)Y_N = X_NCY_N + I_N = \begin{pmatrix} \tilde{C} + I_{N-1} & \mathbf{0} \\ x & \tilde{c} + 1 \end{pmatrix}, \qquad (7.2.25)$$

from which we get immediately (7.2.17).

Thus, assumption (7.1.5) implies that

$$\det(\tilde{C} + I_{N-1}) \neq 0 \tag{7.2.26}$$

and

$$\widetilde{\widetilde{c}} \neq -1. \tag{7.2.27}$$

Similarly, by (7.2.4) and (7.1.6) we have

$$\det(\widetilde{\overline{C}} + I_{N-1}) \neq 0 \tag{7.2.28}$$

and

$$\widetilde{\widetilde{c}} \neq -1. \tag{7.2.29}$$

Therefore, under assumptions (7.1.5)-(7.1.6), by Lemma 7.1.1, the reduced mixed initial-boundary value problem (7.2.7),(7.2.9)-(7.2.10) and (7.2.15) for the variable W is always well-posed, moreover, according to Lemma 7.1.2, by means of (N-1)  $C^1$  boundary controls  $\tilde{H}(t)$  (for instance, take  $h_2(t), \dots, h_n(t)$  with  $h_1(t) \equiv 0$ ) at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly synchronizable. This finishes the proof.

**Remark 7.2.3** As  $t \ge T$ , the synchronizable state  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t, x)$  defined by (7.2.1) satisfies the following wave equation and boundary conditions:

$$\int \frac{\partial^2 \tilde{\tilde{u}}}{\partial t^2} - \frac{\partial^2 \tilde{\tilde{u}}}{\partial x^2} + \tilde{\tilde{a}} \tilde{\tilde{u}} = 0, \qquad (7.2.30)$$

$$\begin{cases} v_{t}^{T} = 0 \\ x = 0 : \tilde{\tilde{u}}_{x} - \tilde{\tilde{c}}\tilde{\tilde{u}}_{t} = 0, \\ \tilde{\tilde{u}}_{x} = 0 \\ \tilde{\tilde{u}}_{x} = 0, \end{cases}$$
(7.2.31)

$$\mathbf{l} x = L : \tilde{\tilde{u}}_x + \tilde{\tilde{c}}\tilde{\tilde{u}}_t = 0, \tag{7.2.32}$$

where  $\tilde{\tilde{a}}$ ,  $\tilde{\tilde{c}}$  and  $\tilde{\tilde{\tilde{c}}}$  are given by (7.2.2)–(7.2.4), respectively. Noting (7.2.27) and (7.2.29), by Lemma 7.1.1, this forward mixed system (7.2.30)-(7.2.32) is also well-

posed. Hence, if we know the initial state of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t, x)$  at the moment t = T:

$$t = T: \quad \widetilde{\widetilde{u}} = \varphi(x), \quad \widetilde{\widetilde{u}}_t = \psi(x), \quad 0 \le x \le L, \quad (7.2.33)$$

we can completely determine the evolution of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  with respect to t.

**Remark 7.2.4** Under assumptions (7.2.2)-(7.2.4), in order to realize the exact boundary synchronization, the (N-1)  $C^1$  boundary controls in H(t) can be chosen in infinitely many ways, therefore, the initial state  $(\varphi, \psi)$  in (7.2.33) will be not unique. Moreover, it can be shown that under additional assumptions  $\tilde{\tilde{c}} \neq 1$  and  $\tilde{\tilde{c}} \neq 1$ , the attainable set of all possible initial data of synchronizable state is the whole space  $C^2[0, L] \times C^1[0, L]$  associated with the conditions of  $C^2$  compatibility at (t, x) = (T, 0)and (T, L), respectively, with boundary conditions (7.2.31)-(7.2.32).

To illustrate this, for any given  $(\varphi, \psi) \in C^2[0, L] \times [C^1[0, L]$  satisfying the conditions of  $C^2$  compatibility with the boundary conditions (7.2.31)-(7.2.32) at (t, x) = (T, 0) and (T, L), respectively, noting  $\tilde{\tilde{c}} \neq 1$  and  $\tilde{\tilde{c}} \neq 1$ , by Remark 1.2, we can solve the backward mixed initial-boundary value problem (7.2.30)-(7.2.33) to get a unique  $C^2$  solution u = u(t, x) and then its initial value at t = 0:

$$t = 0: \quad u = \widehat{\varphi}(x), \quad u_t = \widehat{\psi}(x). \tag{7.2.34}$$

Then, by assumptions (7.2.2)-(7.2.4), it is easy to see that

$$U(t,x) = (u, u, \cdots, u)^{T}(t,x)$$
(7.2.35)

is the C<sup>2</sup> solution on the domain  $R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\}$  to the original system (7.1.1)-(7.1.3) with the null control  $H \equiv 0$  and the initial condition

$$t = 0: U = \widehat{\Phi}(x) = (\widehat{\varphi}(x), \widehat{\varphi}(x), \cdots, \widehat{\varphi}(x))^{T}, U_{t} = \widehat{\Psi}(x) = (\widehat{\psi}(x), \widehat{\psi}(x), \cdots, \widehat{\psi}(x))^{T}.$$
(7.2.36)

Then, by solving the mixed problem (7.1.1)-(7.1.3) and (7.2.36) with null boundary controls, we can reach any given synchronizable state  $(\varphi, \psi)$  at the moment t = T (see also [79], [81]).

## 7.3 Exact boundary synchronization by groups

Similarly to [81], the concept of exact boundary synchronization can be generalized to the exact boundary synchronization by groups. Without loss of generality, we consider here the exact boundary synchronization by 2-groups. For this purpose, setting  $U = \begin{pmatrix} U^{(1)} \\ U^{(2)} \end{pmatrix}$  with  $U^{(1)} = (u_1, \cdots, u_m)^T$  and  $U^{(2)} = (u_{m+1}, \cdots, u_N)^T$ , we look for some boundary controls in H(t), such that  $U^{(1)}$  and  $U^{(2)}$  are independently synchronizable.

**Definition 7.3.1** If there exists T > 0 such that for any given initial state  $(\Phi, \Psi)$ , we can find some  $C^1$  boundary controls with support on [0,T] in H(t), such that the corresponding mixed initial-boundary value problem (7.1.1)-(7.1.4) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  on  $t \ge 0$ , and as  $t \ge T$  we have

$$u_1(t,x) \equiv \dots \equiv u_m(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}_1(t,x), \quad 0 \le x \le L,$$
(7.3.1)

$$u_{m+1}(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{u}_2(t,x), \quad 0 \le x \le L,$$
(7.3.2)

then the coupled system (7.1.1)-(7.1.3) is said to possess the exact boundary synchronization by 2-groups or is said to be exactly synchronizable by 2-groups, and  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t,x) = \begin{pmatrix} \tilde{\widetilde{u}}_1(t,x) \\ \tilde{\widetilde{u}}_2(t,x) \end{pmatrix}$  is called the corresponding synchronizable state by 2groups.

Obviously, if the coupled system possesses the exact boundary synchronization, then it must be exactly synchronizable by 2-groups. However, this trivial situation should be excluded in advance. For this purpose we need to impose suitable assumptions on the synchronizable states by 2-groups  $\tilde{\tilde{u}}_1(t,x)$  and  $\tilde{\tilde{u}}_2(t,x)$  (see also [79]-[81]). We have

**Lemma 7.3.1** Suppose that the coupled system (7.1.1)-(7.1.3) is exactly synchronizable by 2-groups. Suppose furthermore that at least for an initial state  $(\Phi, \Psi)$  the synchronizable states by 2-groups  $\tilde{\tilde{u}}_1(t, x)$  and  $\tilde{\tilde{u}}_2(t, x)$  are linearly independent. Then the coupling matrix  $A = (a_{ij})$  in system (7.1.1) of wave equations should satisfy the following necessary conditions:

$$\sum_{j=1}^{m} a_{ij} \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\widetilde{a}}_{11} & (i=1,\cdots,m), \\ \widetilde{\widetilde{a}}_{21} & (i=m+1,\cdots,N), \end{cases}$$
(7.3.3)

$$\sum_{j=m+1}^{N} a_{ij} \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\widetilde{a}}_{12} & (i=1,\cdots,m), \\ \widetilde{\widetilde{a}}_{22} & (i=m+1,\cdots,N), \end{cases}$$
(7.3.4)

where  $\tilde{\tilde{a}}_{11}$  and  $\tilde{\tilde{a}}_{12}$  are constants independent of  $i = 1, \dots, m$ , while,  $\tilde{\tilde{a}}_{21}$  and  $\tilde{\tilde{a}}_{22}$  are constants independent of  $i = m + 1, \dots, N$ .

For the coupling matrices C and  $\overline{C}$  in boundary conditions (7.1.2)-(7.1.3), we can not get the corresponding necessary conditions under the assumptions of Lemma 7.3.1 (unless we assume that at least for an initial state  $(\Phi, \Psi)$ ,  $\tilde{\tilde{u}}_{1t}(t, x)$  and  $\tilde{\tilde{u}}_{2t}(t, x)$ 

are linearly independent at x = 0 and at x = L, respectively). However, based on Lemma 7.3.1, similarly to [81], we impose the following assumptions:

$$\sum_{j=1}^{m} c_{ij}(\text{resp. } \bar{c}_{ij}) \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\tilde{c}}_{11}(\text{resp. } \widetilde{\tilde{c}}_{11}) & (i = 1, \cdots, m), \\ \widetilde{\tilde{c}}_{21}(\text{resp. } \widetilde{\tilde{c}}_{21}) & (i = m + 1, \cdots, N), \end{cases}$$
(7.3.5)

$$\sum_{j=m+1}^{N} c_{ij}(\text{resp. } \bar{c}_{ij}) \stackrel{\text{def.}}{=} \begin{cases} \widetilde{\tilde{c}}_{12}(\text{resp. } \widetilde{\tilde{c}}_{12}) & (i=1,\cdots,m), \\ \widetilde{\tilde{c}}_{22}(\text{resp. } \widetilde{\tilde{c}}_{22}) & (i=m+1,\cdots,N), \end{cases}$$
(7.3.6)

where  $\widetilde{\widetilde{c}}_{11}$  (resp.  $\widetilde{\widetilde{c}}_{11}$ ) and  $\widetilde{\widetilde{c}}_{12}$  (resp.  $\widetilde{\widetilde{c}}_{12}$ ) are constants independent of  $i = 1, \cdots, m$ , while,  $\widetilde{\widetilde{c}}_{21}$  (resp.  $\widetilde{\widetilde{c}}_{21}$ ) and  $\widetilde{\widetilde{c}}_{22}$  (resp.  $\widetilde{\widetilde{c}}_{22}$ ) are constants independent of  $i = m+1, \cdots, N$ .

**Theorem 7.3.1** Under assumptions (7.1.5)-(7.1.6), suppose furthermore that (7.3.3) -(7.3.6) hold and T > 0 satisfies (7.2.5). For any given initial state  $(\Phi, \Psi)$ , there exist (N-2)  $C^1$  boundary controls with support on [0,T] in H(t) (for instance, take  $h_2(t), \dots, h_m(t), h_{m+2}(t), \dots, h_N(t)$  with  $h_1(t) \equiv h_{m+1}(t) \equiv 0$ ), such that the coupled system (7.1.1)-(7.1.3) possesses the exact boundary synchronization by 2-groups.

**Proof** Let

$$\begin{cases} w_i = u_{i+1} - u_i \quad (i = 1, \cdots, m - 1), \\ w_i = u_{i+2} - u_{i+1} \quad (i = m, \cdots, N - 2). \end{cases}$$
(7.3.7)

Similarly to the proof of Theorem 7.2.1 (see also [79],[81]), under assumptions (7.3.3)–(7.3.4), the original system (7.1.1) for the variable U can be reduced to a self-closing system (7.2.7) of the same kind for the variable  $W = (w_1, \dots, w_{N-2})^T$ , where  $\tilde{A} = (\tilde{a}_{ij})$  is an  $(N-2) \times (N-2)$  matrix with the entries

$$\widetilde{a}_{ij} = \begin{cases} \sum_{p=j+1}^{m} (a_{i+1,p} - a_{ip}) = \sum_{p=1}^{j} (a_{ip} - a_{i+1,p}), & i = 1, \cdots, N-2; \ j = 1, \cdots, m-1, \\ \sum_{p=j+2}^{N} (a_{i+1,p} - a_{ip}) = \sum_{p=m+1}^{j+1} (a_{ip} - a_{i+1,p}), & i = 1, \cdots, N-2; \ j = m, \cdots, N-2 \end{cases}$$
(7.3.8)

Similarly, under assumptions (7.3.5)-(7.3.6), the original boundary conditions (7.1.2)-(7.1.3) for the variable U can be reduced to self-closing boundary conditions (7.2.9)-(7.2.10) of the same kind for the variable W, in which the coupling matrices  $\widetilde{C} = (\widetilde{c}_{ij})$  and  $\widetilde{\overline{C}} = (\widetilde{c}_{ij})$  are  $(N-2) \times (N-2)$  matrices reduced from  $C = (c_{ij})$  and

 $\overline{C} = (\overline{c}_{ij})$  in the following way:

$$\widetilde{c}_{ij} = \begin{cases} \sum_{p=j+1}^{m} (c_{i+1,p} - c_{ip}) = \sum_{p=1}^{j} (c_{ip} - c_{i+1,p}), & i = 1, \cdots, N-2; \ j = 1, \cdots, m-1, \\ \sum_{p=j+2}^{N} (c_{i+1,p} - c_{ip}) = \sum_{p=m+1}^{j+1} (c_{ip} - c_{i+1,p}), & i = 1, \cdots, N-2; \ j = m, \cdots, N-2, \end{cases}$$
(7.3.9)

$$\widetilde{\overline{c}}_{ij} = \begin{cases} \sum_{p=j+1}^{m} (\overline{c}_{i+1,p} - \overline{c}_{ip}) = \sum_{p=1}^{j} (\overline{c}_{ip} - \overline{c}_{i+1,p}), & i = 1, \cdots, N-2; \ j = 1, \cdots, m-1, \\ \sum_{p=j+2}^{N} (\overline{c}_{i+1,p} - \overline{c}_{ip}) = \sum_{p=m+1}^{j+1} (\overline{c}_{ip} - \overline{c}_{i+1,p}), & i = 1, \cdots, N-2; \ j = m, \cdots, N-2 \end{cases}$$

$$(7.3.10)$$

Moreover,

$$\widetilde{H}(t) = (\widetilde{h}_1(t), \cdots, \widetilde{h}_{N-2}(t))$$
(7.3.11)

with

$$\widetilde{h}_{i}(t) = \begin{cases} h_{i+1}(t) - h_{i}(t), & i = 1, \cdots, m - 1, \\ h_{i+2}(t) - h_{i+1}(t), & i = m, \cdots, N - 2. \end{cases}$$
(7.3.12)

**.** .

The initial condition (7.2.15) of W is given by

$$\widetilde{\Phi}(x) = (\widetilde{\varphi}_1(x), \cdots, \widetilde{\varphi}_{N-2}(x))^T, \quad \widetilde{\Psi}(x) = (\widetilde{\psi}_1(x), \cdots, \widetilde{\psi}_{N-2}(x))^T$$
(7.3.13)

with

$$\widetilde{\varphi}_{i}(x) = \begin{cases} \varphi_{i+1}(x) - \varphi_{i}(x), & i = 1, \cdots, m - 1, \\ \varphi_{i+2}(x) - \varphi_{i+1}(x), & i = m, \cdots, N - 2, \end{cases}$$

$$\widetilde{\psi}_{i}(x) = \begin{cases} \psi_{i+1}(x) - \psi_{i}(x), & i = 1, \cdots, m - 1, \\ \psi_{i+2}(x) - \psi_{i+1}(x), & i = m, \cdots, N - 2. \end{cases}$$
(7.3.14)

In this situation, in order to guarantee the well-posedness and the exact boundary null controllability for the reduced system of W, we should also verify the corresponding conditions (7.1.5)-(7.1.6) in Lemma 7.1.1 and Lemma 7.1.2. Similarly to § 7.2, under assumptions (7.3.5)-(7.3.6) we will prove

$$det(C+I_N) = det(\widetilde{\widetilde{C}}+I_2)det(\widetilde{C}+I_{N-2}), \qquad (7.3.15)$$

where  $\widetilde{\widetilde{C}} = (\widetilde{\widetilde{c}}_{ij})$  given by (7.3.5)-(7.3.6) is a 2 × 2 matrix.

In order to get (7.3.15), let

$$C = \left(\frac{C_{11} \mid C_{12}}{C_{21} \mid C_{22}}\right),\tag{7.3.16}$$

where, by(7.3.5)-(7.3.6), the values of the sum of every row of partitioned matrices  $C_{ij}(1 \le i, j \le 2)$  are  $\tilde{\tilde{c}}_{ij}(1 \le i, j \le 2)$ , respectively.

By elementary transformation, similarly to (7.2.21), it is easy to get

where  $x_{11}$  and  $x_{21}$  are some row vectors of order (m-1),  $x_{12}$  and  $x_{22}$  are some row vectors of order (N-m-1). Deleting the m-th and N-th rows and the m-th and N-th columns of the matrix on the right-hand side of (7.3.17), we get

$$\widetilde{C} = \begin{pmatrix} \widetilde{C}_{11} & \widetilde{C}_{12} \\ \widetilde{C}_{21} & \widetilde{C}_{22} \end{pmatrix}.$$
(7.3.18)

Noting (7.2.24), we have

Taking the determinants of both sides of (7.3.19), by Laplace expansion theorem, we immediately get (7.3.15).

Thus, under assumption (7.1.5) we have

$$\det(C + I_{N-2}) \neq 0 \tag{7.3.20}$$

and

$$\det(\widetilde{C} + I_2) \neq 0. \tag{7.3.21}$$

Similarly, it follows from (7.3.5)-(7.3.6) and (7.1.6) that

$$\det(\widetilde{\overline{C}} + I_{N-2}) \neq 0 \tag{7.3.22}$$

and

$$\det(\tilde{\overline{C}} + I_2) \neq 0, \tag{7.3.23}$$

where  $\tilde{\overline{C}} = (\tilde{\overline{c}}_{ij})$  given by (7.3.5)-(7.3.6) is a 2 × 2 matrix.

Therefore, under assumptions (7.1.5)-(7.1.6), according to Lemma 7.1.1, the reduced mixed initial-boundary problem (7.2.7),(7.2.9)-(7.2.10) and (7.2.15) for the variable W is always well-posed, then using Lemma 7.1.2, by means of  $(N-2) C^1$ boundary controls  $\widetilde{H}(t)$  (for instance, take  $h_2(t), \dots, h_m(t), h_{m+2}(t), \dots, h_N(t)$  with  $h_1(t) \equiv h_{m+1}(t) \equiv 0$ ) on [0,T] at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly synchronizable by 2-groups. This completes the proof.

**Remark 7.3.1** As  $t \geq T$ , the synchronizable state by 2-groups  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t,x) = (\tilde{\widetilde{u}}_{2(t,x)}^{(t,x)})$  satisfies the following coupled system of wave equations

$$\frac{\partial^2 \widetilde{\widetilde{U}}}{\partial t^2} - \frac{\partial^2 \widetilde{\widetilde{U}}}{\partial x^2} + \widetilde{\widetilde{A}} \widetilde{\widetilde{U}} = 0, \qquad (7.3.24)$$

$$x = 0: \quad \widetilde{\widetilde{U}}_x - \widetilde{\widetilde{C}}\widetilde{\widetilde{U}}_t = 0, \tag{7.3.25}$$

$$x = L: \quad \widetilde{\widetilde{U}}_x + \widetilde{\widetilde{C}}\widetilde{\widetilde{U}}_t = 0, \qquad (7.3.26)$$

where

$$\widetilde{\widetilde{A}} = \begin{pmatrix} \widetilde{\widetilde{a}}_{11} & \widetilde{\widetilde{a}}_{12} \\ \widetilde{\widetilde{a}}_{21} & \widetilde{\widetilde{a}}_{22} \end{pmatrix},$$
(7.3.27)

and  $\tilde{\tilde{a}}_{11}$ ,  $\tilde{\tilde{a}}_{12}$ ,  $\tilde{\tilde{a}}_{21}$  and  $\tilde{\tilde{a}}_{21}$  are given by (7.3.3)–(7.3.4). Noting (7.3.21) and (7.3.23), by Lemma 1.1 the forward mixed system (7.3.24)-(7.3.26) is well-posed. Hence, if we

know the initial state of  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t,x)$  at the moment t = T:

$$t = T: \quad \widetilde{\widetilde{U}} = (\varphi^{(1)}(x), \varphi^{(2)}(x))^T, \quad \widetilde{\widetilde{U}}_t = (\psi^{(1)}(x), \psi^{(2)}(x))^T, \quad 0 \le x \le L, \quad (7.3.28)$$

then the evolution of  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t,x)$  with respect to t can be completely determined.

**Remark 7.3.2** Under assumptions (7.3.3)-(7.3.6), in order to realize the exact boundary synchronization by 2-groups, one has infinitively many ways to choose (N - 2)  $C^1$  boundary controls at the end x = 0, then the initial synchronizable state in (7.3.28) is not unique. Moreover, similarly to Remark 2.2, suppose furthermore that

$$det(\tilde{\widetilde{C}} - I_2) \neq 0 \tag{7.3.29}$$

and

$$det(\tilde{\overline{C}} - I_2) \neq 0, \tag{7.3.30}$$

namely, 1 is not an eigenvalue of both  $\tilde{\widetilde{C}}$  and  $\tilde{\overline{C}}$ , any given state  $(\varphi^{(1)}, \varphi^{(2)})^T$  and  $(\psi^{(1)}, \psi^{(2)})^T$  in  $(C^2[0, L])^2 \times (C^1[0, L])^2$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (7.3.25)-(7.3.26), belongs to the attainable set of the initial data of synchronizable state by 2-groups.

Thus, at least for some initial states  $(\Phi, \Psi)$ , the synchronizable states by 2-groups  $\tilde{\widetilde{u}}_1(t,x)$  and  $\tilde{\widetilde{u}}_2(t,x)$  are linearly independent on t = T then for  $t \ge T$ . It shows that the corresponding requirement in Lemma 3.1 is reasonable.

**Remark 7.3.3** The exact boundary synchronization by  $k(k \ge 2)$ -groups can be treated in a similar way.

In order to realize the exact boundary synchronization by 2-groups, the number of boundary controls is equal to (N-2), while, in order to realize the exact boundary synchronization by k-groups, the number of boundary controls is equal to (N-k).

## 7.4 Exact boundary null controllability and synchronization by groups

Similarly to [79],[81], under some hypotheses on the coupled system, we can divide the state variables into two groups  $U^{(1)} = (u_1, \dots, u_m)^T$  and  $U^{(2)} = (u_{m+1}, \dots, u_N)^T$ such that  $U^{(1)}$  is exactly null controllable, while  $U^{(2)}$  is exactly synchronizable. **Definition 7.4.1** If there exists T > 0 such that for any given initial state  $(\Phi, \Psi)$ , we can find some boundary controls with support on [0,T] in H(t), such that the corresponding mixed initial-boundary value problem (7.1.1)-(7.1.4) admits a unique  $C^2$  solution  $U = U(t, x) = (u_1(t, x), \dots, u_N(t, x))^T$  on  $t \ge 0$ , and as  $t \ge T$  we have

$$u_1(t,x) \equiv \dots \equiv u_m(t,x) \equiv 0, \quad 0 \le x \le L, \tag{7.4.1}$$

$$u_{m+1}(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L,$$
(7.4.2)

then the coupled system (7.1.1)-(7.1.3) is said to possess the exact boundary null controllability and synchronization by 2-groups or is said to be exactly null controllable and synchronizable by 2- groups, and  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  is called to be the partially synchronizable state.

Obviously, when the coupled system (7.1.1)-(7.1.3) possesses the exact null controllability, it must satisfy the requirement of Definition 7.4.1. This situation should be excluded in advance. Similarly to [81], we have

**Lemma 7.4.1** Under assumptions (7.1.5)-(7.1.6), suppose furthermore that the coupled system (7.1.1)-(7.1.3) is exactly null controllable and synchronizable by 2-groups, but not exactly null controllable, then the coupling matrix A must satisfy the following necessary conditions:

$$\sum_{j=m+1}^{N} a_{ij} = 0 \quad (i = 1, \cdots, m),$$
(7.4.3)

$$\sum_{j=m+1}^{N} a_{ij} \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}} \quad (i=m+1,\cdots,N),$$
(7.4.4)

where  $\tilde{\tilde{a}}$  is a constant independent of  $i = m + 1, \dots, N$ .

Different from the situation in [81], the conclusion of Lemma 7.4.1 is no longer valid in general for the coupling matrices C and  $\overline{C}$  in the coupled dissipative boundary conditions (7.1.2)-(7.1.3). However, based on Lemma 7.4.1, we impose the following similar assumptions:

$$\sum_{j=m+1}^{N} c_{ij}(\text{resp. } \bar{c}_{ij}) = 0 \quad (i = 1, \cdots, m),$$
(7.4.5)

$$\sum_{j=m+1}^{N} c_{ij}(\text{resp. } \bar{c}_{ij}) \stackrel{\text{def.}}{=} \widetilde{\widetilde{c}}(\text{resp. } \widetilde{\widetilde{c}}) \quad (i=m+1,\cdots,N),$$
(7.4.6)

where  $\tilde{\tilde{c}}$  and  $\tilde{\tilde{c}}$  are constants independent of  $i = m + 1, \cdots, N$ .

**Remark 7.4.1** Similarly to Remark 7.2.1, if we suppose furthermore that the constant  $\tilde{\tilde{a}}$  given by (7.4.4) is positive, it is easy to get the necessity of (7.4.5)-(7.4.6) for the exact boundary null controllability and synchronization by 2-groups.

**Theorem 7.4.1** Under assumptions (7.1.5)-(7.1.6), suppose furthermore that (7.4.3)-(7.4.6) hold and T > 0 satisfies (7.2.5). For any given initial state  $(\Phi, \Psi)$ , we can find (N-1)  $C^1$  boundary controls with support on [0,T] in H(t) (for instance, take  $h_1(t), \dots, h_m(t), h_{m+2}(t), \dots, h_N(t)$  with  $h_{m+1}(t) \equiv 0$ ), such that the coupled system (7.1.1)-(7.1.3) possesses the exact boundary null controllability and synchronization by 2-groups.

**Proof** Let

$$\begin{cases} w_i = u_i \quad (i = 1, \cdots, m), \\ w_i = u_{i+1} - u_i \quad (i = m + 1, \cdots, N - 1). \end{cases}$$
(7.4.7)

Similarly to the proof of Theorem 7.2.1 and Theorem 7.3.1 (see [79]-[81]), under assumptions (7.4.3)–(7.4.4), the original system (7.1.1) for the variable U can be reduced to a self-closing system (7.2.7) of the same kind for the variable  $W = (w_1, \dots, w_{N-1})^T$ , where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-1) \times (N-1)$  matrix with

$$\widetilde{a}_{ij} = \begin{cases} a_{ij}, \quad j = 1, \cdots, m, \\ \sum_{p=j+1}^{N} a_{ip} = -\sum_{p=m+1}^{j} a_{ip}, \quad j = m+1, \cdots, N-1, \end{cases} \quad \text{for } i = 1, \cdots, m,$$

$$(7.4.8)$$

$$\widetilde{a}_{ij} = \begin{cases} a_{i+1,j} - a_{ij}, \quad j = 1, \cdots, m, & \text{for } i = m+1, \cdots, N-1. \\ \sum_{p=j+1}^{N} (a_{i+1,p} - a_{ip}) = \sum_{p=m+1}^{j} (a_{ip} - a_{i+1,p}), \quad j = m+1, \cdots, N-1, \end{cases}$$
(7.4.9)

Similarly, by (7.4.5)-(7.4.6), the original boundary conditions (7.1.2)-(7.1.3) for the variable U can be reduced, respectively, to a self-closing boundary conditions (7.2.9)-(7.2.10) of the same kind for the variable W, in which the  $(N-1) \times (N-1)$ coupling matrices  $\tilde{C} = (\tilde{c}_{ij})$  and  $\tilde{C} = (\tilde{c}_{ij})$  are reduced from  $C = (c_{ij})$  and  $\overline{C} = (\bar{c}_{ij})$ in the following way:

$$\widetilde{c}_{ij} = \begin{cases} c_{ij}, \quad j = 1, \cdots, m, \\ \sum_{p=j+1}^{N} c_{ip} = -\sum_{p=m+1}^{j} c_{ip}, \quad j = m+1, \cdots, N-1, \end{cases} \quad \text{for } i = 1, \cdots, m,$$
(7.4.10)

$$\widetilde{c}_{ij} = \begin{cases} c_{i+1,j} - c_{ij}, & j = 1, \cdots, m, \\ \sum_{p=j+1}^{N} (c_{i+1,p} - b_{ip}) = \sum_{p=m+1}^{j} (c_{ip} - c_{i+1,p}), & j = m+1, \cdots, N-1, \end{cases}$$
(7.4.11)

and similar expressions for  $\tilde{c}_{ij}$   $(i, j = 1, \dots, N-1)$ . Moreover,  $\tilde{H}(t)$  is given by (7.2.13) with

$$\widetilde{h}_{i}(t) = \begin{cases} h_{i}(t), & i = 1, \cdots, m, \\ h_{i+1}(t) - h_{i}(t), & i = m + 1, \cdots, N - 1. \end{cases}$$
(7.4.12)

The initial condition (7.2.15) of W is given by

$$\widetilde{\varphi}_{i}(x) = \begin{cases} \varphi_{i}(x), & i = 1, \cdots, m, \\ \varphi_{i+1}(x) - \varphi_{i}(x), & i = m+1, \cdots, N-1, \\ \widetilde{\psi}_{i}(x) = \begin{cases} \psi_{i}(x), & i = 1, \cdots, m, \\ \psi_{i+1}(x) - \psi_{i}(x), & i = m+1, \cdots, N-1. \end{cases}$$
(7.4.13)

By (7.4.5)-(7.4.6), we can prove

$$det(C+I_N) = (\tilde{\tilde{c}}+1)det(\tilde{C}+I_{N-1}).$$
(7.4.14)

In fact, let

$$C = \left(\frac{C_{11} \mid C_{12}}{C_{21} \mid C_{22}}\right),\tag{7.4.15}$$

in which the sums of every row of partitioned matrices  $C_{12}$  and  $C_{22}$  are 0 and  $\tilde{\tilde{c}}$ , respectively.

By elementary transformation, using (7.2.21) and noting (7.4.10)-(7.4.11), we have

$$\begin{pmatrix} -I_m \\ X_{N-m} \end{pmatrix} C \begin{pmatrix} -I_m \\ Y_{N-m} \end{pmatrix}$$

$$= \begin{pmatrix} C_{11} & -C_{12}Y_{N-m} \\ -X_{N-m}C_{21} & X_{N-m}C_{22}Y_{N-m} \end{pmatrix} = \begin{pmatrix} \widetilde{C} & \mathbf{0} \\ x & \widetilde{\tilde{c}} \end{pmatrix},$$
(7.4.16)

where x is a row vector of order (N-1). Thus,

$$\begin{pmatrix} -I_m \\ X_{N-m} \end{pmatrix} (C+I_N) \begin{pmatrix} -I_m \\ Y_{N-m} \end{pmatrix}$$
$$= \begin{pmatrix} -I_m \\ X_{N-m} \end{pmatrix} C \begin{pmatrix} -I_m \\ Y_{N-m} \end{pmatrix} + I_N$$
$$= \begin{pmatrix} \tilde{C}+I_{N-1} & \mathbf{0} \\ x & \tilde{c}+1 \end{pmatrix}.$$
(7.4.17)

Taking the determinants of both sides of (7.4.17), we immediately get (7.4.14).

Thus, by (7.1.5) we have

$$\det(\widetilde{C} + I_{N-1}) \neq 0 \tag{7.4.18}$$

and

$$\tilde{\tilde{c}} \neq -1. \tag{7.4.19}$$

Similarly, by (7.1.6) we can get

$$\det(\widetilde{\overline{C}} + I_{N-1}) \neq 0 \tag{7.4.20}$$

and

$$\tilde{\tilde{c}} \neq -1. \tag{7.4.21}$$

Therefore, under assumptions (7.1.5)-(7.1.6), by Lemma 7.1.1 the mixed initialboundary problem (7.2.7),(7.2.9)-(7.2.10) and (7.2.15) for the variable W is always well-posed, then using Lemma 7.1.2, by means of (N-1)  $C^1$  boundary controls  $\widetilde{H}(t)$ (for instance, take  $h_1(t), \dots, h_m(t), h_{m+1}(t), \dots, h_N(t)$  with  $h_{m+1}(t) \equiv 0$ ) at the end x = 0, the variable W is exactly null controllable, then the variable U is exactly null controllable and synchronizable by 2-groups. This proves Theorem 7.4.1.

**Remark 7.4.2** As  $t \ge T$ , the partially synchronizable state  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  satisfies the wave system (7.2.30)-(7.2.32), where  $\tilde{\widetilde{a}}$ ,  $\tilde{\widetilde{c}}$  and  $\tilde{\widetilde{c}}$  are given by (7.4.4) and (7.4.6). Noting (7.4.19) and (7.4.21), this forward mixed system is also well-posed. Hence, if we know the initial condition (7.2.33) of  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  at the moment t = T, then the evolution of the partially synchronizable state  $\tilde{\widetilde{u}} = \tilde{\widetilde{u}}(t,x)$  with respect to t can be completely determined.

Moreover, suppose furthermore that  $\tilde{\tilde{c}} \neq 1$  and  $\tilde{\tilde{c}} \neq 1$ , any given state  $(\varphi, \psi)$  in  $C^2[0, L] \times C^1[0, L]$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) =

(T,0) and (T,L), respectively, with boundary conditions (7.2.31)-(7.2.32), belongs to the attainable set of the initial data of partially synchronizable state.

### 7.5 Generalized exact boundary synchronization

We now consider the problem of synchronization from a more general mathematical point of view.

Let

$$w_i = \sum_{j=1}^{N} \theta_{ij} u_j \quad (i = 1, \cdots, M)$$
 (7.5.1)

or

$$W = (H)U, \tag{7.5.2}$$

where  $W = (w_1, \cdots, w_M)^T$ ,  $M \leq N - 1$ , and  $(H) = (\theta_{ij})_{M \times N}$  is full-rank.

Suppose that for any given initial state  $(\Phi, \Psi)$ , the mixed initial-boundary value problem (7.1.1)-(7.1.4) for the variable U can be reduced to a self-closing mixed initialboundary values problem of the same kind for the variable W, then, if T > 2L, we can use Lemma 7.1.2 to find  $M C^1$  boundary controls in H(t) with support on [0, T], such that as  $t \ge T$  we have

$$W(t,x) \equiv 0, \quad 0 \le x \le L, \tag{7.5.3}$$

i.e.,

$$(\mathbf{H})U(t,x) \equiv 0, \quad 0 \le x \le L. \tag{7.5.4}$$

Thus, the coupled system (7.1.1)-(7.1.3) is said to have the **generalized exact** boundary synchronization associated with (H) and U = U(t, x) is the corresponding generalized synchronizable state.

As in [81], under hypotheses (7.1.5)-(7.1.6), this definition covers all the cases discussed in the previous sections.

A natural question is under what conditions the original coupled system for the variable U can be reduced to a self-closing coupled system of the same type for the variable W by means of (7.5.2).

**Theorem 7.5.1** For any given coupling  $N \times N$  matrices A, C and  $\overline{C}$ , if there

exist a full-rank matrix  $(\mathbf{H}) = (\theta_{ij})_{M \times N}$  and  $M \times M$  matrices  $\widetilde{A}$ ,  $\widetilde{C}$  and  $\widetilde{\overline{C}}$  such that

$$(\mathbf{H})A = \widetilde{A}(\mathbf{H}), \tag{7.5.5}$$

$$(\mathbf{H})C = \widetilde{C}(\mathbf{H}), \tag{7.5.6}$$

$$(\mathbf{H})\overline{C} = \overline{C}(\mathbf{H}), \tag{7.5.7}$$

then, by (7.5.2) we can get a self-closing coupled system (7.2.7) and (7.2.9)-(7.2.10) (in which the numbers of both wave equations and the boundary conditions on two sides are all equal to M) with  $\tilde{H}(t) = \bigoplus H(t)$  for the variable W. Thus, we can realize the generalized exact boundary synchronization associated with  $\bigoplus$  for the variable U by  $M C^1$  boundary controls on x = 0. Moreover,

$$Spectrum(\widetilde{C}) \subseteq Spectrum(C),$$
 (7.5.8)

$$Spectrum(\overline{C}) \subseteq Spectrum(\overline{C}),$$
 (7.5.9)

where Spectrum(C) denotes the set of all the eigenvalues of matrix C, etc.

**Proof** Similarly to [81], here, we need only to prove (7.5.8), and (7.5.9) can be obtained in a similar way. In fact, for the full-rank matrix (H), by Gram-Schmidt process on its column vectors, there exists a nonsingular matrix  $P \in \mathbb{R}^{N \times N}$  such that

$$(\mathbf{H})P = (I_M, \mathbf{0}). \tag{7.5.10}$$

Let

$$P^{-1}CP \stackrel{\text{def.}}{=} \left( \begin{array}{cc} \widehat{C}_{11} & \widehat{C}_{12} \\ \widehat{C}_{21} & \widehat{C}_{22} \end{array} \right), \tag{7.5.11}$$

where  $\hat{C}_{11}$  and  $\hat{C}_{22}$  are matrices of order M and order (N-M), respectively. It is easy to see that

$$(\underbrace{\mathbf{H}}_{\mathcal{C}} CP = (\underbrace{\mathbf{H}}_{\mathcal{P}} P)(P^{-1}CP) = (\widehat{C}_{11}, \widehat{C}_{12}), \qquad (7.5.12)$$

$$\widetilde{C}(\mathbb{H})P = (\widetilde{C}, \mathbf{0}).$$
 (7.5.13)

Hence, by(7.5.6) we have

$$\widehat{C}_{11} = \widetilde{C}, \quad \widehat{C}_{12} = \mathbf{0}, \tag{7.5.14}$$

then, noting (7.5.11), we get (7.5.8). This ends the proof.

The next question is for any given coupling  $N \times N$  matrices A, C and  $\overline{C}$ , how to guarantee the existence of such  $M \times N$  matrix (H) and  $M \times M$  matrices  $\widetilde{A}$ ,  $\widetilde{C}$  and  $\widetilde{\overline{C}}$  in Theorem 7.5.1.

We first consider the case M = N - 1. Similarly to [81], we have

**Theorem 7.5.2** Suppose that M = N-1. For the coupled system (7.1.1)-(7.1.3) of wave equations, if the coupling matrices A, C and  $\overline{C}$  possess a common real right eigenvector  $X \in \mathbb{R}^N \setminus \{0\}$ :

$$AX = \widetilde{\widetilde{a}}X,\tag{7.5.15}$$

$$CX = \widetilde{c}X, \tag{7.5.16}$$

$$\overline{C}X = \widetilde{\overline{c}}X,\tag{7.5.17}$$

where  $\tilde{\tilde{a}}, \tilde{\tilde{c}}$  and  $\tilde{\tilde{c}}$  are corresponding eigenvalues, respectively, then there exist a  $(N-1) \times N$  full-rank matrix  $(H) = (\theta_{ij})$  and  $(N-1) \times (N-1)$  matrices  $\tilde{A}, \tilde{C}$  and  $\tilde{\overline{C}}$  such that (7.5.5)-(7.5.7) hold. Moreover, X constitutes a system of basic solutions to the linear algebraic system

(H)
$$Y = 0,$$
 (7.5.18)

and

$$Spectrum(C) = \left\{\tilde{\tilde{c}}\right\} + Spectrum(\tilde{C}),$$
 (7.5.19)

$$Spectrum(\overline{C}) = \left\{\widetilde{\widetilde{c}}\right\} + Spectrum(\widetilde{\overline{C}}).$$
 (7.5.20)

**Proof** Similarly, we need only to prove (7.5.19). In fact, for the full-rank matrix  $(\mathbb{H}) \in \mathbb{R}^{(N-1)\times N}$ , by Gram-Schmidt process on its row vectors, it is easy to see that there exists a matrix  $D = (d_1, \dots, d_{N-1})$ , in which  $d_k (1 \le k \le N-1)$  are column vectors of order N, such that

$$(\mathbf{H})D = I_{N-1}.$$
 (7.5.21)

Noting  $(\mathbb{H})X = 0$ , we can take  $Z \in \mathbb{R}^{N \times 1}$  such that

$$Z \in (Span\{d_1, \cdots, d_{N-1}\})^{\perp}$$
(7.5.22)

and

$$Z^T \cdot X = 1. \tag{7.5.23}$$

Then, for any given  $\lambda \in \mathbb{C}$ , noting (7.5.6), (7.5.16) and (7.5.18), we have

$$\begin{pmatrix} \begin{pmatrix} \mathbf{H} \\ Z^T \end{pmatrix} (C - \lambda I_N)(D, X) = \begin{pmatrix} \mathbf{H} CD & \mathbf{H} CX \\ Z^T CD & Z^T CX \end{pmatrix} - \lambda I_N = \begin{pmatrix} \tilde{C} - \lambda I_{N-1} & 0 \\ Z^T CD & \tilde{\tilde{c}} - \lambda \end{pmatrix}$$
(7.5.24)

Taking the determinants of both sides of (7.5.24), we immediately get (7.5.19). The proof of (7.5.20) is completely similar. This finishes the proof.

**Remark 7.5.1** Under assumptions (7.1.5)-(7.1.6), it follows from (7.5.19)-(7.5.20) that the reduced coupled system (7.2.7) and (7.2.9)-(7.2.10) for the variable W is always well-posed.

**Remark 7.5.2** As  $t \ge T$ , by (7.5.4) and (7.5.18), the generalized synchronizable state can be written as  $U(t,x) = \tilde{u}(t,x)X$ , in which  $X = (\alpha_1, \dots, \alpha_N)^T$  is a system of basic solutions to (7.5.18), and  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  satisfies the wave system (7.2.30)-(7.2.32). Under assumptions (7.1.5)-(7.1.6), noting that by (7.5.19)-(7.5.20) we have  $\tilde{\tilde{c}} \ne -1$ and  $\tilde{\tilde{c}} \ne -1$ , according to Lemma 7.1.1, if we know the initial condition (7.2.33) of  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  at the moment t = T, then the evolution of  $\tilde{\tilde{u}} = \tilde{\tilde{u}}(t,x)$  with respect to t can be completely determined.

Moreover, suppose furthermore that  $\tilde{\tilde{c}} \neq 1$  and  $\tilde{\tilde{c}} \neq 1$ , then any given state  $(\varphi, \psi)$  in  $C^2[0, L] \times C^1[0, L]$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (7.2.31)-(7.2.32), can be the initial data of a generalized synchronizable state.

**Remark 7.5.3** In the case of exact boundary synchronization, M = N - 1 and  $X = (1, \dots, 1)^T$  is a system of basic solutions to (7.5.18). The requirement that the coupling matrices A, C and  $\overline{C}$  possess a common right eigenvector X implies (7.2.2)–(7.2.4). Thus, Theorem 7.5.2 leads to the conclusion given in §7.2.

**Remark 7.5.4** In the case of exact boundary null controllability and synchronization by 2-groups, M = N - 1 and  $X = (\overbrace{0, \dots, 0}^{N-m}, \overbrace{1, \dots, 1}^{N-m})^T$  is a system of basic solutions to (7.5.18). The requirement that the coupling matrices A, C and  $\overline{C}$  possess a common right eigenvector X implies (7.4.3)–(7.4.6). Thus, the conclusion given in §7.4 follows from Theorem 7.5.2.

On the other hand, similarly to [81], we have

**Theorem 7.5.3** Suppose that M = N - 1, if the coupled system (7.1.1)-(7.1.3) possesses the generalized exact boundary synchronization associated with (H), but is not exactly boundary null controllable, then there exist a real right eigenvector  $X \in \mathbb{R}^N \setminus \{0\}$  and a real number  $\tilde{a}$  such that (7.5.15) holds.

For the case M = N - 2, we only consider the special situation that

$$(\underline{\mathbf{H}}) = \begin{pmatrix} (\underline{\mathbf{H}})_1 \\ (\underline{\mathbf{H}})_2 \end{pmatrix}_{(N-2) \times N},$$
(7.5.25)

where  $(\mathbb{H})_1$  is a full-rank  $(N_1 - 1) \times N_1$  matrix, while  $(\mathbb{H})_2$  is a full-rank  $(N_2 - 1) \times N_2$  matrix with  $N_1 + N_2 = N$ .

Let

$$A = \begin{pmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{pmatrix}, \quad \widetilde{A} = \begin{pmatrix} \widetilde{A}_{11} & \widetilde{A}_{12} \\ \widetilde{A}_{21} & \widetilde{A}_{22} \end{pmatrix} \text{ etc.}, \quad (7.5.26)$$

where  $A_{11}$  (resp.  $\widetilde{A}_{11}$ ) is an  $N_1 \times N_1$  (resp.  $(N_1-1) \times (N_1-1)$ ) matrix,  $A_{21}$  (resp.  $\widetilde{A}_{21}$ ) is an  $N_2 \times N_1$  (resp.  $(N_2-1) \times (N_1-1)$ ) matrix,  $A_{12}$  (resp.  $\widetilde{A}_{12}$ ) is an  $N_1 \times N_2$  (resp.  $(N_1-1) \times (N_2-1)$ ) matrix,  $A_{22}$  (resp.  $\widetilde{A}_{22}$ ) is an  $N_2 \times N_2$  (resp.  $(N_2-1) \times (N_2-1)$ ) matrix etc. Thus, (7.5.5)–(7.5.7) in Theorem 7.5.1 can be equivalently written as

$$(\mathbf{H})_{i}A_{ij} = \widetilde{A}_{ij}(\mathbf{H})_{j}, \quad i, j = 1, 2,$$

$$(7.5.27)$$

$$(\underline{\mathbf{H}}_{i}C_{ij} = \widetilde{C}_{ij}(\underline{\mathbf{H}}_{j}), \quad i, j = 1, 2,$$

$$(7.5.28)$$

$$(\mathbb{H})_i \overline{C}_{ij} = \overline{C}_{ij} (\mathbb{H})_j, \quad i, j = 1, 2.$$
(7.5.29)

Similarly to [81], we have

**Theorem 7.5.4** Suppose that M = N - 2. For the coupled system of wave equations (7.1.1)-(7.1.3), if there exist real vectors  $X_1 \in \mathbb{R}^{N_1} \setminus \{0\}$  and  $X_2 \in \mathbb{R}^{N_2} \setminus \{0\}$  and real numbers  $\tilde{\tilde{a}}_{ij}$ ,  $\tilde{\tilde{c}}_{ij}$  and  $\tilde{\tilde{c}}_{ij}(i, j = 1, 2)$  such that

$$A_{ij}X_j = \widetilde{a}_{ij}X_i, \quad i, j = 1, 2,$$
 (7.5.30)

$$C_{ij}X_j = \tilde{c}_{ij}X_i, \quad i, j = 1, 2,$$
 (7.5.31)

$$\overline{C}_{ij}X_j = \widetilde{\widetilde{c}}_{ij}X_i, \quad i, j = 1, 2,$$
(7.5.32)

then we can find an  $(N-2) \times N$  full rank matrix  $(H) = \begin{pmatrix} H_1 \\ H_2 \end{pmatrix}$  and  $(N-2) \times (N-2)$  matrices  $\widetilde{A}$ ,  $\widetilde{C}$  and  $\widetilde{\overline{C}}$  such that (7.5.27) - (7.5.29) hold, moreover,  $X_1$  (resp.  $X_2$ ) constitutes a system of basic solutions to the linear algebraic system

$$(\mathbb{H})_1 Y_1 = 0 \text{ (resp. } (\mathbb{H})_2 Y_2 = 0).$$
 (7.5.33)

In addition, let

$$\widetilde{\widetilde{C}} = \begin{pmatrix} \widetilde{\widetilde{c}}_{11} & \widetilde{\widetilde{c}}_{12} \\ \widetilde{\widetilde{c}}_{21} & \widetilde{\widetilde{c}}_{22} \end{pmatrix}, \quad \widetilde{\widetilde{C}} = \begin{pmatrix} \widetilde{\widetilde{c}}_{11} & \widetilde{\widetilde{c}}_{12} \\ \widetilde{\widetilde{c}}_{21} & \widetilde{\widetilde{c}}_{22} \end{pmatrix}.$$
(7.5.34)

We have

$$Spectrum(C) = Spectrum(\widetilde{C}) + Spectrum(\widetilde{\widetilde{C}}), \qquad (7.5.35)$$

$$Spectrum(\overline{C}) = Spectrum(\overline{C}) + Spectrum(\overline{C}).$$
 (7.5.36)

**Proof** We need only to prove (7.5.35). Similarly to the proof of Theorem 7.5.2, by Gram-Schmidt process on the row vectors of matrices  $(\mathbb{H}_1 \text{ and } (\mathbb{H}_2, \text{ respectively, it is easy to see that there exist matrices } D_1 = (d_1, \dots, d_{N_1-1}) \text{ and } D_2 = (d_{N_1}, \dots, d_{N-2}),$  in which  $d_k(1 \leq k \leq N-2)$  are column vectors, such that

$$(\widehat{\mathbf{H}}_1 D_1 = I_{N_1 - 1}, \tag{7.5.37}$$

$$(\mathbf{H})_2 D_2 = I_{N_2 - 1}. \tag{7.5.38}$$

Take  $Z_1 \in \mathbb{R}^{N_1}$  and  $Z_2 \in \mathbb{R}^{N_2}$  such that

$$Z_1 \in (Span\{d_1, \cdots, d_{N_1-1}\})^{\perp}, \ Z_2 \in (Span\{d_{N_1}, \cdots, d_{N-2}\})^{\perp},$$
(7.5.39)

$$Z_1^T \cdot X_1 = 1, \ Z_2^T \cdot X_2 = 1. \tag{7.5.40}$$

For any given  $\lambda \in \mathbb{C}$ , noting (7.5.28) and (7.5.31) we have

$$\begin{pmatrix} \begin{pmatrix} \left( \begin{array}{c} H \\ Z_{1}^{T} \\ \end{array} \right) \\ \begin{pmatrix} \left( \begin{array}{c} H \\ Z_{2}^{T} \\ \end{array} \right) \\ \begin{pmatrix} \left( \begin{array}{c} H \\ Z_{1}^{T} \\ \end{array} \right) \\ Z_{2}^{T} \\ Z_{2}^{T} \\ \end{array} \right) C_{11}(D_{1}, X_{1}) \\ \begin{pmatrix} \left( \begin{array}{c} H \\ Z_{1}^{T} \\ \end{array} \right) C_{12}(D_{2}, X_{2}) \\ Z_{2}^{T} \\ Z_{2}^{T} \\ \end{array} \right) C_{12}(D_{1}, X_{1}) \\ \begin{pmatrix} \left( \begin{array}{c} H \\ Z_{1}^{T} \\ \end{array} \right) \\ Z_{2}^{T} \\ Z_{2}^{T} \\ \end{array} \right) C_{22}(D_{2}, X_{2}) \\ C_{22}(D_{2}, X_{2}) \\ \end{pmatrix} - \lambda I_{N} \\ = \begin{pmatrix} \left( \begin{array}{c} \tilde{C}_{11} & \mathbf{0} & \tilde{C}_{12} & \mathbf{0} \\ \frac{* & \tilde{c}_{11}}{\tilde{C}_{21}} & \mathbf{0} & \tilde{C}_{22} \\ \frac{\tilde{C}_{21}}{\tilde{C}_{21}} & \mathbf{0} & \tilde{C}_{22} \\ \frac{* & \tilde{c}_{11} - \lambda}{\tilde{C}_{21}} & \frac{\tilde{C}_{12} & \mathbf{0}}{\tilde{C}_{22} - \lambda I_{N-m-1}} \\ \frac{* & \tilde{c}_{12}}{\tilde{C}_{21}} & \mathbf{0} \\ \frac{* & \tilde{c}_{21}}{\tilde{C}_{21}} & \mathbf{0} \\ \frac{\tilde{C}_{22} - \lambda I_{N-m-1}}{\tilde{C}_{22} - \lambda} \\ \end{pmatrix},$$

$$(7.5.41)$$

where "\*" denotes some unmarked values. Taking the determinants of both sides of (7.5.41), by Laplace expansion theorem, we immediately get (7.5.35). (7.5.36) can be obtained in a similar way. This ends the proof.

**Remark 7.5.5** Under assumptions (7.1.5)-(7.1.6), by (7.5.35)-(7.5.36) we can guarantee the well-posedness of the reduced system (7.2.7) and (7.2.9)-(7.2.10) for the variable W.

By (7.5.25) and noting that the rank of  $(\mathbb{H}_1 \text{ (resp. } \mathbb{H}_2) \text{ is } N_1 - 1 \text{ (resp. } N_2 - 1),$ the linear algebraic system (7.5.33) possesses a system of basic solutions  $X_1 =$   $(\alpha_1, \cdots, \alpha_{N_1})^T \in \mathbb{R}^{N_1} \setminus \{0\}$  (resp.  $X_2 = (\alpha_{N_1+1}, \cdots, \alpha_N)^T \in \mathbb{R}^{N_2} \setminus \{0\}$ ). Thus, noting (7.5.4), as  $t \geq T$  the generalized synchronizable state  $U(t, x) = (u_1(t, x), \cdots, u_N(t, x))^T$  can be written as

 $u_p(t,x) = \alpha_p \tilde{\tilde{u}}_1(t,x), \quad p = 1, \cdots, N_1, \quad 0 \le x \le L,$  (7.5.42)

$$u_q(t,x) = \alpha_q \widetilde{u}_2(t,x), \quad q = N_1 + 1, \cdots, N, \quad 0 \le x \le L.$$
 (7.5.43)

Completely similarly to [81], we have

**Theorem 7.5.5** Suppose that M = N - 2. If the coupled system (7.1.1)-(7.1.3) possesses the generalized exact boundary synchronization associated with (7.5.25), and at least for an initial state  $(\Phi, \Psi)$ ,  $\tilde{\tilde{u}}_1(t, x)$  and  $\tilde{\tilde{u}}_2(t, x)$  given by (7.5.42)-(7.5.43) for the generalized synchronizable state are linearly independent, then there exist real vectors  $X_1 \in \mathbb{R}^{N_1} \setminus \{0\}$  and  $X_2 \in \mathbb{R}^{N_2} \setminus \{0\}$  and real numbers  $\tilde{\tilde{a}}_{ij}$  (i, j = 1, 2) such that (7.5.30) hold.

**Remark 7.5.6** As  $t \ge T$ , for the generalized sychronizable state  $U(t, x) = \begin{pmatrix} \tilde{\tilde{u}}_1(t, x)X_1\\ \tilde{\tilde{u}}_2(t, x)X_2 \end{pmatrix}$ given by (7.5.42)-(7.5.43), in which  $X_1 = (\alpha_1, \dots, \alpha_{N_1})^T$  (resp.  $X_2 = (\alpha_{N_1+1}, \dots, \alpha_N)^T$ ) is a system of basic solutions to (7.5.33),  $\tilde{\widetilde{U}}(t, x) = \begin{pmatrix} \tilde{\tilde{u}}_1(t, x)\\ \tilde{\tilde{u}}_2(t, x) \end{pmatrix}$  satisfies the coupled system (7.3.24)-(7.3.26) of wave equations, in which  $\tilde{\tilde{a}}_{ij}$ ,  $\tilde{\tilde{c}}_{ij}$  are  $\tilde{\tilde{\tilde{c}}}_{ij}$  (i, j = 1, 2) given by (7.5.30)-(7.5.32) respectively. By (7.1.5)-(7.1.6), and noting (7.5.35)-(7.5.36), according to Lemma 7.1.1, if we know the initial state (7.3.28) of  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t, x)$  at the moment t = T, then the evolution of  $\tilde{\widetilde{U}} = \tilde{\widetilde{U}}(t, x)$  with respect to t can be completely determined.

Moreover, under the additional assumption that 1 is not an eigenvalue of both  $\widetilde{\widetilde{C}}$  and  $\widetilde{\overline{C}}$ , any given state  $(\varphi^{(1)}, \varphi^{(2)})^T$  and  $(\psi^{(1)}, \psi^{(2)})^T$  in  $C^2[0, L] \times C^1[0, L]$ , satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, with boundary conditions (7.3.25)-(7.3.26), can be the initial data of a generalized synchronizable state  $\widetilde{\widetilde{U}} = \widetilde{\widetilde{U}}(t, x)$ . Thus, at least for some initial states  $(\Phi, \Psi)$ , the generalized synchronizable states  $\widetilde{\widetilde{u}}_1(t, x)$  and  $\widetilde{\widetilde{u}}_2(t, x)$  are linearly independent on  $t \geq T$ , which means the corresponding requirement in Theorem 7.5.5 is reasonable.

**Remark 7.5.7** In the case of exact boundary synchronization by 2-groups, M = N - 2 and  $X_1 = (\overbrace{1, \cdots, 1}^m)^T$ ,  $X_2 = (\overbrace{1, \cdots, 1}^{N-m})^T$ . The requirements (7.5.30) - (7.5.32) imply (7.3.3) - (7.3.6), then, Theorem 7.5.4 leads to the conclusion given in §7.3.

**Remark 7.5.8** The situation M = N - k ( $k \ge 2$ ) can be similarly discussed.

## Chapter 8

# Exact Boundary Synchronization for a Coupled System of Quasilinear Wave Equations

#### Contents

8.1 Intr	roduction and Main Result	
8.2 Preliminaries		
8.2.1	Semi-global classical solutions for a coupled system of quasi- linear wave equations	
8.2.2	Estimates on the $C^1$ , $C^2$ norms of solutions and the continuity modulus of the second order partial derivatives of solutions for a coupled system of nonautonomous linear wave equations $\ldots \ldots \ldots$	
8.3 Proof of Theorem 8.1.1		
8.4 Remarks		
8.5 Apj	pendix: Proof of Lemma 8.3.2	

## 8.1 Introduction and Main Result

Synchronization is a widespread natural phenomenon. How to describe the phenomenon of synchronization for a coupled system has been widely studied by biologists, engineers and sociologists (see [60], [104]). The previous research activities, however, only focused on the systems gouverned by ordinary differential equations or complex networks (see [39], [95] and [115]). Recently, several kinds of exact synchronizations are introduced by Li et.al. (see [57], [81]) for a coupled system of 1-D wave
equations with boundary conditions of Dirichlet type, Neumann type, coupled third type and coupled dissipative type in the framework of  $C^2$  solutions, and it is shown that these synchronizations can be realized by means of less boundary controls. For the multi-dimensional case, by indirectly using the HUM method (see [88]–[89]), a similar result for a coupled system of wave equations with Dirichlet boundary controls can be also established in the framework of weak solutions (see [79]). In this chapter, based on the theory of semi-global classical solutions for quasilinear hyperbolic systems, under suitable hypotheses, we will present an iteration procedure given by a unified constructive method to establish the local exact boundary synchronization for a coupled system of 1-D quasilinear wave equations with afore mentioned boundary conditions.

Precisely speaking, we consider the following coupled system of quasilinear wave equations:

$$\frac{\partial^2 U}{\partial t^2} - a^2 (U, U_t, U_x) \frac{\partial^2 U}{\partial x^2} + A(U, U_t, U_x) U = 0, \qquad (8.1.1)$$

where  $U = (u_1, \ldots, u_N)^T$  is an unknown vector function of (t, x),  $A(U, U_t, U_x) = (a_{ij}(U, U_t, U_x))$  is an  $N \times N$  coupling matrix, whose elements are  $C^1$  functions with respect to their arguments,  $a(U, U_t, U_x)$ , the common velocity of propagation for the coupled system, is also a  $C^1$  function with respect to its arguments, and

$$a(0,0,0) > 0. \tag{8.1.2}$$

(8.1.1) can be written as

$$\frac{\partial^2 u_i}{\partial t^2} - a^2(U, U_t, U_x) \frac{\partial^2 u_i}{\partial x^2} + \sum_{j=1}^N a_{ij}(U, U_t, U_x) u_j = 0 \quad (i = 1, \cdots, N).$$
(8.1.3)

For fixing the idea, we study only the case of one-sided controls, the case of twosided controls can be similarly treated. At the end x = 0, we prescribe any one of the following boundary conditions:

$$x = 0: U = H(t) \text{ (Dirichlet type)}, \tag{8.1.4a}$$

$$x = 0: U_x = H(t)$$
(Neumann type), (8.1.4b)

$$x = 0: U_x - B(U)U = H(t)$$
(Coupled third type), (8.1.4c)

where  $B(U) = (b_{ij}(U))$  is an  $N \times N$  boundary coupling matrix with  $C^1$  elements, and

$$H(t) = (h_1(t), \cdots, h_N(t))^T$$
 (8.1.5)

are  $C^2$  (case (8.1.4a)) or  $C^1$  (cases (8.1.4b)–(8.1.4c)) functions of t, the components of which will be totally or partially taken as boundary controls.

At the end x = L, since no boundary controls are concerned, we prescribe any

one of the following homogeneous boundary conditions:

$$x = L : U = 0$$
(Dirichlet type), (8.1.6a)

$$x = L : U_x = 0$$
(Neumann type), (8.1.6b)

$$x = L : U_x + \overline{B}(U)U = 0$$
 (Coupled third type), (8.1.6c)

where  $\overline{B}(U) = (\overline{b}_{ij}(U))$  is an  $N \times N$  boundary coupling matrix with  $C^1$  elements.

The initial condition is given by

$$t = 0: (U, U_t) = (\Phi(x), \Psi(x)), \ 0 \le x \le L,$$
(8.1.7)

in which both  $\Phi = (\varphi_1, \cdots, \varphi_N)^T$  and  $\Psi = (\psi_1, \cdots, \psi_N)^T$  are  $C^2$  and  $C^1$  vector functions of x on [0, L], respectively, such that the conditions of  $C^2$  compatibility at the points (t, x) = (0, 0) and (0, L) are satisfied, respectively. Moreover, we assume that

$$\|(\Phi, \Psi)\|_{(C^{2}[0,L])^{N} \times (C^{1}[0,L])^{N}} \leq \varepsilon.$$
(8.1.8)

where  $\varepsilon > 0$  is suitably small.

**Definition 8.1.1** The coupled system (8.1.1), (8.1.4) and (8.1.6) is said to possess the exact boundary synchronization if there exist two positive constants  $T_0$  and T with  $T > T_0$ , such that for any given initial state  $(\Phi(x), \Psi(x))$  with small norm  $\|(\Phi,\Psi)\|_{(C^{2}[0,L])^{N}\times(C^{1}[0,L])^{N}}$  and satisfying the conditions of  $C^{2}$  compatibility at the point (t, x) = (0, L), we can find some boundary controls with support on  $[0, T_0]$  in H(t), such that the corresponding mixed initial-boundary value problem (8.1.1), (8.1.4) and (8.1.6)-(8.1.7) admits a unique  $C^2$  solution  $U(x,t) = (u_1(t,x), \cdots, u_N(t,x))^T$ on the domain

$$R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\},\tag{8.1.9}$$

and as  $T_0 \leq t \leq T$  we have

$$u_1(t,x) \equiv \dots \equiv u_N(t,x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{u}}(t,x), \quad 0 \le x \le L.$$
 (8.1.10)

 $\widetilde{\widetilde{u}} = \widetilde{\widetilde{u}}(t, x)$ , being a priori unknown, is called the corresponding synchronizable state.

For the exact boundary synchronization for a coupled system of 1-D linear wave equations, it is shown (cf. [81]) that there is a necessary condition that the sum of the elements in every row of the coupling matrices  $A = (a_{ij}), B = (b_{ij})$  and  $\overline{B} =$  $(\bar{b}_{ij})$  should be independent of  $i = 1, \dots, N$ . In this chapter, for the corresponding

quasilinear case, we impose the similar assumption that

$$\sum_{j=1}^{n} a_{ij}(U, U_t, U_x) \stackrel{\text{def.}}{=} \widetilde{\widetilde{a}}(U, U_t, U_x), \qquad (8.1.11)$$

$$\sum_{j=1}^{n} b_{ij}(U) \stackrel{\text{def.}}{=} \widetilde{\widetilde{b}}(U), \qquad (8.1.12)$$

$$\sum_{j=1}^{n} \bar{b}_{ij}(U) \stackrel{\text{def.}}{=} \widetilde{\tilde{b}}(U), \qquad (8.1.13)$$

where  $\tilde{\tilde{a}}, \tilde{\tilde{b}}$  and  $\tilde{\tilde{b}}$  are all independent of  $i = 1, \cdots, N$ .

Obviously, U = 0 is an equilibrium of system (8.1.1) with boundary conditions (8.1.4) (in which  $H \equiv 0$ ) and (8.1.6). Based on the theory of semi-global  $C^2$  solutions, using a unified constructive method for one-sided exact boundary controllability (see [55], [74], [75], [72]), by a suitable iteration procedure as well as certain estimates for coupled system of wave equations, we will establish the local exact boundary synchronization around U = 0. The main result in this chapter is

**Theorem 8.1.1** Suppose that a,  $a_{ij}$ ,  $b_{ij}$  and  $\bar{b}_{ij}$  are all  $C^1$  functions with respect to their arguments on the domain under consideration. suppose furthermore that (8.1.2) and (8.1.11)-(8.1.13) hold. Let

$$T > T_0 > 2 \frac{L}{a(0,0,0)}.$$
 (8.1.14)

For any given initial state  $(\Phi(x), \Psi(x)) \in (C^2[0, L])^N \times (C^1[0, L])^N$ , satisfying simultaneously (8.1.8), the conditions of  $C^2$  compatibility at the point (t, x) = (0, L), as well as the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) for the boundary conditions without control in (8.1.4), there exist (N - 1) boundary controls with support on  $[0, T_0]$  in H(t) (for example, we take  $h_2(t), \dots, h_N(t)$  as controls with  $h_1(t) \equiv 0$ ) with small  $(C^2[0, T])^N$  norm (case (8.1.4a)) or  $(C^1[0, T])^N$  norm (cases (8.1.4b)–(8.1.4c)), such that the mixed initial-boundary value problem (8.1.1), (8.1.4) and (8.1.6)–(8.1.7) admits a unique  $C^2$  solution U = U(t, x) on the domain  $R(T) = \{(t, x) | 0 \leq t \leq T, 0 \leq x \leq L\}$ , and as  $T_0 \leq t \leq T, U = U(t, x)$  possesses the exact boundary synchronization shown by (8.1.10).

The chapter is organized as follows: In § 8.2 we present the existence and uniqueness of semi-global  $C^2$  solution to the coupled system of quasilinear wave equations with boundary conditions of various types, meanwhile, for the sake in what follows, we establish the estimates on the  $C^1$ ,  $C^2$  and continuity modulus of the second order partial derivatives of the  $C^2$  solutions for a coupled system of nonautonomous linear wave equations. In § 8.3, we use a suitable iteration scheme to establish local exact boundary synchronization for a coupled system of quasilinear wave equations. Some remarks are given in § 8.4. Moreover, an appendix (see §8.5) is attached at the end for the precise proof of a technical lemma in the main text.

**Remark 8.1.1** Similar results are still vaild for the coupled boundary conditions of dissipative type, see Remark 8.4.2.

**Remark 8.1.2** Precisely speaking, the conditions of  $C^2$  compatibility at the point (t, x) = (0, L) are given respectively for the boundary conditions (8.1.6a)-(8.1.6c) as follows:

$$\begin{cases} \Phi(L) = 0, \\ \Psi(L) = 0, \\ \Phi''(L) = 0; \end{cases}$$
(8.1.15a)

$$\begin{cases} \Phi'(L) = 0, \\ \Psi'(L) = 0; \end{cases}$$
(8.1.15b)

$$\begin{cases} \Phi'(L) + \overline{B}(\Phi(L))\Phi(L) = 0, \\ \psi'_i(L) + \sum_{k,j=1}^N \frac{\partial \bar{b}_{ij}}{\partial u_k}(\Phi(L))\psi_k(L)\varphi_j(L) + \sum_{j=1}^N \bar{b}_{ij}(\Phi(L))\psi_j(L) = 0, \end{cases}$$
(8.1.15c)

where  $i = 1, \cdots, N$ .

Similarly, for any given H(t), the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) are given by

$$\begin{cases} \Phi(0) = H(0), \\ \Psi(0) = H'(0), \\ a^2(\Phi(0), \Psi(0), \Phi'(0))\Phi''(0) - A(\Phi(0), \Psi(0), \Phi'(0))\Phi(0) = H''(0); \end{cases}$$
(8.1.16a)

$$\begin{cases} \Phi'(0) = H(0), \\ \Psi'(0) = H'(0); \end{cases}$$
(8.1.16b)

$$\begin{cases} \Phi'(0) - B(\Phi(0))\Phi(0) = H(0), \\ \psi'_i(0) - \sum_{k,j=1}^N \frac{\partial b_{ij}}{\partial u_k}(\Phi(0))\psi_k(0)\varphi_j(0) - \sum_{j=1}^N b_{ij}(\Phi(0))\psi_j(0) = h'_i(0), \end{cases}$$
(8.1.16c)

where  $i = 1, \dots, N$ , respectively. We point out that if some components in H(t) are known in advance, then the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0)involves only those formulas in (8.1.16) where the corresponding components of H(t)are given. For example, suppose that  $h_1(t) \equiv 0$ , and the other components of H(t)are taken as boundary controls, the conditions of  $C^2$  compatibility at the point (t, x) =(0,0) are just given by the first formulas of (8.1.16), the right-hand side of which is zero.

**Remark 8.1.3** For the boundary conditions of Neumann type (8.1.4b) and (8.1.6b), by noting (8.1.15b) and (8.1.16b), the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0) and (0, L) are only related to the initial data and do not depend on the coupled system of wave equations. This fact will play an important role in establishing a unified constructive method independent of iteration below.

#### 8.2 Preliminaries

# 8.2.1 Semi-global classical solutions for a coupled system of quasilinear wave equations

In order to get the well-posedness of the mixed initial-boundary value problem (8.1.1), (8.1.4), (8.1.6) and (8.1.7) in the framework of  $C^2$  solutions, we first reduce the system to a first order quasilinear hyperbolic system, then use the corresponding results of semi-global  $C^1$  solutions.

Let

$$V = (v_1, \cdots, v_N)^T = \frac{\partial U}{\partial x}, W = (w_1, \cdots, w_N)^T = \frac{\partial U}{\partial t}.$$
 (8.2.1)

System (8.1.1) can be reduced to the following first order quasilinear system:

$$\begin{cases} \frac{\partial U}{\partial t} = W, \\ \frac{\partial V}{\partial t} - \frac{\partial W}{\partial x} = 0, \\ \frac{\partial W}{\partial t} - a^2(U, W, V) \frac{\partial V}{\partial x} = -A(U, W, V)U. \end{cases}$$
(8.2.2)

Let  $I_N$  be the identity matrix of order N. (8.2.2) can be written in the following

matrix form:

$$\frac{\partial}{\partial t} \begin{pmatrix} U\\V\\W \end{pmatrix} + \begin{pmatrix} 0 & 0 & 0\\ 0 & 0 & -I_N\\ 0 & -a^2(U,W,V)I_N & 0 \end{pmatrix} \frac{\partial}{\partial x} \begin{pmatrix} U\\V\\W \end{pmatrix} = \begin{pmatrix} W\\0\\-A(U,W,V)U\\(8.2.3) \end{pmatrix}.$$

The characteristic equation of (8.2.2) is

$$\det \begin{pmatrix} \lambda I_N & 0 & 0\\ 0 & \lambda I_N & I_N\\ 0 & a^2(U, W, V) I_N & \lambda I_N \end{pmatrix} = \lambda^N |\lambda^2 I_N - a^2 I_N| = 0, \quad (8.2.4)$$

whose solutions, the eigenvalues of system (8.2.2), are all real:

$$\lambda_i^- = -a, \ \lambda_i^0 = 0, \ \lambda_i^+ = a \ (i = 1, \cdots, N),$$
 (8.2.5)

and the corresponding left eigenvectors, which constitute a complete set, can be chosen as

$$l_{i}^{-} = (\mathbf{0}, a(U, W, V)\mathbf{e}_{i}, \mathbf{e}_{i}), l_{i}^{0} = (\mathbf{e}_{i}, \mathbf{0}, \mathbf{0}), l_{i}^{+} = (\mathbf{0}, -a(U, W, V)\mathbf{e}_{i}, \mathbf{e}_{i}) \quad (i = 1, \cdots, N),$$
(8.2.6)

where  $\mathbf{0} = (0, \dots, 0)$  is the zero vector of order N, and  $\mathbf{e}_i = (0, \dots, \stackrel{(i)}{1}, \dots, 0)$  is a unit row vector of order N. Thus, (8.2.2) is a first order quasilinear hyperbolic system.

Let

$$\widetilde{U} = \begin{pmatrix} U \\ V \\ W \end{pmatrix}$$
(8.2.7)

and

$$\begin{cases} V_i^- = l_i^- \widetilde{U} = av_i + w_i, \\ V_i^0 = l_i^0 \widetilde{U} = u_i, \\ V_i^+ = l_i^+ \widetilde{U} = -av_i + w_i \end{cases}$$
(8.2.8)

We have

$$\begin{cases} V^{-} = aV + W, \\ V^{0} = U, \\ V^{+} = -aV + W, \end{cases}$$
(8.2.9)

where 
$$V^{-} = (V_{1}^{-}, \cdots, V_{N}^{-})^{T}$$
,  $V^{0} = (V_{1}^{0}, \cdots, V_{N}^{0})^{T}$  and  $V^{+} = (V_{1}^{+}, \cdots, V_{N}^{+})^{T}$ . Then  

$$\begin{cases}
U = V^{0}, \\
V = \frac{1}{2a}(V^{-} - V^{+}), \\
W = \frac{1}{2}(V^{-} + V^{+}).
\end{cases}$$
(8.2.10)

Under the above transformation, the boundary condition 
$$(8.1.4)$$
 can be correspondingly replaced by

$$x = 0: W = \dot{H}(t),$$
 (8.2.11a)

$$x = 0: V = H(t),$$
 (8.2.11b)

$$x = 0: V - B(U)U = H(t).$$
 (8.2.11c)

Notice (8.2.10), in a neighborhood of  $\widetilde{U} = 0$ , i.e.,  $\widetilde{V} \stackrel{\text{def.}}{=} \begin{pmatrix} V^- \\ V^0 \\ V^+ \end{pmatrix} = 0$ , the boundary conditions (8.2.11) on x = 0 can be rewritten as

$$x = 0: \quad V^+ = -V^- + 2\dot{H}(t),$$
 (8.2.12a)

$$x = 0:$$
  $V^+ = V^- - 2a(V^0, \frac{V^- + V^+}{2}, H(t))H(t),$  (8.2.12b)

$$x = 0: \quad V^{+} = V^{-} - 2a(V^{0}, \frac{V^{-} + V^{+}}{2}, B(V^{0})V^{0} + H(t)) (B(V^{0})V^{0} + H(t)),$$
(8.2.12c)

which can be uniformly expressed as

$$x = 0: P(t, V^{-}, V^{0}, V^{+}) = 0,$$
 (8.2.13)

where

$$P(t, V^{-}, V^{0}, V^{+}) = V^{+} + V^{-} - 2\dot{H}(t) \text{ for } (8.2.12a),$$

$$P(t, V^{-}, V^{0}, V^{+}) = V^{+} - V^{-} + 2a(V^{0}, \frac{V^{-} + V^{+}}{2}, H(t))H(t) \text{ for } (8.2.12b),$$

$$(8.2.14b)$$

$$(8.2.14b)$$

$$P(t, V^{-}, V^{0}, V^{+}) = V^{+} - V^{-} + 2a(V^{0}, \frac{V^{-} + V^{+}}{2}, B(V^{0})V^{0} + H(t)) (B(V^{0})V^{0} + H(t))$$
  
for (8.2.12c).  
(8.2.14c)

For the boundary conditions (8.2.13) on x = 0, regarding  $(t, V^-, V^0, V^+)$  as variables, noting (8.2.9), under the hypothesis that  $||H||_{C^2}(\text{case }(8.1.4a))$  or  $||H||_{C^1}(\text{cases})$ 

(8.1.4b)-(8.1.4c)) and  $\|\widetilde{U}\|_{C^0}$  are suitably small (the validity of this hypothesis will be shown by Lemma 8.2.1), it is easy to see that

$$\det\left(\frac{\partial P_i}{\partial V_j^+}\right) \neq 0, \ i, j = 1, \cdots, N.$$
(8.2.15)

Therefore, using the implicit function theorem, for any given T > 0, on the intervel [0, T] the boundary condition on x = 0 can be locally rewritten around  $\tilde{U} = 0$  to the following unified form:

$$x = 0: V^+ = G(t, V^-, V^0) + \widetilde{H}(t),$$
 (8.2.16)

where G and  $\widetilde{H}$  are  $C^1$  functions with respect to their arguments, and without loss of generality, we assume that

$$G(t,0,0) \equiv 0. \tag{8.2.17}$$

Obviously, the  $C^1$  norm of  $\widetilde{H}(t)$  is small enough.

Similarly, the boundary conditions (8.1.6a)–(8.1.6c) on x = L can be also locally rewritten around  $\tilde{U} = 0$  to the following unified form:

$$x = L: V^{-} = \overline{G}(V^{0}, V^{+}),$$
 (8.2.18)

in which  $\overline{G}$  is a  $C^1$  function with respect to its arguments and

$$\overline{G}(0,0) \equiv 0. \tag{8.2.19}$$

Meanwhile, the initial condition (8.1.7) can be correspondingly written as

$$t = 0: \widetilde{U}(0, x) = \begin{pmatrix} \Phi(x) \\ \Phi'(x) \\ \Psi(x) \end{pmatrix} \stackrel{\text{def.}}{=} \widetilde{U}_0(x), \ 0 \le x \le L.$$
(8.2.20)

By the conditions of  $C^2$  compatibility at the points (t, x) = (0, 0) and (0, L) for the mixed initial-boundary value problem (8.1.1), (8.1.4) and (8.1.6)–(8.1.7), it is easy to see that the conditions of  $C^1$  compatibility at these two points are also satisfied for the mixed initial-boundary value problem (8.2.2), (8.2.20), (8.2.16) and (8.2.18). Based on the theory of semi-global  $C^1$  solutions to the first order quasilinear hyperbolic system with zero eigenvalues (see [73], [70], [81], [111]–[112]), we get immediately the following lemma:

**Lemma 8.2.1** Suppose that on the domain under consideration,  $a, a_{ij}, b_{ij}$  and  $\bar{b}_{ij}$  are  $C^1$  functions with respect to their arguments. Suppose furthermore that the conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, 0) and (0, L),

respectively. For any given and possibly quite large T > 0, the forward mixed initialboundary value problem (8.1.1), (8.1.4) and (8.1.6)–(8.1.7) admits a unique  $C^2$  solution U = U(t, x) with small  $C^2$  norm on the domain

$$R(T) = \{(t, x) | 0 \le t \le T, \ 0 \le x \le L\},$$
(8.2.21)

provided that  $\|(\Phi, \Psi)\|_{(C^2[0,L])^N \times (C^1[0,L])^N}$  and  $\|H\|_{(C^2[0,T])^N}$ , (case (8.1.4a)) or  $\|H\|_{(C^1[0,T])^N}$  (cases (8.1.4b)–(8.1.4c)) are suitably small (possibly depending on T).

#### 8.2.2 Estimates on the $C^1$ , $C^2$ norms of solutions and the continuity modulus of the second order partial derivatives of solutions for a coupled system of nonautonomous linear wave equations

In order to obtain the exact boundary synchronization for the quasilinear system (8.1.1), (8.1.4) and (8.1.6), we have to investigate the  $C^1$ ,  $C^2$  norms of solutions and the continuity modulus of the second order partial derivatives of solutions for a coupled system of nonautonomous linear wave equations. Here, the continuity modulus of a function f(t, x) on R(T) is the following non-negative function:

$$\omega(\eta) = \omega(\eta|f) \stackrel{\text{def.}}{=} \sup_{\substack{|t'-t''| \le \eta, \ |x'-x''| \le \eta \\ (t',x'), \ (t'',x'') \in R(T)}} |f(t',x') - f(t'',x'')|, \quad \eta \ge 0.$$
(8.2.22)

Similarly, the continuity modulus of a vector function f can be defined by

$$\omega(\eta|f) \stackrel{\text{def.}}{=} \max_{i=1,\cdots,n} \omega(\eta|f_i), \qquad (8.2.23)$$

and the continuity modulus of a matrix function  $A = (a_{ij})_{n \times n}$  can be defined by

$$\omega(\eta|A) \stackrel{\text{def.}}{=} \max_{i,j=1,\cdots,n} \omega(\eta|a_{ij}). \tag{8.2.24}$$

In this subsection, we still denote U = U(t, x) as the unknown vector function, a = a(t, x) as the common velocity of propagation and A = A(t, x), B = B(t) and  $\overline{B} = \overline{B}(t)$  as coupling matrices. Consider the following coupled system of nonautonomous wave equations:

$$\frac{\partial^2 U}{\partial t^2} - a^2(t,x)\frac{\partial^2 U}{\partial x^2} + A(t,x)U = F(t,x), \qquad (8.2.25)$$

where  $U = (u_1, \ldots, u_N)^T$  is the unknown vector function of (t, x),  $A(t, x) = (a_{ij}(t, x))$ is an  $N \times N$  matrix with  $C^1$  elements, a(t, x) > 0, the common velocity of propagation for the coupled system, is a  $C^1$  function of (t, x), while  $F(t, x) = (f_1(t, x), \cdots, f_N(t, x))^T$ is also a  $C^1$  vector function of (t, x).

At the end x = 0, corresponding to (8.1.4a)–(8.1.4c), we prescribe any one of the following boundary conditions:

$$x = 0: U = H(t) \text{ (Dirichlet type)}, \tag{8.2.26a}$$

$$x = 0: U_x = H(t)$$
(Neumann type), (8.2.26b)

$$x = 0: U_x - B(t)U = H(t)$$
(Coupled third type), (8.2.26c)

in which  $B(t) = (b_{ij}(t))$  is an  $N \times N$  matrix with  $C^1$  elements, and  $H(t) = (h_1(t), \dots, h_N(t))^T$ are  $C^2$  (case (8.2.26a)) or  $C^1$  (cases (8.2.26b)–(8.2.26c)) functions of t.

At the end x = L, corresponding to (8.1.6a)–(8.1.6c), we prescribe any one of the following boundary conditions:

$$x = L : U = H(t) \text{ (Dirichlet type)}, \tag{8.2.27a}$$

$$x = L : U_x = \overline{H}(t)$$
 (Neumann type), (8.2.27b)

$$x = L : U_x + \overline{B}(t)U = \overline{H}(t)$$
 (Coupled third type), (8.2.27c)

in which  $\overline{B}(t) = (\overline{b}_{ij}(t))$  is an  $N \times N$  matrix with  $C^1$  elements, and  $\overline{H}(t) = (\overline{h}_1(t), \cdots, \overline{h}_N(t))^T$ are  $C^2$  (case (8.2.27a)) or  $C^1$  (cases (8.2.27b)–(8.2.27c)) functions of t.

Denote

$$l = \begin{cases} 1 & \text{case } (8.2.26a), \\ 0 & \text{cases } (8.2.26b) - (8.2.26c) \end{cases}, \quad \bar{l} = \begin{cases} 1 & \text{case } (8.2.27a), \\ 0 & \text{cases } (8.2.27b) - (8.2.27c) \end{cases}, \quad (8.2.28) \\ s = l + 1, \qquad \bar{s} = \bar{l} + 1. \end{cases}$$

Suppose that on the domain under consideration, we have

$$||a||_1 + ||A||_1 + ||B||_1 + ||\overline{B}||_1 \le D < +\infty,$$
(8.2.30)

where D is a positive constant. Here and hereafter, for simplicity we denote  $\|\cdot\|_0$ ,  $\|\cdot\|_1$  and  $\|\cdot\|_2$  as as the corresponding  $C^0$ ,  $C^1$  and  $C^2$  norms, respectively.

For the mixed initial-boundary value problem (8.2.25), (8.2.26)–(8.2.27) and (8.1.7), by the estimates established in [73] and [72] for the  $C^1$  and  $C^2$  norms of solutions, as well as the continuity modulus of the second order partial derivatives of solutions, obtained by a similar manner, we have

**Lemma 8.2.2** For any given T > 0, suppose that  $a, a_{ij}, b_{ij}, \bar{b}_{ij}$  and  $f_i$   $(i, j = 1, \dots, N)$  are all  $C^1$  functions on the domain  $R(T) = \{(t, x) | 0 \le t \le T, 0 \le x \le L\}$ , suppose furthermore that the conditions of  $C^2$  compatibility are satisfied at the points

(t,x) = (0,0) and (0,L), respectively. Then the forward mixed initial-boundary value problem (8.2.25)-(8.2.27) and (8.1.7) admits a unique  $C^2$  solution U = U(t,x) on the domain R(T), and we have

$$\begin{aligned} \|U\|_{(C^{1}[R(T)])^{N}} &\leq C_{1} \left( \|(\Phi, \Psi)\|_{(C^{1}[0,L])^{N} \times (C^{0}[0,L])^{N}} + \|(H,\overline{H})\|_{(C^{l}[0,T])^{N} \times (C^{\bar{l}}[0,T])^{N}} + \|F\|_{(C^{0}[R(T)])^{N}} \right) \tag{8.2.31}$$

and

$$\begin{aligned} \|U\|_{(C^{2}[R(T)])^{N}} &\leq C_{2} \big( \|(\Phi, \Psi)\|_{(C^{2}[0,L])^{N} \times (C^{1}[0,L])^{N}} \\ &+ \|(H, \overline{H})\|_{(C^{s}[0,T])^{N} \times (C^{\overline{s}}[0,T])^{N}} + \|F\|_{(C^{1}[R(T)])^{N}} \big), \end{aligned}$$
(8.2.32)

where  $C_1 = C_1(T, D)$  and  $C_2 = C_2(T, D)$  are positive constants depending only on T and D. Moreover, if

$$F(t,x) \equiv 0, \tag{8.2.33}$$

then for any given  $\eta$  with

$$0 < \eta \le \min_{(t,x) \in R(T)} \frac{L}{a(t,x)},$$
(8.2.34)

we have

$$\omega\left(\eta \left|\frac{\partial^2 U}{\partial t^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U}{\partial x^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U}{\partial t \partial x}\right)\right| \le \Omega(\eta), \quad (8.2.35)$$

where

$$\Omega(\eta) = C_3 \left( \omega \left( \eta \big| \Phi_{xx} \right) + \omega \left( \eta \big| \Psi_x \right) + \omega \left( \eta \big| \frac{d^s H}{dt^s} \right) + \omega \left( \eta \big| \frac{d^{\bar{s}} H}{dt^{\bar{s}}} \right) + \left( \| \Phi \|_2 + \| \Psi \|_1 + \| H \|_s + \| \overline{H} \|_{\bar{s}} \right) \right)$$
$$\cdot \left( \eta + \omega (\eta | a_x) + \omega (\eta | A_x) + \omega (\eta | B_t) + \omega (\eta | \overline{B}_t) \right) \right)$$
(8.2.36)

is a non-negative function of  $\eta$ , and  $\Omega(\eta) \to 0$ , when  $\eta \to 0$ ,  $C_3 = C_3(T, D)$  being a positive constant depending on T and D.

**Remark 8.2.1** For the backward mixed initial-boundary value problem (8.2.25)–(8.2.27) with final condition

$$t = T : (U, U_t) = (\overline{\Phi}(x), \ \overline{\Psi}(x)), \ 0 \le x \le L,$$

$$(8.2.37)$$

satisfying the conditions of  $C^2$  compatibility at the points (t, x) = (T, 0) and (T, L), respectively, similar results can be obtained.

## Chapter 8: Exact Boundary Synchronization for a Coupled System of Quasilinear Wave Equations

#### 8.3 Proof of Theorem 8.1.1

Arbitrarily choose  $H^{(0)}(t)$  (where  $h_1^{(0)}(t) \equiv 0$ ) to be a  $C^2$  (case (8.1.4a)) or  $C^1$  (cases (8.1.4b)–(8.1.4c)) functions of t, satisfying

$$\|H^{(0)}\|_{(C^s[0,T])^N} \le C_4 \varepsilon, \tag{8.3.1}$$

where  $C_4$  is a positive constant,

$$s = \begin{cases} 2, \text{ case } (8.1.4a), \\ 1, \text{ cases } (8.1.4b) - (8.1.4c), \end{cases}$$
(8.3.2)

and  $\varepsilon > 0$ , given by (8.1.8), is a suitably small positive constant, such that the conditions of  $C^2$  compatibility are satisfied at the point (t, x) = (0, 0), and

$$H^{(0)}(t) \equiv 0, \quad T_0 \le t \le T.$$
 (8.3.3)

**Remark 8.3.1** Since we have assumed  $h_1^{(0)}(t) \equiv 0$ , by the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0), the initial condition (8.1.7) can not be arbitrarily given near x = 0. Although this is a special case in Theorem 8.1.1, it is not an issue for the generality of the whole proof.

By Lemma 8.2.1, the initial-boundary value problem (8.1.1), (8.1.4) (in which  $H(t) = H^{(0)}(t)$  and (8.1.6)–(8.1.7) admits a unique  $C^2$  solution  $U = U^{(0)}(t, x)$  on the domain R(T), satisfying

$$\|U^{(0)}\|_{(C^2[R(T)])^N} \le G(\varepsilon), \tag{8.3.4}$$

where  $G(\varepsilon)$  is a  $C^0$  function of  $\varepsilon \ge 0$ , satisfying

$$G(0) = 0, (8.3.5)$$

and, without loss of generality, we may assume that

$$1 \gg G(\varepsilon) \ge C_5 \varepsilon, \tag{8.3.6}$$

where  $C_5$  is a positive constant. Since  $U^{(0)} \in C^2$ , there exists a non negative function  $\Omega_0(\eta)$  of  $\eta$  (with  $\Omega_0(\eta) \to 0$ , as  $\eta \to 0$ ), such that

$$\omega\left(\eta \left|\frac{\partial^2 U^{(0)}}{\partial t^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(0)}}{\partial x^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(0)}}{\partial t \partial x}\right)\right| \le \Omega_0(\eta).$$
(8.3.7)

Here we point out that, being the first step of the iteration,  $U^{(0)}$  is not asked to be synchronized on  $T_0 \leq t \leq T$ .

Let

$$U^{(-1)}(t,x) \stackrel{\text{def.}}{=} U^{(0)}(t,x).$$
(8.3.8)

For any given  $m \ge 0$ , we use an iteration procedure to get  $U^{(m)}(t, x)$  from  $U^{(m-1)}(t, x)$  on R(T):

$$\frac{\partial^2 U^{(m)}}{\partial t^2} - a^2 (U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) \frac{\partial^2 U^{(m)}}{\partial x^2} + A(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) U^{(m)} = 0,$$
(8.3.9)

the boundary condition on x = 0 is given by any one of :

$$x = 0: U^{(m)} = H^{(m)}(t),$$
 (8.3.10a)

$$x = 0: U_x^{(m)} = H^{(m)}(t), \tag{8.3.10b}$$

$$x = 0: U_x^{(m)} - B(U^{(m-1)})U^{(m)} = H^{(m)}(t), \qquad (8.3.10c)$$

the boundary condition on x = L is given by any one of

$$x = L : U^{(m)} = 0, (8.3.11a)$$

$$x = L : U_x^{(m)} = 0, (8.3.11b)$$

$$x = L : U_x^{(m)} - \overline{B}(U^{(m-1)})U^{(m)} = 0, \qquad (8.3.11c)$$

and the initial condition is

$$t = 0: (U^{(m)}, U_t^{(m)}) = (\Phi(x), \Psi(x)), \ 0 \le x \le L.$$
(8.3.12)

In this iteration scheme, for m = 0, since  $H^{(0)}(t)$  has been chosen, the corresponding solution is just  $U^{(0)}(t, x)$ ; while, for each  $m \ge 1$ ,  $H^{(m)}(t)$  is a  $C^2$  (case (8.3.10a)) or  $C^1$  (cases (8.3.10b)–(8.3.10c)) functions of t to be determined. Our goal is to find a suitable  $H^{(m)}(t)$  (where  $h_1^{(m)}(t) \equiv 0$ ), for each  $m \ge 1$ , such that the corresponding mixed initial-boundary value problem (8.3.9)–(8.3.12) admits a  $C^2$  solution  $U^{(m)} = U^{(m)}(t, x)$  on the domain R(T), satisfying the exact boundary synchronization (8.1.10). Notice that in the whole iteration procedure, the form of the conditions of  $C^2$ compatibility (8.1.15)–(8.1.16) at the points (t, x) = (0, 0) and (0, L) never changes (just replacing H(t) by  $H^{(m)}(t)$ ).

If the synchronization sequences  $\{U^{(m)}(t,x)\}(m \ge 1)$  are obtained by means of suitable boundary controls  $H^{(m)}(t)$  (in which  $h_1^{(m)}(t) \equiv 0, \ 0 \le t \le T$ ), and if one can prove that

$$U^{(m)}(t,x) \to U(t,x)$$
 in  $(C^2[R(T)])^N$  as  $m \to +\infty$ , (8.3.13)

then U(t, x) should satisfy (8.1.1) and (8.1.6)–(8.1.7) and possesses the synchronization (8.1.10), and the boundary controls correspondingly to (8.1.4) are given by

$$H(t) = \lim_{m \to +\infty} H^{(m)}(t), \qquad (8.3.14)$$

with  $h_1(t) \equiv 0$  ( $0 \leq t \leq T$ ). Therefore, in order to get Theorem 8.1.1, it suffices to establish the following

**Lemma 8.3.1** Under the assumptions of Theorem 8.1.1, for any given integer  $m \ge 0$ , there exist a boundary control  $H^{(m)}(t)$  (in which  $h_1^{(m)}(t) \equiv 0$  ( $0 \le t \le T$ )) and positive constants  $C_6 = C_6(T_0, T)$ ,  $C_7 = C_7(T_0, T)$  and  $C_8 = C_8(T_0, T)$ , such that the mixed initial-boundary value problem (8.3.9)–(8.3.12) admits a unique  $C^2$  solution  $U^{(m)} = U^{(m)}(t, x)$  on the domain R(T), which satisfies

$$\|U^{(m)}\|_{(C^{2}[R(T)])^{N}} \leq C_{6}G(\varepsilon), \qquad (8.3.15)$$

$$\|U^{(m)} - U^{(m-1)}\|_{(C^1[R(T)])^N} \le C_7^m G^m(\varepsilon),$$
(8.3.16)

$$\omega\left(\eta \left|\frac{\partial^2 U^{(m)}}{\partial t^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(m)}}{\partial x^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(m)}}{\partial t \partial x}\right)\right) \le \Omega_1(\eta),\tag{8.3.17}$$

where

$$\Omega_1(\eta) = \frac{C_8 \left( \omega(\eta | \Phi_{xx}) + \omega(\eta | \Psi_x) + \Omega_0(\eta) + G(\varepsilon) \left( \eta + \omega(\eta | \Gamma^*) \right) \right)}{1 - C_8 G(\varepsilon)}$$
(8.3.18)

is a non-negetive function of  $\eta$  with  $\Omega_1(\eta) \to 0$  as  $\eta \to 0$ , while  $G(\varepsilon)$  and  $\Omega_0(\eta)$  are given by (8.3.4) and (8.3.7),

$$\Gamma^* = \left\{ \frac{\partial a}{\partial u_k}, \frac{\partial a}{\partial u_{kt}}, \frac{\partial a}{\partial u_{kx}}, \frac{\partial a_{ij}}{\partial u_k}, \frac{\partial a_{ij}}{\partial u_{kt}}, \frac{\partial a_{ij}}{\partial u_{kx}}, \frac{\partial b_{ij}}{\partial u_k}, \frac{\partial \overline{b}_{ij}}{\partial u_k}; i, j, k = 1, \cdots, N \right\}.$$
(8.3.19)

Moreover, for any given  $m \ge 1$ ,  $U^{(m)}(t, x)$  possesses the synchronization (8.1.10) for  $T_0 \le t \le T$ .

**Proof:** For m = 0, since  $H^{(0)}(t)$  has been determined, taking  $C_6 \ge 1$ , (8.3.4) immediately yields (8.3.15). Moreover, we have (8.3.16) because of (8.3.8). Combining (8.3.7) and (8.3.18), one gets (8.3.17), provided that  $\varepsilon > 0$  is suitably small and  $C_8 \ge 1$ .

By induction, suppose that for  $m \ge 1$  we have

$$\|U^{(m-1)}\|_{(C^2[R(T)])^N} \le C_6 G(\varepsilon), \tag{8.3.20}$$

$$\|U^{(m-1)} - U^{(m-2)}\|_{(C^{1}[R(T)])^{N}} \le C_{7}^{m-1} (G(\varepsilon))^{m-1},$$
(8.3.21)

$$\omega\left(\eta \left|\frac{\partial^2 U^{(m-1)}}{\partial t^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(m-1)}}{\partial x^2}\right) + \omega\left(\eta \left|\frac{\partial^2 U^{(m-1)}}{\partial t \partial x}\right)\right| \le \Omega_1(\eta).$$
(8.3.22)

Then for any given  $m \ge 1$ , let

$$\begin{cases} w_i^{(m)} = u_{i+1}^{(m)} - u_i^{(m)}, \\ \tilde{h}_i^{(m)} = h_{i+1}^{(m)} - h_i^{(m)}, \end{cases} \quad i = 1, \cdots, N - 1.$$
(8.3.23)

Under assumption (8.1.11), it is easy to see that from the original system (8.3.9) for the variable  $U^{(m)} = (u_1^{(m)}, \cdots, u_N^{(m)})$ , we can get the following self-closing system for the variable  $W^{(m)} = (w_1^{(m)}, \cdots, w_{N-1}^{(m)})$  (see also [81]):

$$\frac{\partial^2 W^{(m)}}{\partial t^2} - a^2 (U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) \frac{\partial^2 W^{(m)}}{\partial x^2} + \widetilde{A}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) W^{(m)} = 0,$$
(8.3.24)

where  $\widetilde{A} = (\widetilde{a}_{ij})$  is an  $(N-1) \times (N-1)$  reduced matrix with

$$\widetilde{a}_{ij}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) = \sum_{p=j+1}^N \left( a_{i+1,p}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - a_{ip}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) \right) \\
= \sum_{p=1}^j \left( a_{i,p}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - a_{i+1,p}(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) \right), i, j = 1, \cdots, N-1 \\$$
(8.3.25)

Similarly, by assumptions (8.1.12)–(8.1.13), from the original boundary conditions (8.3.10)–(8.3.11) for the variable  $U^{(m)}$ , we can get, respectively, the self-closing boundary conditions for the variable  $W^{(m)}$ :

$$x = 0: W^{(m)} = \widetilde{H}^{(m)}(t), \tag{8.3.26a}$$

$$x = 0: W_x^{(m)} = \widetilde{H}^{(m)}(t),$$
 (8.3.26b)

$$x = 0: W_x^{(m)} - \widetilde{B}(U^{(m-1)})W^{(m)} = \widetilde{H}^{(m)}(t)$$
(8.3.26c)

and

$$x = L : W^{(m)} = 0, (8.3.27a)$$

$$x = L : W_x^{(m)} = 0, (8.3.27b)$$

$$x = L : W_x^{(m)} + \overline{B}(U^{(m-1)})W^{(m)} = 0, \qquad (8.3.27c)$$

in which  $\widetilde{H}^{(m)}(t) = (\widetilde{h}_1^{(m)}(t), \cdots, \widetilde{h}_{N-1}^{(m)}(t))^T$ , and  $\widetilde{B} = (\widetilde{b}_{ij})$  and  $\widetilde{\overline{B}} = (\widetilde{\overline{b}}_{ij})$  are  $(N-1) \times (N-1)$  matrices reduced by  $B = (b_{ij})$  and  $\overline{\overline{B}} = (\overline{b}_{ij})$  in the following way:

$$\widetilde{b}_{ij}(U^{(m-1)}) = \sum_{p=j+1}^{n} \left( b_{i+1,p}(U^{(m-1)}) - b_{ip}(U^{(m-1)}) \right)$$

$$= \sum_{p=1}^{j} \left( b_{i,p}(U^{(m-1)}) - b_{i+1,p}(U^{(m-1)}) \right), \quad i, j = 1, \cdots, N-1$$
(8.3.28)

etc.

The initial condition of  $W^{(m)}$  is given by

$$t = 0: (W^{(m)}, W_t^{(m)}) = (\widetilde{\Phi}(x), \widetilde{\Psi}(x)), \ 0 \le x \le L,$$
(8.3.29)

where

$$\begin{cases} \widetilde{\Phi}(x) = (\varphi_2(x) - \varphi_1(x), \cdots, \varphi_N(x) - \varphi_{N-1}(x))^T, \\ \widetilde{\Psi}(x) = (\psi_2(x) - \psi_1(x), \cdots, \psi_N(x) - \psi_{N-1}(x))^T. \end{cases}$$
(8.3.30)

Moreover, it easily follows from the condition of  $C^2$  compatibility at the point (t, x) = (0, L) for  $U^{(m)}$  that the condition of  $C^2$  compatibility at the point (t, x) = (0, L) for  $W^{(m)}$  is still satisfied.

Noting (8.3.23), once we achieve the exact null controllability for  $W^{(m)}$ , we certainly have the exact synchronization for  $U^{(m)}$ . Similarly to the method in [55] and [77], our next step is to construct a  $C^2$  solution  $W^{(m)} = W^{(m)}(t,x)$  on R(T) for each  $m \geq 1$ , satisfying simultaneously the system (8.3.24), the initial condition (8.3.29), the null final condition

$$W^{(m)}(t,x) \equiv 0, \quad T_0 \le t \le T$$
 (8.3.31)

and the boundary condition (8.3.27) on x = L. To this end, we need a suitable treatment on the constructive method for realizing the exact boundary null controllability of  $W^{(m)}$ . In order that, regarding the conditions of  $C^2$  compatibility at the point (t, x) = (0, 0), the artificial boundary conditions on x = 0 do not depend on the iteration (8.3.24), we take *Neumann* type artificial boundary conditions on x = 0, which is independent of m, instead of the commonly used *Dirichlet* type boundary conditions (see Remark 8.1.2–8.1.3). Thus, for the sequence  $\{W^{(m)}\}$ , we can prove the following Lemma 8.3.2 (for the details of its proof, see Appendix 8.5):

**Lemma 8.3.2** For any given  $m \ge 1$ , under the assumptions of Theorem 1.1, there exist a boundary control  $\widetilde{H}^{(m)}(t)$  on x = 0, composed of  $C^2$  (case (8.3.26a)) or  $C^1$  (cases (8.3.26b)–(8.3.26c)) functions of t with support on  $[0, T_0]$ , and the positive constants  $C_9(T_0, T)$ ,  $C_{10}(T_0, T)$  and  $C_{11}(T_0, T)$ , such that the mixed initial-boundary value problem (8.3.24), (8.3.26)–(8.3.27) and (8.3.29) admits a unique  $C^2$  solution  $W^{(m)} = W^{(m)}(t,x)$  on the domain R(T), satisfying (8.3.31) and

$$\|W^{(m)}\|_{(C^2[R(T)])^N} \le C_9 G(\varepsilon), \tag{8.3.32}$$

$$\|W^{(m)} - W^{(m-1)}\|_{(C^{1}[R(T)])^{N}} \le C_{10}C_{7}^{m-1}(G(\varepsilon))^{m}, \qquad (8.3.33)$$

and

$$\omega\left(\eta \left|\frac{\partial^2 W^{(m)}}{\partial t^2}\right) + \omega\left(\eta \left|\frac{\partial^2 W^{(m)}}{\partial x^2}\right) + \omega\left(\eta \left|\frac{\partial^2 W^{(m)}}{\partial t \partial x}\right)\right| \le \Omega_2(\eta), \quad (8.3.34)$$

where

$$\Omega_2(\eta) = C_{11} \bigg( \omega(\eta | \Phi_{xx}) + \omega(\eta | \Psi_x) + \Omega_0(\eta) + G(\varepsilon) \Big( \eta + \Omega_1(\eta) + \Omega(\eta | \Gamma^*) \Big) \bigg),$$
(8.3.35)

and  $G(\varepsilon)$ ,  $\Omega_0(\eta)$  and  $\Omega_1(\eta)$  are given by (8.3.4), (8.3.7) and (8.3.18), respectively.

Noting (8.3.32)–(8.3.34), by the way to obtain the corresponding boundary controls in the constructive method (see [77]), we have

$$\begin{aligned} \|\widetilde{H}^{(m)}\|_{(C^{e}[0,T])^{N}} &\leq C_{12} \|W^{(m)}\|_{(C^{2}[R(T)])^{N}} \leq C_{13}G(\varepsilon), \end{aligned} \tag{8.3.36} \\ \|\widetilde{H}^{(m)} - \widetilde{H}^{(m-1)}\|_{(C^{\bar{e}}[0,T])^{N}} \leq C_{14} \|W^{(m)} - W^{(m-1)}\|_{(C^{1}[R(T)])^{N}} \leq C_{15}C_{7}^{m-1} \big(G(\varepsilon)\big)^{m} \end{aligned} \tag{8.3.36}$$

and

$$\omega\left(\eta \left| \frac{d^e \widetilde{H}^{(m)}}{dt^e} \right) \le C_{16} \Omega_2(\eta), \tag{8.3.38}$$

where

$$e = \begin{cases} 2, & \text{case } (8.3.26a), \\ 1, & \text{cases } (8.3.26b) - (8.3.26c), \end{cases}, \quad \overline{e} = \begin{cases} 1, & \text{case } (8.3.26a), \\ 0, & \text{cases } (8.3.26b) - (8.3.26c), \end{cases}$$

$$(8.3.39)$$

and  $C_{12}$ ,  $C_{13}$ ,  $C_{14}$ ,  $C_{15}$  and  $C_{16}$  are all positive constants depending only on  $T_0$  and T.

Thus, by (8.3.23) and noting  $h_1^{(m)} \equiv 0$ , we get

$$||H^{(m)}||_{(C^r[R(T)])^N} \le C_{17}G(\varepsilon)$$
(8.3.40)

$$\|H^{(m)} - H^{(m-1)}\|_{(C^{\overline{r}}[R(T)])^N} \le C_{18}C_7^{m-1}(G(\varepsilon))^m$$
(8.3.41)

and

$$\omega\left(\eta \left| \frac{d^r H^{(m)}}{dt^r} \right) \le C_{19}\Omega_2(\eta),\tag{8.3.42}$$

where

$$r = \begin{cases} 2, & \text{case } (8.3.10a), \\ 1, & \text{cases } (8.3.10b) - (8.3.10c), \end{cases}, \quad \overline{r} = \begin{cases} 1, & \text{case } (8.3.10a), \\ 0, & \text{cases } (8.3.10b) - (8.3.10c), \end{cases}$$

$$(8.3.43)$$

and  $C_{17}$ ,  $C_{18}$ ,  $C_{19}$  are positive constants depending only on  $T_0$  and T.

Obviously, for  $m \ge 1$ , the previously obtained boundary control  $H^{(m)}$  satisfies the conditions of  $C^2$  compatibility at the points (t, x) = (0, 0) and (0, L) for the mixed initial-boundary problem (8.3.9)–(8.3.12), respectively. Notice (8.3.22), (8.3.40) and (8.3.42), it is easy to see from Lemma 8.2.2 that

$$\|U^{(m)}\|_{(C^2[R(T)])^N} \le C_{20}G(\varepsilon) \tag{8.3.44}$$

and

$$\begin{aligned} &\omega\left(\eta\Big|\frac{\partial^2 U^{(m)}}{\partial t^2}\right) + \omega\left(\eta\Big|\frac{\partial^2 U^{(m)}}{\partial x^2}\right) + \omega\left(\eta\Big|\frac{\partial^2 U^{(m)}}{\partial t\partial x}\right) \\ &\leq C_{21}\bigg(\omega(\eta|\Phi_{xx}) + \omega(\eta|\Psi_x) + \Omega_0(\eta) + G(\varepsilon)\Big(\eta + \Omega_1(\eta) + \Omega(\eta|\Gamma^*)\Big)\bigg), \end{aligned} \tag{8.3.45}$$

where  $C_{20}$  and  $C_{21}$  are positive constants depending only on T and  $T_0$ . Therefore, taking  $C_6 \ge C_{20}$ ,  $C_8 \ge C_{21}$ , we get (8.3.15) and (8.3.17).

On the other hand, let

$$V^{(m)} = U^{(m)} - U^{(m-1)}.$$
(8.3.46)

 $V^{(m)}$  satisfies the following miexd initial-boundary value problem:

$$\frac{\partial^2 V^{(m)}}{\partial t^2} - a^2 (U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) \frac{\partial^2 V^{(m)}}{\partial x^2} + A(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) V^{(m)} 
+ (a^2 (U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - a^2 (U^{(m-2)}, U_t^{(m-2)}, U_x^{(m-2)})) \frac{\partial^2 U^{(m-1)}}{\partial x^2} 
+ \left( A(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - A(U^{(m-2)}, U_t^{(m-2)}, U_x^{(m-2)}) \right) U^{(m-1)} = 0$$
(8.3.47)

with the initial condition

$$t = 0: (V^{(m)}, V_t^{(m)}) = (0, 0), \ 0 \le x \le L,$$
(8.3.48)

and the boundary conditions

$$x = 0: V^{(m)} = H^{(m)} - H^{(m-1)},$$
(8.3.49a)

$$x = 0: V_x^{(m)} = H^{(m)} - H^{(m-1)},$$
(8.3.49b)

$$x = 0: V_x^{(m)} - B(U^{(m-1)})V^{(m)} = \left(B(U^{(m-1)}) - B(U^{(m-2)})\right)U^{(m-1)} + H^{(m)} - H^{(m-1)}$$
(8.3.49c)

and

$$x = L : V^{(m)} = 0, (8.3.50a)$$

$$x = L : V_x^{(m)} = 0, (8.3.50b)$$

$$x = L : V_x^{(m)} + \overline{B}(U^{(m-1)})V^{(m)} = \left(\overline{B}(U^{(m-2)}) - \overline{B}(U^{(m-1)})\right)U^{(m-1)}$$
(8.3.50c)

Noting (8.3.20) - (8.3.21), we have

$$\begin{split} &\| \left( A(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - A(U^{(m-2)}, U_t^{(m-2)}, U_x^{(m-2)}) \right) U^{(m-1)} \|_{(C^0[R(T)])^N} \\ \leq & C_{22} C_7^{m-1} \left( G(\varepsilon) \right)^m, \end{split}$$

$$\begin{aligned} & (8.3.51) \\ &\| (a^2(U^{(m-1)}, U_t^{(m-1)}, U_x^{(m-1)}) - a^2(U^{(m-2)}, U_t^{(m-2)}, U_x^{(m-2)})) \frac{\partial^2 U^{(m-1)}}{\partial x^2} \|_{(C^0[R(T)])^N} \end{aligned}$$

$$\leq C_{23}C_7^{m-1}(G(\varepsilon))^m,\tag{8.3.52}$$

$$\| (B(U^{(m-1)}) - B(U^{(m-2)})) U^{(m-1)} \|_{(C^0[0,T])^N} \le C_{24} C_7^{m-1} (G(\varepsilon))^m,$$
(8.3.53)

$$\| (B(U^{(m-1)}) - B(U^{(m-2)})) U^{(m-1)} \|_{(C^0[0,T])^N} \le C_{25} C_7^{m-1} (G(\varepsilon))^m,$$
(8.3.54)

where  $C_{22}$ ,  $C_{23}$ ,  $C_{24}$  and  $C_{25}$  are positive constants depending only on T and  $T_0$ .

Noting (8.3.41), by Lemma 8.2.2 there exists a positive constant  $C_{26} = C_{26}(T, T_0)$  such that

$$\|V^{(m)}\|_{C^{1}[R(T)]} = \|U^{(m)} - U^{(m-1)}\|_{C^{1}[R(T)]} \le C_{26}C_{7}^{m-1}(G(\varepsilon))^{m}.$$
(8.3.55)

Therefore, choosing  $C_7 \geq C_{26}$ , we get (8.3.16). The proof of Theorem 8.1.1 is complete.

#### 8.4 Remarks

**Remark 8.4.1** Theorem 8.1.1 is still valid in the case of two-sided controls, provided that (8.1.14) is replaced by

$$T > T_0 > \frac{L}{a(0,0,0)},$$
 (8.4.1)

and there is an inhomogeneous control  $\overline{H}(t)$  on the right-hand side of (8.1.6).

**Remark 8.4.2** Similar results hold for the following boundary conditions of coupled dissipative type:

$$x = 0: U_x - C(U)U_t = H(t), (8.4.2)$$

$$x = L : U_x + \overline{C}(U)U_t = 0. \tag{8.4.3}$$

where  $C(U) = (c_{ij}(U))_{N \times N}$  and  $\overline{C}(U) = (\overline{c}_{ij}(U))_{N \times N}$  are  $C^1$  matrix functions with respect to their arguments. However, different from boundary conditions (8.1.4) and (8.1.6), in the case of boundary conditions (8.4.2)–(8.4.3), in order to guarantee the well-posedness and the exact boundary null controllability for the forward mixed initial-boundary value problem, we should impose the following assumptions:

$$\det\left(C(0) + \frac{1}{a(0,0,0)}I_N\right) \neq 0 \tag{8.4.4}$$

and

$$\det\left(\overline{C}(0) + \frac{1}{a(0,0,0)}I_N\right) \neq 0.$$
(8.4.5)

for the boundary coupling matrices C(U) and  $\overline{C}(U)$ . Then, combining the proof of Theorem 8.1.1 and the method presented in [57], one can realize the corresponding local exact boundary synchronization for the coupled system of wave equations (8.1.1) and (8.4.2)-(8.4.3) by (N-1) boundary controls on x = 0, provided that

$$\sum_{j=1}^{n} c_{ij}(U) \stackrel{\text{def.}}{=} \widetilde{\widetilde{c}}(U)$$
(8.4.6)

and

$$\sum_{j=1}^{n} \bar{c}_{ij}(U) \stackrel{\text{def.}}{=} \widetilde{\widetilde{c}}(U), \qquad (8.4.7)$$

where  $\tilde{\widetilde{c}}(U)$  and  $\tilde{\widetilde{c}}(U)$  are independent of  $i = 1, \cdots, N$ .

**Remark 8.4.3** Theorem 8.1.1 and Remark 8.4.2 are also valid for the corresponding nonautonomous quasilinear system, in which

$$a = a(t, x, U, U_t, U_x), \ A = A(t, x, U, U_t, U_x)$$
  

$$B = B(t, U), \ \overline{B} = \overline{B}(t, U) \ (or \ C = C(t, U), \ \overline{C} = \overline{C}(t, U)),$$
(8.4.8)

with suitable regularity with respect to t and x.

**Remark 8.4.4** For the following more general coupled system of quasilinear wave equations:

$$\frac{\partial^2 U}{\partial t^2} - a^2(U, U_t, U_x) \frac{\partial^2 U}{\partial x^2} + F(U, U_t, U_x) = 0, \qquad (8.4.9)$$

where F is a  $C^2$  vector function with respect to its arguments and

$$F(0,0,0) = 0. \tag{8.4.10}$$

Thus, one can find  $C^1$  matrices  $A(U, U_t, U_x)$ ,  $\check{A}(U, U_t, U_x)$  and  $\hat{A}(U, U_t, U_x)$  such that

$$F(U, U_t, U_x) = A(U, U_t, U_x)U + \check{A}(U, U_t, U_x)U_t + \hat{A}(U, U_t, U_x)U_x,$$
(8.4.11)

and the corresponding local exact boundary synchronization can be realized, provided that A,  $\breve{A}$  and  $\hat{A}$  satisfy similar assumptions as in (8.1.11), respectively.

#### 8.5 Appendix: Proof of Lemma 8.3.2

In order to get Lemma 8.3.2, it suffices to prove that for any given integer  $m \ge 1$ , there exists a unique  $C^2$  solution  $W^{(m)} = W^{(m)}(t, x)$  to the coupled system (8.3.24) of wave equations, which satisfies simultaneously the initial condition (8.3.29), the finial condition (8.3.31) and the boundary condition (8.3.27) on x = L as well as the estimates (8.3.32)–(8.3.34).

Let

$$T^{(m)} = \int_0^{T_0} \inf_{0 \le x \le L} a(U^{(m-1)}(t,x), U_t^{(m-1)}(t,x), U_x^{(m-1)}(t,x)) dt.$$
(8.5.1)

Due to (8.1.14) and (8.3.20), for suitably small  $\varepsilon > 0$  we have

$$T^{(m)} > 2L.$$
 (8.5.2)

Choose  $T_1^{(m)}$  and  $T_2^{(m)}$  such that

$$\int_{0}^{T_{1}^{(m)}} \inf_{0 \le x \le L} a(U^{(m-1)}(t,x), U_{t}^{(m-1)}(t,x), U_{x}^{(m-1)}(t,x)) dt$$

$$= \int_{T_{2}^{(m)}}^{T_{0}} \inf_{0 \le x \le L} a(U^{(m-1)}(t,x), U_{t}^{(m-1)}(t,x), U_{x}^{(m-1)}(t,x)) dt$$

$$= L.$$
(8.5.3)

By (8.1.14) and (8.3.20), for  $\varepsilon > 0$  small enough, we have

$$0 < T_1^{(m)} < \frac{T_0 - \delta}{2} < \frac{T_0}{2} < \frac{T_0 + \delta}{2} < T_2^{(m)} < T_0,$$
(8.5.4)

in which

$$\delta = \frac{1}{2} \left( T_0 - 2 \frac{L}{a(0,0,0)} \right). \tag{8.5.5}$$

(i) We first consider the forward mixed initial-boundary value problem for system (8.3.24) with the initial condition (8.3.29), the boundary condition (8.3.27) on x = L and the following artificial *Neumann* boundary condition

$$x = 0: W_x^{(m)} = \mathfrak{F}(t), \quad 0 \le t \le \frac{T_0}{2},$$
(8.5.6)

where  $\mathfrak{F}(\cdot)$  is an arbitrarily given  $C^1$  function of t, satisfying the following conditions of  $C^2$  compatibility at the point (t, x) = (0, 0):

$$\begin{cases} \mathfrak{F}(0) = \widetilde{\Phi}'(0), \\ \mathfrak{F}'(0) = \widetilde{\Psi}'(0) \end{cases}$$
(8.5.7)

(see Remark 8.1.2) and verifying

$$\|\mathfrak{F}\|_{C^1[0,\frac{T_0}{2}]} \le \varepsilon, \tag{8.5.8}$$

$$\Omega(\eta | \dot{\mathfrak{F}}) \le \Omega_0(\eta). \tag{8.5.9}$$

Due to (8.5.7)-(8.5.9), this artificial boundary condition (8.5.6) can be chosen to be independent of the iteration (i.e., independent of m), namely, we can take the same artificial boundary condition (8.5.6) in every iteration step. Meanwhile, in order that the constants, obtained from all the estimates of the solutions in the iteration procedure, to be also independent of m, each iteration should be discussed on the same domain, i.e., on

$$R_f = \left\{ (t, x) \middle| 0 \le t \le \frac{T_0}{2}, 0 \le x \le L \right\}$$
(8.5.10)

or

$$R_b = \left\{ (t, x) \middle| \frac{T_0}{2} \le t \le T, 0 \le x \le L \right\}.$$
(8.5.11)

All these are different from the previous works on the constructive method.

Let

$$\widetilde{\Gamma}^* = \left\{ \frac{\partial a}{\partial u_k}, \frac{\partial a}{\partial u_{kt}}, \frac{\partial a}{\partial u_{kx}}, \frac{\partial \widetilde{a}_{ij}}{\partial u_k}, \frac{\partial \widetilde{a}_{ij}}{\partial u_{kt}}, \frac{\partial \widetilde{a}_{ij}}{\partial u_{kx}}, \frac{\partial \widetilde{b}_{ij}}{\partial u_k}, \frac{\partial \overline{b}_{ij}}{\partial u_k} \right\},$$

$$i, j, k = 1, \cdots, N.$$
(8.5.12)

Obviously, there exists a positive constant  $\widetilde{C}_1$  such that

$$\omega(\eta|\widetilde{\Gamma}^*) \le \widetilde{C}_1 \omega(\eta|\Gamma^*). \tag{8.5.13}$$

Here and hereafter,  $\widetilde{C}_i(i = 1, 2 \cdots)$  denote positive constants,  $\widetilde{C}_i(T_0)$  denote positive constants depending only on  $T_0$ , and  $\widetilde{C}_i(T, T_0)$  denote positive constants depending only on T and  $T_0$ .

By Lemma 8.2.2, there exists a unique  $C^2$  solution  $W^{(m)} = W_f^{(m)}(t,x)$  on the domain  $R_f$  (see (8.5.10)). In particular, noting (8.3.20), (8.3.22) and the properties of continuity modulus (see [84]), we have

$$\|W_f^{(m)}(t,x)\|_2 \le \widetilde{C}_2(T_0)\varepsilon \le \widetilde{C}_3(T_0)G(\varepsilon), \quad \forall (t,x) \in R_f,$$
(8.5.14)

$$\omega\left(\eta \left| \frac{\partial^2 W_f^{(m)}}{\partial t^2} \right) + \omega\left(\eta \left| \frac{\partial^2 W_f^{(m)}}{\partial x^2} \right) + \omega\left(\eta \left| \frac{\partial^2 W_f^{(m)}}{\partial t \partial x} \right) \right.\right. \\ \leq \widetilde{C}_4(T_0) \left( \omega(\eta \left| \widetilde{\Phi}_{xx} \right) + \omega(\eta \left| \widetilde{\Psi}_x \right) + \Omega_0(\eta) + \varepsilon \left(\eta + \Omega_1(\eta) + \left(1 + G(\varepsilon)\right) \Omega(\eta \left| \widetilde{\Gamma}^* \right) \right) \right) \\ \leq \widetilde{C}_5(T_0) \left( \omega(\eta \left| \Phi_{xx} \right) + \omega(\eta \left| \Psi_x \right) + \Omega_0(\eta) + G(\varepsilon) \left(\eta + \Omega_1(\eta) + \Omega(\eta \left| \Gamma^* \right) \right) \right).$$

$$(8.5.15)$$

Thus we can uniquely determine the value of  $(W_f^{(m)}, W_{fx}^{(m)})$  on x = L:

$$x = L : (W_f^{(m)}, W_{fx}^{(m)}) = (a^{(m)}(t), \bar{a}^{(m)}(t)), \quad 0 \le t \le \frac{T_0}{2}, \tag{8.5.16}$$

and we have

$$\|(a^{(m)}(t), \overline{a}^{(m)}(t))\|_{(C^{2}[0, \frac{T_{0}}{2}])^{N} \times (C^{1}[0, \frac{T_{0}}{2}])^{N}} \leq \widetilde{C}_{6}(T_{0})G(\varepsilon), \qquad (8.5.17)$$

$$\omega(\eta | \ddot{a}^{(m)}) + \omega(\eta | \dot{\overline{a}}^{(m)})$$

$$\leq \widetilde{C}_7(T_0) \bigg( \omega(\eta | \Phi_{xx}) + \omega(\eta | \Psi_x) + \Omega_0(\eta) + G(\varepsilon) \Big( \eta + \Omega_1(\eta) + \Omega(\eta | \Gamma^*) \Big) \bigg). \quad (8.5.18)$$

On the other hand, denoting

$$Z_f^{(m)} = W_f^{(m)} - W_f^{(m-1)}, \ m = 0, 1, 2, \dots$$
(8.5.19)

it is easy to see that  $Z_f^{(m)}$  satisfies the following mixed initial-boundary value problem:

with the initial condition

$$t = 0: (Z_f^{(m)}, Z_{ft}^{(m)}) = (0, 0), \ 0 \le x \le L,$$
(8.5.21)

the boundary condition on x = 0

$$x = 0: Z_{fx}^{(m)} = 0, (8.5.22)$$

and the boundary condition on x = L:

$$x = L : Z_f^{(m)} = 0, (8.5.23a)$$

$$x = L : Z_{fx}^{(m)} = 0, (8.5.23b)$$

$$x = L : Z_{fx}^{(m)} + \widetilde{\overline{B}}(U^{(m-1)}) Z_f^{(m)} = \left(\widetilde{\overline{B}}(U^{(m-2)}) - \widetilde{\overline{B}}(U^{(m-1)})\right) W_f^{(m-1)}.$$
 (8.5.23c)

Noting (8.3.20)–(8.3.21), (8.3.25) and (8.5.14), we get

$$I_{0} \stackrel{\text{def.}}{=} \left\| \left( a^{2} (U^{(m-2)}, U_{t}^{(m-2)}, U_{x}^{(m-2)}) - a^{2} (U^{(m-1)}, U_{t}^{(m-1)}, U_{x}^{(m-1)}) \right) \frac{\partial^{2} W_{f}^{(m-1)}}{\partial x^{2}} \right\|_{(C^{0}[R_{f}])^{N}} \leq \widetilde{C}_{8}(T_{0}) C_{7}^{m-1} \left( G(\varepsilon) \right)^{m},$$

$$(8.5.24)$$

$$I_{1} \stackrel{\text{def.}}{=} \| \left( \widetilde{A}(U^{(m-1)}, U_{t}^{(m-1)}, U_{x}^{(m-1)}) - \widetilde{A}(U^{(m-2)}, U_{t}^{(m-2)}, U_{x}^{(m-2)}) \right) W_{f}^{(m-1)} \right) \|_{(C^{0}[R_{f}])^{N}} \leq \widetilde{C}_{9}(T_{0}) C_{7}^{m-1} \left( G(\varepsilon) \right)^{m},$$

$$(8.5.25)$$

$$I_{2} \stackrel{\text{def.}}{=} \| \left( \widetilde{\overline{B}}(U^{(m-1)}) - \widetilde{\overline{B}}(U^{(m-2)}) \right) W_{f}^{(m-1)} \right) \|_{(C^{0}[R_{f}])^{N}} \leq \widetilde{C}_{10}(T_{0}) C_{7}^{m-1} \left( G(\varepsilon) \right)^{m}.$$
(8.5.26)

Therefore, by Lemma 8.2.2, there exists a positive constant  $\widetilde{C}_{11}(T_0)$  independent of m, such that

$$\|W^{(m)} - W^{(m-1)}\|_{(C^{1}[R_{f}])^{N}} \leq \widetilde{C}_{11}(T_{0})C_{7}^{m-1}(G(\varepsilon))^{m}.$$
(8.5.27)

In particular, we have

$$\|(a^{(m)}(t) - a^{(m-1)}(t), \overline{a}^{(m)}(t) - \overline{a}^{(m-1)}(t))\|_{(C^{1}[0, \frac{T_{0}}{2}])^{N} \times (C^{0}[0, \frac{T_{0}}{2}])^{N}} \leq \widetilde{C}_{12}(T_{0})C_{7}^{m-1}(G(\varepsilon))^{m}.$$
(8.5.28)

(ii) On the domain  $R_b$  (see (8.5.11)), let  $W_b^{(m)}(t,x) \equiv 0$ . Obviously,  $W = W_b^{(m)}(t,x)$  verifies system (8.3.24), finial condition (8.3.31) and boundary conditions (8.3.27).

(iii) Thus, we can find the vector function  $(c^{(m)}(t), \bar{c}^{(m)}(t)) \in (C^2[0,T])^N \times (C^1[0,T])^N$  (for example, by polynomial interpolation), such that

$$(c^{(m)}(t), \bar{c}^{(m)}(t)) = \begin{cases} (a^{(m)}(t), \bar{a}^{(m)}(t)), & 0 \le t \le \frac{T_0 - \delta}{2}, \\ (0, 0), & \frac{T_0 + \delta}{2} \le t \le T, \end{cases}$$
(8.5.29)

and  $(c^{(m)}(t), \bar{c}^{(m)}(t))$  verifies the boundary condition (8.3.27) at x = L on the whole intervel  $0 \le t \le T$ , moreover, we have

$$\begin{aligned} \|(c^{(m)}(t), \overline{c}^{(m)}(t))\|_{C^{2}[0,T] \times C^{1}[0,T]} &\leq \widetilde{C}_{13}(T, T_{0})G(\varepsilon), \end{aligned} \tag{8.5.30} \\ \|(c^{(m)}(t) - c^{(m-1)}(t), \overline{c}^{(m)}(t) - \overline{c}^{(m-1)}(t))\|_{C^{2}[0,T] \times C^{1}[0,T]} &\leq \widetilde{C}_{14}(T, T_{0})C_{6}^{m-1}(G(\varepsilon))^{m}, \end{aligned} \tag{8.5.31} \\ \omega(\eta|\ddot{c}^{(m)}) + \omega(\eta|\dot{c}^{(m)}) \\ &\leq \widetilde{C}_{15}(T, T_{0}) \bigg( \omega(\eta|\Phi_{xx}) + \omega(\eta|\Psi_{x}) + \Omega_{0}(\eta) + G(\varepsilon) \Big(\eta + \Omega_{1}(\eta) + \Omega(\eta|\Gamma^{*})\Big) \bigg). \end{aligned} \tag{8.5.32}$$

We now change the status of t and x, and consider the following leftward mixed initial-boundary value problem on the domain R(T) for system (8.3.24) with the initial condition

$$x = L : W_l^{(m)} = c^{(m)}(t), W_{lx}^{(m)} = \bar{c}^{(m)}(t), \quad 0 \le t \le T$$
(8.5.33)

and the boundary conditions

$$t = 0: W_{lt}^{(m)} = \widetilde{\Psi}(x), \quad 0 \le x \le L,$$
 (8.5.34)

$$t = T : W_{lt}^{(m)} = 0, \quad 0 \le x \le L$$
 (8.5.35)

where  $\widetilde{\Psi}(x)$  is given by (8.3.30).

Obviously, the corresponding conditions of  $C^2$  compatibility are satisfied at the points (t, x) = (0, L) and (T, L), respectively. Hence, by Lemma 8.2.2, there exists a

unique  $C^2$  solution  $W^{(m)} = W_l^{(m)}(t, x)$  on the domain R(T), such that

$$\|W_{l}^{(m)}(t,x)\|_{C^{2}[R(T)]} \leq \widetilde{C}_{16}(T,T_{0})G(\varepsilon), \quad \forall (t,x) \in R(T),$$

$$\omega\left(\eta \left|\frac{\partial^{2}W_{l}^{(m)}}{\partial t^{2}}\right) + \omega\left(\eta \left|\frac{\partial^{2}W_{l}^{(m)}}{\partial x^{2}}\right) + \omega\left(\eta \left|\frac{\partial^{2}W_{l}^{(m)}}{\partial t\partial x}\right)\right.\right)$$

$$\leq \widetilde{C}_{17}(T,T_{0})\left(\omega(\eta|\Phi_{xx}) + \omega(\eta|\Psi_{x}) + \Omega_{0}(\eta) + G(\varepsilon)\left(\eta + \Omega_{1}(\eta) + \Omega(\eta|\Gamma^{*})\right)\right).$$

$$(8.5.37)$$

Thus,  $Z_l^{(m)} = W_l^{(m)} - W_l^{(m-1)}$  satisfies the mixed initial-boundary value problem of system (8.5.20) with the homogeneous boundary condition

$$t = 0: Z_x^{(m)} = 0, (8.5.38)$$

$$t = T : Z_x^{(m)} = 0 \tag{8.5.39}$$

and the initial condition

$$x = L : (Z^{(m)}, Z_x^{(m)}) = (c^{(m)}(t) - c^{(m-1)}(t), \overline{c}^{(m)}(t) - \overline{c}^{(m-1)}(t)).$$
(8.5.40)

Combining (8.5.31), (8.5.36) and Lemma 8.2.2, we get

$$\|W_l^{(m)}(t,x) - W_l^{(m-1)}(t,x)\|_{(C^2[R(T)])^N} \le \widetilde{C}_{18}(T,T_0)C_7^{m-1}(G(\varepsilon))^m.$$
(8.5.41)

(iv) By the uniqueness of the  $C^2$  solution for one-sided mixed initial-boundary value problem (see [72], [84], [111] and [112]), the  $C^2$  function  $W^{(m)} = W_l^{(m)}(t, x)$  on R(T) satisfies the system (8.3.24), the initial condition (8.3.30), the finial condition (8.3.31) and the boundary condition (8.3.27) on x = L. Moreover, noting (8.5.36)–(8.5.37) and (8.5.41), we get (8.3.32)–(8.3.34). This concludes the proof of Lemma 8.3.2.

### Bibliography

- O. Aamo. Disturbance rejection in 2x2 linear hyperbolic systems. Automatic Control, IEEE Transactions on, PP(99):1, 2012.
- [2] Fatiha Alabau-Boussouira. Indirect boundary stabilization of weakly coupled hyperbolic systems. SIAM J. Control Optim., 41(2):511–541 (electronic), 2002.
- [3] Fatiha Alabau-Boussouira. A two-level energy method for indirect boundary observability and controllability of weakly coupled hyperbolic systems. SIAM J. Control Optim., 42(3):871–906 (electronic), 2003.
- [4] Fatiha Alabau-Boussouira. Controllability of cascade coupled systems of multidimensional evolution PDEs by a reduced number of controls. C. R. Math. Acad. Sci. Paris, 350(11-12):577–582, 2012.
- [5] Fatiha Alabau-Boussouira. Insensitizing exact controls for the scalar wave equation and exact controllability of 2-coupled cascade systems of PDE's by a single control. *Math. Control Signals Systems*, 26(1):1–46, 2014.
- [6] Fatiha Alabau-Boussouira and Matthieu Léautaud. Indirect controllability of locally coupled wave-type systems and applications. J. Math. Pures Appl. (9), 99(5):544–576, 2013.
- [7] S. Amin, Hante F., and A. Bayen. *Hybrid Systems: Comp*, chapter On stability of switched linear hyperbolic conservation laws with reflecting boundaries, pages 602–605. Springer-Verlag, 2008.
- [8] Fabio Ancona and Andrea Marson. On the attainable set for scalar nonlinear conservation laws with boundary control. SIAM J. Control Optim., 36(1):290– 312 (electronic), 1998.
- [9] Fabio Ancona and Andrea Marson. Asymptotic stabilization of systems of conservation laws by controls acting at a single boundary point. In *Control methods in PDE-dynamical systems*, volume 426 of *Contemp. Math.*, pages 1– 43. Amer. Math. Soc., Providence, RI, 2007.

- [10] Georges Bastin and Jean-Michel Coron. On boundary feedback stabilization of non-uniform linear 2 × 2 hyperbolic systems over a bounded interval. Systems Control Lett., 60(11):900–906, 2011.
- [11] Georges Bastin and Jean-Michel Coron. Stability and Boundary Stabilization of 1-D Hyperbolic Systems. Preprint, 2015.
- [12] Georges Bastin, Jean-Michel Coron, and Brigitte d'Andréa Novel. Using hyperbolic systems of balance laws for modeling, control and stability analysis of physical networks. In Lecture notes for the Pre-Congress Workshop on Complex Embedded and Networked Control Systems, Seoul, Korea, 2008.
- [13] Alberto Bressan and Giuseppe Maria Coclite. On the boundary control of systems of conservation laws. SIAM J. Control Optim., 41(2):607–622 (electronic), 2002.
- [14] Eduardo Cerpa and Jean-Michel Coron. Rapid stabilization for a Kortewegde Vries equation from the left Dirichlet boundary condition. *IEEE Trans. Automat. Control*, 58(7):1688–1695, 2013.
- [15] Marco Cirinà. Boundary controllability of nonlinear hyperbolic systems. SIAM J. Control, 7:198–212, 1969.
- [16] J.-M. Coron, B. d'Andréa-Novel, and G. Bastin. A lyapunov approach to control irrigation canals modeled by saint-venant equations. *Proceedings of the 1999 European Control Conference, Karlsruhe, Germany*, 1999.
- [17] Jean-Michel Coron. On the null asymptotic stabilization of the two-dimensional incompressible Euler equations in a simply connected domain. SIAM J. Control Optim., 37(6):1874–1896 (electronic), 1999.
- [18] Jean-Michel Coron. Local controllability of a 1-D tank containing a fluid modeled by the shallow water equations. *ESAIM Control Optim. Calc. Var.*, 8:513– 554 (electronic), 2002. A tribute to J. L. Lions.
- [19] Jean-Michel Coron. Control and nonlinearity, volume 136 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2007.
- [20] Jean-Michel Coron and Georges Bastin. Dissipative boundary conditions for one-dimensional quasi-linear hyperbolic systems: Lyapunov stability for the  $C^1$ -norm. *Preprint*, 2014.
- [21] Jean-Michel Coron, Georges Bastin, and Brigitte d'Andréa Novel. Dissipative boundary conditions for one-dimensional nonlinear hyperbolic systems. SIAM J. Control Optim., 47(3):1460–1498, 2008.

- [22] Jean-Michel Coron, Brigitte d'Andréa Novel, and Georges Bastin. A strict Lyapunov function for boundary control of hyperbolic systems of conservation laws. *IEEE Trans. Automat. Control*, 52(1):2–11, 2007.
- [23] Jean-Michel Coron, Oliver Glass, and Zhiqiang Wang. Exact boundary controllability for 1-D quasilinear hyperbolic systems with a vanishing characteristic speed. SIAM J. Control Optim., 48(5):3105–3122, 2009/10.
- [24] Jean-Michel Coron and Qi Lü. Local rapid stabilization for a Korteweg-de Vries equation with a Neumann boundary control on the right. J. Math. Pures Appl. (9), 102(6):1080–1120, 2014.
- [25] Jean-Michel Coron and Hoai-Minh Nguyen. Dissipative boundary conditions for nonlinear 1-d hyperbolic systems: sharpconditions through an approach via time-delay systems. *Preprint*, 2014.
- [26] Jean-Michel Coron, Rafael Vazquez, Miroslav Krstic, and Georges Bastin. Local exponential  $H^2$  stabilization of a 2 × 2 quasilinear hyperbolic system using backstepping. SIAM J. Control Optim., 51(3):2005–2035, 2013.
- [27] René Dáger and Enrique Zuazua. Wave propagation, observation and control in 1-d flexible multi-structures, volume 50 of Mathématiques & Applications (Berlin) [Mathematics & Applications]. Springer-Verlag, Berlin, 2006.
- [28] J. de Halleux, C. Prieur, J.-M. Coron, B. d'Andréa-Novel, and G. Bastin. Boundary feedback control in networks of open channels. *Automatica*, 39(8):1365 – 1376, 2003.
- [29] F. Di Meglio. Dynamics and control of slugging in oil production. PhD thesis, MINES ParisTech, 2011.
- [30] F. Di Meglio, D. Bresch-Pietri, and U. J. F. Aarsnes. An adaptive observer for hyperbolic systems with application to underbalanced drilling. In *Proceeding* of the 2014 IFAC World Congress, pages 11391–11397, 2014.
- [31] Florent Di Meglio, Rafael Vazquez, and Miroslav Krstic. Stabilization of a system of n + 1 coupled first-order hyperbolic linear PDEs with a single boundary input. *IEEE Trans. Automat. Control*, 58(12):3097–3111, 2013.
- [32] Ababacar Diagne, Georges Bastin, and Jean-Michel Coron. Lyapunov exponential stability of 1-D linear hyperbolic systems of balance laws. *Automatica* J. IFAC, 48(1):109–114, 2012.
- [33] S. Djordjevic, O.H. Bosgra, P.M.J. Van den Hof, and D. Jeltsema. Boundary actuation structure of linearized two-phase flow. In American Control Conference (ACC), 2010, pages 3759 –3764, 30 2010-july 2 2010.

- [34] Vasile Dragan and Aristide Halanay. Stabilization of linear systems. Systems & Control: Foundations & Applications. Birkhäuser Boston, Inc., Boston, MA, 1999. Translated from the 1994 Romanian original.
- [35] S. Dudret, K. Beauchard, F. Ammouri, and Rouchon P. Stability and asymptotic observers of binary distillation processes described by nonlinear convection/diffusion models. *Proceedings of the 2012 American Control Conference*, *Montréal, Canada*, pages 3352–3358, 2012.
- [36] Thomas Duyckaerts, Xu Zhang, and Enrique Zuazua. On the optimality of the observability inequalities for parabolic and hyperbolic systems with potentials. Ann. Inst. H. Poincaré Anal. Non Linéaire, 25(1):1–41, 2008.
- [37] O. Yu. Émanuilov. Boundary controllability of semilinear evolution equations. Uspekhi Mat. Nauk, 44(3(267)):185–186, 1989.
- [38] Xiaoyu Fu, Jiongmin Yong, and Xu Zhang. Exact controllability for multidimensional semilinear hyperbolic equations. SIAM J. Control Optim., 46(5):1578–1614 (electronic), 2007.
- [39] Hirokazu Fujisaka and Tomoji Yamada. Stability theory of synchronized motion in coupled-oscillator systems. Progr. Theoret. Phys., 69(1):32–47, 1983.
- [40] Olivier Glass. On the controllability of the 1-D isentropic Euler equation. J. Eur. Math. Soc. (JEMS), 9(3):427–486, 2007.
- [41] Olivier Glass. On the controllability of the non-isentropic 1-D Euler equation. J. Differential Equations, 257(3):638–719, 2014.
- [42] J. M. Greenberg and Tatsien Li. The effect of boundary damping for the quasilinear wave equation. J. Differential Equations, 52(1):66–75, 1984.
- [43] Qilong Gu and Tatsien Li. Exact boundary controllability for quasilinear wave equations in a planar tree-like network of strings. Ann. Inst. H. Poincaré Anal. Non Linéaire, 26(6):2373–2384, 2009.
- [44] Martin Gugat. Boundary controllability between sub- and supercritical flow. SIAM J. Control Optim., 42(3):1056–1070 (electronic), 2003.
- [45] Martin Gugat. Optimal boundary control in flood management. In Control of coupled partial differential equations, volume 155 of Internat. Ser. Numer. Math., pages 69–94. Birkhäuser, Basel, 2007.
- [46] Martin Gugat, Markus Dick, and Günter Leugering. Gas flow in fan-shaped networks: classical solutions and feedback stabilization. SIAM J. Control Optim., 49(5):2101–2117, 2011.

- [47] Martin Gugat and Michaël Herty. Existence of classical solutions and feedback stabilization for the flow in gas networks. ESAIM Control Optim. Calc. Var., 17(1):28–51, 2011.
- [48] Martin Gugat and Günter Leugering. Global boundary controllability of the de St. Venant equations between steady states. Ann. Inst. H. Poincaré Anal. Non Linéaire, 20(1):1–11, 2003.
- [49] Lina Guo and Zhiqiang Wang. Exact boundary observability for nonautonomous first-order quasilinear hyperbolic systems. *Math. Methods Appl. Sci.*, 31(16):1956–1971, 2008.
- [50] Harry Hochstadt. Integral equations. John Wiley & Sons, New York-London-Sydney, 1973. Pure and Applied Mathematics.
- [51] T. Horsin. On the controllability of the Burgers equation. ESAIM Control Optim. Calc. Var., 3:83–95 (electronic), 1998.
- [52] Long Hu. Sharp time estimates for exact boundary controllability of quasilinear hyperbolic systems. *Preprint*, 2014.
- [53] Long Hu and Florent Di Meglio. Finite-time backstepping stabilization of  $3 \times 3$  hyperbolic systems. Accepted by IEEE European Control Conference, 2015.
- [54] Long Hu, Florent Di Meglio, Rafael Vazquez, and Miroslav Krstic. Boundary control design of homodirectional and general heterodirectional linear hyperbolic PDEs. *Preprint*, 2015.
- [55] Long Hu, Fanqiong Ji, and Ke Wang. Exact boundary controllability and exact boundary observability for a coupled system of quasilinear wave equations. *Chin. Ann. Math. Ser. B*, 34(4):479–490, 2013.
- [56] Long Hu, Tatsien Li, and Peng Qu. Exact boundary synchronization for a coupled system of 1-D quasilinear wave equations. *Preprint*, 2014.
- [57] Long Hu, Tatsien Li, and Bopeng Rao. Exact boundary synchronization for a coupled system of 1-D wave equations with coupled boundary conditions of dissipative type. *Commun. Pure Appl. Anal.*, 13(2):881–901, 2014.
- [58] Long Hu, Rafael Vazquez, Florent Di Meglio, and Miroslav Krstic. Boundary exponential stabilization of 1-D inhomogeneous quasilinear hyperbolic systems. *Preprint*, 2015.
- [59] Long Hu and Zhiqiang Wang. On boundary control of a hyperbolic system with a vanishing characteristic speed. *Preprint*, 2014.
- [60] Christiaan Huygens. *Œuvres Complètes de Christiaan Huygens*, volume 19. Swets & Zeitlinger B.V, Amsterdam, 1967.

- [61] Fritz John. Formation of singularities in one-dimensional nonlinear wave propagation. Comm. Pure Appl. Math., 27:377–405, 1974.
- [62] Jerzy Klamka. Controllability of dynamical systems. Mat. Stosow., (9(50)):57– 75, 2008.
- [63] Paul Koosis. The logarithmic integral. I, volume 12 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 1988.
- [64] Miroslav Krstic and Andrey Smyshlyaev. Backstepping boundary control for first-order hyperbolic PDEs and application to systems with actuator and sensor delays. Systems Control Lett., 57(9):750–758, 2008.
- [65] Miroslav Krstic and Andrey Smyshlyaev. Boundary control of PDEs, volume 16 of Advances in Design and Control. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2008. A course on backstepping designs.
- [66] I. Lasiecka and R. Triggiani. Exact controllability of semilinear abstract systems with application to waves and plates boundary control problems. *Appl. Math. Optim.*, 23(2):109–154, 1991.
- [67] I. Lasiecka, R. Triggiani, and Peng-Fei Yao. Inverse/observability estimates for second-order hyperbolic equations with variable coefficients. J. Math. Anal. Appl., 235(1):13–57, 1999.
- [68] Guenter Leugering and E. J. P. Georg Schmidt. On the modelling and stabilization of flows in networks of open canals. SIAM J. Control Optim., 41(1):164– 180, 2002.
- [69] Tatsien Li. Global classical solutions for quasilinear hyperbolic systems, volume 32 of RAM: Research in Applied Mathematics. Masson, Paris; John Wiley & Sons, Ltd., Chichester, 1994.
- [70] Tatsien Li. Exact boundary controllability for quasilinear wave equations. J. Comput. Appl. Math., 190(1-2):127–135, 2006.
- [71] Tatsien Li. Exact boundary observability for 1-D quasilinear wave equations. Math. Methods Appl. Sci., 29(13):1543–1553, 2006.
- [72] Tatsien Li. Controllability and observability for quasilinear hyperbolic systems, volume 3 of AIMS Series on Applied Mathematics. American Institute of Mathematical Sciences (AIMS), Springfield, MO; Higher Education Press, Beijing, 2010.
- [73] Tatsien Li and Yi Jin. Semi-global C<sup>1</sup> solution to the mixed initial-boundary value problem for quasilinear hyperbolic systems. *Chinese Ann. Math. Ser. B*, 22(3):325–336, 2001.

- [74] Tatsien Li and Bopeng Rao. Local exact boundary controllability for a class of quasilinear hyperbolic systems. *Chinese Ann. Math. Ser. B*, 23(2):209–218, 2002. Dedicated to the memory of Jacques-Louis Lions.
- [75] Tatsien Li and Bopeng Rao. Exact boundary controllability for quasi-linear hyperbolic systems. SIAM J. Control Optim., 41(6):1748–1755 (electronic), 2003.
- [76] Tatsien Li and Bopeng Rao. Exact controllability for first order quasilinear hyperbolic systems with vertical characteristics. Acta Math. Sci. Ser. B Engl. Ed., 29(4):980–990, 2009.
- [77] Tatsien Li and Bopeng Rao. Strong (weak) exact controllability and strong (weak) exact observability for quasilinear hyperbolic systems. *Chin. Ann. Math. Ser. B*, 31(5):723–742, 2010.
- [78] Tatsien Li and Bopeng Rao. Asymptotic controllability for linear hyperbolic systems. Asymptot. Anal., 72(3-4):169–187, 2011.
- [79] Tatsien Li and Bopeng Rao. Exact synchronization for a coupled system of wave equations with Dirichlet boundary controls. *Chin. Ann. Math. Ser. B*, 34(1):139–160, 2013.
- [80] Tatsien Li and Bopeng Rao. A note on the exact synchronization by groups for a coupled system of wave equations. *Mathematical Methods in the Applied Sciences*, 38(2):241–246, 2015.
- [81] Tatsien Li, Bopeng Rao, and Long Hu. Exact boundary synchronization for a coupled system of 1-D wave equations. *ESAIM Control Optim. Calc. Var.*, 20(2):339–361, 2014.
- [82] Tatsien Li and Lixin Yu. Exact controllability for first order quasilinear hyperbolic systems with zero eigenvalues. *Chinese Ann. Math. Ser. B*, 24(4):415–422, 2003.
- [83] Tatsien Li and Lixin Yu. Exact boundary controllability for 1-D quasilinear wave equations. SIAM J. Control Optim., 45(3):1074–1083, 2006.
- [84] Tatsien Li and Wen Ci Yu. Boundary value problems for quasilinear hyperbolic systems. Duke University Mathematics Series, V. Duke University, Mathematics Department, Durham, NC, 1985.
- [85] Tatsien Li, Wen Ci Yu, and Wei Xi Shen. Second initial-boundary value problems for quasilinear hyperbolic-parabolic coupled systems (in Chinese). *Chinese* Ann. Math., 2(1):65–90, 1981.

- [86] Tatsien Li and Bingyu Zhang. Global exact controllability of a class of quasilinear hyperbolic systems. J. Math. Anal. Appl., 225(1):289–311, 1998.
- [87] Peter Linz. Analytical and numerical methods for Volterra equations, volume 7 of SIAM Studies in Applied Mathematics. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 1985.
- [88] J.-L. Lions. Contrôlabilité exacte, perturbations et stabilisation de systèmes distribués. Tome 1, volume 8 of Recherches en Mathématiques Appliquées [Research in Applied Mathematics]. Masson, Paris, 1988. Contrôlabilité exacte. [Exact controllability], With appendices by E. Zuazua, C. Bardos, G. Lebeau and J. Rauch.
- [89] J.-L. Lions. Exact controllability, stabilization and perturbations for distributed systems. SIAM Rev., 30(1):1–68, 1988.
- [90] Qi Lü. Exact controllability for stochastic transport equations. SIAM J. Control Optim., 52(1):397–419, 2014.
- [91] Sorin Micu. On the controllability of the linearized Benjamin-Bona-Mahony equation. SIAM J. Control Optim., 39(6):1677–1696 (electronic), 2001.
- [92] Boris S. Mordukhovich and Jean-Pierre Raymond. Dirichlet boundary control of hyperbolic equations in the presence of state constraints. *Appl. Math. Optim.*, 49(2):145–157, 2004.
- [93] Boris S. Mordukhovich and Jean-Pierre Raymond. Neumann boundary control of hyperbolic equations with pointwise state constraints. SIAM J. Control Optim., 43(4):1354–1372 (electronic), 2004/05.
- [94] A. Pazy. Semigroups of linear operators and applications to partial differential equations, volume 44 of Applied Mathematical Sciences. Springer-Verlag, New York, 1983.
- [95] Louis M. Pecora and Thomas L. Carroll. Synchronization in chaotic systems. *Phys. Rev. Lett.*, 64(8):821–824, 1990.
- [96] Vincent Perrollaz. Exact controllability of scalar conservation laws with an additional control in the context of entropy solutions. *SIAM J. Control Optim.*, 50(4):2025–2045, 2012.
- [97] Vincent Perrollaz and Lionel Rosier. Finite-time stabilization of 2×2 hyperbolic systems on tree-shaped networks. SIAM J. Control Optim., 52(1):143–163, 2014.

- [98] Tie Hu Qin. Global smooth solutions of dissipative boundary value problems for first order quasilinear hyperbolic systems. *Chinese Ann. Math. Ser. B*, 6(3):289–298, 1985. A Chinese summary appears in Chinese Ann. Math. Ser. A 6 (1985), no. 4, 514.
- [99] Lionel Rosier and Pierre Rouchon. On the controllability of a wave equation with structural damping. Int. J. Tomogr. Stat., 5(W07):79–84, 2007.
- [100] David L. Russell. Controllability and stabilizability theory for linear partial differential equations: recent progress and open questions. SIAM Rev., 20(4):639– 739, 1978.
- [101] Denis Serre. Systems of conservation laws: a challenge for the XXIst century. In Mathematics unlimited—2001 and beyond, pages 1061–1080. Springer, Berlin, 2001.
- [102] Joel Smoller. Shock waves and reaction-diffusion equations, volume 258 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Science]. Springer-Verlag, New York-Berlin, 1983.
- [103] Andrey Smyshlyaev and Miroslav Krstic. Adaptive control of parabolic PDEs. Princeton University Press, Princeton, NJ, 2010.
- [104] Steven Strogatz. Sync: The emerging science of spontaneous order. Hyperion, 2003.
- [105] Rafael Vazquez and Miroslav Krstic. Control of 1-D parabolic PDEs with Volterra nonlinearities. I. Design. Automatica J. IFAC, 44(11):2778–2790, 2008.
- [106] Rafael Vazquez and Miroslav Krstic. Control of 1D parabolic PDEs with Volterra nonlinearities. II. Analysis. Automatica J. IFAC, 44(11):2791–2803, 2008.
- [107] Rafael Vazquez and Miroslav Krstic. Marcum Q-functions and explicit kernels for stabilization of linear hyperbolic systems with constant coefficients. Systems & Control Letters, 68:33 – 42, 2014.
- [108] Rafael Vázquez, Emmanuel Trélat, and Jean-Michel Coron. Control for fast and stable laminar-to-high-Reynolds-numbers transfer in a 2D Navier-Stokes channel flow. Discrete Contin. Dyn. Syst. Ser. B, 10(4):925–956, 2008.
- [109] Ke Wang. Exact boundary controllability for a kind of second-order quasilinear hyperbolic systems. *Chin. Ann. Math. Ser. B*, 32(6):803–822, 2011.
- [110] Ke Wang. Exact boundary controllability of nodal profile for 1-D quasilinear wave equations. *Front. Math. China*, 6(3):545–555, 2011.
- [111] Zhiqiang Wang. Exact controllability for nonautonomous first order quasilinear hyperbolic systems. Chinese Ann. Math. Ser. B, 27(6):643–656, 2006.
- [112] Zhiqiang Wang. Exact boundary controllability for non-autonomous quasilinear wave equations. *Math. Methods Appl. Sci.*, 30(11):1311–1327, 2007.
- [113] Zhiqiang Wang. Global exact controllability for quasilinear hyperbolic systems of diagonal form with linearly degenerate characteristics. *Nonlinear Anal.*, 69(2):510–522, 2008.
- [114] Zhiqiang Wang and Lixin Yu. Exact boundary controllability for a onedimensional adiabatic flow system. Appl. Math. J. Chinese Univ. Ser. A, 23(1):35–40, 2008.
- [115] Chai Wah Wu. Synchronization in complex networks of nonlinear dynamical systems, volume 76. World Scientific, 2007.
- [116] C.-Z. Xu and G. Sallet. Exponential stability and transfer functions of processes governed by symmetric hyperbolic systems. ESAIM: Control, Optimisation and Calculus of Variations, 7:421–442, 2002.
- [117] Li Xin Yu. Semi-global C<sup>1</sup> solutions to mixed initial-boundary value problems for a kind of quasilinear hyperbolic systems. *Chinese Ann. Math. Ser. A*, 25(5):549–560, 2004.
- [118] Lixin Yu. Exact boundary controllability for a kind of second-order quasilinear hyperbolic systems and its applications. *Math. Methods Appl. Sci.*, 33(3):273– 286, 2010.
- [119] Lixin Yu. Exact boundary observability for a kind of second order quasilinear hyperbolic systems and its applications. *Nonlinear Anal.*, 72(12):4452–4465, 2010.
- [120] Qi Zhang. Exact boundary controllability with less controls acting on two ends for quasilinear hyperbolic systems (in Chinese). Appl. Math. J. Chinese Univ. Ser. A, 24(1):65–74, 2009.
- [121] Yi Zhou and Zhen Lei. Local exact boundary controllability for nonlinear wave equations. *SIAM J. Control Optim.*, 46(3):1022–1051 (electronic), 2007.
- [122] E. Zuazua. Exact controllability for semilinear wave equations in one space dimension. Ann. Inst. H. Poincaré Anal. Non Linéaire, 10(1):109–129, 1993.
- [123] Enrike Zuazua. Exact controllability for the semilinear wave equation. J. Math. Pures Appl. (9), 69(1):1–31, 1990.

## Boundary controllability, stabilization and synchronization for 1-D hyperbolic balance laws

## Abstract :

This thesis is devoted to three topics in the control field, namely, exact boundary controllability, boundary stabilization and exact boundary synchronization, for hyperbolic systems of balance laws. For the exact boundary controllability part, we first improve the boundary control time for hyperbolic systems of conservation laws with general boundary conditions and show that this control time is sharp. Then for a coupled hyperbolic system with zero characteristic speed, we prove that it is impossible to achieve the corresponding exact boundary controllability even with inner couplings in the equation. However, one can stabilize the system in infinite time by means of boundary feedback laws. For the boundary stabilization part, we show how to stabilize both the  $n \times n$  linear and quasilinear hyperbolic systems by means of one-sided closed-loop boundary controls. For that a backstepping method is developed. For the exact boundary synchronization part, we first recall both the exact boundary controllability and observability results for coupled systems of quasilinear wave equations. Then several kinds of exact synchronizations are introduced for a coupled system of 1-D linear and quasilinear wave equations with boundary conditions of Dirichlet type, Neumann type, coupled third type and coupled dissipative type in the framework of  $C^2$  solutions. We show that all these synchronizations can be realized by means of few boundary controls.

**Keywords :** Exact boundary controllability, boundary stabilization, boundary synchronization, 1-D hyperbolic balance laws, coupled wave systems

## Contrôle frontière, stabilisation et synchronisation pour des systèmes de lois de bilan en dimension un d'espace

## Résumé :

Cette thèse est consacrée à trois sujets dans le domaine du contrôle, qui sont la contrôlabilité exacte frontière, la stabilisation frontière et la synchronisation exacte frontière, des systèmes hyperboliques de lois de bilan. Pour la partie sur la contrôlabilité exacte frontière, on améliore le temps de contrôlabilité exacte pour les systèmes hyperboliques de lois de conservation pour des conditions aux limites générales. On montre aussi que ce temps est optimal. En ce qui concerne les systèmes hyperboliques couplés avec une vitesse caractéristique nulle, nous prouvons que l'on n'a pas la contrôlabilité exacte, même avec des couplages internes dans les équations. Cependant, on montre que l'on peut stabiliser les systèmes par les lois de rétroaction à la frontière du domaine. Dans la deuxième partie, nous nous intéressons à la stabilisation frontière des systèmes hyperboliques de lois de bilan. En utilisant une approche "backstepping", on montre comment stabiliser des systèmes d'abord dans les cas linéaires puis dans les cas quasi-linéaires. La troisième partie concerne la synchronisation exacte frontière. Nous rappelons d'abord les résultats de contrôlabilité et d'observabilité exacte frontière pour les systèmes couplés d'équations des ondes quasi-linéaires. Puis nous introduisons plusieurs types de synchronisations pour un système d'équations des ondes linéaires, puis quasi-linéaires, couplées avec des conditions aux limites de type Dirichlet, de type Neumann, de type Robin et de type dissipatif dans le cadre de solutions de classe  $C^2$ . Nous montrons que toutes ces synchronisations peuvent être réalisées en imposant peu de contrôles aux frontières.

Mot clés : Contrôlabilité exacte frontière, stabilisation frontière, synchronisation frontière, systèmes hyperboliques de lois de bilan 1-D, systèmes d'équations des ondes couplées